

Effort–Reward Imbalance, Overcommitment, Perceived Control and Health Behaviours in Central and Eastern Europe

Sung-Wei Chen

Thesis submitted for the degree of Doctor of Philosophy Department of Epidemiology and Public Health University College London

Declaration

I, Sung–Wei Chen, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Abstract

Aims Health behaviours – alcohol drinking, smoking, poor diet and physical inactivity – are influenced by various psychosocial factors. Despite evidence linking work stress and personality constructs independently to health behaviours, only limited literature is available on the relationship between work stress, personality and health behaviours. The aims of the thesis are: (1) to examine the potential role of overcommitment (OC) personality in the relationship between work stress defined by the Effort–Reward Imbalance (ERI) model and health behaviours; (2) to investigate the potential role of perceived control (PC) in the relationship between ERI, OC and health behaviours.

Methods This project used data from the HAPIEE (Health, Alcohol and Psychosocial factors In Eastern Europe) study, which randomly selected people aged 45 to 69 years from population registers in Russia, Poland and the Czech Republic. A two–wave cohort study for drinking and smoking outcomes (n= 7,513) and a cross–sectional study for dietary outcomes (n= 11,012) were analysed by logistic regression and structural equation modelling.

Results In terms of the potential role of OC in the relationship between ERI and health behaviours, OC and ERI may have bi–directional relationship; the effect of OC on ERI was stronger than the other direction in the middle–aged and older populations. Thus, antecedent role of OC in the relation between ERI and health behaviours was statistically significant, but mediator role of OC was not. With regards to the potential role of PC in the relationship between OC, ERI and health behaviours, both ERI and PC partially mediated the effects of OC on health behaviours; ERI and PC may have bi–directional relationship.

Conclusion This thesis will contribute to deeper understanding of intersecting pathways by which work stress (ERI) and personality constructs (OC and PC) jointly influence health behaviours, thereby providing insight into research, practice and policy.

Acknowledgements

I owe my completion of this thesis to guidance and support from several people.

Firstly, I would like to thank my supervisors Dr. Hynek Pikhart and Dr. Anne Peasey, who provided me with the opportunity to study in a supportive and inspiring environment, guided me through various challenges from research and London life, and demonstrated a logical and organized way of Western wisdom. I deeply appreciate their continuing encouragement and advice throughout my study. Thanks particularly to Prof. Martin Bobak, who have supervised me and offered excellent guide, thoughtful insight and revolutionary idea in an incredibly interesting way.

I am extremely grateful to Prof. Tarani Chandola for careful evaluation and helpful comment for my thesis. The most thrilling and enlightening experience in UCL is to attend the courses by Sir Michael Marmot. Thanks to Dr. Anne McMunn and Dr. Annie Britton for kind help and guidance in our department. I never forget that Mrs. Sandy Persaud always gave enthusiastic warmth in each encounter. Thanks to the HAPIEE team members: Denes, Peijue, Olga, Andrea, Pia, Gyuri, Milagros, Dorina, Margherita and Yaoyue for brainstorming discussion and generous help. I am grateful to UCL colleagues; in particular, David, Knott, Linda, and Saddie as good friends and tutors of the Great British culture.

I deeply appreciate my bosses and colleagues in National Taiwan University Hospital: Prof. Susan Shur-Fen Gau, Prof. Kuan-Yu Hung, Prof. Hsi-Yu Yu, Prof. Jui-Sheng Sun, Prof. Chien-Chang Wu and Dr. Jen-Hui Chan for creating the precious opportunity to study abroad with job security. Thanks to Prof. Shen-Ing Liu and Prof. Ping-Lung Hsin for giving inspiration and good recommendation for my study.

Finally, I would thank my parents and grandparents who studied in Austria and Japan, inspiring my curiosity to a novel world. Without love and patience from my wife Shih-Min and two daughters, I could not be where I am now. Thanks for keeping dad running enthusiastically despite lagging a bit behind.

Index

Chapte	er 1. Introduction	
Chapte	er 2. Literature Review22	
2.1 ⊦	lealth Behaviours22	
2.1.1	Health behaviours – definition and their impact on health 22	
2.1.2	Health behaviours and health in Central and Eastern Europe24	
2.2 P	Psychosocial Factors	
2.2.1	Psychosocial factors – definition and their impact on health	
2.2.2	Psychosocial factors and health in Central and Eastern Europe33	
2.2.3	Psychosocial factors and health behaviours	
2.3 V	Vork Stress and Health Behaviours41	
2.3.1	The Demand–Control model41	
2.3.2	The Effort-Reward Imbalance model44	
2.3.3	Empirical studies on work stress and health behaviours46	
2.4 C	Overcommitment Personality and Health Behaviours52	
2.4.1	Overcommitment – origins from Type A behaviour53	
2.4.2	Empirical studies on Type A behaviour and health behaviours57	
2.4.3	Overcommitment – origins from perceived control	
2.4.4	Empirical studies on perceived control and health behaviours	
2.4.5	Relationship between overcommitment, perceived control and health	
behaviours		
2.5 C	Overcommitment, Effort–Reward Imbalance, and Health Behaviours75	
2.5.1	Modifying or main effect of OC in ERI–outcome relationship75	
2.5.2	Antecedent role of OC in ERI–outcome relationship	
2.5.3	Mediator role of OC in ERI-outcome relationship80	
2.5.4	Reciprocal relationship between OC and ERI across life-course 83	
2.6 C	Overcommitment, Effort–Reward Imbalance, Perceived Control, and	
Health Behaviours		
2.6.1	Transactional Model of Stress	
2.6.2	Reciprocal relationship between perceived control and ERI	

2.6.3 Interaction between perceived control and ERI	94
2.6.4 Relationship between OC, ERI, perceived control and health	
behaviours	96
2.7 Summary of Literature Review	98
2.7.1 Gaps identified in the existing research	98
2.7.2 Importance of proposed project	. 101
Chapter 3. Aims, Objectives and Hypotheses	. 102
3.1 Aims	. 102
3.2 Objectives and Hypotheses	. 102
Chapter 4. Methods	. 106
4.1 Study Description	. 106
4.2 Study Samples	. 107
4.3 Description of Variables	. 110
4.3.1 Outcome variables	. 111
4.3.2 Main exposure variables	. 114
4.3.3 Covariates	. 118
4.4 Statistical Power	. 119
4.5 Statistical Analysis	. 120
4.6 Methodology for Structural Equation Modelling	. 121
4.6.1 Introduction to path analysis	. 122
4.6.2 Introduction to confirmatory factor analysis	. 125
4.6.3 Basic steps in Structural Equation Modelling	. 127
4.6.4 Structural Equation Modelling in cohort studies	. 130
4.6.5 Structural Equation Modelling with categorical outcomes	. 133
Chapter 5. Drinking Outcomes	. 137
5.1 Descriptive Statistics	. 137
5.1.1 Descriptive characteristics of study populations	. 137
5.1.2 Descriptive characteristics of drinking outcomes	. 139
5.2 Potential Role of OC in ERI–Drinking Relationship	. 146
5.2.1 Associations between ERI and drinking outcomes	. 146

5.2.2 Associations between OC and drinking outcomes147
5.2.3 Antecedent or mediator role of OC in ERI-drinking relationship 149
5.2.4 Modifying role of OC in ERI–drinking relationship157
5.3 Potential Role of PC in Relation between OC, ERI, and Drinking 159
5.3.1 Associations between PC and drinking outcomes
5.3.2 Mediator roles of PC and ERI in OC-drinking relationship160
5.3.3 Modifying role of PC in ERI–drinking relationship
5.4 Main Findings for Drinking Outcomes 169
Chapter 6. Smoking Outcomes
6.1 Descriptive Statistics
6.1.1 Descriptive characteristics of study populations
6.1.2 Descriptive characteristics of smoking outcomes
6.2 Potential Role of OC in ERI–Smoking Relationship
6.2.1 Associations between ERI and smoking outcomes
6.2.2 Associations between OC and smoking outcomes
6.2.3 Antecedent or mediator role of OC in ERI-smoking relationship 181
6.2.4 Modifying role of OC in ERI–smoking relationship
6.3 Potential Role of PC in Relation between OC, ERI, and Smoking 192
6.3.1 Associations between PC and smoking outcomes
6.3.2 Mediator roles of PC and ERI in OC-smoking relationship194
6.3.3 Modifying role of PC in ERI–smoking relationship
6.4 Main Findings for Smoking Outcomes
Chapter 7. Dietary Outcomes
7.1 Descriptive Statistics
7.1.1 Descriptive characteristics of study populations
7.1.2 Descriptive characteristics of dietary outcomes
7.2 Associations between ERI, OC, and Dietary Outcomes
7.2.1 Associations between ERI and dietary outcomes
7.2.2 Associations between OC and dietary outcomes
7.2.3 Modifying role of OC in ERI-diet relationship

7.3 Potential Role of PC in Relation between OC, ERI, and Diet	219	
7.3.1 Associations between PC and dietary outcomes	219	
7.3.2 Modifying role of PC in ERI–diet relationship	221	
7.3.3 Mediator roles of PC and ERI in OC-diet relationship	222	
7.4 Main Findings for Dietary Outcomes	229	
Chapter 8. General Discussion	231	
8.1 Summary for Results	231	
8.2 Methodological Issues	235	
8.2.1 Representativeness of study sample and selection bias	235	
8.2.2 Information bias	237	
8.2.3 Confounding	239	
8.2.4 Limitations of two-wave cohort study designs	240	
8.2.5 Limitations of cross-sectional study designs	241	
8.2.6 Random error	242	
8.3 Discussion of Results	242	
8.3.1 Work stress and health behaviours	243	
8.3.2 Overcommitment personality and health behaviours	244	
8.3.3 Potential role of OC in relationship between ERI and health		
behaviours	245	
8.3.4 Perceived control and health behaviours	248	
8.3.5 Potential role of PC in relationship between OC, ERI and he	alth	
behaviours		
Chapter 9. Implications for Research, Practice and Policy	251	
9.1 Associations between Work Stress and Health Behaviours	251	
9.1.1 Organisational intervention for health behaviours	252	
9.1.2 Organisational intervention for the ERI model	253	
9.2 Potential Role of OC in Relationship between ERI and Health		
Behaviours253		
9.2.1 Cognitive behaviour therapy and personality	254	
9.2.2 Individual intervention targeting cognitive-behaviour mecha	nisms via	

which personality influences work stress255		
9.3	Potential Role of PC in Relation between OC, ERI, and Health	
Beł	naviours	256
9.3.	.1 Stress management intervention and Transactional Model of Stress	i
		256
9.3.	2 Stress management intervention for the ERI model	257
9.4	A Multi–Level and Integrated Perspective for Psychosocial Factors at	
Wo	rk and Health Behaviours	258
Cha	apter 10. Conclusions	260
Reference		262
Арр	pendix	304
1.	Personal questionnaire in the HAPIEE study wave 1	
2.	Questionnaire for working individuals in the HAPIEE study wave 1	
3.	Dietary questionnaire in the HAPIEE study wave 1	
4.	Path analyses for three separate drinking outcomes	
5.	Published paper relevant for the thesis	

List of tables

Table 2.1	Big Five personality traits and their dimensions56	
Table 2.2	Different constructs in the concepts of control64	
Table 2.3	Empirical studies supporting perceived control as a mediator in the effects	
of works st	tress on outcomes91	
Table 2.4	Empirical studies supporting work stress as a mediator in the effects of	
perceived	control on outcomes93	
Table 2.5	Empirical studies supporting interaction between perceived control and	
work stres	s95	
Table 4.1	The CAGE questionnaire112	
Table 4.2.	Nine individual HDI components as dichotomous variables114	
Table 4.3	The Effort–Reward Imbalance questionnaire116	
Table 4.4	The perceived control score117	
Table 4.5	Power calculation for the two study samples (N= 6000 and 4000)120	
Table 5.1	Descriptive statistics of study sample by country and gender138	
Table 5.2	Descriptive statistics of drinking outcomes by country and gender140	
Table 5.3	Crude associations between ER ratio and drinking outcomes in country-	
specific str	rata142	
Table 5.4	Crude associations between OC and drinking outcomes in country-specific	
strata		
Table 5.5	Bivariate analyses for relationships between covariates and drinking	
outcomes in men144		
Table 5.6	Bivariate analyses for relationships between covariates and drinking	
outcomes in women145		
Table 5.7	Associations between ER ratio at wave 1 and drinking outcomes at wave	
2		
Table 5.8	Associations between OC at wave 1 and drinking outcomes at wave	

2148
Table 5.9 Measurement model of the SEM for antecedent or mediator role of OC in
ERI-drinking relation151
Table 5.10 Structural model of the SEM for antecedent or mediator role of OC in
ERI-drinking relation in men153
Table 5.11 Structural model of the SEM for antecedent or mediator role of OC in ERI–
drinking relation in women155
Table 5.12 Evaluation for modifying role of OC in ERI–drinking relationship157
Table 5.13 Measurement model of the SEM for modifying role of OC in ERI-drinking
relation158
Table 5.14Associations of PC at wave 1 with drinking outcomes at wave 2159
Table 5.15 Measurement model of the SEM for mediator roles of PC and ERI in OC-
drinking relation162
Table 5.16 Structural model of the SEM for mediator roles of PC and ERI in OC-
drinking relation in men164
Table 5.17 Structural model of the SEM for mediator roles of PC and ERI in OC-
drinking relation in women166
Table 5.18 Evaluation for modifying role of PC in ERI–drinking relationship168
Table 5.19 Measurement model of the SEM for modifying role of PC in ERI-drinking
relation169
Table 6.1 Descriptive statistics of smoking outcomes by country and gender173
Table 6.2 Crude associations between ER ratio and smoking outcomes in country-
specific strata
Table 6.3 Crude associations between OC and smoking outcomes in country-
specific strata175
Table 6.4 Bivariate analyses between covariates and smoking outcomes in men

Table 6.5 Bivariate analyses between covariates and smoking outcomes in women

477
Table 6.6 Associations between ER ratio at wave 1 and smoking outcomes at wave
2179
Table 6.7 Associations between OC at wave 1 and smoking outcomes at wave
2
Table 6.8 Results of path analysis for antecedent or mediator role of OC in ERI-
smoking status relationship in men184
Table 6.9 Results of path analysis for antecedent or mediator role of OC in ERI-
smoking status relationship in women
Table 6.10 Results of path analysis for antecedent or mediator role of OC in ERI-
smoking intensity relationship in men188
Table 6.11 Results of path analysis for antecedent or mediator role of OC in ERI-
smoking intensity relationship in women190
Table 6.12 Evaluation for modifying role of OC in ERI–smoking relationship192
Table 6.13 Associations between PC at wave 1 and smoking outcomes at wave
2193
Table 6.14 Results of path analysis for mediator roles of PC and ERI in OC–smoking
status relationship in men196
Table 6.15 Results of path analysis for mediator roles of PC and ERI in OC–smoking
status relationship in women198
Table 6.16 Results of path analysis for mediator roles of PC and ERI in OC-smoking
intensity relationship in men
Table 6.17 Results of path analysis for mediator roles of PC and ERI in OC–smoking
Table 6.17 Results of path analysis for mediator roles of PC and ERI in OC–smoking
Table 6.17 Results of path analysis for mediator roles of PC and ERI in OC–smoking intensity relationship in women
Table 6.17Results of path analysis for mediator roles of PC and ERI in OC–smokingintensity relationship in women

the preven	tion of chronic diseases (2003)209
Table 7.4	Overall scores of Healthy Diet Indicator and proportions of subjects in
different sc	ores
Table 7.5	Crude associations between exposure variables and HDI outcome in
country-sp	ecific strata211
Table 7.6	Crude associations between exposure variables and HDI outcome in
gender-sp	ecific strata212
Table 7.7	Associations between HDI outcome and covariates by ordinal logistic
regression	
Table 7.8	Associations between ER ratio and dietary outcomes215
Table 7.9	Associations between OC and dietary outcomes217
Table 7.10	Evaluation for modifying role of OC in ERI-diet relationship218
Table 7.11	Associations between perceived control and dietary outcomes220
Table 7.12	Evaluation for modifying role of PC in ERI–diet relationship221
Table 7.13	Results of path analysis for mediator roles of PC and ERI in OC-diet
relationship	o in men
Table 7.14	Results of path analysis for mediator roles of PC and ERI in OC-diet
relationship	o in women
Table 7.15	Results of path analysis for mediator roles of PC and ERI in OC-diet
relationship	o in pooled data of men and women228

List of figures

Figure 2.1	Trends in life expectancy at birth (years) for selected countries (1970-
2010)	
Figure 2.2	Theoretical framework for social determinants of health31
Figure 2.3	Mean levels of control for 7 population samples plotted against all-cause
mortality for	the countries from which the population samples were drawn36
Figure 2.4	Conceptual framework for social contexts of health behaviours
Figure 2.5	Low control, high demand and low work support cause adverse health
outcomes	
Figure 2.6	High effort and low reward at work cause adverse health outcomes45
Figure 2.7	Theoretical framework of Theory of Planned Behaviour68
Figure 2.8	Perceived behavioural control mediated the effects of personality traits on
physical act	ivity73
Figure 2.9	The Michigan model describes individual and situational factors affecting
the process	of work stress78
Figure 2.10	Transactional Model of Stress87
Figure 2.11	Potential relation between OC, ERI, PC and health behaviours based on
Transaction	al Model of Stress97
Figure 3.1	A conceptual framework of the thesis105
Figure 4.1	The study sites in the HAPIEE study107
Figure 4.2	Flow chart for selection of the study subsamples110
Figure 4.3	The single-mediator model adopted to introduce path analysis123
Figure 4.4	A measurement model with one factor and three indicators126
Figure 4.5	An autoregressive model with a cross-lagged panel design131
Figure 4.6	A three–wave cohort design to estimate a mediation model131
Figure 4.7	A two-wave cohort design to estimate a mediation model132
Figure 4.8	A measurement model for relationships between a latent continuous

factor and th	nree observed categorical variables134
Figure 5.1	Measurement model and structural model specified for the SEM for
antecedent	or mediator role of OC in ERI–drinking relation149
Figure 5.2	Results of the SEM for antecedent or mediator role of OC in ERI-drinking
relation in m	en154
Figure 5.3	Results of the SEM for antecedent or mediator role of OC in ERI-drinking
relation in w	omen156
Figure 5.4	Measurement model and structural model specified for the SEM for
mediator rol	es of PC and ERI in OC–drinking relationship161
Figure 5.5	Results of the SEM for mediator roles of PC and ERI in OC-drinking
relation in m	en165
Figure 5.6	Results of the SEM for mediator roles of PC and ERI in OC-drinking
relation in w	omen167
Figure 6.1	Hypothetical model specified for the path analysis for antecedent or
mediator rol	e of OC in ERI–smoking relation182
	e of OC in ERI-smoking relation
Figure 6.2	-
Figure 6.2 smoking sta	Results of path analysis for antecedent or mediator role of OC in ERI-
Figure 6.2 smoking sta Figure 6.3	Results of path analysis for antecedent or mediator role of OC in ERI– tus relationship in men
Figure 6.2 smoking sta Figure 6.3 smoking sta	Results of path analysis for antecedent or mediator role of OC in ERI– tus relationship in men
Figure 6.2 smoking sta Figure 6.3 smoking sta Figure 6.4	Results of path analysis for antecedent or mediator role of OC in ERI– tus relationship in men
Figure 6.2 smoking sta Figure 6.3 smoking sta Figure 6.4 smoking inte	Results of path analysis for antecedent or mediator role of OC in ERI– tus relationship in men
Figure 6.2 smoking sta Figure 6.3 smoking sta Figure 6.4 smoking inte Figure 6.5	Results of path analysis for antecedent or mediator role of OC in ERI– tus relationship in men
Figure 6.2 smoking sta Figure 6.3 smoking sta Figure 6.4 smoking inte Figure 6.5 smoking inte	Results of path analysis for antecedent or mediator role of OC in ERI– tus relationship in men
Figure 6.2 smoking sta Figure 6.3 smoking sta Figure 6.4 smoking inte Figure 6.5 smoking inte Figure 6.6	Results of path analysis for antecedent or mediator role of OC in ERI- tus relationship in men
Figure 6.2 smoking sta Figure 6.3 smoking sta Figure 6.4 smoking inte Figure 6.5 smoking inte Figure 6.6 PC and ERI	Results of path analysis for antecedent or mediator role of OC in ERI- tus relationship in men
Figure 6.2 smoking sta Figure 6.3 smoking sta Figure 6.4 smoking inte Figure 6.5 smoking inte Figure 6.6 PC and ERI Figure 6.7	Results of path analysis for antecedent or mediator role of OC in ERI- tus relationship in men

status relation	ship in women199
Figure 6.9 R	esults of path analysis for mediator roles of PC and ERI in OC-smoking
intensity relation	onship in current smokers in men201
Figure 6.10	Results of path analysis for mediator roles of PC and ERI in OC–smoking
intensity relation	onship in current smokers in women203
Figure 7.1 H	lypothetical model specified for path analysis for mediator roles of PC
and ERI in OC	–diet relation223
Figure 7.2 R	Results of path analysis for mediator roles of PC and ERI in OC-diet
relationship in	men225
Figure 7.3 R	Results of path analysis for mediator roles of PC and ERI in OC-diet
relationship in	women
Figure 7.4 R	Results of path analysis for mediator roles of PC and ERI in OC-diet
relationship in	pooled data of men and women228

List of abbreviations

CEE	Central and Eastern Europe
CHD	Coronary heart disease
CI	Confidence interval
DC model	Demand–Control model
ERI model	Effort–Reward Imbalance model
ER ratio	Effort-reward ratio
HAPIEE study	Health, Alcohol and Psychosocial factors In Eastern Europe study
HDI	Healthy Diet Indicator
LOC	Locus of control
OC	Overcommitment
OR	Odds ratio
PC	Perceived control
PBC	Perceived behavioural control
SCM	Social Cognitive Model
SEM	Structural Equation Modelling
ТРВ	Theory of Planned Behaviour
WHO	World Health Organisation

Chapter 1. Introduction

Health behaviours, such as alcohol drinking, smoking, poor diet and physical inactivity, have been found to increase the risks of chronic diseases – major causes of morbidity and mortality across the world.¹ Health behaviours are influenced by a wide range of psychosocial factors – measurements that link psychological phenomena to social environment and physiology, such as chronic stress, personality constructs, psychological distress, and protective aspects of social environment.^{2,3} Despite empirical support linking each psychosocial factor independently to health behaviours, few studies have attempted to examine potential relationships between different psychosocial factors – combined influences of work stress and personality on human behaviours have attracted researchers' interest and debate. A profound understanding of these relationships is crucial to promote accumulation of knowledge and to inform effective interventions on health behaviours.

Stressors are demands made by internal or external environment that upset balance in an individual, thereby affecting physical and psychological well-being and requiring one's action to restore balance.⁴ Work stress, as defined by the Demand–Control (DC) model and the Effort–Reward Imbalance (ERI) model, has been repeatedly reported to predict worse profiles of health behaviours and health outcomes in empirical studies.^{5,6} The DC model suggests that job task profiles defined by low control and high demand (job strain) may elicit sustained stress reactions. The ERI model proposes that violation of social reciprocity in terms of high effort and low reward at work may elicit negative emotions and sustained stress responses.⁷

Personality represents a dynamic organisation, inside the person, of psychophysical systems that create a person's characteristic patterns of behaviours, thoughts and feelings.⁸ Various personality constructs have been repeatedly reported to predict health behaviours in several empirical studies.^{9,10} Overcommitment (OC)

personality was proposed in the ERI model; OC displays attitude, behaviour and emotion characterized by excessive striving at work and strong motivation for approval. OC was originated from the concept of Type A behaviour, which was derived from perceived control (PC). Rosenman proposed that Type A persons have higher need for control over environment and tend to perceive lower PC; their response is enhanced coping to assert and maintain control over environment.¹¹ Although the effects of OC on health behaviours have rarely been examined, many studies have supported the effects of OC–related personality (Type A behaviour, Neuroticism, and Hostility) on health behaviours.^{12,13,14}

Despite empirical evidence linking work stress and OC–related personality independently to health behaviours, little literature is available on the relationship between work stress, OC personality and health behaviours. The potential role of OC in ERI–outcome relationship was originally suggested as main effect or modifying effect.¹⁵ However, this original assumption on OC remains inconclusive in existing literature and appears relatively simple compared to accumulated research on diverse roles of personality in stress processes (modifying, antecedent, mediator, or direct effects).¹⁶ To evaluate the potential role of OC in ERI–outcome relationship more rigorously, these four possible roles should be examined simultaneously.

It is plausible to suggest a potential role of perceived control (PC) in the relationship between OC, ERI and health behaviours; PC might mediate the effects of OC on health behaviours. Rosenman proposed that Type A persons have higher need for control (OC) over environment and tend to perceive lower PC.¹¹ Greenberger and Strasser suggested that the higher need for control a person has, the lower PC one perceives.¹⁷ In fact, it is suggested that social–cognitive constructs (e.g. PC) provide a more active and specific process account of individual differences that complements the broader and more static personality traits (e.g. OC).¹⁸

Based on gaps identified in existing research, the aims of the thesis are: (1) To examine the potential role of OC personality in the relationship between work stress

defined by the ERI model and health behaviours, including modifying, antecedent, mediator, and direct effect of OC. (2) To investigate the potential role of PC in the relationship between ERI, OC and health behaviours. A two–wave cohort study for drinking and smoking outcomes (n= 7513) and a cross–sectional study for dietary outcomes (n= 11012) are conducted in the middle–aged and older populations in Central and Eastern Europe. It is hoped that this thesis would contribute to growing understanding of the combined influences of work stress (ERI) and personality constructs (OC and PC) on health behaviours.

The thesis is structured as follows. The second chapter provides a general literature review on psychosocial factors and health behaviours, particularly focusing on the associations between work stress and health behaviours, the associations between OC–related personality and health behaviours, and the associations between PC and health behaviours. Furthermore, the chapter discusses the potential role of OC in the relationship between ERI and health behaviours, and the potential role of PC in the relationship between OC, ERI and health behaviours.

The third chapter outlines the aims, objectives and hypotheses of the thesis. The fourth chapter gives detailed explanation for the methods of the thesis, including description of the HAPIEE (Health, Alcohol and Psychosocial factors In Eastern Europe) study, study samples, description of variables used in this project, statistical power, and statistical analysis. Specific details in the methodology for Structural Equation Modelling (SEM) are also provided.

Chapters five to seven describe the analytical methods and the results for drinking outcomes, smoking outcomes and dietary outcomes, respectively. In general, each result chapter is divided into three parts. First, descriptive statistics for study populations and outcomes are shown. Second, the potential role of OC in ERI– outcome relationship is analyzed, including direct, antecedent, mediator and modifying effects of OC. Third, the potential role of PC in the relationship between OC, ERI and outcomes is examined. Finally, main findings relevant to the hypotheses are

summarized.

Chapter eight focuses on general discussion of the thesis findings. The main results for three health-behaviour outcomes are summarized. The methodological issues are also addressed. Next, the results are discussed according to the following topics: (1) work stress and health behaviours; (2) OC personality and health behaviours; (3) the potential role of OC in the relationship between ERI and health behaviours; (4) PC and health behaviours; (5) the potential role of PC in the relationship between OC, ERI and health behaviours.

Chapter nine provides an overall discussion regarding the implications for research, practice and policy based on the findings obtained from the thesis. Finally, Chapter ten summarizes the general conclusions of the whole thesis.

Chapter 2. Literature Review

2.1 Health Behaviours

2.1.1 Health behaviours – definition and their impact on health

It has been increasingly recognized that individuals can contribute to their health by adopting or avoiding particular behaviours. Since the 1970s, health behaviours have become a paramount issue in epidemiology through a series of empirical studies. For instance, the impacts of health behaviours on mortality and morbidity were investigated in Alameda County in the United State (1979).¹⁹ Health behaviours were added to traditional physiological and environmental factors to predict health outcomes in the British Whitehall II study (1991).²⁰ In contemporary research, health behaviours – including alcohol drinking, smoking, poor diet and physical inactivity – are considered the main causes of morbidity and mortality (chronic disease such as heart disease, stroke, or cancer) across the world.¹

While health behaviours are undoubtedly important, their definitions are diverse. Conner and Norman defined health behaviours as: "any activity undertaken for the purpose of preventing or detecting disease or for improving health and well-being".²¹ By this definition, there are a wide range of health behaviours including medical service usage, compliance with treatment, and self-directed health behaviours (e.g. smoking). The definitions have evolved with mounting evidence for the impacts of psychosocial factors on health behaviours. Emphasizing psychological factors, Gochman defined health behaviours as: "personal attributes like beliefs, expectations, values, and other cognitive elements; personality characteristics, including emotional and affective states and traits; and overt behaviour patterns, actions, and habits that relate to health maintenance, health restoration, and health improvement".²² Emphasizing social factors, Cockerham proposed the collective patterns of health behaviours – health lifestyles, which are based on individuals' life choices from available options according to their life chances (social structures, collective patterns of living related to societies, institutions, or social classes that constrain or enable individuals to act).²³

Despite diversity in the definitions, a common way of classifying health behaviours in epidemiology is to distinguish between risky behaviours and protective behaviours.²⁴ Risky behaviours have harmful effects on health or predispose individuals to diseases; protective behaviours enhance health or protect individuals from diseases. It depends on existing evidence to define whether a specific behaviour is a risky or protective behaviour; empirical studies have shown that risky behaviours such as alcohol drinking, smoking, poor diet and physical inactivity can increase the risks of chronic diseases across the world, including Central and Eastern Europe (CEE).^{25,26} Because of their impacts on morbidity and mortality, the term health behaviours adopted in this thesis represent four risky health behaviours: alcohol drinking, smoking, poor diet and physical inactivity. The four health behaviours are discussed in Chapter 2, but physical inactivity will be excluded from my analyses due to the limitations of available data.

The four risky health behaviours and their impacts on health are summarized in the next part of this section. Although not all researchers agree on the J–shape relationship between higher alcohol consumption and poorer health outcomes, moderate alcohol consumption is generally associated with decreased mortality due to cardio-protective effects, but high alcohol consumption is associated with increased mortality and morbidity.^{27,28} For acute effects, drinking progressively impairs cognition, attention, judgement and coordination, resulting in increased risks of accident, injury, violence, and suicide. For chronic effects, high alcohol consumption adversely affects nearly every organ of the body, resulting in coronary heart disease, cardiomyopathy, liver cirrhosis, gastritis, pancreatitis, infection, neurological disorders, psychiatric disorders, and cancers of upper digestive tract, liver and breast.^{29,30}

The toxic components in tobacco smoking include hydrogen cyanide, carbon monoxide, and nitrogen oxide, which cause damage of cells and tissues in a variety

of organs. Smoking is found to increase the risks of coronary heart disease, myocardial infarction, stroke, chronic obstructive pulmonary disease, and cancers of lung, mouth, pharynx, larynx and oesophagus.^{31,32} For example, after long duration of smoking (20 to 50 years), mortality rate of middle–aged current smokers was found to increase to 3 times higher than that of non–smokers.³³ Lung cancer death rates are 10–12 times higher in current smokers than in non–smokers across the world.³⁴

Dietary intakes are reported to predict a variety of chronic diseases; adherence to a healthy diet is generally associated with reduced mortality and morbidity.^{35,36} For example, high intakes of fruit and vegetable (major sources of vitamins and minerals) may reduce the risks of coronary heart disease, stroke, hypertension, diabetes, obesity and cancer.³⁷ In contrast, high intakes of saturated fat and cholesterol are associated with high levels of blood lipids (low–density lipoprotein fraction of cholesterol and triglycerides), which increase the risks of coronary heart disease and atherosclerosis. High intakes of free sugars are also associated with increased risk of coronary heart disease.³⁸

Engaging in regular physical activity can increase metabolism of fats and carbohydrates, increase artery diameter and coronary blood flow, and lower blood pressure, thereby resulting in reduced cardiovascular morbidity and mortality.³⁹ For adults and elders, at least 150 minutes of moderate–intensity aerobic physical activity weekly or at least 75 minutes of vigorous–intensity aerobic physical activity weekly is recommended by WHO.⁴⁰ In contrast, physical inactivity poses physiological influence on cardiovascular system, leading to increased risks for coronary heart disease, myocardial infarction, hypertension, stroke, obesity, hyperlipidaemia, diabetes mellitus, and sudden death.^{41,42}

2.1.2 Health behaviours and health in Central and Eastern Europe

(1) Mortality gap between Eastern Europe and Western Europe

Since the 1970s, life expectancy continued to rise in Western Europe while it began to fall in CEE. After 1989, there was more divergence in mortality within CEE countries; life expectancy continued to decrease and fluctuate in Russia and Former Soviet Union, but it increased gradually in the Czech Republic, Poland, Hungary, and Slovenia (Figure 2.1). In the mid–1990s, there was a life expectancy gap of 6 years for men and 5 years for women between Eastern and Western Europe. This East–West mortality gap was mainly attributable to cardiovascular diseases (54%) and external causes of death (23%) in middle–aged populations, particularly in men.⁴³ Figure 2.1 Trends in life expectancy at birth (years) for selected countries (1970–2010). Sources: WHO Regional Office for Europe (2011).⁴⁴



Bobak and Marmot (1996) proposed that the East–West mortality gap might be mainly explained by health behaviours (e.g. drinking, smoking or diet) and psychosocial factors (e.g. work stress, perceived control, or social support). Material factors (e.g. environmental pollution or poor medical care) might contribute to only 20% of this gap. Psychosocial factors are important in determining the inequalities in health among and within countries.⁴⁵ Between CEE countries, the mortality changes after 1989 were associated with both changes in income and income inequality. In poor countries, income was associated with mortality primarily via material factors (e.g. malnutrition or unclean water); in rich countries, mortality was more strongly related to income inequality mainly via psychosocial factors.⁴⁶ Within CEE countries, the social gradient in morbidity and mortality has been identified as the gradient observed in the Whitehall II study for British civil servants.⁴⁷ A gradient in mortality among those not poor argues for the importance of psychosocial factors linked to social position.⁴⁸

McKee and Shkolnikov (2001) indicated that the leading causes of high mortality in CEE were injury, violence and cardiovascular disease, particularly in men before age 65. High alcohol consumption, binge drinking, smoking and poor nutrition were considered important underlying factors. As men with least educational levels and least social support were affected the most, they suggested that psychosocial factors might play pivotal roles in explaining health behaviours and mortality gap.⁴⁹

Mackenbach et al (2008) reported that mortality and poor self-rated health were substantially higher in groups of lower socioeconomic status among 22 European countries, but the magnitude of inequalities between higher and lower socioeconomic status was much larger in CEE countries. The authors reported that these country variations were attributable to causes of death related to health behaviours (smoking or drinking) and quality of medical care; psychosocial factors were not measured in this study.⁵⁰ In summary, health behaviours appear to be major pathways that partially explain the East–West mortality gap.

(2) Health behaviours and health in Central and Eastern Europe

In general, CEE has higher prevalence of risky health behaviours than Western Europe. From 1960s to 1980s, there had been a dramatic rise in alcohol consumption and cigarettes smoking in CEE.⁵¹ Between 1995 and 2005, Eurocadet Project across

30 European countries found that the highest daily consumption of alcohol per capita was in men in the Czech Republic (56.9 g/l), followed by Luxembourg and Hungary (46.1 g/l). The highest prevalence of current smokers was in men in Latvia (61.6%), Lithuania (58.8%) and Estonia (55.1%). The lowest levels of physical activity were among women in Hungary, Poland and the Czech Republic, and among men in Hungary, Estonia and Bulgaria. In terms of low fruit and vegetable consumption (< 150 g/d), Slovakia, the Czech republic and Latvia were among the fourth to sixth in Europe.⁵² Notably, this project excluded Former Soviet Union, which might have even worse profiles of health behaviours than other CEE countries (e.g. 63.0% of Russian men were current smokers in 2004).⁵³

The evidence on the links between four health behaviours and health in CEE is summarized. In terms of alcohol drinking, there has been strong evidence suggesting that drinking explained the mortality fluctuations in Russians over the past 20 years. Gorbachev's anti–alcohol campaign (1984 to 1987) was associated with increased life expectancy by 3.2 years for men and 1.3 years for women.⁵⁴ However, increased alcohol consumption was associated with rising mortality after the 1990s during social transition.⁵⁵ Several studies found that heavy drinking and binge drinking increased cause–specific mortality in cardiovascular disease, accident, violence and liver cirrhosis in Russia.^{56,57} Bobak et al found that Russian men had higher prevalence of binge drinking and alcohol–related problems than Czech and Polish men, despite lower annual intake of alcohol.⁵⁸ In the Czech republic, Poland and Hungary, the temporal relationships between social policies (e.g. anti–alcohol campaign), alcohol consumption and mortality were also observed in several longitudinal studies.^{59,60}

With regards to smoking, it has been reported that smoking accounts for 30% of all deaths at 35–69 years old and 14% at older ages in CEE; mortality from smoking– attributable diseases in Russia is among the highest of the world.⁶¹ From 1960 to 1989, trends in cigarette sales were associated with mortality rates from lung cancer in CEE – the highest rates in Europe.⁶² In publications from 2006, mortality rates from lung

cancer among male current smokers in CEE (20% to 28%) were significantly higher than those in Western Europe (16%).⁶³ In Russia, the dramatic rise of smoking prevalence during the 1990s and 2000s was explained by expansion of trans--national tobacco companies.⁶⁴ Perlman and Bobak reported that current smokers had increased risks of mortality in men (OR= 1.80) and women (OR= 2.63) in the Russia Longitudinal Monitoring Survey.⁶⁵ In the Czech Republic, smoking prevalence in men declined from 50% in 1985 to 44% in 1992, and this decline was accompanied by reduced cardiovascular mortality at the same period.⁶⁶ Notably, smoking prevalence among women in CEE countries has been increasing since the 1980s.⁶⁷

In terms of diet, high consumption of saturated fat and low intake of fruit/vegetable are common in CEE, probably due to poverty, social norms or winter shortages of food.⁶⁸ Between 2002 and 2005, Boylan et al found higher intakes of saturated fat and sugar but lower intakes of fruit/vegetable in Russia, the Czech Republic and Poland compared to WHO dietary guidelines for the prevention of chronic diseases.^{69,70} In Russia, most people maintained adequate levels of nutrition, but their dietary patterns were characterized by high levels of animal fat, low levels of high–quality protein, and low intakes of fruit and vegetable.⁷¹ In Poland and the Czech Republic, dietary patterns were similar to Russia before 1989; however, ecological studies found that changes in diet quality (e.g. decreased consumption of carbohydrate and saturated fat and increased intake of fruit/vegetable) were associated with decline in cardiovascular mortality in the late 1990s.^{72,73}

With regards to physical activity, Steptoe and Wardle showed that 70% of young adults in Eastern Germany, Poland and Hungary had lack of regular exercise (sports or physically active games less than 4 times over past 2 weeks) compared to 64% of Western counterparts.⁷⁴ Palosuo reported that 43% of men and 59% of women in Russia had leisure–time exercise less than once a month, compared to 12% of Finnish counterparts.⁷⁵ In Russia, only 21% of men and 12% of women engaged in regular leisure–time exercise (sports) in the 1990s.⁷⁶ A national survey in Poland showed that

only 14% of the population reported regular leisure–time exercise in the 1990s.⁷⁷ In the Czech Republic, a national survey reported that only 27% of men and 18% of women engaged in regular leisure–time exercise in the 1990s.⁷⁸

(3) Social determinants of health behaviours in Central and Eastern Europe

Cockerham explained the patterns of health behaviours in CEE by the social contexts under communist regimes and social transformations after 1989.⁷⁹ Using the works of sociologists Max Weber and Pierre Bourdieu as theoretical grounds, he defined collective patterns of health behaviours as health lifestyles based on life choices from options available according to life chances in social structures.^{80,81} Weber proposed that life choices (from individual) and life chances (from social structures) interact in a dialectical way; individual's life choices are constrained or enabled by life chances based on socioeconomic factors (class and status). Bourdieu proposed that knowledge of social structures produce enduring orientations towards routine actions (habitus – schemes of perception, thought and action). Habitus provides a process assimilating social structures into individual subjectivity; people choose lifestyles without free will, as habitus predisposes them to limited choices.

To demonstrate Cockerham's analyses in CEE, health lifestyles of Russian working–class men are taken for example. Heavy episodic drinking (e.g. high doses of vodka in a short time) was a strong tradition of Russian peasant culture, which spread into cities as industrialization transformed peasants into industrial workers. Traditionally, heavy drinking took place only on holidays, but it gradually became common throughout the year. In social drinking, one is expected to drink as much as the others regardless of one's own will. Social norms and interpersonal dynamics may force one's choice to drink. Similarly, there were social norms for smoking and eating more fat and less fruit and vegetable among these men.⁸²

In socialism, there is the priority of state goals (e.g. military or heavy industry) over personal needs (e.g. health care). This paternalism of the state induced a false

sense of security; people believed that the state would take care of their health. Although the state invested in secondary prevention (e.g. contact with physicians for early detection of diseases), primary prevention (e.g. people take responsibility for their health lifestyles) was not impimented or encouraged. In summary, unhealthy lifestyles of Russian working–class men were determined by habitus derived from the wider society – social norms, experience and reality of class circumstances (deemphasizing the individual and over-emphasizing the state).⁸³

According to sociological theories, Cockerham elaborated social determinants of health behaviours in the CEE contexts. As mentioned earlier, the Eastern–West mortality gap was mainly attributable to psychosocial factors and health behaviours based on epidemiological evidence. Indeed, Cockerham's works implied that various psychosocial factors embedded in social contexts might influence individual's health behaviours. The associations between psychosocial factors and health behaviours will be discussed in detail in the following sections.

2.2 Psychosocial Factors

2.2.1 Psychosocial factors – definition and their impact on health

Psychosocial factors were defined by Hemingway and Marmot (1999) as measurements that potentially link psychological phenomena to social environment and physiological changes.⁸⁴ Martikainen et al (2002) defined psychosocial factors as pertaining to the influence of social factors on an individual's mind or behaviour, and to the interrelation of behavioural and social factors. This definition might have important implications for social epidemiologists and health researchers, because it implies that psychosocial factors can be viewed as: (1) mediators in the effects of social structural factors on individual health outcomes, or (2) modified by the social structures and contexts in which they exist.⁸⁵

There have been numerous empirical studies presenting causal pathways from social structures to individual health.⁸⁶ The macro–level social, political and economic context contributes to unequal distribution of resource, power and prestige within a society, which then affects the meso–level socioeconomic position (social position) defined mainly by occupation, education and income. A social gradient in health across a society was reported in the British Whitehall II study and later replicated across the world; those in higher social position on individual health are mediated by psychosocial, behavioural (health behaviour), material, and biological factors.⁸⁹ In the theoretical framework for social determinants of health (Figure 2.2), psychosocial factors are regarded as mediators in the effects of social position on individual health. Figure 2.2 Theoretical framework for social determinants of health. Source: WHO Commission on Social Determinants of Health (2008).⁹⁰



At micro–level individuals, psychosocial factors can influence health directly via psychobiological processes or indirectly via choices of health behaviours. Psychobiological processes are the pathways via which psychosocial factors stimulate central nervous system activation of autonomic, endocrine, immune and inflammatory responses.⁹¹ First, social processes influence individual's psychological processes, involving the brain structures in limbic system and prefrontal cortex to influence lower neural pathways (hypothalamic–pituitary–adrenal axis). Second, these neural pathways regulate autonomic nervous systems by neurotransmitters (e.g. catecholamine) and hormones (e.g. cortisol), thereby influencing peripheral physiological activities. Finally, physiological consequences are tissue damage in inflammation, inhibited immune function, metabolic changes, and oxidative stress reactions which contribute to a variety of chronic diseases.⁹²

Based on existing literature, psychosocial factors related to health are categorized by Steptoe et al.⁹³ First, chronic stress exposures such as work stress, neighborhood stress, caregiver strain, economic hardship, and life events have been shown to increase risk of cardiovascular disease, depression and poor health.^{94,95,96} In terms of work stress, the Demand–Control model and the Effort–Reward Imbalance model have gained strong support to predict a variety of health outcomes.^{97,98} Second, several personality constructs have been found to reduce (e.g. perceived control or self-efficacy) or increase (e.g. Type A behaviour, hostility or neuroticism) the risk of morbidity and mortality.^{99,100} Third, psychological distress and depression have been reported to be associated with cardiovascular morbidity and mortality.¹⁰¹ In contrast, positive affect was shown to improve health.¹⁰² Fourth, protective aspects of social environment such as social support, social network and social capital have been found to reduce morbidity and mortality.^{103,104}

Although numerous studies have linked each psychosocial factor independently to health outcomes, relatively few studies have attempted to examine potential interrelationships between different psychosocial factors (e.g. work stress and personality) in relation to health behaviours. It is widely recognized that psychosocial factors rarely occur in isolation, so it is important to expand the breadth of studies to address this multiplicity of psychosocial factors – this thesis will focus on this issue.¹⁰⁵

2.2.2 Psychosocial factors and health in Central and Eastern Europe

As mentioned earlier, Bobak and Marmot (1996) proposed that the East–West mortality gap might be mainly explained by health behaviours (e.g. drinking, smoking or diet) and psychosocial factors (e.g. work stress or perceived control).¹⁰⁶ McKee and Shkolnikov (2001) also indicated that health behaviours and psychosocial factors played pivotal roles in explaining the East–West mortality gap.¹⁰⁷ The links between health behaviours and health in CEE have been introduced in Section 2.1.2. In this section, the relations between psychosocial factors and health in CEE are discussed.

Previous research has showed that psychosocial factors such as work stress, depression, low perceived control, and poor social support are serious risk factors for morbidity and mortality in CEE countries.^{108,109} In the following paragraphs, the two psychosocial factors related to this thesis – work stress and perceived control, and their influence on health in the CEE contexts are introduced. Note that the impact of OC on health has rarely been examined in CEE.

(1) Work stress and health in Central and Eastern Europe

The social contexts of working conditions in CEE are described. From 1947 to 1989, the communists adopted centrally planned economies across CEE countries. The socialist enterprise was state–owned and oriented to an input–output plan rather than any market. The enterprise played a role in implementing the state's social welfare policies; efficiency and productivity were not major concerns for managers. This legacy of socialist era may influence employee's passive attitude and low efficiency even in post–communistic period.¹¹⁰

There have been rapid and profound changes in labour markets in CEE since 1989; new economic mechanisms created quite different contexts from communistic regimes. After 1989, labour market differences between CEE and Western Europe were substantial in terms of gross domestic product per capita (GDP – a country's

standard of living) and unemployment rate. From 1990 to 1995, CEE countries which joined European Union later (e.g. Czech Republic or Poland) went through transitional recession with GDP decreased by 4.7%; after 1995, economic began to recover with annual GDP growth at 3.7%. However, Former Soviet Union still lagged behind. Employees often had low wage levels, unstable pay, dual earning careers and holding a third job, partcularly in those with low social position.¹¹¹

In terms of unemployment, cumulated decline in employment from 1990 to 2003 was 17.0% in CEE countries, in contrast to cumulated increase in employment by 7.3% in European Union. Despite growth of productivity in CEE after 1995, restructuring processes led to layoffs of redundant workers, resulting in considerable job losses in agricultural and industrial sectors. High unemployment rates also aggravated job insecurity in active employees.¹¹²

Globalisation led to deregulation of labour markets and increased competition; many organisations undertook restructuring and downsizing.¹¹³ Global division of labour has transformed CEE countries into manufacturing locations as subcontractors for Western European and US firms. They were transformed from unskilled and labor– intensive to skilled and capital–intensive production (e.g. automotive or information technology).¹¹⁴ To be globally competitive in production efficiency, organisations might adopt strategies to meet aggressive production goals, probably resulting in high levels of work stress.¹¹⁵ For example, information technology employees in CEE were found to have stressful work conditions, low wage and temporary employment.¹¹⁶

As the biggest contribution to the East–West mortality gap was from working– aged men, psychosocial factors at work might contribute to this gap.¹¹⁷ Several studies have shown that work stress measured by the DC and the ERI models predicted cardiovascular morbidity and mortality, ^{118, 119} poor self-rated health, ¹²⁰ menstrual pain, ¹²¹ depression, ¹²² and high alcohol consumption in CEE.¹²³ The adverse effects of high ER ratio on health in CEE were at least as strong as those found in Western countries.¹²⁴ Laszlo et al compared the prevalence of job insecurity

(one item in the ERI model) in 23,245 adults among 16 European countries; the highest levels of job insecurity were in Poland (41.7%), Czech Republic (41.0%) and Hungary (40.4%), compared to the lowest in Spain (14.2%) and France (17.6%). Job insecurity was associated with poor self-rated health across European countries.¹²⁵ Therefore, work stress appears to be a serious public health issue in CEE.

(2) Perceived control and health in Central and Eastern Europe

In socialism, there is the priority of state goals (e.g. military or heavy industry) over personal needs (e.g. health care). This paternalism of the state induced a false sense of security; people believed that the state would take care of their health. Although the state invested in secondary prevention (e.g. visiting physician for early detection of diseases), primary prevention (e.g. people take responsibility for their health lifestyles) was not encouraged or implemented.¹²⁶ Perceived control (PC) is defined as: the extent to which one can intentionally produce desired outcomes and prevent undesired ones.¹²⁷ The paternalism of socialism may reduce people's PC for their health behaviours and health.

Since 1989, socio-economic transformation in CEE might challenge one's beliefs about how the world works, which were primarily based on economic and political understanding in communist periods. The past is not an adequate guide to the present, and some people might have low sense of control (PC) over life. The stressors from transformation not only challenged individuals to cope actively and to find opportunities, but also threatened some people to suffer from uncertainty and low PC over life.¹²⁸

Due to transformational contexts in CEE, reserachers were particularly interested in one psychosocial factor – PC and its relation to health. Several empirical studies have examined the relationships between PC and health outcomes in CEE. Bobak et al conducted a cross–sectional survey (n= 5,330) in 7 CEE countries (Russia, Lithuania, Latvia, Estonia, Poland, Czech Republic and Hungary). They found that low PC was significantly associated with poor self–rated health; PC can partially mediate

the effects of material deprivation on poor health.¹²⁹ Lundberg et al observed that the levels of PC in Russia were significantly lower (n= 9,237) than those in Sweden (n= 1,007); lower PC was associated with poorer self-rated health in both countries.¹³⁰

Carlson used data from the 1992 World Value Survey, showing that lower PC was associated with poorer self-rated health within and between 23 national samples of men and women; PC can partially explain the East–West divide in self-rated health.¹³¹ At the population level, Pikhart found lower levels of PC were associated with higher rates of all–cause mortality in 7 CEE countries; Pikhart indeed provided the ecological evidence that group levels of PC predicted population rates of mortality (Figure 2.3). Figure 2.3 Mean levels of control for 7 population samples plotted against all–cause mortality for the countries from which these population samples were drawn. Source: Pikhart (2002).¹³²



2.2.3 Psychosocial factors and health behaviours

Across multiple health behaviours, the risk patterns by social position remain relatively constant: people with higher levels of income, occupation or education tend to engage in fewer risky health behaviours than those with lower social position.¹³³ Social epidemiologists attempted to identify the psychosocial factors linking social position to health behaviours. Sorensen and colleagues in Harvard School of Public
Health proposed a conceptual framework for the social context of health behaviours, which combined the strength of several disciplinary perspectives in the framework of social ecological model (Figure 2.3). First, they adopted the rich tradition of psychological research, building on behavioural theories (e.g. Social Cognitive Models) to identify critical psychosocial factors predicting health behaviours – self-efficacy, attitude or intention.¹³⁴ Second, they incorporated the input from social epidemiology, in which numerous studies have supported the causal pathways from social position to health behaviours. Finally, they identified mediating mechanisms that are important to behaviour change and are potentially modifiable by interventions based on theory and research.



Figure 2.4 Conceptual framework for social contexts of health behaviours. Source:

Social ecological model has become a main model for health behaviours since the 1980s, when the limited effectiveness of individual interventions to change health behaviours led to a paradigm shift to consider broader social contexts in which people live and work. Social ecological model provides a comprehensive framework for understanding the multiple and interacting determinants of health behaviours and can be used to develop interventions targeting at several levels. A general acceptance of this model is reflected by international authorities guiding public health.^{136,137} The multi-level determinants of health behaviours are intrapersonal, interpersonal, organisational, community, and public policy levels. Public policy level includes local and state policies and laws that regulate or support healthy actions for disease prevention. Community level involves social networks and norms, which exist among individuals, groups and organisations. Interpersonal level refers to interpersonal processes – family and friends that provide social identity and social support.¹³⁸ Organisational and intrapersonal levels are introduced in more details below.

Organisational level means the level of workplace; an organisation is a group of people intentionally organized to accomplish an overall set of goals (e.g. products or services). Given how important work is in everyday life, workplace is both a resource for health promotion and a source of stress exposure influencing health. Since the 1970s, organisational policies and practices are often the target of health promotion. For examples, worksites may offer smoke–free office buildings, smoking cessation classes, facilities for physical activity, healthy food served at cafeterias, or health examinations for employees.¹³⁹ Most large corporations now provide health promotion programs for employees. In contrast, accumulating evidence has shown that work stress can influence health directly via psychobiological processes or indirectly via health behaviours (for more details, see Section 2.3).¹⁴⁰

Intrapersonal level refers to the level of psychological factors in an individual. Although there have been a wide range of psychological models predicting health behaviours, no single theoretical model has dominated research and practice in health behaviours. Glanz et al conducted reviews of publications from 1986 to 2005; they found that the most commonly used theories of health behaviours at intrapersonal level are Social Cognitive Models (SCMs). SCMs specify cognitive and affective factors as proximal determinants of health behaviours; the common construct across all SCMs is self-efficacy (a component of PC), belief that one can successfully perform the behaviour.¹⁴¹ In addition, specific personality traits (Type A behaviour, Neuroticism or Hostility) have received empirical support to predict health behaviours. Various personality factors – personality traits and social–cognitive constructs have been found to predict health behaviours (for more details, see Section 2.4).

The distinction between personality traits and social-cognitive constructs (e.g. PC or self-efficacy) is introduced here. By integrating diverse perspectives in psychology, "personality" is defined by Carver and Scheier as: a dynamic organisation, inside the person, of psychophysical systems that create the person's characteristic patterns of behaviours, thoughts and feelings. First, characteristic patterns suggest continuity and consistency uniquely identified in an individual. Personality traits - proposed by dispositional, biological, and psychoanalytic perspectives are biologically based temperaments less susceptible to the influence of environments. Second, dynamic organisation implies ongoing readjustment and adaptation in an individual; social learning, cognitive, socio-cognitive, and humanistic perspectives view personality as an accumulated set of thoughts and behaviours learned from environments.¹⁴² Social-cognitive constructs (expectancy or appraisal) are more susceptible to the influence of environments (e.g. social position or work stress).^{143,144} Social-cognitive constructs provide a more active and specific process account of individual differences that complements the broader and more static description in personality traits; they provide better prediction for behaviours and more modifiable targets for interventions.¹⁴⁵

In social ecological model, there has been consistent support for the multi–level influences (independent effects of various psychosocial factors) on health behaviours, but the interactions between different levels – such as organisational level and intrapersonal level deserve further research.¹⁴⁶ In this thesis, the dimension between organisational and intrapersonal levels will be addressed. To further elaborate this issue, literature review will particularly focus on work stress, personality constructs

(personality traits and social-cognitive constructs), and their influences on health behaviours in the following sections.

2.3 Work Stress and Health Behaviours

For most population in early, middle and old adulthood, work plays a significant role as it is generally a prerequisite for a regular income, an opportunity for learning and achievement, and a variety of life opportunities. It is mainly through work that core social identity outside family and social status are acquired. Thus, the qualities of work conditions in terms of prospect and security are crucial for health and well-being.¹⁴⁷ In contrast, physical and psychological hazards at work may lead to adverse health consequences. With technological progression, the nature of work has undergone fundamental changes from industrial production to service sector; current jobs are often sedentary works involving information processing and coordination, rather than physically strenuous works. Thus, work stress becomes a central concern in modern societies; most employees are exposed to psychological demands rather than physical hazards.¹⁴⁸ Among theoretical models of work stress, the Demand-Control (DC) model and the Effort-Reward Imbalance (ERI) model have gained considerable attention and support.¹⁴⁹

2.3.1 The Demand–Control model

Karasek and Theorell traced the history of modern work patterns to industrial revolution in the 19th century. The dramatic changes of work structures generated the political and economic power of enterprises, which exert strict control over employees' work processes. Karl Marx (1867) indicated the alienating and dehumanizing nature of work patterns. Frederick Taylor (1911) wrote "Principles of Scientific Management"; workers' tasks were simplified into element skills required, which were recombined into

complex tasks by a machine–paced assembly line. A set of managers and engineers control workers' behaviours in a way that coordinate their specialized tasks. This precise division of labour can increase work speed and produce maximum profit; however, workers may lose control and get high demands, as Karasek confirmed by empirical research in machine–paced operatives and low–status service operatives.¹⁵⁰ Since the 1960s, social movements against work conditions occurred in United States and Europe. Nevertheless, the new division of labour was formed on a global scale, with similar work patterns replicated across the world.

Karasek and Theorell proposed the DC model based on accumulating literature on psychological demands and control. In the research on demands, Selye (1936) proposed a U–shaped association between demands and performance; some level of demands is necessary for effective performance and job satisfaction, but higher demands are disastrous.¹⁵¹ Hinkle (1968) conducted the first prospective study on the link between high demands and the risk of myocardial infarction.¹⁵² Since then, there have been a growing number of studies on psychological demands and health.¹⁵³ In the research on control, Karasek attributed his concepts to the similar origins as perceived control: White's effectance motivation, Rotter's locus of control, and Bandura's self-efficacy (for more details, see Section 2.4.3). He emphasized social learning processes: active learning occurs in situations requiring high demands and high control in order to choose how best to cope with a new stressor.

The DC model is initially composed of two dimensions. One dimension is control (decision latitude) indicating employees' control over their tasks and how these tasks are executed. Control consists of skill discretion (a variety of tasks, low repetitiveness, occasions for creativity, and opportunities to learn new things) and decision authority (ability to make decisions about their own job, and ability to influence team and company). Another dimension concerns psychological demands representing psychological stressors in work environment (time pressures, pace of work, interruption rate, conflicting demands, amount of work, degree of concentration

required). Interactions of high and low levels of control and demands generate four psychosocial work characteristics. Job strain (high demands and low control) is the worst situation; when individuals have low control to cope with overwhelmingly high demands, the stress would produce adverse health outcomes. Passive job (low demands and low control) is the second worst situation; employees become passive at work. Active job (high demands and high control) is a favorable situation; when individuals have high control to cope with high demands, they can learn actively and develop competence to deal with challenges. Low strain (low demands and high control) is a favorable situation regarding health outcomes.

Figure 2.5 Low control, high demand and low work support cause adverse health outcomes. Source: Adapted from Johnson and Hall (1988).¹⁵⁴



Johnson and Hall added a third dimension – workplace support from supervisors and colleagues into the original framework, because accumulating research showed that social support can influence health outcomes or buffer stress–health relationships. Social support generally includes instrumental and emotional support. Instrumental support is extra resources or direct assistance in work tasks given by supervisors or colleagues. Emotional support is social and emotional integration and trust between colleagues and supervisors. The combination of low control, high demands, and low workplace support (called iso-strain) is a stressful psychosocial work condition producing adverse health outcomes (Figure 2.4).

2.3.2 The Effort–Reward Imbalance model

The ERI model was proposed by Johannes Siegrist (1990) based on medical sociology and the concept of social reciprocity, a basic principle of social exchange process. Social reciprocity lies at the core of employment contract, which assumes that tasks and obligations to be performed by employees in exchange for adequate rewards from employers. Contractual reciprocity operates via norms of return expectancy. Violation of reciprocity in terms of high effort and low reward can elicit strong negative emotions, which subsequently trigger sustained stress responses (sustained autonomic and neuroendocrine activation) involved in the pathogenesis of coronary heart disease and adverse health outcomes. In contrast, adequate reward can promote positive emotions, well-being and good health.¹⁵⁵ Notably, the ERI model particularly emphasizes the social roles of work; the workplace provides opportunities to acquire self-regulatory needs in terms of self-efficacy (e.g. successful job performance), self-esteem (e.g. recognition form supervisors or colleagues), and self-integration (e.g. belonging to a social group).¹⁵⁶

The ERI model is composed of three dimensions. The first dimension is extrinsic effort that represents time pressure, interruptions, responsibility, overtime work, physical demands, and increasing demands. The second dimension is reward that represents salary, esteem (respect from superiors, colleagues and work, adequate support, and unfair treatment) and social status control (promotion prospects, adequate position, adequate work prospects, undesirable change, and job security). Effort–Reward ratio is the ratio of the score for extrinsic effort (E) to the score for reward (R). ER ratio > 1 is considered to indicate effort–reward imbalance (Figure 2.5).

The ERI model incorporated a personality dimension into a situational model. Overcommitment (OC) is a personality construct reflecting the personal need for control in dealing with work demands. A high OC person displays a pattern of attitudes, behaviours, and emotions characterized by an excessive striving at work and a strong motivation for esteem and approval at work. High OC people strive towards high achievement, have difficulty withdrawing from work, and maintain excessive effort under inadequate reward, thereby resulting in prolonged non–reciprocal exchange.¹⁵⁷ Figure 2.6 High effort and low reward at work cause adverse health outcomes. Source: Adapted from Siegrist (2000).



The following three hypotheses are derived from the ERI model: (1) Extrinsic ERI hypothesis: the mismatch between high effort and low reward can lead to adverse health outcomes. (2) Intrinsic effort hypothesis: a high level of OC per se can increase the risk of adverse health outcomes; that is, there is a direct effect of OC on health. (3) Interaction hypothesis: those who are characterized by both conditions (1) and (2) have even higher risks of adverse health outcomes; that is, there is, there is interaction between OC and ERI.¹⁵⁸

Some researchers argue that the ERI model has more general explanatory value than the DC model in current working contexts. The DC model was initially addressed in blue–collar industrial workers and it reflected social concerns emphasizing workers' control in the 1970s.¹⁵⁹ In this era of globalization, tight managerial control in industry is shifted to flexibility, self-regulation and decentralization.¹⁶⁰ Empirical evidence supports that the ERI model emphasizing "reward" in career prospects and esteem is more sensitive in explaining work stress in modern occupations than the DC model.^{161,162} In addition, although the two models overlap to some extent in "extrinsic effort" and "demands", they identify different aspects of work stress in "reward" and "control". Evidence has supported that combination of the ERI and the DC models, or at least dimensions of them (e.g. ERI and control), can produce stronger predictive power on health outcomes than adopting either model alone.¹⁶³

2.3.3 Empirical studies on work stress and health behaviours

Many studies have showed that the DC and the ERI models can predict mental disorders (e.g. depression, anxiety), cardiovascular diseases (e.g. coronary heart diseases, hypertension, stroke), metabolic diseases (e.g. diabetes, hyperlipidaemia), musculoskeletal diseases, poor self-rated health and sickness absence.^{164,165,166} For example, Kivimäki et al conducted a meta-analysis on 13 European cohort studies; the hazard ratio for coronary heart disease was 1.23 (95% CI= 1.10–1.37) for job strain versus no strain.¹⁶⁷ Stansfeld and Candy conducted a meta-analysis on 11 cohort studies; the effects of job strain (OR= 1.82, 95% CI= 1.06–3.10) and ERI (OR= 1.84, 95% CI= 1.45–2.35) were substantial on depression and anxiety.¹⁶⁸ Van Vegchel et al reviewed 45 studies and found that most studies supported predictive validity of ER ratio for various health outcomes.¹⁶⁹

The impacts of work stress (psychosocial factor) on health are mainly explained by direct psychobiological processes or by indirect pathways via health behaviours.

Siegrist reviewed 46 studies and found moderate support for consistent associations between the DC/ERI models and health behaviours; relatively strong relationships were found in alcohol drinking and overweight (a proxy measure for diet and physical activity) in men.¹⁷⁰ Heikkilä et al conducted a meta-analysis based on 11 European cross–sectional studies (n= 118,701) and 4 cohort studies (n= 43,971); individuals with job strain were more likely than those with no strain to have co–occurrence of several health behaviours– heavy drinking, current smokers, physical inactivity and overweight (OR= 1.25, 95% Cl= 1.12–1.39).¹⁷¹ Kouvonen et al reported the dose–response relationship between the extent of work stress (measured by the DC/ERI models) and the number of health behaviors (alcohol, smoking, physical inactivity, and overweight) in the Finnish Public Sector Study (n= 36,127).^{172,173}

The potential mechanisms linking work stress to health behaviours are suggested based on existing evidence. In terms of biological pathways, work stress might lead to biological responses (e.g. dysfunction of mesolimbic dopamine system in the brain), which cause substance addictions (alcohol drinking or smoking).¹⁷⁴ Work stress can influence physiological responses (e.g. increased activities of hypothalamus– pituitary–adrenal axis and elevated levels of cortisol and insulin), resulting in food choice towards high-fat and high-carbohydrate content.¹⁷⁵ In terms of psychological pathways, work stress has been found to predict psychological distress like anxiety and depression; individuals might engage in risky health behaviour (emotion-focused coping) to temporarily relieve or avoid their psychological distress and to distract their attention from stressful situation.¹⁷⁶

In the following paragraphs, empirical evidence on the relationships between the DC/ERI models and the four health behaviours will be discussed.

(1) Work stress and alcohol drinking

In terms of the DC model, Heikkilä et al conducted a meta-analysis on 12 crosssectional (n= 142,140) and 4 longitudinal studies (n= 48,646) on the DC model and alcohol drinking. Compared to moderate drinkers (1–210 g/week of ethanol in men; 1– 140 g/week in women), heavy drinkers (>= 280 g/week in men; >= 210 g/week in women) had significantly higher odds of job strain (OR= 1.12, 95% CI= 1.00-1.26).¹⁷⁷ Siegrist also found 4 out of 6 longitudinal studies supporting the link between the DC model and alcohol consumption.¹⁷⁸ In general, the moderate associations between the DC model and alcohol drinking have received support. Most of studies reported that job strain (high demands and low control) was associated with high alcohol consumption or alcohol abuse.^{179,180} However, several studies had slightly different findings. For example, Gimeno et al reported that passive job (low demands and low control) was related to heavy drinking in a US cross–sectional study (n= 3,099).¹⁸¹ Niedhammer et al found that low control was associated with high alcohol consumption in men, but low work support was related to high alcohol consumption in women in a French cross–sectional survey (n= 20,625).¹⁸² It is possible that the associations between the dimensions of work stress and drinking are relatively specific for each study population, thereby leading to differences between various studies.

In terms of the ERI model, several studies have showed promising results to support the links between high ER ratio and drinking outcomes. Head et al reported that high ER ratio was associated with alcohol dependence in men (OR= 1.93) after adjustment for age and employment grade in British Whitehall II cohort study (n= 7,372); this association in women was not as remarkable as that in men. In contrast to men, women with higher employment grade tended to drink more.¹⁸³ Puls et al found that ERI was associated with high alcohol consumption in a German cross–sectional study.¹⁸⁴ Bobak et al conducted a cross–sectional study in men (n= 694) in 3 CEE countries; they found that high ER ratio was associated with binge drinking (OR= 1.36), problem drinking (OR= 1.37), negative consequences of alcohol (OR= 1.22), high annual intake of alcohol (OR= 1.29), and high annual number of drinking sessions (OR= 1.34).¹⁸⁵ To establish stronger evidence on the link between the ERI model and drinking, more longitudinal studies are still needed.

(2) Work stress and smoking

In terms of the DC model, Albertson et al reviewed 22 prospective studies on the DC model and smoking outcomes. There was strong evidence for the effect of high demands on smoking intensity (the amount smoked) in current smokers. High control increased the probability of smoking cessation, and high demands increased the probability of smoking cessation, work social support was positively associated with smoking cessation and negatively associated with smoking intensity and relapse. Heikkilä et al conducted a meta-analysis on 15 cross–sectional data (n= 166,130) and 6 longitudinal data (n= 52,024) from European studies. Current smokers had higher odds of job strain than never–smokers (OR= 1.11, 95% Cl= 1.03–1.18); there was no difference in job strain between ex–smokers and never–smokers. For smoking intensity, current smokers with job strain smoked three cigarettes more per week than those without job strain. However, there was no clear evidence for longitudinal associations between job strain and taking up or quitting smoking (changes in smoking status).¹⁸⁷

In terms of the ERI model, Kouvonen et al reported that high ER ratio was associated with being current smokers (OR= 1.28) in a Finnish cross–sectional study (n= 46,190); among current smokers, high ER ratio was associated with high smoking intensity (OR= 1.19).¹⁸⁸ Peter et al observed a positive association between ER ratio and smoking intensity in a German cross–sectional study of middle-aged men.¹⁸⁹ In addition, an Australian cross–sectional study (n= 1,101) showed that higher ER ratio was associated with higher smoking intensity in women, but not in men.¹⁹⁰ However, Ota et al found that ER ratio at baseline did not predict smoking cessation at 2–year follow-up in 1,423 middle-aged men in Japan.¹⁹¹ Despite promising results to support the associations between ERI and smoking outcomes in several cross–sectional studies, more longitudinal studies are needed to provide better evidence.

(3) Work stress and diet

The associations between chronic stress and dietary outcomes have been extensively studied in epidemiological and laboratory studies.^{192,193} In terms of the DC model, Hellerstedt and Jeffery found that high demands were associated with high fat intakes in men, but not women, in a US cross-sectional study (n= 3,843).¹⁹⁴ In a British cohort Study (n= 3,397), passive job was associated with unhealthy food habits (e.g. not eating vegetables/fruit at least twice a day, or not choosing wholegrain bread and low-fat milk).¹⁹⁵ In a Finnish cohort study (n= 6,243), low job strain was associated with a healthier diet (fresh vegetables and fruit daily, whole grain bread daily, fish at least twice a week, using vegetable-based margarine, and usually using oil in cooking) among women, but not men.¹⁹⁶ A Finnish cross-sectional study (n= 6,369) reported that job strain was associated with frequent use of packed meals (less in line with nutritional recommendations) among men.¹⁹⁷ In Japan, job strain was positively associated with more fat and cholesterol intakes among men in a cohort study (n= 25,104);¹⁹⁸ job strain was associated with less vegetables (n= 6,759) and more calorie intakes (n= 1,183) in two cross-sectional studies.^{199,200} However, no significant association between job strain and diet was reported in a cohort study in Netherlands (n= 3,309).²⁰¹

Despite empirical support on the link between the DC model and diet, no research is available on the ERI model and diet. The ERI model has been found to predict other health behaviours (drinking or smoking), and it is reasonable to suggest a potential link between ERI and diet. In addition, the measurements of dietary outcomes varied considerably between previous studies (e.g. foods, nutrients, or food habits); those studies with less precise measures (e.g. fewer food items) should be interpreted cautiously. The method of diet quality takes into account the intakes of various foods and nutrients, thereby providing more accurate pictures of diet than single food/nutrient intake. Diet quality is often defined by the adherence to dietary guidelines associated with health outcomes, such as the WHO guidelines for the prevention of

chronic diseases.²⁰² Thus, empirical studies regarding work stress and diet quality would provide more solid evidence for this topic.

(4) Work stress and physical activity

With regard to the DC model, Fransson et al conducted a meta-analysis based on 14 European cohort studies (n= 170,162); in prospective analyses, the odds for physical inactivity during leisure time were higher for those with job strain (OR= 1.21, 95% CI= 1.11-1.31) and passive job (OR= 1.20, 95% CI= 1.11-1.30) compared to those with no strain.²⁰³ Kirk and Rhode conducted a systematic review and found that job strain was related to physical inactivity during leisure time in 6 out of 8 cross– sectional and prospective studies.²⁰⁴ For example, Lallukka et al reported that job strain was prospectively associated with physical inactivity (OR= 1.88) among white– collar men in British Whitehall II Study (n= 3,397) and women (OR= 1.25) in Helsinki Health Study (n= 6,070).²⁰⁵ Gimeno et al found that passive job was related to physical inactivity at 5–year follow-up (OR= 1.16) in men in Whitehall II Study (n= 4,291).²⁰⁶

In terms of the ERI model, Kouvonen et al reported that high ER ratio was associated with physical inactivity among women (OR= 1.08) and men (OR= 1.17) in a Finnish cross–sectional study (n= 35,918).²⁰⁷ In contrast, Kuper et al reported that higher ER ratio was prospectively associated with more physical activity measured by time spent in moderate to vigorous activity in Whitehall II study.²⁰⁸ It is implied that high ERI might decrease physical activity in some people (e.g. those prone to depression) but increase physical activity in others (e.g. those with active coping adopt physical activity to reduce stress).²⁰⁹ Further evidence is needed regarding the relationship between the ERI model and physical activity.

In summary, there has been adequate support for the moderate associations between the DC model and four health behaviours. Additionally, existing studies have shown promising results to support the relationship between the ERI model and health behaviours, but no literature is available on ERI–diet relationship. There are some

limitations to the generality of existing literature on work stress and health behaviours. First, there have been relatively few longitudinal and intervention studies in this topic, and further evidence with methodology better than cross–sectional design is needed. Second, the associations between the work stress and specific health behaviours varied across sexes and populations; this inconsistency may be explained by sex and other psychosocial factors (mediators or modifiers) like personality constructs.²¹⁰ To predict health behaviours more accurately, it is helpful to incorporate personality constructs into the investigation of relationships between work stress and health behaviours.

2.4 Overcommitment Personality and Health Behaviours

In the ERI model, Siegrist proposed the personality construct of overcommitment (OC) which describes individual attitudes, behaviours and emotions reflecting excessive work–related striving and high need for approval and esteem; they have difficulty withdrawing from work and maintain excessive effort under inadequate reward.²¹¹ Siegrist initially described OC as "need for control" – a distinct individual pattern of coping with work demands, which evolved from Type A behaviour.²¹² Rosenman traced the origins of Type A behaviour to perceived control (PC) in psychology; Type A persons have higher need for control over environment and tend to perceive lower PC, and their response is enhanced coping to assert and maintain control over environment.²¹³ Thus, the origins of OC might be traced to Type A behaviour (see Section 2.4.1) and PC (see Section 2.4.3).

Siegrist and colleagues initially assessed "need for control" by a 29–item scale with 6 dimensions (need for approval, competitiveness, disproportionate irritability, inability to withdraw from work, hard work, and perfectionism).²¹⁴ However, several studies cannot replicate the factorial structure of need for control.^{215,216} A shorter version of OC score was then developed by exploratory and confirmatory factor

analyses, consisting of inability to withdraw from work (5 items) and disproportionate irritability (1 item). In terms of internal consistency, coefficient alpha for the 6-item OC measure were ranged from 0.79 to 0.82 in the European samples.²¹⁷ In terms of discriminate validity, OC score correlated very weakly with other Big Five personality traits except Neuroticism.²¹⁸

Personality psychology is in debate over broad versus specific personality traits. Although more and more specific personality traits have been identified, new personality constructs are suggested to be compared with existing personality traits for a possible common core.²¹⁹ Due to theoretical links, OC may overlap with other personality traits related to Type A behaviour – Hostility (negative attitude and mistrust towards others, predisposing a person to anger and aggression) and Neuroticism (stable and pervasive individual differences in the tendency to experience negative emotions). Researchers proposes that OC represents an aspect of Neuroticism manifested in the work context. In empirical studies, OC correlated significantly with Type A behaviour (r= 0.39) and Neuroticism (r= 0.30–0.38).^{220,221,222} Neuroticism correlated significantly with Hostility (r= 0.66) and Type A behaviour (r= 0.34).^{223,224}

2.4.1 Overcommitment – origins from Type A behaviour

The origins of OC personality are traced to Type A behaviour and PC. In this section, Type A behaviour and related personality traits (Hostility and Neuroticism) are introduced, and their relationships with health outcomes are summarized.

(1) Type A behaviour

Historically, research on the impact of personality on health had a rich and long tradition in medicine and psychology; Type A behaviour has been one of the most influential constructs in the studies regarding psychosocial factors and health. Friedman and Rosenman (1959) firstly described Type A behaviour as an emotion–

action complex characterized by hostility and aggression, sense of time urgency, competitiveness, and ambitiousness.²²⁵ Rosenman et al (1975) conducted the Western Collaborative Group Study in 3,524 employed men aged 39–59 years old, a prospective study with a follow–up period of 8.5 years. The study found that Type A behaviour was strongly associated with the incidence of coronary heart disease (CHD) (ORs= 1.87 in younger group and 1.98 in older group) after adjustment for classical risk factors (diabetes, blood pressure, smoking or blood lipids).²²⁶ Subsequently, these results have been replicated in the Framingham Heart Study and other prospective studies.²²⁷ In the late 1970s, the review panel of US National Heart, Lung and Blood Institute endorsed Type A behaviour as an independent risk factor for CHD.

However, since the late 1980s, several studies failed to show any association between Type A behaviour and CHD. For examples, a longer follow–up of Western Collaborative Group Study observed no association between Type A behaviour and CHD mortality.²²⁸ In the Framingham Heart Study, Type A behaviour was associated with incidence of angina pectoris, but not myocardial infarction or fatal cardiac events.²²⁹ Therefore, researchers suggested that previous evidence on Type A behaviour and CHD should be interpreted more cautiously. Several explanations for inconsistent associations between Type A behaviour and CHD were raised; one main explanation is that the global Type A construct is too broad, and only specific components may be pathogenic to CHD. Hostility has been identified as the "toxic" element in Type A behaviour, as this element can predict CHD most strongly.²³⁰

(2) Hostility

Hostility is a multi–faceted construct incorporating cognitive (cynicism and negative beliefs about human nature), affective (anger, annoyance, and resentment), and behavioural components (aggression, antagonism and uncooperativeness). These attitudes and cognitions predispose a person to intensive emotion (anger) coupled with physiological arousal, leading to verbal or physical aggression.²³¹ There

are several widely–used measures of Hostility (e.g. Buss–Durkee scale or Cook– Medley Scale).^{232,233} A large number of prospective studies and meta-analyses have supported the association between Hostility and CHD.^{234,235} A meta-analysis by Chida and Steptoe found that Hostility was associated with more CHD events in the 25 healthy population studies (hazard ratio= 1.19; 95% CI= 1.05–1.35) and with poor prognosis in the 19 CHD population studies (hazard ratio= 1.24; 95% CI= 1.08–1.42). Notably, the harmful effects of Hostility on CHD events in men were greater than those in women.²³⁶

There have been several potential pathways via which Hostility might affect the risk of CHD. Firstly, Hostility might simply be a marker for an "inborn structural weakness" of cardiovascular system, which causes both CHD and Hostility. Secondly, Hostility influences the body on a daily basis, forming a pattern of intense responsiveness to physical and mental stressors, which then increases the risk of atherosclerosis.²³⁷ Thirdly, Hostility may have a negative impact on social relationship, resulting in lack of social support.²³⁸ Fourthly, Hostility may be associated with health behaviours – smoking or alcohol consumption.²³⁹ Finally, life course perspective views Hostility as the product of person and environment; low socioeconomic position in childhood and early adulthood can predict high levels of Hostility.²⁴⁰

The Edinburgh Artery Study (1991) of 1,592 community-dwelling people was designed to gather information on risk factors of cardiovascular disease, with the measures of Hostility and Big Five personality traits administered simultaneously; the study found that Hostility was strongly associated with Neuroticism.²⁴¹ Subsequently, Smith's review focused on the problems about measurements of Hostility, reporting that several items in Cook–Medley Hostility Inventory (e.g. cynicism or social avoidance) overlapped with Neuroticism.²⁴² Felsten reported that Neuroticism correlated strongly with Hostility measured by Buss–Durkee Hostility Inventory in men (r= 0.66) and women (r= 0.63).²⁴³ Since that time, researchers have recognized that Neuroticism is a personality trait closely related to Hostility and Type A behaviour.

(3) Neuroticism

Neuroticism or negative affectivity, proposed by Eysenck and Eysenck (1964), reflects the stable and pervasive individual differences in the tendency to experience negative emotional states, including anxiety, anger, guilt and distress.²⁴⁴ High Neuroticism individuals tend to be worried, easily upset, often depressed, and to focus on negative aspects of self, others and the world.²⁴⁵ This dimension has been an established part of the most widely–used model of Big Five personality traits (Table 2.1). Neuroticism has been assessed by several measures: NEO Five-Factor Inventory or Eysenck Personality Questionnaire.^{246,247}

Table 2.1	Big Five personality traits and their dimensions	
	Big i we personality traits and their dimensions	

Personality traits	Dimensions
1. Neuroticism	Anxiety, anger–hostility, depression, self- consciousness, impulsiveness, vulnerability
2. Extraversion	Warmth, gregariousness, assertiveness, activity, excitement-seeking, positive emotions
3. Openness to experience	Fantasy, aesthetics, feelings, actions, ideas, values
4. Agreeableness	Straightforward, trust, altruism, compliance, modesty, and tender mindedness
5. Conscientiousness:	Competence, order, dutifulness, achievement striving, self-discipline, deliberation

An important issue is the conceptualization and measurement of Neuroticism. This global trait includes several more specific characteristics, including depression, anxiety, irritability, anger, or low self-esteem. Scales implying the measurement of specific dimensions are often psychometrically indistinguishable from the measures of broader traits. For instance, individuals with depressive or anxiety disorders scored high on the measures of Neuroticism, and high levels of Neuroticism were associated with increased risk of depressive and anxiety.²⁴⁸ Those studies on the relationships between Neuroticism and various diseases would involve the effects of undiagnosed depressive or anxiety disorders, and vice versa.²⁴⁹

High levels of Neuroticism were associated with adverse psychosocial outcomes, such as higher stress, poor mental health, poor social relationship, poor work performance, counterproductive work behaviour, and occupational injury.^{250,251} Also, many prospective studies have showed that Neuroticism predicted a wide range of health problems, including cardiovascular diseases and all–cause mortality.^{252,253} Finally, numerous studies have indirectly supported the effects of Neuroticism on health; various measurements of specific dimensions (depression, anxiety or low self–esteem) predicted subsequent CHD,²⁵⁴ atherosclerosis,²⁵⁵ diabetes,²⁵⁶ and all–cause mortality in cohort studies.^{257,258}

2.4.2 Empirical studies on Type A behaviour and health behaviours

To my best knowledge, the effects of OC on health behaviours have rarely been reported in empirical studies.²⁵⁹ Due to theoretical links, evidence for the effects of Type A behaviour and related personality (Neuroticism and Hostility) on health behaviours can be used to partially support the links between OC and health behaviours. I summarise empirical studies on the associations of Type A behaviour with four health behaviours – alcohol drinking, smoking, diet, and physical activity.

(1) Type A behaviour and alcohol drinking

Type A behaviour has received moderate support in relation to alcohol drinking. Friedman and Rosenman found that those with Type A behaviour had higher alcohol consumption than Type B behaviour (a behavioural pattern characterized by absence of Type A behaviour; Type B persons tend to be relaxed and easy–going).²⁶⁰ Koskenvuo et al reported that Type A persons drank alcohol more frequently than Type B persons in a Finnish cross–sectional study of 11,364 adults.²⁶¹ In the US cross–sectional study of 12,866 men, Folsom et al reported that those with Type A behaviour consumed 30% more alcohol and drank more frequently than Type B persons.²⁶² In

contrast, Glynn et al observed that Type A behaviour was not associated with high alcohol consumption in a cross–sectional study in the US (n= 1,556 men).²⁶³

Hostility has been found to be associated with high alcohol consumption in numerous large population–based studies. In a Finnish cohort study of 2,125 men aged 42–60 years, Everson et al reported that alcohol consumption significantly mediated the effects of Hostility on mortality and myocardial infarction.²⁶⁴ Pulkki et al found that Hostility assessed at 12–21 years old predicted the frequency of alcohol use after 9 years of follow-up in a Finnish cohort study (n= 1,219).²⁶⁵ Siegler et al found that Hostility assessed at late adolescence subsequently predicted high alcohol consumption in adulthood in the US (n= 4,710).²⁶⁶ Scherwitz et al reported that Hostility was associated with increased alcohol consumption in a cross–sectional study of 5,115 young adults in the US.²⁶⁷ Whiteman et al observed that Hostility was related to high alcohol consumption in a British cross–sectional study (n= 1,592).²⁶⁸

Neuroticism has received moderate support in relation to alcohol consumption. Kuntsche et al found that high Neuroticism was associated with high alcohol consumption in a cross–sectional study of 2,090 university students in Switzerland.²⁶⁹ Almada et al reported that high Neuroticism was associated with high alcohol consumption in the US cross–sectional study of 1,871 middle-aged men.²⁷⁰ However, some studies reported negative findings on Neuroticism and alcohol drinking.^{271,272} A meta–analysis of 124 studies reported that Neuroticism predicted "emotion–focused coping" to minimize negative emotions via emotional expression, withdrawl or avoidance (r= 0.22–0.41), such as alcohol abuse (r= 0.28).²⁷³ Neuroticism is associated with high rates of stress exposure and intense emotional and physiological reactivity to stress, so they tend to minimize unpleasant arousal via avoidance or drinking. As drinking is only one of maladaptive coping strategies used by Neuroticism, effects of Neuroticism on alcohol drinking might not be very strong.²⁷⁴

(2) Type A behaviour and smoking

Type A behaviour has received strong support in relation to smoking. Jenkins et al observed that Type A behaviour was found in 53% of heavy smokers (> 20 cigarettes per day), 47% of light smokers (< 20 cigarettes per day), and 41% of never smokers; Type A behaviour was associated with smoking status after 4 years of follow–up in 2,318 middle–aged men in the US.²⁷⁵ Shekelle et al found that Type A behaviour was positively correlated with smoking intensity (the number of cigarettes smoked per day) in a cross–sectional study of 4,108 adults in the US, but the magnitude of correlation was not large.²⁷⁶ In a Finnish population–based cohort study (n= 1,125) with 9 years of follow–up, Pulkki et al reported that Type A behaviour mediated 28.5% and 20.5% of the effects of low education on smoking in men and women, respectively.²⁷⁷

Hostility has been found to be associated with smoking in several large population–based studies. Siegler et al found that high Hostility measured in young adulthood predicted the risk of being current smokers after 22 years of follow–up in 4,710 people in the US.²⁷⁸ Pulkki et al found that high Hostility measured at 12–21 years old predicted smoking intensity after 9 years of follow-up in 1,219 Finnish people.²⁷⁹ Everson et al found that smoking significantly mediated the effects of Hostility on mortality in a Finnish cohort study of 2,125 middle–aged men.²⁸⁰ Scherwitz et al observed that high Hostility was associated with a 1.5 times higher prevalence of current smokers in the US cross–sectional study of 5,115 adults.²⁸¹ Schrijvers et al found that high Hostility was associated with being current smokers in a cross–sectional study of 3,494 adults in the Netherlands.²⁸² In a British cross–sectional study (n= 1,592), high Hostility was associated with being current smokers.²⁸³

Neuroticism has been moderately supported to be associated with smoking. In a British cohort study (n= 5,362), high Neuroticism measured at age 16 was associated with being current smokers in adulthood.²⁸⁴ Goodwin and Hamilton reported that higher Neuroticism was associated with greater risk of cigarette smoking in the US cross–sectional study of 3,032 adults.²⁸⁵ In a twin study of 1,551 adults in Australia,

the association between Neuroticism and smoking was explained by genetic and environmental sources of co-variation.²⁸⁶ Notably, the meta-analysis of 22 studies published between 1972 and 2001 reported that high Neuroticism was associated with an increased likelihood of being current smokers, but the effect size was modest.²⁸⁷

(3) Type A behaviour and diet

Type A behaviour has received moderate support in relation to dietary outcomes. In a cross–sectional study in Northern Ireland (n= 551), Barker et al found that Type A behaviour in men had moderate but significant associations with intakes of saturated fat and cholesterol. In women, Type A behaviour had a weak association with sugar intake.²⁸⁸ In contrast, Gallacher et al found no association between Type A behaviour and fat intake, but Type A behaviour was associated with low fruit/vegetable intake in a British cross–sectional study of 532 middle-aged men.²⁸⁹ In a cohort study of 10,602 men, Type A behaviour was significantly associated with high consumption of saturated fat, cholesterol, and vegetable in Northern Ireland and France.²⁹⁰

Hostility has been found to be associated with dietary outcomes in several studies. For example, Iribarren et al found that higher Hostility was associated with less intakes of polyunsaturated fat after 2 years of follow–up in 3,581 young adults in the US.²⁹¹ Scherwitz et al observed that higher Hostility was strongly associated with greater caloric intake in a cross–sectional study of 5,115 young adults in the US.²⁹² In the US cohort study of 629 adults, Hostility was associated with less likelihood of monitoring and controlling for dietary patterns after 1 year of follow–up.²⁹³ There has been indirect support for the effects of Hostility on diet–related outcomes; for instance, high Hostility predicedt high body–mass index after 22 years in 4,710 young adults in the US.²⁹⁴

Neuroticism has been reported to be associated with dietary outcomes in several studies. In the Helsinki Birth Cohort Study (n= 1,681), higher Neuroticism was found to be associated with lower fish and vegetable consumption in women.²⁹⁵ De Bruijn et al found that the effects of high Neuroticism on low fruit consumption in 405 adults in

the Netherlands.²⁹⁶ Vollrath et al conducted a cohort study on personality and food consumption in 327 Norwegian children aged 6–12 years old; girls with lower Conscientiousness and higher Neuroticism consumed more sweet drinks, and boys with higher Conscientious and lower Neuroticism consumed more fruits and vegetables.²⁹⁷ Note that literature has indirectly supported the effects of Neuroticism on diet–related outcomes such as increased body–mass index.^{298,299}

(4) Type A behaviour and physical activity

Type A behaviour has generated mixed support in relation to physical activity. Pulkki et al found that Type A behaviour mediated 17.7% of the effects of low education on physical inactivity after 9 years of follow–up in women in a Finnish cohort study (n= 1,125).³⁰⁰ In contrast, some components of Type A behaviuor subsequently predicted high levels of physical activity in 2,031 young adults from the Young Finns Study.³⁰¹ Abbott et al observed that those with Type A behaviour perceived themselves to be more physically active, even though objective estimates of physical activity were not associated with Type A behaviour.³⁰²

Hostility has received moderate support to predict physical inactivity. Schrijvers et al observed that Hostility was associated with physical inactivity in a cross–sectional study of 3,494 adults in the Netherlands.³⁰³ Maier and James found that greater Hostility was associated with lesser physical activity in 859 college students in the US.³⁰⁴ In a 9–year follow–up study of 2,125 middle-aged men in the US, physical inactivity was found to mediate the effects of Hostility on mortality.³⁰⁵ In a 7–year follow–up study of 1,022 adults in the US, physical inactivity was found to significantly mediate the impacts of Hostility on recurrent CVD events.³⁰⁶ However, there have been several studies with negative findings on Hostility and physical inactivity.^{307,308}

Neuroticism has received strong support to be associated with physical inactivity in numerous studies. In a Norwegian population-based cross-sectional study (n= 38,743), Brunes et al found that high Neuroticism was associated with physical

inactivity.³⁰⁹ In a Dutch population–based survey (n= 19,288), low Neuroticism was found to be associated with high physical activity.³¹⁰ Droomers et al found that high Neuroticism was associated with physical inactivity in a Dutch cross–sectional study of 2,598 adults and elders.³¹¹ Tolea et al reported that high Neuroticism predicted low muscle strength in the US cohort study (n= 1,220); physical inactivity partly mediated this association.³¹² Notably, a meta-analysis of 33 studies found that high Neuroticism was negatively associated with physical activity (r= -0.11).³¹³

In summary, Type A behaviour and related personality (Hostility and Neuroticism) are significant, albeit not strong, predictors for four health behaviours. It is important to note that other personality predictors for health behaviours include: Conscientiousness and Extraversion. Rhodes and Smith conducted a meta-analysis of 33 studies on personality traits and physical activity; the effects of Conscientiousness (r= 0.20) and Extraversion (r= 0.23) on physical activity were slightly stronger than that of Neuroticism (r= -0.11). A meta-analysis of 194 studies reported that Conscientiousness was most consistently associated with all health behaviours; Neuroticism received moderate support to predict all health behaviours.³¹⁴

2.4.3 Overcommitment – origins from perceived control

OC was originated from the concepts of Type A behaviour, which was derived from perceived control (PC) in psychology. Rosenman proposed that Type A persons have higher need for control over environment and tend to perceive lower PC; their response is enhanced coping to assert and maintain control over environment. The concepts of PC and its relationships with health behaviours are introduced below.

(1) Origins of perceived control

Perceived control, also named personal control or sense of control, has emerged broadly from social science and psychology over a century.³¹⁵ In social science,

concepts of control appeared in Marx and Durkheim's works. Karl Marx's concept of alienation means that workers lose control of their lives and destinies by being deprived of the right to be director of their actions. Durkheim's description of anomie indicates that society undergoes significant changes with a discrepancy between values commonly possessed and what is actually achievable in daily life, leading to people's feelings of purposelessness and powerlessness.³¹⁶

In psychology, "control" has been one of the most pervasive and enduring ideas across diverse schools. In psychoanalysis, Freud emphasized ego's primary task in reducing conflict between external reality, superego and id. Hartmann (1939) then proposed a conflict–free sphere of ego, acting through cognitive processes to adapt to environment.³¹⁷ This idea of an autonomous ego was adopted by Robert White (1959), who proposed that people have "effectance motivation" to be effective in dealing with their environment.³¹⁸ Social psychologist Lewin (1936) argued that people strive to control the world rather than just react to it.³¹⁹ Humanistic psychologist, Deci and Ryan (1985), proposed autonomy and self-determination that described individual's fundamental motivation to act as a causal agent on environment.³²⁰

While early psychological theories emphasized control as a motivation to master environment, since the 1960s cognitive psychology directed researchers to focus on cognitive processes (belief, expectancy, or perception) that describe how individuals interact with environment.³²¹ Expectancy is the judgment about the likelihood that a given behaviour will attain the outcome; as a step away from learning perspectives to social–cognitive perspectives, expectancy emphasizes mental representation in this process. People think over available evidence (e.g. past outcomes or current situations) and judge the likelihood of future outcomes; this expectancy then influences success or failure of outcomes.³²² Rotter (1966) proposed the expectancy – locus of control (LOC), the belief about one's behaviour over the outcomes. Bandura (1977) proposed the expectancy – self-efficacy, the belief of one's ability to perform the behaviour.³²³ Self-efficacy and LOC are the two components of PC.

(2) Definition and measurement of perceived control

Perceived control was defined by Skinner (1988) as: "the extent to which one can intentionally produce desired outcomes and prevent undesired ones".³²⁴ PC is viewed as a "self–outcome relation" that was integrated from a "self–behaviour relation" (self-efficacy) and a "behaviour–outcome relation" (LOC).³²⁵ Thus, self-efficacy and LOC are viewed as the two components of PC.³²⁶ Skinner's works for PC provided an integrative framework to organize heterogeneous constructs in the concepts of control, some of which were often used in epidemiological and health research (Table 2.2). The heterogeneity among these constructs interfered with the accumulation of research findings.³²⁷ Skinner's integrative framework can be used to locate parallel constructs; for instance, if a construct is defined in a way as a behaviour–outcome relation, its associations with other measurements would be similar to LOC.

Table 2.2Different constructs in the concepts of control328,329,330,331,332

Construct	Definitions
Locus of control (LOC) (Rotter 1966)	The belief about contingency between one's action and actual outcome. Internal LOC refers to the conviction that outcomes are contingent upon one's own behaviour, whereas external LOC refers to the conviction that outcomes are not contingent upon one's action, but upon chance or powerful others.
Self-efficacy (Bandura 1977)	The belief in one's capabilities to successfully execute the behaviour required to produce certain outcomes.
Mastery (Pearlin & Schooler 1978)	A perception that reflects one's personal control over life outcomes. The extent to which one regards one's life chances as being under one's own control in contrast to being fatalistically ruled.
Learned helplessness (Overmeier & Seligman 1967)	The acquisition of expectancy is based on interaction between exposure and response to it. Learned helplessness occurs when an individual has learned that there is no relationship between his responses and outcome.
Self-control (Rosenbaum et al, 1982)	Self-control refers to the ability to monitor and inhibit one's own emotions, thoughts, and behaviours. Primary control involves taking action to get desired outcomes, and secondary control refers to changing oneself to adjust to the environment.
Sense of coherence (Antonovsky 1993)	A global orientation that indicates the extent to which one has a pervasive, enduring and dynamic feeling of confidence that one's internal and external environments are predictable.

The heterogeneous concepts of PC have generated various measurements.^{333,334}

In health research, the most widely used measurement for PC is Lachman and Weaver's General Perceived Control Scale based on the concepts of mastery and PC. Pearlin and Schooler's concept of mastery is similar to Skinner's definition of PC, as mastery (Table 2.2) also indicates "self–outcome relation".³³⁵ General Perceived Control scale has 2 dimensions: (1) personal mastery represents one's sense of effectiveness or efficacy in carrying out goals; (2) perceived constraint represents one's beliefs of the obstacles or factors beyond one's control that interfere with reaching goals.³³⁶ This thesis will adopt this measure for PC, because it has been validated in many empirical studies including CEE populations (for more details, please see Section 2.2.2). For example, Bobak et al conducted a cross–sectional survey (n= 5,330) in 7 CEE countries; they found that lower PC was associated with poor self–rated health.³³⁷ Lachman and colleagues found that higher PC was associated with better health status, fewer chronic diseases, fewer functional limitations, and more regular exercise in several large–sample cohort studies in the US.^{328,339,340}

Due to the evidence supporting the effects of PC on health, epidemiologist Leonard Syme (1989) suggested that PC provides a parsimonious concept to understand why higher rates of disease are found among seemingly unrelated factors: poor social support, life events, migration, or low control at work. PC can transcend research boundaries to develop interdisciplinary integration in experimental, clinical and epidemiological studies.³⁴¹ In experimental studies, PC has been reported to suppress autonomic arousal, cardiovascular activation, stress hormone release, and pain perception in animals and humans.³⁴² In clinical and epidemiological research, as mentioned earlier, empirical evidence showed that higher PC predicted lower morbidity and mortality. It is suggested that PC can influence health directly by psychobiological processes and indirectly via health behaviours.

2.4.4 Empirical studies on perceived control and health behaviours

PC has been widely applied in predicting health behaviours in psychology. Skinner suggested that when people perceive high control, they initiate action, exert effort, try hard, and persist in the face of failures. When people perceive control as impossible, they withdraw, escape, or become fearful and depressed.³⁴³ PC was integrated into Transactional Model of Stress by Skinner and Wellborn; individuals' appraisals of whether the stressor is controllable and whether their resources are adequate to exercise control subsequently influence coping. Appraisals of high control lead to active coping, such as information seeking, planning, efforts, and direct action. Appraisals of low control result in escape, passivity, and risky health behaviours.³⁴⁴

Bandura proposed potential mechanisms via which self-efficacy might affect health behaviours: (1) Cognition: those with high self-efficacy tend to anticipate success scenarios and to create effective means for exercising control over actions (e.g. drinking abstinence or dietary choices). (2) Motivation: those with high self-efficacy can motivate themselves and guide their actions anticipatorily through forethought; their expectancies for positive outcomes are high, and their motivations to execute actions are strong. (3) Affection: Self-efficacy influences whether a stressor is cognitively constructed in a way good for emotion. Self-efficacy regulates emotional states by supporting effective actions. (4) Selection processes: self-efficacy affects the types of environments or activities (e.g. drinking occasions) people choose to undertake or to avoid.³⁴⁵

Bandura's theory of self-efficacy has a profound impact on Social Cognitive Models (SCMs). Glanz et al conducted reviews of publications from 1986 to 2005 and found that SCMs are the most commonly used theories in predicting health behaviours at intrapersonal level.³⁴⁶ SCMs specify cognitive and affective factors as proximal determinants of health behaviours based on the assumptions of self–regulation involving cognitive evaluation of beliefs, goal setting, and ongoing evaluation of goal–

directed behaviours.³⁴⁷ Socio–cognitive constructs are assumed to mediate the effects of social determinants on health behaviours, and they are suggested to be more modifiable than personality traits. Notably, all SCM models have a common construct – self-efficacy, belief that one can successfully perform the behaviour.³⁴⁸

The roles of self-efficacy in SCMs are summarized. Health Belief Model proposes that health behaviours are determined by two cognitions: perceptions of illness threat and evaluation of behaviours to counteract this threat. Self-efficacy and intention were added to the model to improve predictive power.³⁴⁹ Protection Motivation Theory suggests that primary determinants of performing health behaviours are threat appraisal and coping appraisal. Coping appraisals include self-efficacy and behaviour–outcome expectancy (one's expectancy that carrying out the behaviour can remove the threat). Social Cognitive Theory suggests that behaviour is determined by intention to perform the behaviour, behaviour–outcome expectancy, and self-efficacy. Trans-theoretical Model of Change proposes that different cognitive factors are important at different stages: pre-contemplation (not thinking about change), contemplation (aware of need to change), preparation, action, and maintenance; one major factor influencing stage transitions is self-efficacy.³⁵⁰

Theory of Planned Behaviour (TPB) is the most widely accepted SCM in current literature. It proposes that the proximal determinant of behaviour is intention, which is predicted by attitude (beliefs about perceived consequences of the behaviour), subjective norm (perceptions of whether salient groups or others think the person should perform the behaviour), and perceived behavioural control (PBC; perception of the extent to which performance of the behaviour is easy or difficult). In particular, PBC was derived from the concept of self-efficacy (Figure 2.7).³⁵¹ TPB has received strong support in predicting health behaviours. Armitage and Conner conducted a meta-analysis based on 185 studies; PBC significantly predicted health behaviours directly or indirectly via intention, and PBC emerged as the strongest predictor of intention.³⁵²

In summary, all SCMs emphasize pivotal roles of the two components of PC (self-

efficacy and behaviour-outcome expectancy) in predicting health behaviours. In the following review, PC and its components (self-efficacy, LOC, and PBC derived from self-efficacy) have been extensively supported to predict four health behaviours – alcohol drinking, smoking, diet, and physical activity.

Figure 2.7 Theoretical framework of Theory of Planned Behaviour. Source:





(1) Perceived control and alcohol drinking

Lower levels of PC and its components have received empirical support to predict high alcohol consumption, binge drinking and problem drinking in a wide range of Western populations from college students to elders. In terms of PC, Perlman et al observed that high PC was related to low alcohol consumption in a Russian cross– sectional study of 1,599 men.³⁵³ Troein et al showed that low PC was associated with high alcohol consumption in a Swedish cross–sectional study of 453 men.³⁵⁴

In terms of PC's components, low self-efficacy has been found to predict problem drinking, binge drinking, and relapse of alcohol abuse in college students and adults (n= 273 to 359) in cohort studies from US, UK and Australia.^{355,356,357} Grembowski et al found that low self-efficacy was related to heavy alcohol consumption in a cross–sectional study of 2,524 American elders.³⁵⁸ In terms of TPB, low PBC was reported to predict high alcohol consumption and binge drinking in college students (n= 289 to 513) in cohort studies from UK and Australia.^{359,360} A randomized controlled study

showed that TPB–based intervention (e.g. to promote PBC) reduced binge drinking after 1 month in 467 college students from Estonia, Finland and UK.³⁶¹

(2) Perceived control and smoking

Lower levels of PC and its components (self-efficacy, LOC and PBC) have been found to predict more smoking status as current smokers, higher smoking intensity (number of cigarettes per day), and less smoking cessation from adolescents to adults. In terms of PC, Sigrun et al showed that higher PC at age 14 predicted less smoking intensity at age 17 in a longitudinal study in Iceland (n= 1,293 adolescents).³⁶² Devogli et al found that low PC was associated with more current smokers in a cross–sectional study in Italy (n= 4,002 adults).³⁶³ Low PC was associated with high smoking intensity in a Swedish cross–sectional study of 453 middle-aged men.³⁶⁴

In terms of PC's components, Diclemente et al reported that lower self-efficacy predicted more current smokers and less smoking cessation after 5 months of follow– up in 957 American adults.³⁶⁵ There are intervention studies in Norway and US (n= 244 to 642), showing that increased self-efficacy can lead to smoking cessation.^{366,367} Low self-efficacy and external LOC were associated with current smokers in a cross– sectional study of 885 Korean adolescents.³⁶⁸ Bennett et al found that external LOC was associated with current smokers in a cross–sectional study of 11,401 British adults.³⁶⁹ With regard to TPB, lower levels of PBC were reported to predict more current smokers in several cohort studies (n= 346, 4079 and 14,434) from Canada, Netherlands and China.^{370,371,372} Moan and Rise observed that high levels of PBC predicted smoking cessation after 6 months of follow–up in 698 adults in Norway.³⁷³

(3) Perceived control and diet

Lower levels of PC and its components (self-efficacy, LOC or PBC) have been repeatedly reported to predict less consumption of fruit/vegetable and more consumption of saturated fat and sugar in Western populations from students to elders.

In a British cross–sectional study (n= 372), Barker et al found that low PC was associated with unhealthy dietary patterns, such as less consumption of vegetables, wholegrain bread and vegetarian food, and more consumption of chips, meat, crisps, snacks, white bread, and sugar.³⁷⁴

In terms of PC's components, a review of 14 prospective and 21 cross–sectional studies found that lower self-efficacy predicted less intakes of fruit/vegetable and more intakes saturated fat and sugar.³⁷⁵ Grembowski el al reported that low self-efficacy was associated with high intakes of saturated fat in the US cross–sectional study of 2,524 elders.³⁷⁶ There have been large–sample cross–sectional studies (n= 7,115 to 13,045) showing that Internal LOC was related to more fruit/vegetable and less saturated fat/sugar intakes across 18 Europe countries.^{377,378} Finally, low levels of PBC in TPB predicted low intake of fruit/vegetable and high intake of saturated fat in adults in cohort studies from the UK (n= 413) and the US (n= 609).^{379,380}

(4) Perceived control and physical activity

Low levels of PC and its components have been found to predict physical inactivity from adolescents to adults across Western populations. In terms of PC, Lachman and colleagues found that higher PC was associated with more time in vigorous to moderate levels of physical activity in adults in two large–sample cross–sectional studies (n= 3,848 and 4,242) in the US.^{381,382}

In terms of PC's components, Hagger et al conducted a meta–analysis of 72 studies and found that self-efficacy and PBC in TPB are the strongest determinants for intention and behaviour of physical activity.³⁸³ Several longitudinal studies (n= 328 to 389) showed that higher self-efficacy predicted more time in physical activity in college students and adults from US, Australia and Finland.^{384,385,386} Steptoe and Wardle found that internal LOC was associated with high levels of physical activity in a cross–sectional study of 7,115 students across 18 European countries.³⁸⁷ Rhodes

and Courneya reported that PBC predicted intention and behaviour of physical activity after 2 weeks in 300 students and 272 cancer survivors in Canada.³⁸⁸

There are some limitations for the interpretation of existing literature. While most evidence has supported the effects of PC or its components on health behaviours, some studies reported negative findings.^{389,390} Godin and Kok's meta-analysis found that PBC contributed an additional 12% of variance to predict health behaviours. The effects of PC on health behaviours appear significant, but other psychosocial factors (e.g. social norm, attitude, or social support) may confound the relationship between PC and health behaviours.³⁹¹ For example, Perlman et al found the effect of PC on health behaviours in Russian populations was weaker than that of Western European populations; the findings were explained by the differences in social norm and attitude embedded in Russian contexts.³⁹² Research regarding the effects of PC on health behaviours should take into account other psychosocial factors.

2.4.5 Relationship between overcommitment, perceived control and health behaviours

Siegrist initially developed OC as need for control. High OC persons have higher need for control over environment and tend to perceive lower PC; their response is enhanced coping to assert control over environment. In this section, the potential relation between OC, PC and health behaviours is elaborated based on two approaches in dispositional perspective on personality (need and motive; personality trait). Dispositional perspective proposes that people display consistency in actions, thoughts and feelings. Motives are fundamental desires and personality traits channel how these desires are expressed; they are different but complementary.³⁹³

(1) The approach of need and motive

In this approach, a need is an unsatisfactory internal condition that motivates

behaviour; a need is an internal force that determines how people seek out or respond to the environment. A motive is a cognitive–affective cluster organized around readiness for a preferred experience or goal; a motive takes the underlying need and move it a step closer to the behaviour. Henry Murray (1938) proposed that some needs are based on biological nature (e.g. needs for food or water), while others are based on psychological makeup (e.g. needs for achievement, autonomy, or affiliation).³⁹⁴

Skinner (1995) proposed a meta-theory to explain the widespread effects of PC across life domains: PC reflects the fundamental need for control in all humans.³⁹⁵ This assumption is based on the accumulating literature in psychology. White (1959) proposed effectance motivation that all people have an inborn need to build an increased competence to deal with environment.³⁹⁶ Piaget (1976) proposed that infants enjoy and detect contingencies in environment as soon as they have motor control over behaviours. Children and adults have their needs for competence met in playgrounds and work, respectively.³⁹⁷ DeCharms (1968), Deci and Ryan (1985) described a need for autonomy – the intrinsic motivation to be the origin of one's own behaviour and to choose one's course of action.^{396,399}

Greenberger and Strasser proposed a dynamic model of PC based on reactance theory, learned helplessness and two–process model of PC.^{400,401} Cognitive appraisal for PC is suggested as a function of two dimensions:

PC = The amount of control possessed / The amount of control desired

The amount of control desired is viewed as "need for control". Given the same amount of control possessed, the higher need for control a person has, the lower PC one perceives.⁴⁰² If individuals perceive lower control than they need, they are motivated to seek control by three responses: (1) Direct reaction: individuals attempt to restore control directly at environmental source. (2) Indirect reaction: they attempt to adapt to environment and change themselves by cognitive adjustment or action; for example, they may drink alcohol to satisfy feeling of control immediately. (3) Learned helplessness: after failure in direct and indirect reactions, the only way is not trying. In
short, high need for control (OC) might decrease one's PC, which might subsequently affect risky health behaviours.

(2) The approach of personality trait

By this approach, it is plausible to suggest that PC can mediate the effects of OC personality on health behaviours. In health psychology, numerous studies have adopted the analytical framework with a hierarchical structure, in which the effects of higher–order personality traits (e.g. Five–Factor Model such as Neuroticism) on health behaviours are mediated by lower–order, socio–cognitive constructs (e.g. self-efficacy or PBC in TPB).^{403,404,405} For examples, McEachan et al used the hierarchical framework in which PBC and TPB variables mediated the effects of personality traits on intention and behaviour of physical activity (Figure 2.8).⁴⁰⁶

Figure 2.8 Perceived behavioural control mediated the effects of personality traits on physical activity. Source: McEachan, Sutton and Myers (2010).



The personality traits can clarify which personality factors predict health outcomes, but do less in describing how these factors are associated with cognitive, emotional, and behavioral processes that affect health outcomes. The Five–Factor Model focuses on the structure of personality that people "have", rather than personality processes that people "do". In contrast, social–cognitive constructs can provide a more active and specific process account of individual differences that complements the broader and more static personality traits. Socio-cognitive constructs are considered mediating pathways for the effects of personality traits on health outcomes; they may provide better prediction for behaviours and more modifiable targets for interventions.⁴⁰⁷

A similar hierarchical relationship was proposed in Transactional Model of Stress. Primary appraisal is where the individual evaluates and gives personal meaning to a stressor, and considers the significance of "what is at stake" in terms of harm, threat, loss or challenge. Secondary appraisal addresses the question "what can I do about it" by evaluating one's ability to change the situation and to manage emotional reaction (e.g. PC or self-efficacy); it is where the individual evaluates the availability of coping options and resources to deal with the stressor.⁴⁰⁸ Higher–order personality traits are suggested to influence primary appraisal, secondary appraisal (PC) and coping, which then affect health behaviours (for more details, see Section 2.6).⁴⁰⁹

Section 2.4 is summarized. First, OC was originated from Type A behaviour and highly correlated with Hostility and Neuroticism, all of which have received empirical support to predict health behaviours. Second, Type A behaviour was derived from PC; PC has also received strong support to predict health behaviours. Similar to Type A behaviour, high OC persons may have higher need for control and tend to perceive lower PC, thereby engaging in risky health behaviours.⁴¹⁰ Despite little evidence on the effects of OC on health behaviours, the literature on the effects of Type A behaviour and PC on health behaviours is used to partially support these relationships.

2.5 Overcommitment, Effort–Reward Imbalance, and Health Behaviours

In the ERI model, Siegrist proposed the personality construct of overcommitment (OC) which describes individual attitudes, behaviours and emotions reflecting excessive work–related striving and high need for approval and esteem. Siegrist proposed 3 hypotheses for the ERI model: (1) Extrinsic ERI hypothesis: high effort and low reward lead to adverse health outcomes. (2) OC hypothesis: a high level of OC can increase risks of adverse health outcomes; there are main effects of OC on adverse health outcomes. (3) ERI x OC interaction hypothesis: those who are characterized by both condition (1) and (2) have even higher risks of adverse health outcomes; there are modifying roles of OC in ERI–outcome relationship.⁴¹¹

Current literature has been inconsistent on potential roles of OC in ERI–outcome relationship; however, research on the influence of personality on health has been accumulated in interrelated fields of behavioral medicine, health psychology and psychosomatic medicine since Friedman and Rosenman's Type A behaviour (1959). Compared to current understanding on diverse roles of personality in stress–outcome processes, original assumption of OC (main or modifying effects) appears relatively simple. To gauge potential roles of OC in ERI–outcome relationship, it would be helpful to review a wider range of literature regarding OC–related personality traits (Type A behaviour, Neuroticism and Hostility). In the following sections, potential roles of OC in ERI–outcome relationship are discussed in detail.

2.5.1 Modifying or main effect of OC in ERI–outcome relationship

Main effect implies that two or more predictor variables (e.g. ERI and OC) contribute independently to explaining variance in an outcome. Modifying effect (effect modification) means that the magnitude and direction of the effect of a predictor (e.g. ERI) on an outcome depends on the level of another predictor (e.g. OC). There are different types of modifying effects. For example, "buffering" implies that an adaptive

personality trait protects people from adverse effects of risk exposure. On the other hand, "vulnerability" implies that a maladaptive personality trait combined with a risk factor predict a disproportionately adverse outcome compared to the additive effect. Vulnerability is the suggested mechanism for the interaction between OC and ERI.

Van Vegchel et al reviewed 45 studies on the ERI model published from 1986 to 2003, including outcomes of physical health (e.g. cardiovascular diseases), psychosomatic health (e.g. depression), behaviour (e.g. sickness absence), and job-related well–being (e.g. burnout). First, the review found that extrinsic ERI hypothesis was strongly supported in 48 out of 52 studies; some studies included several outcomes and were counted twice. Second, OC hypothesis was examined in 27 of 52 studies, and "main effects" of OC on outcomes were supported in 17 out of 27 studies (63%). Third, OC x ERI interaction hypothesis was examined in only 12 of 52 studies, and "modifying roles" of OC were supported in only 3 out of 12 studies (25%).⁴¹² As interaction hypothesis was rarely examined, they suggested that strong conclusions regarding modifying role of OC cannot be made. This review also noted that the potential roles of OC in the relationship between ERI and health–behaviour outcomes have not been tested.

Parkes reviewed 33 longitudinal studies from 2000 to 2009 on the relationships between work stress and personality constructs (e.g. OC, Neuroticism or Hostility).⁴¹³ First, "main effects" of OC were supported in 7 out of 8 studies (88%). Among men, high OC was consistently predictive of adverse health outcomes, including depression, anxiety, poor subjective health, and CHD; the reported risk ratios were generally moderate (1.5–2.0).^{414,415} Among women, the effects of OC were less consistent. Main effects of Neuroticism were reported in 9 out of 11 studies (82%); main effects of Hostility were supported in 2 out of 2 studies. Second, "modifying effects" of OC were not supported in 2 out of 2 studies.^{416,417} In contrast, modifying effects of Neuroticism and Hostility were supported in 4 out of 5 studies; those with higher levels of Neuroticism or Hostility were more vulnerable to work stress.^{418,419,420}

The reviews by Van Vegchel et al and Parkes are summarized. First, most studies have supported "main effects" of OC and related personality traits; however, this evidence might not really confirm main effects of OC (OC and ERI contribute independently to explaining variance in an outcome), as other two possibilities had not been tested: antecedent or mediator roles of OC in ERI–outcome relationship. For mediation analyses, confirmation of the effect of OC on outcome is merely a first step for OC–ERI–outcome (OC as antecedent) or ERI–OC–outcome (OC as mediator) causal chains.⁴²¹ Thus, previous evidence for "main effects" of OC might partially support or at least not exclude other two possibilities – antecedent or mediator. Second, modifying role of OC has not been supported by the limited amount of literature. Failure to consider interaction may lead to this modifying role remaining hidden in most studies; more research to test modifying role is needed before drawing any conclusion.

As mentioned before, research on the influence of personality on health has been accumulated in psychology and medicine. Compared to current understanding on diverse roles of personality in stress processes, original assumption of the roles of OC (main or modifying effect) appears relatively simple. A more sophisticated model – the Michigan model has been particularly influential in guiding research into the combined effects of personality constructs and work stress. Objective work environments influence subjective perceptions of work stress, which affect short–term psycho–biological responses, leading to long–term health problems. The influence of personality can operate at several points in the stress process, including modifying, mediator, bidirectional or direct effects (Figure 2.9).⁴²²

Figure 2.9 The Michigan model describes individual and situational factors affecting

the process of work stress. Source: Israel, et al (1996).423



The Michigan model attempted to capture potentially diverse roles of personality in work–stress processes. For example, if personality affects a person's perceptions to objective work stressors or if personality creates one's objective work stressors, personality is suggested to be an antecedent in the effects of work stress on outcomes. The model also incorporates bi–directional pathways and feedback loops. For example, if objective or perceived work stressors would affect personality which then influences health, personality is suggested to be a mediator in the effects of work stress on outcomes. Based on the framework provided by the Michigan model, four possible roles of personality can be proposed: modifying, antecedent, mediator, or direct effect. In this thesis, I attempt to test these four potential roles of OC in ERI– outcome relationship.

2.5.2 Antecedent role of OC in ERI–outcome relationship

Antecedent role of OC (OC influences ERI that then affects outcomes) has ever been supported by theoretical explanations and empirical studies. Indeed, Siegrist implied the possibility of antecedent role of OC in ERI–outcome relationship; individuals with high OC might expose themselves more often to high demands (efforts) at work, or they exaggerate their efforts beyond what is formally needed, thereby resulting in continued imbalance between high effort and low reward.⁴²⁴ In contrast, individuals with low OC tend to reduce their efforts or change jobs in order to avoid effort–reward imbalance. OC might explain the duration of exposure to ERI work stress. Moreover, several researchers who investigated the ERI model suggested testing the possibility of antecedent role of OC in future research.⁴²⁵

In terms of psychological theories, personality can influence work stress via several mechanisms: (1) Stressor creation: high Neuroticism individuals may create objective work stressors for themselves by provoking interpersonal conflicts or poor work performance. High Hostility persons might have antagonistic behaviours which elicit negative behaviours from others. (2) Perception: personality influences one's perception to objective work environments. High Neuroticism individuals tend to perceive their jobs as having high levels of stressors. High Hostility persons tend to perceive threat or hostile intent from others. (3) Self selection: Type A persons may select themselves into highly competitive jobs, because they tend to set task goals too high for their abilities, leading to more failures and dissatisfaction. High Neuroticism people choose less complex jobs or they are less attractive candidates for better jobs. (4) Reaction: high Neuroticism people react exaggeratedly to work stressors in their psycho–biological processes and health behaviours; they tend to use emotion–focused coping, such as alcohol drinking.^{426,427}

In terms of empirical evidence, antecedent role of OC-related personality in work stress-outcome relationship has been partially supported (in particular the effect from

personality traits to work stress). For example, Hintsa et al reported that specific dimensions of Type A behaviour (high aggression, hard-driving, and time urgency) measured at adolescence subsequently predicted both high ER ratio and high job strain at adulthood in the Cardiovascular Risk in Young Finns study (n= 752).⁴²⁸ Their team in University of Helsinki also found that high Neuroticism measured at adolescence predicted both high ER ratio and high job strain after 15 years (n= 621); Neuroticism predicted low control (β = -0.129, p= 0.012), high job strain (β = 0.337, p= 0.001), and low rewards (β = -0.195, p= 0.001).^{429,430}

With regards to empirical studies regarding the effect of OC personality on work stress, Allisey et al found that high OC was associated with high effort (r= 0.40) and low reward (r= –0.31) in an Australian cross–sectional study (n= 897).⁴³¹ Rennesund and Saksvik reported that high OC was associated with high job strain in a cross–sectional study in Norway (n= 924).⁴³² In addition, there is a growing body of evidence showing that personality traits influence the way people perceive environmental stressors (e.g. daily stressors) and subsequent responses to stressors in various experimental and epidemiological studies.^{433,434}

Note that the above studies only examined the pathway from OC (or related personality traits) to perceived work stress (ERI or job strain), but the pathway from perceived work stress to outcomes has not been tested simultaneously. Nevertheless, there have been many studies supporting the effects of work stress on health outcomes (Section 2.3.3). Thus, it is of value to examine the causal path "OC–work stress–outcome" simultaneously in a longitudinal study.

2.5.3 Mediator role of OC in ERI–outcome relationship

Mediator role of OC (ERI influences OC that then affects outcomes) appears possible based on theoretical explanations and empirical studies. Personality is defined by Carver and Scheier (2000) as: "a dynamic organisation, inside the person,

of psychophysical systems that create a person's characteristic patterns of behaviours, thoughts and feelings". This definition attempts to integrate diverse perspectives in psychology. Characteristic patterns suggest continuity and consistency uniquely identified in an individual; dispositional, biological, and psychoanalytic perspectives argue that personality traits are biologically based "temperaments" not susceptible to influence of environments (e.g. work stress) and do not change over time.⁴³⁵ The perspectives have been challenged by current literature showing that personality traits can change over time. Thus, personality is now defined as a dynamic organisation which implies ongoing readjustment and adaptation in an individual; social learning, cognitive, and socio–cognitive perspectives view personality as accumulated sets of thoughts and behaviours which are learned from or changed by environments.⁴³⁶

Continuity and change are often indexed by correlation between personality scores across two time points (e.g. test-retest correlation). The meta-analysis of 92 longitudinal studies reported that personality traits (measured by Big Five personality traits) change moderately before age 30 (test-retest correlation increase from 0.41 at childhood to 0.55 at age 30) and become increasingly stable but still change mildly across adulthood (test-retest correlation increase from 0.55 at age 30 to 0.70 at age 50).⁴³⁷ The most remarkable changes occur in young adulthood involving more lifechanging roles (e.g. a new job) and identity decisions than any other period, but modest changes continue into middle-aged (40-60 years old) and older populations (> 60 years old). Note that personality traits can change over time, but they are more stable than other psychological constructs except intelligence. In addition, literature has showed that personality traits can be changed by stressors from environments and uncontrollable situations (e.g. low social position or chronic diseases). For example, the onset of heart disease, diabetes, hypertension, and cancer can predict subsequent changes in Big Five personality traits in a meta-analysis of three US cohort studies (n= 17,493; mean age= 55.8 years).⁴³⁸

There have been several empirical studies supporting that psychosocial work

conditions predicted changes in OC–related personality (Neuroticism). A 3–year longitudinal study in the Netherland (n= 576) found that perceived work stress and life satisfaction can predict moderate changes in personality traits in adults (average age 43.9 and 41.7 years for men and women); positive work experience was related to personality maturation (e.g. decreased levels of Neuroticism). ⁴³⁹ An 8–year longitudinal study in the US (n= 1,130) reported that as work satisfaction increased, the levels of Neuroticism decreased.⁴⁴⁰ Robert et al reported that de-investment in work (counterproductive behaviours such as fighting with co-workers or breaking safety rules) was associated with increased levels of Neuroticism in an 8–year longitudinal study of 907 young adults in New Zealand.⁴⁴¹

In particular, empirical studies have found that OC personality was changed by work environments. DeJonge et al tested the stability of ERI constructs over time in a Dutch cohort study (n= 650); they reported that test–retest reliability for OC scale was 0.53 over 1–year interval and 0.45 over 2–year interval.⁴⁴² Tsutsumi et al found that during 1 year of organisational changes, OC scale changed significantly and moderately in 544 Japanese employees; the magnitude of changes in OC scale was less than that in situation–specific components in the ERI model (effort and reward).⁴⁴³

Some limitations should be noted in the above studies. First, the aforementioned studies only examined the pathway from work conditions to OC–related personality, but the pathway from OC to outcomes has not been tested simultaneously. However, many studies have supported the effects of OC–related personality on health behaviours (Section 2.4.2). It is of value to examine the causal path "work stress–OC– outcome" simultaneously in a cohort study. Second, the above studies measured exposure factors in work conditions, but work stress has not been measured by the ERI or DC models. Third, although OC personality may be changed by work stress, the magnitude of personality change is expected to be small if the samples come from middle–aged (40–60 years old) and older populations (> 60 years old).

2.5.4 Reciprocal relationship between OC and ERI across life-course

Antecedent (Section 2.5.2) and mediator (Section 2.5.3) roles of OC in ERIoutcome relationship may coexist. To fully elucidate how personality traits and stress processes interact over time, there is a need for researchers to adopt the life course approach, in which personality and (work) environments might have "bi–directional causal relationship" across life span.^{444,445} Social environments in childhood (e.g. rearing styles or learning experiences) and adulthood (e.g. work environments) might alter an individual's personality traits. In contrast, personality traits may shape mastery of educational and work tasks (e.g. occupational attainment or job performance), cultivation of social relationship, and maintenance of physical and mental health (e.g. responses to stressors); thus, personality may influence an individual to select, encounter and create different environmental stressors.^{446,447}

In empirical studies, Roberts et al found that personality traits measured at age 18 predicted objective and subjective work experiences at age 26; those with high Neuroticism at age 18 experienced difficult transitions into employment, occupied lower prestige jobs, were less satisfied with jobs, and reported financial difficulties at age 26. In contrast, work experiences were related to changes in personality traits between 18 and 26 years old; higher occupational status, more satisfying jobs, and having financial security at age 18 were associated with decreased levels in Neuroticism at 26.⁴⁴⁸ Sutin et al showed that career success (e.g. more job satisfaction or higher incomes) predicted decreased levels of Neuroticism after 10 years; personality traits predicted changes in career success after 10 years.⁴⁴⁹ Note that the above evidence comes from the samples in their young adulthood.

Importantly, Sutin and Costa adopted a longitudinal cross–lagged analysis in the US (n= 722) and found that Big Five personality traits have significant effects on work stress defined by the DC model; low levels of Neuroticism were associated with increases in job control and decreases in demands after 10 years. However, work

stress (job strain) was found to have only small effects on personality traits after 10 years.⁴⁵⁰ The results may be somewhat explained by the middle–aged sample (mean age 52.3 years old at wave 2), as the meta-analysis of 92 longitudinal studies reported that personality traits become increasingly stable across adulthood (test–retest correlation 0.55 at age 30 and 0.70 at age 50).⁴⁵¹ This study is particularly relevant to this thesis, as my samples in the HAPIEE study are also in middle–aged adulthood.

Note that another role of OC not mentioned is a *confounder* between work stress and health behaviours. Earlier studies tended to adjust personality as a confounder for self-reported bias, suggesting that individuals with Neuroticism tend to report high levels of distress even without objective stressors.⁴⁵² However, the measure of work stress is to assess "perceived" rather than "objective" work stressor, personality traits can cause underlying construct – perception (perceived work stress). If OC personality is viewed as the antecedent or mediator in the causal path from work stress to health behaviours, OC should *not* be viewed as a confounder.

The literature review in Section 2.5 tentatively suggests four potential roles of OC in the relationship between work stress (ERI) and health behaviours: modifying role, antecedent, mediator, and direct effect of OC. Note that these potential roles of OC may not be mutually exclusive (e.g., OC is possible to have both antecedent role in ERI–drinking relation and direct effect on drinking). Finally, main effect of OC on health behaviours can be considered after the first three roles are excluded.

2.6 Overcommitment, Effort–Reward Imbalance, Perceived Control, and Health Behaviours

The relationship between ERI, OC and health behaviours has been addressed. In this chapter, the potential role of PC in the relationship between ERI, OC and health behaviours is considered by theoretical framework of Transactional Model of Stress.

2.6.1 Transactional Model of Stress

Transactional Model of Stress, proposed by Richard Lazarus and Susan Folkman (1984), is the most widely–accepted model to clarify complex relationship between the environmental stressor and the person in relation to health outcomes (health behaviours, emotional well–being, and functional status).⁴⁵³ This model has been enriched by accumulating literature in personality psychology, cognitive psychology, and health psychology. In this model, stressful experiences are constructed as "person–environment transactions". When a person faces a stressor in environment, one would evaluate potential threat (primary appraisal) and one's ability to alter the situation and manage negative emotional reaction (secondary appraisal). Both appraisals can affect one's coping efforts (problem management and emotional regulation), which subsequently influence health behaviours and health outcomes.⁴⁵⁴ The processes are described in detail in the following paragraphs.

Stressors are demands made by internal or external environment that upset balance or homeostasis in an individual, thereby affecting physical and psychological well-being and requiring one's action to restore balance or equilibrium.⁴⁵⁵ Stressors can contribute to diseases via direct physiological effects or indirect effects via health behaviours. This model emphasizes that individual's cognitive appraisals, rather than objective stressor, would influence health behaviours and health outcomes.

Cognitive appraisals refer to cognitive processes that incorporate not only information from the stressor but also information inside the person. Primary appraisal is where the individual evaluates and gives personal meaning to the stressor, and considers the significance of "what is at stake" in terms of harm, threat, loss or challenge. Two basic primary appraisals are: (1) perceptions of *susceptibility* to the threat, which refer to beliefs about the likelihood of getting a threat (e.g. a stressful condition or an illness); (2) perceptions of *severity* of the threat, which refer to feelings about the seriousness of a threat and its possible consequences (e.g. death, disability,

or negative effects on work and social relationship). Appraisals of high severity and susceptibility of a threat not only prompt efforts to cope with the stressor, but also generate psychological distress and physiological responses.⁴⁵⁶ As primary appraisal is one's subjective evaluation for susceptibility or severity of objective stressor, I suggest that perceived "severity" of work stress measured by the ERI model (effort–reward ratio) or the DC model (job strain) should be viewed as primary appraisal.

Secondary appraisal is an assessment of a person's availability of coping options and resources to deal with the stressor; it addresses the question "what can I do about it" by evaluating one's ability to change situation and to manage emotional reaction.⁴⁵⁷ Key examples of secondary appraisals in psychology are: perceived ability to change the situation (PC over the threat), perceived ability to manage one's emotional reactions to the threat (PC over emotional feeling), and expectations about the effectiveness of one's coping effort (coping self-efficacy).⁴⁵⁸ As mentioned previously, PC is defined as the extent to which one can intentionally produce desired outcomes and prevent undesired ones. Thus, PC should be viewed as secondary appraisal.

Coping effort is defined as the constantly changing cognitive and behavioural efforts a person makes to manage specific external/internal demands that are appraised as taxing/exceeding the resources of the person. The basic approach to classifying coping strategies is the dichotomy between problem–focused coping (directed at changing the stressful situation, including active coping, problem solving, and information seeking) versus emotion–focused coping (directed at changing the stressful situation, including emotional expression, avoidance, and denial). When a stressor is appraised as controllable and a person has high self-efficacy, one is more likely to engage in problem–focused coping. By contrast, when a stressor is perceived as highly threatening and uncontrollable, one tends to use emotion–focused coping.^{459,460}

Avoidance and denial (emotion-focused coping) can shift one's attention away from the stressor and temporarily minimize psychological distress by avoiding

thoughts and feelings about the stressor (e.g. avoiding others, refusing to think about the threat, or hiding feelings); however, avoidance and denial may lead to intrusive thoughts that generate psychological distress over time and keep people from developing adaptive coping strategies. Avoidance coping is temporarily useful, but it is ineffective when confronting a stressor that poses a real threat – something that will have to be dealt with eventually.⁴⁶¹ Engaging in risky health behaviours (e.g. drinking or smoking) is suggested to relieve psychological distress temporarily and to distract one's attention from stressful situation; it may however cause long–term social and health problems. Thus, engaging in risky health behaviours is viewed as an emotion–focused coping, like avoidance and denial.⁴⁶²

Coping outcomes, influenced by coping efforts, indicate a person's adaptation to a stressor. Three main categories of coping outcomes are emotional well-being, functional status (health status or disease progression), and health behaviours. The theoretical framework of Transactional Model of Stress is illustrated (Figure 2.10). Figure 2.10 Transactional Model of Stress. Source: Adapted from Glanz and



Schwartz (2008).463

Transactional Model of Stress has been enriched by literature from personality psychology.⁴⁶⁴ In contrast to situation–specific coping efforts, Lazarus conceptualized coping styles as stable dispositional characteristics reflecting generalized tendencies to interpret and respond to stress in particular ways.⁴⁶⁵ Coping styles are enduring personality traits to drive primary and secondary appraisals and coping efforts; specific effects of a stressful event on adjustment may partly depend on a person's coping styles. The most widely studied coping style is dispositional optimism – the tendency to have positive rather than negative generalized expectancies for outcomes; these expectancies are relatively stable over time and across situations.⁴⁶⁶ Evidence has supported that dispositional optimism influences the transactional process at primary appraisal, secondary appraisal and coping efforts.⁴⁶⁷ For example, Taylor et al found that dispositional optimism was associated with lower perceived risk of disease (primary appraisal), higher PC over disease (secondary appraisal), more problem–focused coping, and less risk health behaviors.⁴⁶⁸

In addition to dispositional optimism, several studies reported that other personality traits (e.g. Neuroticism) can influence the transactional process at several points – primary appraisal, secondary appraisal and coping efforts (Figure 2.10). First, personality traits may alter subjective appraisal of ongoing stressor (primary appraisal); for example, high Neuroticism person tends to interpret neutral or ambiguous stimuli as a threatening event. Second, personality traits (Neuroticism) may reduce one's PC (secondary appraisal) which then affects health behaviours.⁴⁶⁹ Third, personality traits may influence available choice of one's coping efforts; for example, high Neuroticism person tends to use more emotion–focused coping (e.g. risky health behaviours) in order to relieve higher levels of psychological distress.⁴⁷⁰

As Neuroticism is highly correlated with OC, it is plausible that OC may influence the transactional process at primary appraisal, secondary appraisal and coping efforts. Lazarus and Folkman have never mentioned "overcommitment" but they described "commitment", which denotes enduring motivational and cognitive process on what is

important and what has meaning for the person. In a stressful encounter, commitment determines what is at stake – to what extent the stressor harms or threatens the person (commitment influences primary appraisal). The greater the strength of a commitment, the more vulnerable a person is to psychological stress in the area of that commitment. Additionally, commitment impels a person toward a course of action that can reduce threat and sustain coping effort in the face of obstacles (commitment influences secondary appraisal).⁴⁷¹

After personality traits were taken into account, the unique roles of PC (secondary appraisal) are considered. First, empirical studies have shown that PC can mediate the impacts of Five Factor personality traits on health behaviours.^{472,473,474} Compared to static and broad personality traits, PC has better prediction for health behaviours by providing active and specific cognitive processes. Besides, PC is more modifiable (better target for intervention) than personality traits.⁴⁷⁵ Second, empirical evidence has found that PC is changeable by stressors from environments (e.g. social position, work stress, or chronic illness); PC can mediate the effects of stressors on health outcomes.^{476,477} Thus, Steptoe proposed that PC occupies a central position in the interplay between psychosocial demands (e.g. stressors from environments) and resources (e.g. personality traits) and poses effects on health.⁴⁷⁸

In summary, potential roles of PC (secondary appraisal) are considered in the relationship between ERI (primary appraisal), OC (personality traits), and health behaviours by the Transactional Model of Stress. This integration of the ERI model and Transactional Model of Stress would shed light on practical implications for both organisational and individual interventions for work stress.

2.6.2 Reciprocal relationship between perceived control and ERI

In Transactional Model of Stress, Lazarus identified primary appraisal and secondary appraisal, both of which were suggested to engage in a reciprocal and

dynamic relationship with each other.⁴⁷⁹ Due to the dynamic nature of stress process, it is often difficult to determine whether primary appraisal causes secondary appraisal, or secondary appraisal affects primary appraisal. Both occur as part of a complex process, and both are required to shape individual's responses to a stressful encounter. Thus, Lazarus proposed that each is dependent on the other, and they should be regarded as part of the same process.⁴⁸⁰

In this thesis, work stress measured by ERI is viewed as primary appraisal, and PC is considered secondary appraisal. In the following paragraphs, empirical evidence on the potentially reciprocal relationship between ERI and PC, together with their relationships with outcomes, will be reviewed. This review is divided into two parts: (1) PC acts as a mediator in the effects of work stress on outcomes; (2) work stress acts as a mediator in the effects of PC on outcomes.

(1) Perceived control acts as a mediator in the effects of work stress on outcomes

The possibility of PC as a mediator in the effects of work stress on outcomes has been supported by theoretical and empirical evidence. In terms of theories, Kohn and Schooler focused on the impact of working conditions on individual's cognition and psychological health; work characteristics that allow for employee's use of independent judgment in complex matters can promote the development of PC and general intellectual functioning.⁴⁸¹ Pearlin et al proposed that the accumulation of experiences in which one successfully controls work environments may lead to increased perceptions of mastery (similar to PC).⁴⁸² Bandura proposed that individuals may learn and emulate skills and beliefs from workplace and bring them to other life situations; work experience may shape one's self-efficacy.⁴⁸³ Siegrist suggested that work role is crucial to fulfill one's self-regulatory needs, because work offers opportunities to acquire self-efficacy (e.g. successful performance), self-esteem (e.g. recognition) and self-integration (e.g. belonging to a group). Work stress defined by ERI may impair one's self-efficacy, self-esteem and self-integration.⁴⁸⁴

In terms of empirical evidence, my review has found 8 empirical studies (listed in Table 2.3) supporting that PC or its components can partially mediate the effects of works tress (job strain, job insecurity, or job uncertainty) on health outcomes (health behaviours, self-rated health, or psychological distress). The exposure of work stress was measured by the DC model in 3 cohort studies^{485,486,487} and by other measures of work stress (e.g. job insecurity or uncertainty) in 5 studies.^{488,489,490,491,492} Note that no such literature is available on the ERI model. In addition, 6 out of 8 studies have sample size less than 500, so the results should be interpreted carefully; a cohort study with larger sample size is still needed in this topic. In general, my review shows that PC and its components may partially mediate the impacts of work stress on health outcomes.

Table 2.3 Empirical studies supporting perceived control as a mediator in the

Authors (year)	Sample	Study type	Work stress measure	Perceived control measure	Outcome	Findings on the roles of perceived control
Payne et al (2002)	213 adults, UK	Cohort study, 1 week	DC model	Self- efficacy	Health behaviour	Self-efficacy partially mediates the impacts of job strain on physical activity
Payne et al (2005)	286 adults, UK	Cohort study, 2 weeks	DC model	PBC in TPB	Health behaviour	PBC partially mediates the impact of demands on physical activity
Wickrama et al (2008)	318 men, USA	Cohort study, 10 years	DC model	Perceived control	Self-rated health	PC partially mediates the effect of work stress on outcomes
Plotnikoff et al (2010)	612 adults, Canada	Cohort study, 1 Year	Perceived work stress	Self- efficacy	Health behaviour	Self-efficacy partially mediates the impact of work stress on physical activity
Vander Elst et al (2011)	211 adults, Belgium	Cross- sectional study	Job insecurity	Perceived control	Psychologi cal distress	PC partially mediates the effect of job insecurity on outcomes
Paulsen et al (2005)	553 adults, Australia	Cohort study, 1.5 years	Job uncertainty	Perceived control	Emotional exhaustion	PC partially mediates the effect of job uncertainty on outcomes
Ito et al (2001)	204 adults, Canada	Cross- sectional study	Job uncertainty	Perceived control	Emotional exhaustion	PC partially mediates the effect of job uncertainty on outcomes
Bordia et al (2004)	222 adults, Australia	Cross- sectional study	Organisati onal changes	Perceived control	Psychologi cal distress	PC partially mediates the effect of work stress on outcomes

effects of works stress on outcomes

(2) Work stress acts as a mediator in the effects of perceived control on outcomes

The possibility of work stress as a mediator in the effects of PC on outcomes has been supported by theoretical and empirical evidence. By socio–cognitive theories, it is reasonable to assume that self-efficacy can affect perceived work stress which then influences health outcomes.⁴⁹³ Bandura proposed that self-efficacy affects cognitive processes, persistency of motivation, affective states and selection processes, all of which contribute to the extent of one's performances (e.g. health behaviours or coping efforts). People with high self-efficacy perceive and appraise stressors with the confidence that they can exercise control over them; they have strong beliefs in capabilities to approach difficult tasks as challenges to be overcome, rather than as threats to be avoided. Thus, they can invest high efforts in what they do and even heighten their efforts in the face of failures.⁴⁹⁴ It is suggested that higher self-efficacy might initially result in lower level of perceived work stress; subsequently, more active coping efforts might further reduce both objective and perceived work stress.

In terms of empirical studies, Spector conducted a meta-analysis and found that higher PC was associated with lower levels of work stress – measured by role conflict in 8 studies and role ambiguity in 14 studies.⁴⁹⁵ In addition, my review found 6 empirical studies (Table 2.4) supporting that PC or its components can affect measures of work stress (e.g. the DC model or other measures), which then influence various outcomes.^{496,497,498,499,500} For example, Judge et al reported that higher self-efficacy in early adulthood predicted better profiles of perceived work conditions (e.g. autonomy or task variety), resulting in higher levels of job satisfaction in middle adulthood.⁵⁰¹ In general, my review implies that work stress may partially mediate the impacts of PC on outcomes. Nevertheless, there has been literature with negative findings. For example, in a 2–wave cohort study in Belgium (n= 536), PC was found to mediate the effect of job insecurity on emotional exhaustion, but job insecurity did not mediate the effect of PC on emotional exhaustion.⁵⁰²

Table 2.4 Empirical studies supporting work stress as a mediator in the effects of

Authors (year)	Sample	Study type	Work stress measure	Perceived control measure	Outcome	Findings on the roles of perceived control
Judge et al (2000)	258 adults, USA	Cohort study, 30 years	Perceived work conditions	Self- efficacy, LOC	Job satis- faction	Perceived work conditions mediated the effect of LOC/self- efficacy on outcomes
Schwarzer et al (2008)	458 teachers, Germany	Cohort study, 1 year	Job demands, perceived work stress	Self- efficacy	Burnout	High self-efficacy decreased perceived work stress, which then reduced burnout
Spreitzer et al (2002)	350 adults, USA	Cohort study, 1 year	Threat in organization downsizing	Perceived control	Voluntary turnover	High PC reduced threat in downsizing, which then decreased turnover
Xanthopoulou et al (2009)	163 adults, Netherland	Cohort study, 1.5 year	DC model, Job control, workplace support	Self- efficacy	Work engage- ment	High self-efficacy increased job control and workplace support, which then increased work engagement
Hoge et al (2004)	205 adults, Germany	Cross– sectional study	Perceived work stress	Sense of coherence	Physical & mental health	High sense of coherence decreased perceived work stress, which improved health
Rennesund et al (2010)	924 adults, Norway	Cross– sectional study	DC model, job strain	Self- efficacy	Work per- formance	High self-efficacy decreased job strain, which promoted work performance

perceived control on outcomes

Several limitations need to be taken into account in the above review. First, the exposure of work stress was assessed by the DC model or other measurements, but no literature is available on the ERI model. While health and job–related outcomes were measured, there were no studies on health–behaviour outcomes. In addition, 5 out of 6 studies have sample size less than 500, and the results should be interpreted carefully; a large cohort study is still needed in this topic.

While PC may mediate the effects of work stress on outcomes, work stress may also mediate the effects of PC on outcomes; the two are not mutually exclusive. In Transactional Model of Stress, primary appraisal (e.g. perceived work stress such as ERI) and secondary appraisal (e.g. PC) are engaged in a reciprocal relationship.⁵⁰³ In social cognitive theory, Bandura emphasized that the reciprocal causation of the characteristics of persons (self-efficacy) and their environments is better captured by

the transactional perspective; people are both producers and products of social environment.⁵⁰⁴ Based on the theoretical and empirical evidence, I tentatively suggest that there is a reciprocal relationship between PC and ERI.

2.6.3 Interaction between perceived control and ERI

In heath and psychological research, interaction hypothesis (interaction between personality and stressor from environment) is a widely–used approach that cannot be ignored in this thesis. Modifying effect means that the magnitude and direction of the effect of a predictor (e.g. work stress) on an outcome depends on the level of another predictor (e.g. PC). There are different forms of interaction. "Buffering" implies that a high level of adaptive personality construct (e.g. PC) protects individuals from adverse effects of risk exposure (e.g. work stress). "Person-environment fit" indicates neither high nor low levels of a personality construct are necessarily maladaptive; adverse outcomes arise from a lack of fit between personality and environment. For instance, job control and locus of control may be congruent or incongruent, with favorable or unfavorable effects on outcomes, respectively.⁵⁰⁵

In terms of empirical evidence, my review found 7 cohort or cross–sectional studies supporting that PC or its components can modify the effects of work stress on outcomes (Table 2.5). There are 3 studies foucused on the DC model,^{506,507,508} 3 study on other measurements of work stress,^{509,510,511} and one study on the ERI model.⁵¹² Note that there are 3 large–sample studies (n > 1000). For example, Bethge and Radoschewski reported that internal locus of control acted as a buffer between the effect of ERI on impaired work ability in a German cross–sectional study (n= 1,348).⁵¹³

Some limitations of my review should be noted. First, the findings on the forms of modifying effects are mixed and inconsistent; both "buffering" and "person– environment fit" have received support, and both two–way (e.g. job control x PC) and three–way interactions (e.g. demand x job control x PC) have been reported. Second,

there have been other studies reporting negative findings on modifying role of PC in work stress–outcome relationship.^{514,515} For instance, Marchand et al found that LOC did not modify the effect of job strain on psychological distress in a 7–year cohort study in Canada (n= 6,359).⁵¹⁶ Moreover, there has been literature supporting mediator rather than modifying role of PC when both possibilities were tested.⁵¹⁷ In a cross–sectional study in Belgium (n= 211), PC did not buffer the effects of job insecurity on outcomes, but PC can mediate the effects of job insecurity on outcomes.

Table 2.5 Empirical studies supporting interaction between perceived control and work stress

Authors	Sample	Study type	Work stress measure	PC measure	Outcome	Findings on the roles of perceived control
Olsson et al (2009)	2,246 adults, Sweden	Cohort study, 9 years	DC model	Sense of coherence	Self-rated health	SOC modified the effect of work stress on self-rated health
Rodriguez et al (2001)	542 Europeans	Cohort study, 1 year	DC model	LOC	Job satis- faction	LOC modified the effect of work stress on job satisfaction
Parkes (1991)	590 civil servants, UK	Cohort study, 1 year	DC model	LOC	Affective distress	LOC modified the effect of work stress on affective distress
Jimmieson et al (2004)	213 adults, Australia	Cohort study, 2 years	Work load, role ambiguity	Self- Efficacy	Job satis- faction	Self-efficacy modified the effect of work stress on job satisfaction
Brockner et al (2004)	1,067 adults, US	Cohort study, 1 year	Threat from organisation downsizing	PC	Work performa- nce	PC modified effect of threat from downsizing on work performance
Lu et al (2000)	581 managers in Taiwan / UK	Cross- sectional study	Perceived work stress	LOC	Job satis- faction, well-being	LOC modified the effect of work stress on job satisfaction/ well-being
Bethge et al (2010)	1,348 adults, Germany	Cross- sectional study	ERI model	LOC	Impaired work ability	Internal LOC acted buffer effect of high ERI on reduced work ability

It is premature to conclude whether PC can modify the effects of work stress on outcomes by existing literature. Based on the Transactional Model of Stress, reciprocal relationship between ERI and PC, rather than interaction between the two, is assumed as my hypothesis. However, considerable findings from previous studies imply that the possibility of interaction between ERI and PC cannot be ignored and should be empirically tested in this thesis.

2.6.4 Relationship between OC, ERI, perceived control and health behaviours

In this chapter, I attempt to hypothesize the potential relationship between ERI, OC, PC and health behaviours by integrating the theoretical frameworks of Transactional Model of Stress and the ERI model. In Transactional Model of Stress, primary appraisal is one's subjective evaluation for susceptibility or severity of objective stressor. I suggest that perceived severity of work stress measured by the ERI model (ER ratio) should be viewed as primary appraisal. The reason is that the lack of reciprocity between high effort and low reward together – rather than either effort or reward alone, defines a state of emotional distress with autonomic arousal and strain reactions in the ERI model.

Secondary appraisal is one's assessment of availability of coping options and resources to deal with the stressor. Main examples of secondary appraisal in psychology are perceived ability to change the situation (PC over the threat), perceived ability to manage one's emotional reactions to the threat (PC over emotional feeling), and expectations about the effectiveness of one's coping effort (coping self-efficacy).⁵¹⁸ Thus, PC should be viewed as secondary appraisal.

Notably, my interpretation and integration for the ERI model and Transactional Model of Stress (ER ratio is primary appraisal) is slightly different from Siegrist's original ideas on the link between these two models. Siegrist (1996) recognized the influence of Lazarus's Transactional Model of Stress on the ERI model.⁵¹⁹ He wrote: "negative emotions are the result of a multistage appraisal process, which includes the taxing of stressor properties and of a person's coping repertoire under exposure..... This theory would predict cognitive and behavioural adjustment to a high-cost / low-gain condition as a consequence of cognitive appraisal processes". Siegrist has ever suggested that "effort / reward" were conceptually similar to "primary appraisal / secondary appraisal" in early works. On the other hand, Siegrist tried to distinguish between the ERI model and Transactional Model of Stress; he suggested that rapid

and direct pathways of affective information processing may bypass neocortical-limbic structures, so chronically everyday experience of ERI at work may not necessarily be subjected to conscious appraisal.

In Transational Model of Stress, primary appraisal and seconday appraisal subsequently affect coping efforts (health behaviours) and health outcomes. In particular, primary appraisal (ERI) and secondary appraisal (PC) engage in a reciprocal relationship. Additionally, personality traits (e.g. OC) can influence the transactional process at primary appraisal, secondary appraisal and coping efforts. The potential relationships between the constructs in this thesis are summarized in Figure 2.11, which is relatively simplified but not different from Figure 2.10.

Figure 2.11 Potential relation between OC, ERI, PC and health behaviours based on Transactional Model of Stress.



Two points should be noted in the above model. First, the directions of the effects are categorized into positive association (+) and negative association (-). In Section 2.4.4, a substantial amount of literature has supported that lower PC is associated with

worse profiles of health behaviours. In Section 2.4.5, the relationship between OC, PC and health behaviours has been addressed; higher OC is associated with lower level of PC, which is associated with worse profiles of health behaviours. PC is suggested to mediate the effects of OC on health behaviours.

Second, it is suggested that OC personality and ERI might have bi-directional causal relationship in Section 2.5.4. Based on existing evidence, the effect of OC personality on ERI would be stronger than the other causal direction in the middle-aged and older populations. In a parsimonious model, the cross-sectional "snapshot" of the bi-directional relationship between OC and ERI would be: ERI mediates the effect of OC personality on health behaviours. Transactional Model of Stress and personality psychology also supports this parsimonious model.

2.7 Summary of Literature Review

2.7.1 Gaps identified in the existing research

For several decades, researchers have been intrigued by the debate over the joint influences of work environment and personality constructs on human behaviours. A profound understanding of the complex relationship between the environment and the person is crucial to promote a steady accumulation of knowledge and to inform effective interventions on health behaviours. In Chapter 2, literature review has summarised the evidence related to different relationships between work stress, personality constructs (OC personality and PC), and health behaviours. Several gaps in existing literature have been identified and listed below.

(1) Very limited literature on the links between OC, ERI and health behaviours

Health behaviours are influenced by a wide range of psychosocial factors. The approach of many epidemiological studies has been to identify independent

contribution of each psychosocial factor. It is widely recognized that psychosocial factors rarely occur in isolation, but few studies have attempted to examine how two or more psychosocial factors interact to influence health outcomes. It is necessary to expand the breadth of studies by addressing this multiplicity of psychosocial factors.⁵²⁰ Work stress (ERI and job strain) or OC–related personality (Type A behaviour, Neuroticism, Hostility, and PC) have been repeatedly reported to independently predict health behaviours (described in Section 2.3.3, 2.4.2 and 2.4.4). Despite considerable evidence linking ERI and OC–related personality independently to health behaviours, there has been very limited literature on the links between OC, ERI and health behaviours in studies where all of them are considered simultaneously.

In the review of 45 studies on the ERI model by Van Vegchel et al, there have been only two studies regarding health–behaviour outcomes (smoking and drinking); extrinsic ERI hypothesis was supported, but the potential role of OC (main or modifying effect) was not tested.^{521,522} Since this review published, another two studies from Japan and Australia have reported negative findings on main effects of OC on smoking; however, modifying effect of OC has not been examined.^{523,524} As very limited literature is available on the links between OC, ERI and health behaviours, it is of value to investigate this topic.

(2) Inconclusive findings regarding potential role of OC in ERI-outcome relationship

Current literature on potential role of OC in ERI–outcome relationship has been inconclusive. The main effect of OC has received strong support from empirical studies, while modifying effect of OC has gained weak support from limited literature (as described in Section 2.5.1). In the wider fields of psychology and medicine, current understanding of diverse roles of personality in stress processes includes "antecedent and mediator roles" of OC in ERI–outcome relationship, but these possibilities have not been examined for the ERI model. To evaluate potential roles of OC in ERI–

outcome relationship more rigorously, four possible roles of OC should be tested: modifying role, antecedent role, mediator role, or direct effect.

(3) No literature on the links between OC, ERI, PC and health behaviours

Perceived control (PC) and its components have gained strong support to predict health behaviours from theoretical models and empirical studies. PC is suggested to mediate the effects of OC personality on health behaviours (as described in Section 2.4.3 and 2.4.5). By Transactional Model of Stress, the potential role of PC (secondary appraisal) can be integrated into the relationship between OC, ERI (primary appraisal) and health behaviours (as described in Section 2.6.4). The integration of the ERI model and Transactional Model of Stress would enlighten practical implications for interventions for work stress. To my knowledge, no literature is available on the links between OC, ERI, PC and health behaviours in studies where all of them are considered simultaneously.

(4) Limitations of small–sample studies and cross–sectional design

Most evidence regarding the relationships between work stress, personality constructs (e.g. OC and PC), and health outcomes (e.g. health behaviours) comes from studies with sample size less than 1000. In addition, about half of these studies have cross–sectional design. These studies cannot provide strong evidence of modifying, antecedent or mediator roles of personality constructs in stress processes with a clear temporal relationship. Thus, a large–sample cohort study is needed to help establish sequential and ultimately causal nature of relationships between work stress, personality constructs and health behaviours, thereby providing more solid evidence for potential roles of personality in work stress–outcome relationship.

2.7.2 Importance of proposed project

Based on the gaps identified, this thesis aims to investigate the relationship between OC, ERI, PC and health behaviours in a large–sample and two–wave cohort study in CEE populations, particularly focusing on two aspects: the potential role of OC in the relationship between ERI and health behaviours; the potential role of PC in the relationship between OC, ERI and health behaviours. It is hoped that this thesis will contribute to growing understanding of intersecting pathways by which work stress (ERI) and personality constructs (OC and PC) jointly influence health behaviours.

The unique context of social transformation in CEE provides a natural setting for investigating the relationship between work stress, personality constructs and health behaviours. The East–West mortality gap in Europe, a major public health concern in CEE, was previously hypothesized to be mainly explained by psychosocial factors and health behaviours. It is crucial to understand the interactions between multiple levels of psychosocial factors (work stress and personality) and their combined influences on health behaviours in order to develop more effective interventions aimed at promoting health behaviours and subsequent health in CEE. Thus, the aims of this thesis might contribute to deeper understanding on the East–West mortality gap. However, I should state explicitly that this thesis using data collected from CEE populations provides an opportunity to investigate the general topic of interest – the relationship between work stress, personality constructs and health behaviours; the specific topic on contextual importance of CEE will not be addressed in detail in the following parts of the thesis.

Chapter 3. Aims, Objectives and Hypotheses

3.1 Aims

This chapter outlines the main aims, objectives, hypotheses, and conceptual framework of the thesis. In the previous chapter, the gaps in knowledge related to the relationship between work stress, personality constructs, and health behaviours were identified; the current project aims to address some of these gaps. Thus, the two aims of the thesis are:

1. To examine the relationship between effort-reward imbalance (ERI), overcommitment (OC), and health behaviours in the Central and Eastern European (CEE) populations, particularly focusing on the potential role of OC in the relationship between ERI and health behaviours.

2. To additionally investigate the potential role of perceived control (PC) in the relationship between ERI, OC and health behaviours in the CEE populations.

Three health–behaviour outcomes (alcohol drinking, smoking, and diet) will be used in this thesis. The specific objectives and relevant hypotheses related to these three outcomes are listed below.

3.2 Objectives and Hypotheses

In relation to the first aim, the focus is on the associations between ERI and health behaviours, on the associations between OC and health behaviours, and on assessing whether OC has potential role of antecedent, mediator, modifier, or direct effect in the relationship between ERI and health behaviours. The objectives and hypotheses are:

Objective 1

To assess crude and adjusted associations between ERI and three health behaviours

- alcohol drinking, smoking, and diet, respectively.

→ Hypothesis 1: Higher ER ratio is associated with higher levels of alcohol drinking after adjustment for covariates.

→ Hypothesis 2: Higher ER ratio is associated with higher levels of smoking after adjustment for covariates.

→ Hypothesis 3: Higher ER ratio is associated with less healthy diet after adjustment for covariates.

Objective 2

To assess crude and adjusted associations between OC and three health behaviours

- alcohol drinking, smoking, and diet, respectively.

→ Hypothesis 4: Higher OC is associated with higher levels of alcohol drinking after adjustment for covariates.

→ Hypothesis 5: Higher OC is associated with higher levels of smoking after adjustment for covariates.

→ Hypothesis 6: Higher OC is associated with less healthy diet after adjustment for covariates.

Objective 3

To evaluate the potential role of OC (antecedent, mediator, modifier, or direct effect) in the relationship between ERI and health behaviours.

 \rightarrow Hypothesis 7: OC and ERI have bi–directional relationship, but it is predicted that the effect of OC on ERI is stronger than the other direction in the middle-aged and older populations. OC might have antecedent role in ERI–drinking relationship.

 → Hypothesis 8: OC and ERI have bi-directional relationship, but it is predicted that the effect of OC on ERI is stronger than the other direction in the middle-aged and older populations. OC might have antecedent role in ERI-smoking relationship.
→ Hypothesis 9: OC and ERI have bi-directional relationship, but it is predicted that

the effect of OC on ERI is stronger than the other direction in the middle-aged and older populations. OC might have antecedent role in ERI–diet relationship.

The second aim focuses on the associations between PC and health behaviours, and on the potential role of PC in the relationship between ERI, OC and health behaviours. The objectives and hypotheses are listed below:

Objective 4

To assess crude and adjusted associations between PC and three health behaviours – alcohol drinking, smoking, and diet, respectively.

→ Hypothesis 10: Lower PC is associated with higher levels of alcohol drinking after adjustment for covariates.

 \rightarrow Hypothesis 11: Lower PC is associated with higher levels of smoking after adjustment for covariates.

→ Hypothesis 12: Lower PC is associated with less healthy diet after adjustment for covariates.

Objective 5

To examine the potential role of PC (mediator or modifier) in the relationship between ERI, OC and health behaviours.

 \rightarrow Hypothesis 13: PC and ERI partially mediate the effects of OC on alcohol

drinking. In addition, PC and ERI might have bi-directional relationship.

 \rightarrow Hypothesis 14: PC and ERI partially mediate the effects of OC on smoking. In

addition, PC and ERI might have bi-directional relationship.

→ Hypothesis 15: PC and ERI partially mediate the effects of OC on diet. In addition,

PC and ERI might have bi-directional relationship.

The conceptual framework (Figure 3.1) illustrates the relationships between the three psychosocial factors (OC, ERI, and PC) and health behaviours in the wider context of social determinants of health. Potential confounders that are known risk factors for health behaviours and are associated with exposure variables in the source population include: age, social position (education and occupation), material factors (deprivation), other psychosocial factors (marital status, depression, and social isolation), and health status (self–rated health); these factors will be adjusted in the regression analyses.

Figure 3.1 A conceptual framework of the thesis



The assumed relationship between OC, ERI, PC, and health behaviours is based on Transactional Model of Stress. The directions of the effects are categorized into positive association (+) and negative association (-).

Chapter 4. Methods

4.1 Study Description

This thesis is based on the data from the HAPIEE (Health, Alcohol and Psychosocial factors In Eastern Europe) study, which is a prospective cohort study designed to examine the impact of classical and non–conventional risk factors (e.g. psychosocial factors) on cardiovascular and other non–communicable diseases in CEE. The baseline data collection (wave 1) was carried out between 2002 and 2005 in six towns in Czech Republic, Novosibirsk in Russia, and Krakow in Poland. The wave 2 data collection was extended to include Lithuania (not included in the thesis) and conducted between 2006 and 2008.

In terms of study populations, the HAPIEE study originally had the 3 cohorts: (1) six towns in Czech Republic, (2) Novosibirsk in Russia, and (3) Krakow in Poland (Figure 4.1). The six Czech towns, with a total population of over 0.6 million, cover a variety of socioeconomic profiles. For example, *Hradec Kralove* is a prosperous city with chemical industry, electronics manufacturing and Information technology, with low unemployment rate (6.5%). *Havirov/Karvina* is a large city with the highest unemployment rate in the country (19.6%). *Kromeriz* and *Jihlava* are both towns with a variety of production industries.

Krakow is the second largest city – a science and technology centre in Poland with a population of one million. Krakow is more prosperous than the Polish average, the unemployment rate in 2007 was 4.8% compared to the national average (13%). The study selected four different districts ranging from blue–collar districts to middle– class districts, which should represent various socioeconomic spectrums.

Novosibirsk, the third largest city in Russia with a population of 1.4 million, is an industrial city with electric power, gas and water supply, and mechanical engineering. Novosibirsk is typical for urban populations in Russia in terms of social development. Two districts of the city with different socioeconomic profiles were selected.



Figure 4.1 The study sites in the HAPIEE study



The reported response rates were 61% in Poland and Russia and 55% in Czech Republic. As a small proportion of non–respondents had died or moved away after the sample was selected but before being invited to the study, they were ineligible for inclusion; thus, the real response rates may be higher (estimated at least 68%, 71%, and 60% for Poland, Russia, and the Czech Republic). In examining a subsample of non–respondents, they were more likely to be younger and male, with lower levels of education, with higher prevalence of smoking, and with poorer self–rated health.⁵²⁵ All participants gave written informed consent, and all procedures were approved by University College London (UCL) Hospital and local ethical committees.

4.2 Study Samples

The cohorts consisted of random samples of men and women aged 45–69 years at baseline, stratified by gender and by 5–year age groups and selected from population registers. Of the 28,947 subjects at wave 1, *ineligible subjects* such as retired persons (14,060), unemployed (1,178), housewives (307) and those with unknown employed status (131) were excluded, as only those employed at the time of the study completed a module on work characteristics (e.g. questionnaire for the ERI model) but others did not (e.g. retired persons completed a module on retirement and quality of life). In the 13,271 *eligible subjects* who were employed at the time of the study, those with missing values in exposure variables – ERI/OC (425) and PC (93) at wave 1 – were excluded. The remaining 12,753 subjects were used to generate two subsamples for the analyses of different outcomes.

For the analyses of drinking and smoking outcomes (available at both wave 1 and 2), 3,450 subjects who were lost to follow–up at wave 2 were excluded. Then, those with missing values in exposure variables and outcomes (drinking and smoking outcomes) at wave 2 (1,158) were excluded. Finally, 632 subjects with missing covariates were excluded. This subsample with complete information for exposures and drinking/smoking outcomes at wave 1 and 2, and with complete information for covariates at wave 1, consisted of 7,513 subjects (3,782 men and 3,731 women).

For the analyses of dietary outcomes (available at wave 1 only), 621 subjects with missing values for more than 15 questions in the Food Frequency Questionnaire were excluded. Next, 1,120 subjects with missing covariates were excluded. This subsample with complete information for exposure variables, dietary outcomes, and covariates at wave 1 was composed of 11,012 subjects (5,735 men and 5,277 women).

As a form of sensitivity analysis, bivariate analyses were conducted among study samples and excluded subjects due to missing values in exposures, outcomes and covariates. In the subsample for drinking/smoking outcomes, excluded subjects (n= 13271 - 7513 = 5758) were more likely to be male (55.1% versus 50.3%), with lower educational level (29.0% versus 34.9% at university degree), with higher alcohol consumption (4139 ± 9376 versus 3652 ± 8725 g/year), with more current smokers (36.1% versus 30.9%), and with poorer self-rated health (2.8 ± 0.8 versus 2.6 ± 0.7)
than the study sample. In the subsample for dietary outcomes, excluded subjects (n= 13271 - 11012 = 2259) were more likely to be older (55.1 ± 6.1 versus 53.9 ± 5.8) and male (53.6% versus 52.1%), with lower educational level (29.6% versus 33.4% at university degree), with more current smokers (34.8% versus 32.9%), and with poorer self–rated health (2.9 ± 0.7 versus 2.7 ± 0.7) than the study sample.



Figure 4.2 Flow chart for selection of the study subsamples

4.3 Description of Variables

The baseline data collection included questionnaires, physical examination, and blood sample. The questionnaire covered socioeconomic status, demographics,

material factors, psychosocial factors, health behaviours and health. Within psychosocial factors, employed subjects completed a module on work characteristics (e.g. questionnaire for the ERI model), and retired persons completed a module on retirement and quality of life. The variables used in the analyses of the thesis are described below. All questionnaires related to the thesis are presented in the Appendix.

4.3.1 Outcome variables

The three health behaviours – drinking, smoking, and diet were adopted as outcome variables in this thesis. As the measurement of physical activity (time spent in physical activity) was relatively simple in the HAPIEE study, physical inactivity was not included in these analyses. The three outcome variables are described below.

(1) Drinking outcomes

Several measures of alcohol consumption were derived from the graduated frequency questionnaire (GFQ). The frequency of drinking occasions was assessed by 9 mutually exclusive categories, ranging from "never" to "daily". The amounts of ethanol consumed per occasion were assessed by 6 mutually exclusive categories at wave 1 (ranging from "<1" to "10 and above" drinks) and 3 mutually exclusive categories at wave 2 (ranging from "<2" to "5 and above" drinks). The amounts were expressed in local units; 1 drink means 0.5L of beer, 0.2L of wine, and 0.05L of spirits, which approximately equal to 20g ethanol. Total annual consumption of alcohol was calculated from the frequency of drinking occasions and the amounts per occasion.⁵²⁶

Based on information from GFQ, three drinking outcomes were obtained: (1) Binge drinking: a dichotomous variable was defined by drinking at least 100g in men or 60g in women of ethanol per drinking session at least once a week, with all other respondents reporting alcohol intakes below these limits in reference category. (2) Heavy drinking: it was defined as a dichotomous variable by the cutoff point: 350

g/week or more of ethanol in men and 210 g/week or more in women, respectively.⁵²⁷ (3) Problem drinking: the CAGE questionnaire (Table 4.1) was used to screen for problem drinking; it contained 4 items with 2 responses (0= no; 1= yes). With a cut-off point of 2, previously reported sensitivity ranged from 0.78 to 0.81 and specificity ranged from 0.76 to 0.96 in relation to alcohol abuse or dependence.⁵²⁸ In the HAPIEE study, both GFQ–based variables and problem drinking were strongly associated with separately taken measures of alcohol consumption and serum gamma–glutamyl transferase.⁵²⁹

Table 4.1 The CAGE questionnaire

	Items					
1.	Have you ever felt you should cut down on your drinking?					
2.	Have people ever annoyed you by criticising your drinking?					
3.	Have you ever felt bad or guilty about your drinking?					
4.	Have you ever had a drink first thing in the morning to steady your nerves or get rid of a hangover?					

(2) Smoking outcomes

Smoking outcomes included smoking status and smoking intensity. Smoking status was measured in a standard way using the questions: "do you smoke cigarettes?" The four answers were: (1) yes, regularly, at least one cigarette a day on average; (2) yes, occasionally, less than one cigarette a day; (3) no, I smoked in the past but I stopped; (4) no, I have never smoked. Those who gave the first two answers were classified as *current smokers*; those who gave the third answer were classified as *past smokers*, and those who gave the last answer were *life–long non–smokers*. For these analyses, the outcome of smoking status was dichotomized as: current smokers (1) and current non–smokers (0) which included past smokers and life–long non–smokers.

Among current smokers, smoking intensity was assessed by the question: "how many cigarettes a day do you smoke now?" For these analyses, the outcome of

smoking intensity was categorized as: 1= light smoker (1–9 cigarettes a day), 2= medium smoker (10–19 cigarettes a day), and 3= heavy smoker (20 or more cigarettes a day). This classification for the levels of smoking intensity has been adopted in several previous studies.^{530,531}

(3) Dietary outcomes

Dietary data were collected using the Food Frequency Questionnaire (FFQ) adapted from Willett et al,⁵³² which was used previously in the Whitehall II Study.⁵³³ FFQ is the primary method for measuring average long-term diet in epidemiologic applications. Due to country-specific dishes, Czech, Polish and Russian FFQs consisted of 136, 148 and 147 food items, respectively. For each food item, a country-specific portion size was specified according to the McCance and Widdowson Food Composition Database and local food composition tables.⁵³⁴ Subjects were asked how often they had consumed that amount of food during the last 3 months, with 9 responses ranged from "less than once per month" to "6 or more times per day". By multiplying frequency of food consumed per day with nutrient content of specified portion size, nutrient intakes (quantity per day) were calculated. This methodology was described in detail by Boylan et al.⁵³⁵

Diet quality is often defined by the adherence to dietary guidelines associated with health outcomes (e.g. chronic diseases or mortality).⁵³⁶ The Healthy Diet Indicator (HDI) was constructed to reflect the adherence to pre–defined dietary recommendations of World Health Organisation (WHO) for the prevention of chronic diseases.⁵³⁷ This approach was developed by Huijbregts et al to identify diet quality associated with chronic diseases.⁵³⁸ The measurement of diet quality takes into account intakes of various foods and nutrients, thereby providing more accurate pictures of diet than single food/nutrient intake. From the WHO guideline, nine nutrient/food intakes were selected as follows: (1) nutrient density evaluated by percentages of total energy intakes (nutrient intakes divided by total energy intakes)

from saturated fats, polyunsaturated fats, total carbohydrates, free sugars, and protein; (2) nutrient intakes of non–starch polysaccharides (NSP), cholesterol, and sodium; (3) food intakes of fruit and vegetable. Macronutrients including fats, protein and carbohydrates are major sources of energy; fruit and vegetable represent main sources for micronutrients (vitamins and minerals). Next, a dichotomous variable was generated for each nutrient/food intake; if one's intake was within the WHO recommended range this variable was coded as 1 (healthy intake), otherwise it was coded as 0 (unhealthy intake). The HDI score was then calculated as the sum of nine dichotomous variables, with values ranged from 0 to 9 (Table 4.2).

Individual HDI components as dichotomous variables	1= Within the WHO recommended range 0= Otherwise
1. Saturated fat	1= < 10% total energy intake
	0= > 10% total energy intake
2. Polyunsaturated fat	1= 6–10% total energy intake
	0= < 6% or > 10% total energy intake
3. Protein	1= 10–15 % total energy intake
	0= < 10% or > 15% total energy intake
4. Total carbohydrate	1= 55–75% total energy intake
	0= < 55% or > 75% total energy intake
5. Free sugars	1= < 10% total energy intake
	0= > 10% total energy intake
Non–starch polysaccharides	1= > 20 g/day
	0= < 20 g/day
7. Cholesterol	1= < 300 mg/day
	0= > 300 mg/day
8. Sodium	1= < 2000 mg/day
	0= > 2000 mg/day
9. Fruit & vegetable	1= > 400 g/day
	0= < 400 g/day

Table 4.2. Nine individual HDI components as dichotomous variables

4.3.2 Main exposure variables

(1) Effort–Reward Imbalance (ERI) at work

The ERI model was operationalized as a standardized self-reported measure containing 23 Likert-scaled items (Table 4.3), defining three unidimensional scales: extrinsic effort (6 items), reward (11 items), and overcommitment (6 items). In the

HAPIEE study, the ERI questionnaire was translated into all three languages and then back translated to confirm the accuracy of original translations.

"Extrinsic effort" was measured by six items on demanding aspects of work environment (3 items measuring quantitative load, 1 item measuring qualitative load, 1 item on physical load, and 1 item on increase in total load over time). For each item, the rating procedure was given on a 5-point scale: (1) no; (2) yes, not at all distressed; (3) yes, somewhat distressed; (4) yes, rather distressed; (5) yes, very distressed. "Reward" was assessed by eleven items (1 item on financial reward, 5 items measuring esteem reward, and 5 items on promotion prospects and job security). In terms of rating procedure, 4 of 11 items were rated in the same way as extrinsic effort. The other items were rated on a 5-point scale: (1) yes; (2) no, not at all distressed; (3) no, somewhat distressed; (4) no, rather distressed; (5) no, very distressed. Positively and negatively worded items were included to control for response biases; negatively worded items were reversely coded to ensure all responses in the same direction.

The extent of imbalance between extrinsic effort and reward was measured by effort–reward (ER) ratio; extrinsic effort score was in the numerator, and reward score was multiplied by a correction factor to adjust for unequal number of items in the denominator.⁵³⁹ High ER ratio (> 1) indicates high levels of work stress, in which a high amount of effort spent is not met by the rewards received or expected. In my analyses, average scores were calculated if a minimum of 5 out of 6 questions on extrinsic effort (the average score of non-missing items multiplied by 6) and 9 out of 11 questions on reward contained valid answers.⁵⁴⁰

Overcommitment (OC) is the cognitive-motivational pattern of coping with demands characterized by an excessive work-related overcommitment and a high need for approval. OC was assessed by 6 items; for examples, I get easily overwhelmed by time pressures at work; as soon as I get up in the morning I start thinking about work problems; people close to me say I sacrifice too much for my job. Each item was rated on a 4-point scale (strongly disagree, disagree, agree, and

strongly agree). In my analyses, average scores were calculated if a minimum of 5 out of 6 questions on OC contained valid answers (the average score of non-missing items multiplied by 6).

By the commonly used analyses based on score distribution, ER ratio and OC were divided into thirds to indicate low (tertile 1), intermediate (tertile 2), and high levels (tertile 3), respectively.⁵⁴¹

 Table 4.3
 The Effort–Reward Imbalance questionnaire

Extrinsic E	ffort
-------------	-------

- 1. There is constant time pressure in my job due to a heavy workload
- 2. There are many interruptions and disturbances in my job
- 3. I have a lot of responsibility in my job
- 4. There is pressure in my job to work overtime
- 5. My job is physically demanding
- 6. Over the past few years, my job has become more and more demanding

Reward

- 1. Are you treated unfairly at work?
- 2. Are the promotion prospects in your job poor?
- 3. Do you expect to experience an undesirable change in your work situation?
- 4. Is your own job security poor?
- 5. Do you receive the respect you deserve from your work colleagues?
- 6. Do you receive the respect you deserve from your supervisors?
- 7. Do you experience adequate support in difficult situations?
- 8. Does your current job adequately reflect your knowledge, skills and training?
- 9. Does your salary/income adequately reflect all your past efforts and achievements?
- 10. Considering all your efforts and achievements, do you receive the respect and prestige you deserve at work?

11. Considering all your efforts and achievements, are your work prospects adequate?

Overcommitment

1. I get easily overwhelmed by time pressures at work

2. As soon as I get up in the morning I start thinking about work problems

3. When I get home, I can easily relax and 'switch off' work

- 4. People close to me say I sacrifice too much for my job
- 5. Work rarely lets me go, it is still on my mind when I go to bed
- 6. If I postpone something that I was supposed to do today, I'll have trouble sleeping at

night

(2) Perceived control (PC)

The PC score was based on 11 questions (Table 4.4) adapted from the Whitehall II Study and MacArthur Foundation Programme on Midlife Development.⁵⁴² This instrument was similar to the perceived constraints of General Perceived Control Scale developed by Lachman and Weaver.⁵⁴³ In terms of external validity, this PC score has been found to be associated with several socioeconomic indicators and self-rated health in the context of CEE countries.⁵⁴⁴

In the PC score, the items 2 to 4 represented "control over health", while other items represented "control over life". The subjects were asked to what extent they agree or disagree with the statements, with the answers recorded on a 6–point scale (0 meaning low control; 5 meaning high control). All negative–worded items (items 5 and 7–11) were reverse coded to ensure that all responses were in the same direction. The final score ranged from 0 (no control) to 55 (maximum control). In my analyses, scores were calculated if a minimum of 9 out of 11 questions contained valid answers (the average score of non-missing items multiplied by 11). By the approach based on score distribution, PC score was divided into thirds to indicate low (tertile 1), intermediate (tertile 2), and high levels (tertile 3), respectively.

Table 4.4 The perceived control score

- 6. Over the next 5–10 years I expect to have many more good things than bad things happen
- 7. I often have the feeling that I am being treated unfairly
- 8. In the past 10 years, my life has been full of changes without my knowing what would happen next
- 9. I very often have the feeling that there's little meaning in the things I do in my daily life
- 10. I sometimes feel as if I've done all there is to do in life
- 11. I gave up trying to make big improvements or changes in my life a long time ago

^{1.} At home I feel I have control over what happens in most situations

^{2.} Keeping healthy depends on things that I can do

^{3.} There are certain things I can do for myself to reduce the risk of a heart attack

^{4.} There are certain things I can do for myself to reduce the risk of getting cancer

^{5.} I feel that what happens in my life is often determined by factors beyond my control

4.3.3 Covariates

(1) Demographics

The subjects were aged 45-69 years old at baseline and were grouped into 5-year age groups. Due to small proportion in 65-69 age group (retired persons were not included), it was incorporated into 60-69 age group. Gender and marital status (married/cohabiting, single, and divorced/separated/widowed) were recorded.

(2) Social position

Education, occupational grade, and material deprivation were used as indicators of social position. *Education* was categorized as: primary/less, vocational (apprenticeship), secondary (A-level equivalent), and university degree. *Occupational grade* was obtained by combing 2 questions about position (higher manager, manager/supervisor, employee, and self-employed) and description of job (sedentary occupation, standing occupation, physical work, and manual work), and it was then categorized as: manager/professional (derived from position), non-manual workers, and manual workers (derived from physical work and manual work).

Material deprivation was assessed by 3 questions about how often the subject's household had difficulties to buy enough food or clothes and to pay bills for electricity, heating and housing. The answers were "never or almost never", "sometimes", "often", and "always" (coded 0 to 3); a deprivation score was derived as the sum of three responses.⁵⁴⁵ The score was dichotomized into low (0–3.9) and high deprivation (4–9).

(3) Other psychosocial factors

Depressive symptoms were measured by the Center for Epidemiologic Studies Depression scale (CES-D), consisting of 20 self-reported items. Each item was based on a question "how often you have felt this way during the past week", rated from 0 (rarely/none of the time) to 3 (most/all of the time).⁵⁴⁶ If at least 16 out of 20 items

were answered, the mean of valid questions was multiplied by 20, so the final score has values between 0 and 60. CESD >= 16 was defined as having depression.⁵⁴⁷

Social isolation was constructed by combining 2 questions about the frequency of contact with friends or relatives. People were classified as *socially isolated* if having regular contact with friends or relatives less than once a month.

(4) Self-rated health

It was assessed by a standard single question with answers on a 5-point scale (1= very good, 2= good, 3= average, 4= poor, and 5= very poor).⁵⁴⁸ A dichotomized measure of self-rated health was used: very good/good/average and poor/very poor.

4.4 Statistical Power

Since this thesis was based on existing data in the HAPIEE study, sample size could not be influenced; however, statistical power can be estimated. Power is the probability of rejecting a false null hypothesis; it should be close to one. Power analysis was calculated using the formula for logistic regression in software G Power 3.1 based on the following assumptions.⁵⁴⁹ (1) Alpha value – the probability of rejecting a true null hypothesis – was set as 0.05. (2) Sample sizes were set as 6000 and 4000, as there were two subsamples on dietary outcomes and drinking/smoking outcomes and men and women were analysed separately. (3) Baseline probability (P0) is the outcome probability at the mean of the predictor X; P0 was set from 0.05 to 0.60 due to the prevalence of health behaviours (outcomes) ranging from 5% to 60%. (4) There is loss of power in multivariate analysis when adjusting for covariates. The adjustment was given for sample size ($N_m = N/1 - R^2$) where R was the correlation coefficient between X (ER ratio or PC) and covariates (R is estimated as 0.20).

Table 4.5 reports the statistical power of the thesis for several odds ratios and various baseline probabilities at the significance level of 0.05. In the subsample on dietary outcomes (sample size= 6000), the results of power calculation shows that

statistical power is over 99% for odds ratio larger than 1.3 in baseline probability of 0.05. In the subsample on drinking/smoking outcomes (sample size= 4000), the results of power calculation shows that statistical power is over 95% for odds ratio larger than 1.3 in baseline probability of 0.05.

Baseline	Examples of outcomes	Odds Ra	tio (OR)		
Probability	-	1.1	1.2	1.3	1.4
Sample size	e = 6000				
0.60	Polyunsaturated fat	94 %	> 99 %	> 99 %	> 99 %
	Fruit and vegetable				
0.30	Heavy drinking in men	91 %	> 99 %	> 99 %	> 99 %
	Smoking status in men				
0.20	Heavy drinking in women	82 %	> 99 %	> 99 %	> 99 %
	Smoking status in women				
0.10	Binge drinking in men	58 %	98 %	> 99 %	> 99 %
	Problem drinking in men				
0.05	Binge drinking in women	36 %	87 %	99 %	> 99 %
	Problem drinking in women				
	Healthy Diet Indicator 5–9				
	Saturated fat				
Sample size					
0.60	Polyunsaturated fat	82 %	> 99 %	> 99 %	> 99 %
	Fruit and vegetable				
0.30	Heavy drinking in men	77 %	> 99 %	> 99 %	> 99 %
	Smoking status in men				
0.20	Heavy drinking in women	66 %	99 %	> 99 %	> 99 %
	Smoking status in women				
0.10	Binge drinking in men	43 %	92 %	> 99 %	> 99 %
	Problem drinking in men				
0.05	Binge drinking in women	26 %	71 %	95 %	> 99 %
	Problem drinking in women				
	Healthy Diet Indicator 5–9				
	Saturated fat				

4.5 Statistical Analysis

The statistical analyses for drinking, smoking and dietary outcomes are described in detail in Chapter 5, 6 and 7, respectively. In general, descriptive statistics with the percentages and means for the covariates and the outcomes by country and by gender were presented. Bivariate analyses for the associations between covariates and outcomes were conducted in men and women, respectively; chi–squared tests were used to examine the significance of differences between categories of the variable. To assess whether data of three countries would be pooled for further analyses, crude associations between exposure variables and outcomes in country–specific strata were assessed. By comparing log likelihoods for the model with the interaction term (between country and exposure variable) and the model without, likelihood–ratio (LR) test was used to test the significance of this interaction term.

The associations between exposure variables (ER ratio, OC and PC) and outcomes (drinking, smoking and diet) were evaluated, respectively. For binary categorical outcomes, the associations between exposure variables and outcomes were assessed by binary logistic regression; for ordinal categorical outcomes (smoking intensity and HDI), the associations between exposure variables and outcomes were assessed by ordinal logistic regression. These associations were assessed after adjustment for age and country (model 1) and after additional adjustment for other covariates (model 2). All above analyses were conducted by STATA 11 (Stata Corp LP, College Station, USA).

For assessing modifying role, log likelihoods for the model with the corresponding interaction term and the model without were compared, and LR test was used to test the significance of this interaction term. For assessing antecedent or mediator roles, the techniques of structural equation modelling (SEM) were applied by software Mplus 7; specific SEM models used for each study outcome are described in detail in Chapter 5, 6 and 7. The general methodology of SEM that will be applied in my analyses is explained in the next section.

4.6 Methodology for Structural Equation Modelling

Structural equation modelling (SEM) comprises two components, a measurement

model (derived from confirmatory factor analysis) and a structural model (derived from path analysis). First, a measurement model relates one or several observed variables to a latent variable, which refers to a theoretical construct that cannot be directly measured. This origin is traced to Spearman (1904) who developed the techniques of factor analysis.⁵⁵⁰ Second, a structural model specifies relation among these latent variables and regressions of latent variables on observed variables. This origin comes from path analysis proposed by Wright (1921).⁵⁵¹ Factor analysis and path analysis were integrated by Joreskog (1970), who developed the first SEM software LISREL.⁵⁵² The 1980s and 1990s witnessed the development of computer programs and rapid expansion of SEM techniques. For example, Muthen (1984) developed the software Mplus and extended the applications of SEM to non-normal data (e.g. categorical variables).⁵⁵³ SEM has also been applied in advanced statistical literature, such as generalized linear models or multilevel analysis.

4.6.1 Introduction to path analysis

Path analysis is a statistical technique that uses both bivariate analysis and linear regression analysis to test causal relation among the variables specified in a model. It involves 3 major steps: (1) a path diagram is drawn based on a theory or a set of hypotheses; (2) path coefficients are calculated using regression techniques; (3) total effect is decomposed into direct and indirect effects. Justification for adopting path analysis in this thesis is that, compared to regression analysis, path analysis allows simultaneous examination of several causal processes underlying observed relationships and comparison of relative importance of each path. Here, the single–mediator model is adopted to introduce path analysis and mediation analysis (Figure 4.3) with the following three equations:

 $Y = i_1 + c X + e_1$ $Y = i_2 + c' X + b M + e_2$

 $M = i_3 + a X + e_3$

Where Y is the dependent variable, X is the antecedent variable, and M is the mediator variable. Path coefficients (a, b and c') obtained by regression analyses are numerical estimates of the causal relationships between variables; they are interpreted as the amount of expected change in dependent variable due to one–unit change in independent variable. The coefficient c represents the strength of prediction of Y from X, with the strength of M–to–Y relation removed. Next, b is the coefficient for the strength of prediction of Y from M, with the strength of X–to–Y relation removed; a is the coefficient for the strength of prediction of Y from the strength of prediction of M from X. The intercepts in each equation (i₁, i₂ and i₃) represent the average score of each variable, respectively. The errors in each equation (e₁, e₂ and e₃) refer to the part of relationships that cannot be predicted.⁵⁵⁴ Figure 4.3 The single–mediator model adopted to introduce path analysis



In the model, total effect of X on Y (c) is decomposed into two parts: direct effect and indirect effect. First, a direct effect of X on Y with the strength of mediated relation removed, is represented and quantified by c'. Second, an indirect (mediated) effect of X on Y transmitted via the mediator variable, is quantified by (ab) or (c – c'). The numerical values of the mediated effect is computed by either the product of coefficients (ab) or the difference in coefficients (c – c').⁵⁵⁵

A mediated effect should be evaluated by the effect size and the statistical

significance. Measures of effect size provide an indication of the size and meaningfulness of an effect that does not depend on sample size. First, a unstandardized or standardized path coefficient can serve as an effect size measure for the path. Second, another common measure for effect size is the proportion of total effect that is mediated (ab / ab + c'); for instance, a researcher can state that a mediated effect explains 30% of the total effect of a predictor on an outcome.

The tests for statistical significance aid in the evaluation of whether a mediated effect is larger than expected by chance alone. One can test the null hypothesis that the indirect effect coefficient is zero in the population from which the sample data were drawn. For significance testing, test statistic is computed by dividing the product of coefficients (indirect effect) by its standard error. Sobel (1982) derived the asymptotic standard error of indirect effect using the multivariate delta method; the standard error of the mediated effect is:⁵⁵⁶

$$\sqrt{\alpha^2 \sigma_\beta^2 + \beta^2 \sigma_\alpha^2}$$

Where α is the unstandardized regression coefficient for predicting M from X, σ_{α}^{2} is the standard error for that coefficient, β is the unstandardized regression coefficient for predicting Y from M controlling for X, and σ_{β}^{2} is the standard error for that coefficient. The indirect effect is divided by the standard error, which is then compared to a standard normal distribution to test for significance (*H*o: $\alpha\beta = 0$).⁵⁵⁷

The product of two normally distributed random variables is normally distributed only in special cases, which explains the inaccuracy for assessing significance of mediation based on normal distribution (e.g. Sobel test).⁵⁵⁸ The simulation study showed that sample sizes of 1,000 were needed for product of coefficients methods to have Type I error rates below 0.05 and adequate power to detect small effects.⁵⁵⁹ Due to inaccuracy in assessing significance of mediation based on normal distribution, MacKinnon et al recommended to evaluate significance testing by "distribution of

product" approach or "bootstrap method" in the studies with small sample sizes or with more complicated models (e.g. multiple mediators or categorical outcomes).⁵⁶⁰ These two approaches are discussed below.

First, "distribution of product" approach bases inference on a mathematical derivation of the distribution of product of two normally–distributed variables; the distribution of product of regression coefficients is often asymmetric with high kurtosis. MacKinnon et al conducted extensive simulations to estimate the empirical sampling distribution of product. On the basis of these empirical sampling distributions, critical values of the product distribution for different significance levels were determined.⁵⁶¹

Second, "bootstrap method" is a nonparametric resampling procedure widely used for testing mediation; it does not impose the assumption of normal distribution. The bootstrap method is a computationally intensive method that involves repeatedly sampling from the data set and estimating the indirect effect in each resampled data set. By repeating this process typically at least 1,000 times, an empirical approximation of sampling distribution of the product of regression coefficients is built and used to estimate confidence intervals and significance levels for the indirect effect.⁵⁶² In this thesis, I decided to test the significance of indirect effect by bootstrap method, because these SEM models are complicated models (categorical outcomes and multiple mediators) and the assumption of normal distribution may not be met.

4.6.2 Introduction to confirmatory factor analysis

As mentioned earlier, SEM comprises two components, a measurement model (derived from confirmatory factor analysis) and a structural model (derived from path analysis). Confirmatory factor analysis (CFA) analyzes a priori measurement model in which the number of factors and their correspondence with indicators are explicitly specified. A measurement model with one factor and three indicators is adopted to introduce parameter estimation in CFA (Figure 4.4).



Figure 4.4 A measurement model with one factor and three indicators

In the above measurement model, a latent continuous variable (factor η) is denoted by the three observed variables (indicators y_1 , y_2 and y_3). This measurement model is also expressed by the following 3 equations

$$y_1 = a_1 + \lambda_1 \eta + e_2$$

 $y_2 = a_2 + \lambda_2 \eta + e_2$

 $y_3 = a_3 + \lambda_3 \eta + e_3$

Where a_1 to a_3 are intercepts that give the expected value of each y when the latent variable η is zero; e_1 to e_3 are unique variances. Factor loadings (λ_1 , λ_2 and λ_3) are used to estimate the direct effects of a factor on each indicator, respectively, and are interpreted as regression coefficients. For example, if unstandardized factor loading is 1.5 for direct effect from the factor to the indicator, then 1–unit increase in the factor is associated with 1.5 unit of increase in the indicator. In general, one factor loading is fixed to 1.0 to scale the corresponding factor in unstandardized solution and is not tested for statistical significance due to no standard error.

Standardized factor loadings are estimated correlation between the indicators and its factor, when indicators are specified to load on a single factor. Therefore, squared standardized factor loadings are proportions of explained variance (R²). For instance,

if a standardized factor loading is 0.7, the factor explain 49% ($0.7^2 = 0.49$) of the variance of the indicator. Ideally, a CFA model can explain the majority of the variance ($\mathbb{R}^2 > 0.50$) of each indicator (standardized factor loading > 0.7).

Each measurement error term (e_1 , e_2 , and e_3) for each indicator represents unique variance, which is the indicator variance not explained by the factor. Like disturbances in path analyses, measurement errors are proxy variables for all sources of residual variation not explained by the model. Two types of unique variance are: random error (score unreliability) and all sources of systematic variance not due to the factor. The ratio of an unstandardized measurement error variance over the observed variance of the corresponding indicator equals the proportion of unexplained variance, and one minus this ratio is the proportion of explained variance (R^2 , squared standardized factor loadings). Thus, unique variance or proportion of unexplained variance is estimated as $(1 - R^2)$.

4.6.3 Basic steps in Structural Equation Modelling

Kline proposed that six basic steps are required in the applications of SEM.⁵⁶³ Specification, the first step, means representation of a researcher's hypotheses in the form of a structural equation model. A researcher assumes the relationships among observed variables and latent variables based on literature, and draws a measurement model (confirmatory factor analysis) and a structural model (path analysis) to represent the presumed relationships. Specification includes selection of variables, directionality of causal effects, parameter status, or type of structural models. Specification requires a series of thoughtful decision; for example, these options should be considered when specifying the directionality of a causal effect: (1) to specify and test alternative models, each with different causal directionality between the two variables; (2) to include reciprocal effects to cover both possibilities; (3) to specify a model without directionality between the two variables, which are specified to be

correlated with each other.

The second step, identification, means going from the known information to the unknown parameters. A model is "identified" if it is theoretically possible to derive a unique estimate for every parameter (unknown information) based on the number of elements in observed variance – covariance matrix (known information). Identification is a property of a model, rather than the data. The difference between the known versus unknown information typically equals degrees of freedom, which should not be less than zero. There are several rules of identification for different structural and measurement models that should be paid attention before conducting SEM.

The third step is measure selection and data collection. A researcher should select a good measure for intended construct based on score reliability and validity. Before analyzing SEM, original data should be screened for problems of co–linearity, outlier, missing data, and normality. The default estimation in SEM is maximum likelihood (ML) which assumes all variables are continuous and normally distributed. If this assumption is violated (e.g. categorical outcomes), a researcher needs to consider other techniques described later.

The fourth step, estimation, involves using an SEM computer tool to conduct the following analyses: (1) Evaluation for model fit, which means to determine how well the model explains the data. If the initial model does not fit the data very well, the researcher should go to the fifth step "re-specification". (2) Interpretation for parameter estimates, which includes magnitude, directionality and significance of each path coefficient, or decomposition of total effect. (3) Consider an equivalent model, which explains the data as well as the researcher's preferred model with a different configuration of hypothesized relationships among the same variables.

The fifth step is re-specification. A researcher usually arrives at this step because the fit of the initial model is poor. Then, a new model should be re–specified based on theoretical consideration. Any re–specified model should go through the previous steps from identification to estimation.

The sixth step is to completely and accurately describe the SEM analysis in written reports. Researchers can refer to published guidelines for reporting results of SEM.⁵⁶⁴

The evaluation of model fit in the fourth step is introduced in the following paragraphs. The original test for overall model fit is chi-squared test, a test of whether the covariance matrix implied by hypothesized model is close enough to sample covariance matrix. However, chi-squared test is too sensitive to sample size. In a large sample (> 5000), chi-squared test may be significant even though only trivial differences exist between observed and predicted covariance.

RMSEA (Root Mean Square Error of Approximation) is a parsimony-adjusted index that theoretically follows a non-central chi-squared distribution, where noncentrality parameter allows for discrepancies between model-implied and sample covariance up to the level of expected values of chi-squared or degrees of freedom. RMSEA is scaled as a badness-of-fit index where declining values indicate improving fit (zero for a perfect fit). RMSEA less than 0.06 indicates "good fit", and RMSEA 0.06– 0.08 is considered "acceptable fit".

CFI (Comparative Fit Index) is an incremental fit index which measures the relative improvement in the fit of hypothesized model compared to that of a baseline model (independent model that assumes zero covariance among observed variables). TLI (Tucker-Lewis Index) compares the mean square (the sum of squares divided by degrees of freedom) for the hypothesized model to that for a baseline model. TLI gives the distance between the baseline and target models as a proportion of distance between baseline and true models. Both CFI and TLI are goodness-of-fit indexes where increasing values indicate better fit (ranged between 0 and 1).⁵⁶⁵

Researchers have not reached common consensus on the criteria for model fit. Hu and Bentler proposed a set of practical criteria for "good fit" of the model, including: RMSEA less than 0.06, CFI more than 0.95, and TLI more than 0.95. In this thesis, the proposed cutoff values are adopted as the guidelines for overall fit of the model.⁵⁶⁶

4.6.4 Structural Equation Modelling in cohort studies

SEM in longitudinal data has a rich history with the development of new analytical models and technological innovations to collect data over time. Autoregressive models were seen as the "gold standard" of methodology for analyzing longitudinal data prior to the development of latent growth curve models.⁵⁶⁷ To date, autoregressive models are still useful for many important questions in longitudinal studies. "Autoregressive" means regressed on itself, so each variable is predicted by the same variable at an early wave; the stability of this variable (the extent to which the mean of a measure is the same across time) is evaluated by the strength of path coefficient of the same variable, any unmeasured exogenous variable (confounder) that correlates with the predictor and the dependent variable can be controlled.

An autoregressive model simply assesses how a construct changes over time. The common application of an autoregressive model with a cross–lagged panel design examines how a construct changes and covariates across time "with other constructs". The cross–lagged panel design involves the following features. First, each variable is modeled with an autoregressive structure; for example, OC at wave 2 is predicted by OC at wave 1. Second, longitudinal relation consistent with longitudinal mediation are present among the variables. For instance, the cross–lagged effects of "OC at wave 1 on ERI at wave 2" and "ERI at wave 1 on OC at wave 2" are measured. These cross–lagged relationships help to identify the directionality of potentially causal relationships. Third, although this model would not include contemporaneous causal relation among variables at the same wave, covariances among the variables at wave 1 and covariances among residual variances of the variables at wave 2 are included. The model recognizes that there are correlation among variables at the same wave but direction of relationships are unknown (Figure 4.5).⁵⁶⁸

Figure 4.5 An autoregressive model with a cross-lagged panel design



e = residual variance of the variable

In terms of mediation analyses, ideally, a 3-wave cohort design is required to estimate a mediation model; the exposure variable should precede the mediator in time, and the mediator should precede the outcome in time. For example, if the mediation process is suggested as $OC \rightarrow ERI \rightarrow$ outcome, the effect of OC at wave 1 on ERI at wave 2 (X), and the effect of ERI at wave 2 on outcome at wave 3 (Y) would be assessed. This mediator effect of ERI is estimated by multiplying the two cross-lagged effects (X * Y) (Figure 4.6). However, multiphase longitudinal studies are relatively rare in occupational health research, and a 2-wave cohort design still can test the significance of partial mediation.





The distinction between partial mediation and full mediation is summarized. Baron and Kenny proposed a simple procedure to test whether a variable B acts as a mediator of the effect of a predictor A on an outcome C. The following assumptions need to be satisfied: (1) the association between A and C is statistically significant; (2) A and B are related; (3) B is significantly associated with C, after control for A; (4) the association *ac* between A and C is weaker when B is controlled, compared with the situation when B is not controlled. If *ac* becomes not significant after control for B, B fully mediates the relationship between A and C. If *ac* is weaker but still significant, B partially mediates the relationship between A and C.⁵⁶⁹



Figure 4.7 A two–wave cohort design to estimate a mediation model

In terms of a 2–wave cohort design, Taris and Kompier (2006) proposed that as mediation is a causal chain involving at least two causal relationships, these causal relationships can be tested separately using only two phases. For example, if the mediation process is suggested as: $OC \rightarrow ERI \rightarrow$ outcome, the effect of OC at wave 1 on ERI at wave 2 (X) and the effect of ERI at wave 1 on outcome at wave 2 (Z) can be assessed (Figure 4.7). Partial mediation applies if both causal relationships are confirmed; the product of two path coefficients (X * Z) can estimate the strength of the mediator effect.⁵⁷⁰ Note that full mediation cannot be examined in a two–wave design; with only 2 phases of data, it is impossible to test whether the relationship between OC at wave 1 and outcome at wave 3 (Figure 4.6) is fully mediated by ERI at wave 2.

Although a 3–wave cohort study provides the best estimation for mediation, a 2– wave cohort design can offer indication for the presence of partial mediation. The evaluation for partial mediation by a psychosocial factor is reasonable, as there are often several psychosocial factors identified to influence a health outcome. Cole and Maxwell (2004) suggested that a 2–wave cohort design yielded better evidence than a cross–sectional study or a half–longitudinal design (one of the associations between predictor, mediator and outcome is cross–sectional) based on two reasons.⁵⁷¹ First, failing to control for prior levels of dependent variables often creates problems of unmeasured confounders. Second, a cross–sectional design is limited by difficulties in determining causal sequence. In a cross–sectional mediation study with 3 variables, for example, there are 6 possible causal sequences ($3 \times 2 \times 1$). Many researchers test only one causal sequence that fits their proposed theory, but the other sequences are neglected. Indeed, it is difficult to argue a priori that one particular causal sequence is plausible while others are not, if temporal sequence cannot be determined.⁵⁷²

4.6.5 Structural Equation Modelling with categorical outcomes

The most common method used in SEM to account for nonlinear relationship between observed categorical variables with a latent continuous variable is summarized; the hypothetical model is illustrated (Figure 4.8). A single latent variable (factor η) underlies the relationship among 3 observed categorical variables (y_1 , y_2 and y_3). To solve the problem of nonlinear relationships, it is assumed that there are 3 continuous latent response variables (y_1^* , y_2^* and y_3^*) underlying each observed categorical variable. Note that factor loadings are represented by λ_1 , λ_2 and λ_3 ,

respectively; 1–unit increase in the latent factor (η) results in λ_1 , λ_2 and λ_3 unit of increase in latent response variables y_1^* , y_2^* and y_3^* , respectively. The next question is how to obtain the latent response variable from the observed categorical variable.⁵⁷³ Figure 4.8 A measurement model for relationships between a latent continuous factor and three observed categorical variables



Muthen described the approach to estimating SEM models with any combination of continuous, dichotomous or ordinal categorical outcomes.^{574,575} In the formulation, a continuous latent response variable y* expresses the amount required to respond in certain categories of an observed categorical variable. For simplicity, consider the one factor model for the continuous latent response variable y*_i for individual i:

$$y_i^* = \nu + \lambda \ \eta_i + \epsilon_i,$$

Where v is an intercept parameter, λ is a factor loading, η is a factor variable, and ϵ is a residual. The expectation μ^* and variance σ^* of y* are

$$\mu^* = \nu + \lambda \ \alpha,$$

$$\sigma^* = \lambda^2 \ \psi + \theta,$$

Where α is the mean of η , ψ is the variance of η , and θ is the variance of the residual ϵ . Next, the latent response variable y* is related to the observed categorical variable y via the threshold model:

$$y_{i} = \begin{cases} C_{i} - 1, & \text{if } \tau_{i, C_{i} - 1} < y_{i}^{*} \\ C_{i} - 2, & \text{if } \tau_{i, C_{i} - 2} < y_{i}^{*} \le \tau_{i, C_{i} - 1} \\ \vdots \\ 1, & \text{if } \tau_{i, 1} < y_{i}^{*} \le \tau_{i, 2} \\ 0, & \text{if } y_{i}^{*} \le \tau_{i, 1} \end{cases}$$

Where C denotes the number of categories in an observed categorical variable; τ_i denotes the location of the cut point (called threshold), with $\tau_0 = -\infty$ and $\tau_c = \infty$. This also leads to the conditional probability expression:

$$P(y \ge c|\eta) = F[-(\tau_c - \nu - \lambda \eta) \ \theta^{-1/2}],$$

Where *F* is typically chosen as a standard normal (probit) or logistic distribution function depending on the distributional assumption for ϵ . By the above procedures, an observed categorical variable is linked with a latent response variable by a conditional probability model (probit or logistic function).⁵⁷⁶

For mediation analysis in categorical variables, probit models are preferable to logistic modes for better mathematical tractability in rescaling and estimation. Probit models assume that transformation function is the cumulative density function of standard normal distribution, but logit models assume that transformation function is the logistic function. Probit and logit models are almost identical and they produce similar results; however, their β coefficients are scaled differently. In probit models, random error is assumed to be distributed normally with variance 1; in logit models, random error is assumed to be distributed logistically with variance $\pi^2/3.^{577}$ That is, probit analysis sets the standard deviation of error as 1, but logit analysis sets the standard deviation of error as 1, but logit analysis sets the standard deviation of error as 1, but logit analysis sets the standard deviation of error as 1.814 (π / $\sqrt{3}$). Thus, probit coefficient multiplied by 1.8 is approximately the same as logit coefficient, but they vary slightly due to small

differences between logistic distribution and standard normal distribution.578

For path analysis in linear regression, total effects (c) can be decomposed into direct effect (c') and indirect effects (a x b). In logit or probit models, (c – c') is not always equal to (a x b) due to fixed residual variance.⁵⁷⁹ Importantly, Mplus software uses probit regression to estimate thresholds for categorical outcomes and provides more complicated iterative approaches based on multivariate probit distribution; as the model is standardized as part of the analysis, scaling problems can be solved. Thus, (c – c') becomes approximately equal to (a x b). However, this solution cannot make mediation analysis in logit or probit models as accurate as linear regression. Probit models produce more similar results to linear regression than logit models.⁵⁸⁰

Another challenge for SEM with categorical variables is estimation; maximum likelihood (ML) estimator assuming that data follow normal distribution is not accurate. One solution is limited–information estimator which uses a summary (e.g. variance) of available data. To correct for non-normal distribution of observed data, a weight matrix is used in conjunction with a least squares estimator, which chooses parameter values to minimize the distance between what is observed (data) and what is expected (model-implied covariance matrix); this combination is weighted least squares (WLS). The data matrix is an asymptotic correlation matrix of latent response variable: tetrachoric correlation matrix for dichotomous variables and polychoric correlation matrix for ordinal categorical variables. In Mplus, WLS estimator applied to ordered categorical outcomes is mean– and variance–adjusted weighted least squares (WLSMV) in probit models. WLSMV generally performs well if the sample size is larger than 200 and the distribution on ordered variables is not markedly skewed.⁵⁸¹

Chapter 5. Drinking Outcomes

The aims of analyses for drinking outcomes in Chapter 5, in line with the aims listed in Chapter 3, include: (1) to examine the potential role of OC in ERI–drinking relationship, including modifying, antecedent, mediator, or direct effects; (2) to investigate the potential role of PC in the relationship between OC, ERI and drinking outcomes. The analyses use data from a 2–wave cohort study (3782 men and 3731 women aged 45–69), part of the HAPIEE study, which has been described in detail in Chapter 4 Methodology.

The results are presented in three parts. First, descriptive statistics for covariates and drinking outcomes by country and by gender are presented. Second, the associations of ERI and OC at wave 1 with drinking outcomes at wave 2, respectively, are assessed by logistic regression. Structural equation modelling (SEM) with an autoregressive and cross–lagged model is applied to examine antecedent or mediator roles of OC in ERI–drinking relationship. Modifying roles of OC in ERI–drinking relationship are also tested. Third, the associations between PC at wave 1 and drinking outcomes at wave 2 are evaluated by logistic regression. SEM with an autoregressive and cross–lagged model is applied to examine antecedent or mediator roles of OC in ERI–drinking relationship. Modifying roles of OC in ERI–drinking and drinking outcomes at wave 2 are evaluated by logistic regression. SEM with an autoregressive and cross–lagged model is used to examine the relationship between OC, ERI, PC, and drinking outcomes. Modifying effects of PC in ERI–drinking relation are also tested.

5.1 Descriptive Statistics

5.1.1 Descriptive characteristics of study populations

In this sample of 7513 subjects (3782 men and 3731 women), the means of age at wave 1 are 54.8 years (standard deviation= 6.0) in men and 53.2 years (standard deviation= 5.4) in women. The average follow–up periods between wave 1 and wave 2 are 3.5 years (standard deviation= 0.7) in men and 3.6 years (standard deviation=

0.6) in women. Descriptive statistics with percentages and means for covariates by country and by gender are presented (Table 5.1).

	Czech Repu	blic	Russia		Poland	
Variable	Men	Women	Men	Women	Men	Women
	(n= 1082)	(n= 1099)	(n= 1402)	(n= 1394)	(n= 1298)	(n=1238)
Age, N (%)						
45 – 49	279 (25.8)	372 (33.9)	300 (21.4)	408 (29.3)	357 (27.5)	461 (37.2)
50 – 54	306 (28.3)	422 (38.4)	353 (25.2)	453 (32.5)	380 (29.3)	423 (34.2)
55 – 59	313 (28.9)	202 (18.4)	387 (27.6)	314 (22.5)	317 (24.4)	230 (18.6)
60 – 69	184 (17.0)	103 (9.3)	362 (25.8)	219 (15.7)	244 (18.8)	124 (10.0)
Education, N (%)						
Primary or less	30 (2.8)	82 (7.5)	81 (5.8)	54 (3.9)	45 (3.5)	47 (3.8)
Vocational	391 (36.1)	277 (25.2)	324 (23.1)	454 (32.6)	256 (19.7)	141 (11.4)
Secondary	382 (35.3)	547 (49.8)	464 (33.1)	395 (28.3)	396 (30.5)	517 (41.8)
University	279 (25.8)	193 (17.6)	533 (38.0)	491 (35.2)	601 (46.3)	532 (43.0)
Occupation, N (%)						
Manager/ profession	295 (27.3)	186 (16.9)	391 (27.9)	276 (19.8)	393 (30.3)	223 (18.0)
Non-manual worker	461 (42.6)	727 (66.2)	479 (34.2)	843 (60.5)	650 (50.1)	846 (68.3)
Manual worker	326 (30.1)	186 (16.9)	531 (37.9)	275 (19.7)	254 (19.6)	170 (13.7)
Marital status, N (%)						
Married/ cohabiting	926 (85.6)	790 (71.9)	1279 (91.2)	885 (63.5)	1188 (91.5)	875 (70.7)
Single	30 (2.8)	31 (2.8)	35 (2.5)	85 (6.1)	43 (3.3)	104 (8.4)
Divorce/ widowed	126 (11.6)	278 (25.3)	88 (6.3)	424 (30.4)	67 (5.2)	259 (20.9)
Deprivation, N (%)						
Low (0 – 3.9)	946 (87.4)	909 (82.7)	1011 (72.1)	740 (53.1)	1057 (81.4)	929 (75.0)
High (4 – 9)	136 (12.6)	190 (17.3)	391 (27.9)	654 (46.9)	241 (18.6)	309 (25.0)
Depression, N (%)						
CESD < 16	974 (90.0)	885 (80.5)	1214 (86.6)	1013 (72.7)	1118 (86.1)	935 (75.5)
CESD >= 16	108 (10.0)	214 (19.5)	188 (13.4)	381 (27.3)	180 (13.9)	303 (24.5)
Social isolation, N (%)						
No (>= once a month)	699 (64.6)	789 (71.8)	632 (45.1)	655 (47.0)	597 (46.0)	614 (49.6)
Yes (< once a month)	383 (35.4)	310 (28.2)	770 (54.9)	739 (53.0)	701 (54.0)	624 (50.4)
Self-rated health, N (%)						
Very good – average	1019 (94.2)	1045 (95.1)	1290 (92.0)	1147 (82.3)	1220 (94.0)	1165 (94.1)
Poor – very poor	63 (5.8)	54 (4.9)	112 (8.0)	247 (17.7)	78 (6.0)	73 (5.9)

Table 5.1 Descriptive statistics of study sample by country and gender (N= 7513)

There are gender differences across three countries. Compared to women, men have higher proportions in age group over 55, university-educated, manager/

profession and manual workers, and married/ cohabiting; men also have lower deprivation, less depressive symptoms, and more social isolation than women.

There are country differences observed in both genders (Table 5.1). For age, Russian samples are older than Czech and Polish samples. University education in Czech samples is the least common of all countries, and Czech samples have the greatest gender inequality in education. High deprivation in Russia is the most prevalent of all countries. The proportions of depression and social isolation in Czech Republic are the lowest among all countries.

5.1.2 Descriptive characteristics of drinking outcomes

Descriptive statistics of three drinking outcomes (binge drinking, heavy drinking, and problem drinking) by country and by gender at wave 1 and 2, respectively, are shown (Table 5.2). There are gender differences across three countries. Men have remarkably higer proportions in binge drinking, heavy drinking, and problem drinking than women.

In addition, there are country differences seen in Table 5.2. At wave 1, the reported levels of annual alcohol intake in Czech men and women are about 1.5–2.0 times the levels in Russian and Polish people. The percentages of binge drinking are highest for Russian sample (14.3%) in men and highest for Czech sample (4.9%) in women. The percentages of heavy drinking are highest for Russian sample in men (38.7%) and women (16.3%). The percentages of problem drinking are highest for Russian sample in men (18.6%) and highest for Czech sample in women (4.1%). At wave 2, the highest percentages of three drinking outcomes in men and women, respectively, are in the same countries as those at wave 1.

	Czech		Russia		Poland	
Variables	Men	Women	Men	Women	Men	Women
	(n= 1082)	(n= 1099)	(n= 1402)	(n= 1394)	(n= 1298)	(n=1238)
Wave 1						
Annual alcohol intake, mean (g)	6146.1	1683.5	4992.8	756.7	3628.8	769.1
Annual drinking occasion, mean	142.6	64.5	82.8	24.7	94.5	34.8
Dose per occasion, mean (g)	43.1	26.1	60.3	30.6	38.4	22.1
1. Binge drinking, n (%)						
Yes	81 (7.5)	54 (4.9)	200 (14.3)	51 (3.7)	52 (4.1)	40 (3.2)
No	1001 (92.5)	1045 (95.1)	1202 (85.7)	1343 (96.3)	1245 (95.9)	1198 (96.8)
2. Heavy drinking, n (%)						
Yes	237 (21.9)	136 (12.4)	543 (38.7)	227 (16.3)	218 (16.8)	124 (10.0)
No	845 (78.1)	963 (87.6)	859 (61.3)	1167 (83.7)	1080 (83.2)	1114 (90.0)
3. Problem drinking, n (%)						
Yes	118 (10.9)	45 (4.1)	261 (18.6)	51 (3.6)	131 (10.1)	38 (3.1)
No	964 (89.1)	1054 (95.9)	1141 (81.4)	1343 (96.4)	1167 (89.9)	1200 (96.9)
Wave 2						
Annual alcohol intake, mean (g)	6024.5	1480.8	4625.3	809.1	4008.6	871.9
Annual drinking occasion, mean	130.4	58.3	79.2	25.8	94.1	37.1
Dose per occasion, mean (g)	46.2	25.4	58.4	31.4	42.6	23.5
1. Binge drinking, n (%)						
Yes	98 (9.1)	49 (4.5)	195 (13.9)	59 (4.2)	83 (6.4)	46 (3.7)
No	984 (90.9)	1050 (95.5)	1207 (86.1)	1335 (95.8)	1215 (93.6)	1192 (96.3)
2. Heavy drinking, n (%)						
Yes	306 (28.3)	178 (16.2)	513 (36.6)	287 (20.6)	301 (23.2)	183 (14.8)
No	776 (71.7)	921 (83.8)	889 (63.4)	1107 (79.4)	997 (76.8)	1055 (85.2)
3. Problem drinking, n (%)						
Yes	95 (8.8)	47 (4.3)	269 (19.2)	58 (4.2)	153 (11.8)	47 (3.8)
No	987 (91.2)	1052 (95.7)	1133 (80.8)	1336 (95.8)	1145 (88.2)	1191 (96.2)

Table 5.2 Descriptive statistics of drinking outcomes by country and gender

The ways of pooling the data in subsequent analyses are described. First, men and women are analysed separately as most studies on the associations between psychosocial factors and health outcomes. Second, crude associations between exposure variables (ER ratio and OC) at wave 1 and drinking outcomes at wave 2 in country–specific strata are assessed. Next, logistic regression analyses are conducted for 3 drinking outcomes, respectively, regressed by country, ER–ratio tertile and interaction term between country and ER–ratio tertile. By comparing the log likelihoods for the model with this interaction term and the model without, likelihood–ratio (LR) test is used to test the significance of this interaction term (Table 5.3). In a similar way, the interaction term between country and OC tertile is evaluated (Table 5.4).

There are country differences seen in Table 5.3. In men, the association between ER ratio and binge drinking in Czech Republic is the strongest of all countries. In women, the association between ER ratio and heavy drinking in Poland is the strongest of all countries; the association between ER ratio and problem drinking in Russia is stronger than other countries. Overall, crude associations between ER ratio at wave 1 and drinking outcomes at wave 2 are not very different across country–specific strata (all p–values by LR test > 0.12); no significant interaction between country and ER ratio is found.

 Table 5.3
 Crude associations between ER ratio and drinking outcomes in country–

specific strata

Strata	ER ratio	Heavy drinking	Binge drinking	Problem drinking
	Tertile▲	OR (95% CI)	OR (95% CI)	OR (95% CI)
Men				
Czech	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.31 (0.91 – 1.89)	1.78 (0.89 – 3.55)	1.25 (0.65 – 2.41)
	Tertile 3	1.63 (1.16 – 2.29)*	3.24 (1.72 – 6.12)*	1.96 (1.08 – 3.55)*
	P for trend	0.004	< 0.001	0.014
	OR by 1 tertile	1.27 (1.07 – 1.50)*	1.81 (1.35 – 2.41)*	1.43 (1.07 – 1.91)
Russia	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.26 (0.99 – 1.60)	1.75 (1.22 – 2.51)*	1.35 (1.00 – 1.83)
	Tertile 3	1.53 (1.15 – 2.03)*	2.81 (1.90 - 3.95)*	2.56 (1.84 - 3.56)
	P for trend	0.002	< 0.001	< 0.001
	OR by 1 tertile	1.24 (1.08 – 1.42)*	1.68 (1.38 – 2.04)*	1.58 (1.34 – 1.87)'
Poland	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.27 (0.90 – 1.78)	1.14 (0.61 – 2.15)	1.86 (1.09 – 3.17)
	Tertile 3	1.47 (1.08 – 2.01)*	1.87 (1.08 – 3.23)*	3.16 (2.03 – 4.27)
	P for trend	0.016	0.017	< 0.001
	OR by 1 tertile	1.21 (1.04 – 1.41)*	1.40 (1.06 – 1.84)*	1.79 (1.43 – 2.25)
Interaction	,	(, , , , , , , , , , , , , , , , , , ,	(,	(, , , , , , , , , , , , , , , , , , ,
country x ERI	LR test	P= 0.984	P= 0.673	P= 0.701
Women				
Czech	Tertile 1	1.00	1.00	1.00
	Tertile 2	0.85 (0.58 – 1.25)	1.53 (0.67 – 3.50)	1.52 (0.64 – 3.64)
	Tertile 3	1.23 (0.86 – 1.78)	2.71 (1.25 – 5.73)*	1.94 (0.85 - 4.41)
	P for trend	0.173	0.004	0.111
	OR by 1 tertile	1.14 (0.95 – 1.37)	1.64 (1.14 – 2.36)*	1.37 (0.93 – 2.02)
Russia	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.04 (0.78 – 1.40)	1.68 (0.99 – 2.83)	1.64 (0.95 – 2.79)
	Tertile 3	1.08 (0.81 – 1.42)	1.93 (1.09 – 3.40)*	2.74 (1.24 – 5.88)
	P for trend	0.726	0.017	0.004
	OR by 1 tertile	1.04 (0.89 – 1.20)	1.39 (1.06 – 1.83)*	1.63 (1.16 – 2.26)
Poland	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.65 (1.13 – 2.43)*	1.88 (0.81 – 5.49)	1.08 (0.46 – 2.52)
	Tertile 3	1.86 (1.30 – 2.66)*	2.67 (1.11 – 5.95)*	1.99 (0.98 – 4.07)
	P for trend	0.001	0.012	0.040
	OR by 1 tertile	1.33 (1.12 – 1.58)*	1.63 (1.06 – 2.45)*	$1.46(1.02 - 2.09)^{\circ}$
			· · · · · · · · · · · · · · · · · · ·	
Interaction				

▲ Country– and gender–specific tertiles of ER ratio. * P value < 0.05.

Country differences are found in Table 5.4. In men, the association between OC and binge drinking in Czech Republic is stronger than other two countries. In women, the association between OC and binge drinking in Poland is the strongest of all countries; the association between OC and problem drinking in Poland is stronger than other two countries. Overall, crude associations between OC at wave 1 and drinking outcomes at wave 2 are not very different across country–specific strata (all p–values

by LR test > 0.16). Due to no significant interactions between country and exposure variables (ER ratio and OC), data for three countries are pooled for further analyses. Table 5.4 Crude associations between OC and drinking outcomes in country–

Strata	OC score Tertile▲	Heavy drinking OR (95% CI)	Binge drinking OR (95% CI)	Problem drinking OR (95% CI)
Men	Terme	01((3578 01)		
Czech	Tertile 1	1.00	1.00	1.00
020011	Tertile 2	0.97 (0.71 – 1.33)	1.40 (0.79 – 2.48)	1.53 (0.84 – 2.77)
	Tertile 3	1.22 (0.88 – 1.69)	2.52 (1.42 – 4.38)*	2.17 (1.21 – 3.91)*
	P for trend	0.203	< 0.001	0.008
	OR by 1 tertile	1.11 (0.94 – 1.31)	1.62 (1.24 – 2.13)*	1.47 (1.10 – 1.95)*
Russia	Tertile 1	1.00	1.00	1.00
1 tuoola	Tertile 2	1.14 (0.89 – 1.47)	1.49 (1.04 – 2.13)*	1.32 (0.97 – 1.80)
	Tertile 3	1.26 (0.95 – 1.67)	1.71 (1.15 – 2.50)*	1.55 (1.11 – 2.16)*
	P for trend	0.090	0.003	0.006
	OR by 1 tertile	1.12 (0.98 – 1.29)	1.32 (1.09 – 1.59)*	1.25 (1.07 – 1.47)*
Poland	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.06 (0.79 – 1.42)	1.03 (0.59 – 1.80)	1.51 (0.99 – 2.31)
	Tertile 3	1.16 (0.85 – 1.59)	1.08 (0.63 – 1.81)	2.10 (1.36 – 3.23)*
	P for trend	0.361	0.887	0.001
	OR by 1 tertile	1.08 (0.92 – 1.26)	1.04 (0.79 – 1.35)	1.45 (1.17 – 1.79)*
Interaction	,			- (- /
country x OC	LR test	P= 0.922	P= 0.162	P= 0.788
Women				
Czech	Tertile 1	1.00	1.00	1.00
	Tertile 2	0.95 (0.65 – 1.39)	0.90 (0.47 – 1.72)	1.58 (0.71 – 3.56)
	Tertile 3	1.04 (0.73 – 1.47)	1.40 (0.72 – 2.69)	1.91 (0.85 – 4.39)
	P for trend	0.803	0.276	0.118
	OR by 1 tertile	0.99 (0.82 – 1.18)	1.18 (0.84 – 1.67)	1.37 (0.92 – 2.02)
Russia	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.05 (0.79 – 1.39)	1.17 (0.69 – 1.99)	1.39 (0.74 – 2.62)
	Tertile 3	1.17 (0.87 – 1.57)	1.22 (0.71 – 2.10)	1.47 (0.79 – 2.81)
	P for trend	0.594	0.485	0.237
	OR by 1 tertile	1.04 (0.90 – 1.19)	1.10 (0.85 – 1.42)	1.20 (0.89 – 1.63)
Poland	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.03 (0.74 – 1.44)	1.12 (0.48 – 2.61)	1.22 (0.55 – 2.70)
	Tertile 3	1.33 (0.94 – 1.87)	2.54 (1.16 – 5.42)*	2.75 (1.28 – 5.76)*
	P for trend	0.118	0.016	0.004
	OR by 1 tertile	1.15 (0.97 – 1.37)	1.62 (1.09 – 2.42)*	1.66 (1.12 – 2.44)*
Interaction				
country x OC	LR test	P= 0.537	P= 0.363	P= 0.402

specific strata

▲ Country– and gender–specific tertiles of OC score. * P value < 0.05.

Bivariate analyses for the associations between covariates and three drinking outcomes are conducted. Chi–squared tests are used to examine the significance of differences between categories of the variable. P–values for heterogeneity are obtained in all categorical variables; p–values for trend are tested in ordinal categorical variables (age and education). In men, bivariate analyses between covariates at wave 1 and three drinking outcomes at wave 2 are shown (Table 5.5). More heavy drinking is significantly (p < 0.05) associated with younger age. More binge drinking is significantly associated with younger age, less education, being manual workers, and higher deprivation. More problem drinking is significantly associated with younger age, less education, being manual workers, higher deprivation, and more depression.

Table 5.5 Bivariate analyses for relationships between covariates and drinking outcomes in men (n= 3782)

Covariates	Heavy drinking (%)	Binge drinking (%)	Problem drinking (%)	
Covariates	n= 1120	n= 376	n= 517	
Age: 45 – 49	34.4	12.1	16.7	
50 – 54	28.9	8.8	13.2	
55 – 59	29.0	9.2	12.3	
60 – 69	25.1	7.7	10.1	
P for heterogeneity	< 0.001	0.006	< 0.001	
P for trend	< 0.001	0.004	< 0.001	
Education: Primary/ less	29.4	11.9	20.1	
Vocational	31.8	11.7	15.0	
Secondary	28.7	9.6	14.0	
University	28.6	7.6	10.4	
P for heterogeneity	0.299	0.004	< 0.001	
P for trend	0.150	0.001	< 0.001	
Occupation class				
Manager/ profession	29.6	9.6	11.5	
Non-manual worker	28.1	8.1	10.9	
Manual worker	31.4	11.3	18.0	
P value	0.146	0.015	< 0.001	
Marital status				
Married/ cohabiting	29.3	9.3	13.4	
Single	29.7	15.3	10.8	
Divorce/ widowed	31.3	10.0	11.3	
P value	0.760	0.098	0.455	
Deprivation: Low (0-3.9)	28.9	8.9	11.9	
High (4–9)	31.9	11.8	18.1	
P value	0.086	0.012	< 0.001	
Depression: CESD < 16	28.7	9.3	11.8	
CESD >= 16	31.8	11.3	18.2	
P value	0.121	0.179	< 0.001	
Social isolation				
No (>= once a month)	29.9	9.8	13.0	
Yes (< once a month)	29.0	9.2	13.4	
P value	0.524	0.464	0.702	
Self-rated health				
Very good – average	29.6	9.6	13.3	
Poor – very poor	27.6	8.1	11.5	
P value	0.471	0.416	0.403	
Table 5.6 reports the bivariate analyses between covariates at wave 1 and drinking outcomes at wave 2 in women. More heavy drinking is significantly (p < 0.05) associated with younger age and higher education. More binge drinking is significantly associated with younger age. Finally, more problem drinking is significantly associated with younger age and more depression.

Table 5.6 Bivariate analyses for relationships between covariates and drinking outcomes in women (n=3731)

Covariates	Heavy drinking (%)	Binge drinking (%)	Problem drinking (%)
Covallates	n= 648	n= 154	n= 153
Age: 45 – 49	23.0	6.3	4.5
50 – 54	22.4	5.1	4.1
55 – 59	20.8	3.3	3.6
60 – 69	13.2	0.7	0.7
P for heterogeneity	< 0.001	< 0.001	0.002
P for trend	< 0.001	< 0.001	0.001
Education: Primary/ less	16.2	3.5	3.0
Vocational	20.7	3.9	3.6
Secondary	20.3	4.9	4.0
University	23.8	4.9	3.5
P for heterogeneity	0.008	0.533	0.804
P for trend	0.029	0.196	0.957
Occupation class			
Manager/ profession	23.6	4.8	4.3
Non-manual worker	21.3	4.6	3.2
Manual worker	18.4	4.3	4.8
P value	0.060	0.897	0.098
Marital status			
Married/ cohabiting	20.9	4.6	3.9
Single	23.7	4.2	1.7
Divorce/ widowed	21.3	4.5	3.6
P value	0.598	0.958	0.219
Deprivation: Low (0-3.9)	21.0	4.8	3.6
High (4–9)	21.6	4.2	3.8
P value	0.705	0.410	0.843
Depression: CESD < 16	20.4	4.4	3.2
CESD >= 16	22.8	5.6	5.9
P value	0.125	0.142	< 0.001
Social isolation			
No (>= once a month)	21.8	4.8	3.8
Yes (< once a month)	20.4	4.4	3.7
P value	0.294	0.599	0.883
Self-rated health			
Very good – average	21.0	4.5	3.6
Poor – very poor	23.0	5.9	3.7
P value	0.339	0.184	0.982

5.2 Potential Role of OC in ERI–Drinking Relationship

The focus of this section is, as the first aim of the thesis, on the associations between ERI and drinking outcomes, on the associations between OC and drinking outcomes, and on assessing whether OC has the potential role of antecedent, mediator, modifier, or direct effect in the relationship between ERI and drinking. As there is no significant interaction between country and exposure variables, data for the three countries are pooled for further analyses.

5.2.1 Associations between ERI and drinking outcomes

The associations between ER ratio at wave 1 and three binary drinking outcomes (heavy drinking, binge drinking, and problem drinking) at wave 2 are assessed using three logistic regression analyses, separately for men and women. These associations are assessed after adjustment for age and country (Model 1) and after additionally adjustment for other covariates (Model 2).

Table 5.7 presents the associations between ER ratio at wave 1 and three drinking outcomes at wave 2 using three logistic regression analyses, respectively. In men, Model 1 shows that the odds of having heavy drinking, binge drinking and problem drinking are 1.41, 2.32 and 1.89, respectively, for highest versus lowest tertile of ER ratio. Model 2 reports that the odds of having heavy drinking, binge drinking and problem drinking are 1.33, 2.29 and 1.79 for highest versus lowest tertile of ER ratio. The adjusted OR changes for heavy drinking, binge drinking and problem drinking are 1.35, respectively, by 1–tertile increase in ER ratio (p < 0.05).

In women, Model 1 shows that the odds of having heavy drinking, binge drinking and problem drinking are 1.23, 1.98 and 2.16, respectively, for highest versus lowest tertile of ER ratio. Model 2 shows the odds of heavy drinking, binge drinking and problem drinking are 1.29, 2.06 and 1.82 for highest versus lowest tertile of ER ratio. The adjusted OR changes for heavy drinking, binge drinking and problem drinking are

146

1.13, 1.42 and 1.36, respectively, by 1-tertile increase in ER ratio (p < 0.05).

2

Model	ER ratio	Heavy drinking	Binge drinking	Problem drinking
	Tertile▲	OR (95% CI)	OR (95% CI)	OR (95% CI)
Men (n= 3782)				
Model 1*	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.24 (1.04 – 1.46)*	1.55 (1.16 – 2.07)*	1.25 (0.98 – 1.59)
	Tertile 3	1.41 (1.17 – 1.68)*	2.32 (1.73 – 3.09)*	1.89 (1.51 – 2.38)*
	P for trend	0.001	< 0.001	< 0.001
	OR by 1 tertile	1.19 (1.08 – 1.30)*	1.52 (1.32 – 1.75)*	1.39 (1.24 – 1.56)*
Model 2**	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.21 (1.01 – 1.46)*	1.51 (1.12 – 2.04)*	1.27 (0.99 – 1.63)
	Tertile 3	1.33 (1.12 – 1.57)*	2.29 (1.69 - 3.07)*	1.79 (1.40 – 2.28)*
	P for trend	0.002	< 0.001	< 0.001
	OR by 1 tertile	1.15 (1.06 – 1.26)*	1.49 (1.29 – 1.73)*	1.35 (1.19 – 1.52)*
Women (n= 3731)				
Model 1*	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.05 (0.87 – 1.27)	1.49 (0.99 – 2.24)	1.48 (0.93 – 2.36)
	Tertile 3	1.23 (1.02 – 1.49)*	1.98 (1.34 – 2.93)*	2.16 (1.34 – 3.47)*
	P for trend	0.054	0.001	0.002
	OR by 1 tertile	1.11 (1.00 – 1.22)	1.40 (1.15 – 1.69)*	1.42 (1.13 – 1.80)*
Model 2**	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.11 (0.91 – 1.37)	1.52 (0.97 – 2.37)	1.29 (0.82 – 2.15)
	Tertile 3	1.29 (1.04 – 1.58)*	2.06 (1.34 – 3.16)*	1.82 (1.13 – 2.94)*
	P for trend	0.025	0.001	0.010
	OR by 1 tertile	1.13 (1.02 – 1.26)*	1.42 (1.16 – 1.75)*	1.36 (1.08 – 1.72)*

* Model 1: adjusted for age and country. ** Model 2: additionally adjusted for other covariates such as education, occupation, marital status, deprivation, depression, social isolation, and self-rated health. A Gender–specific tertile of ER ratio: in men, tertile 1 (0.20-0.32), tertile 2 (0.32-0.47), and tertile 3 (> 0.47); in women, tertile 1 (0.20-0.31), tertile 2 (0.31-0.46), and tertile 3 (> 0.46). * P value < 0.05.

5.2.2 Associations between OC and drinking outcomes

The associations between OC at wave 1 and three drinking outcomes (heavy drinking, binge drinking, and problem drinking) at wave 2 are assessed using three logistic regression analyses, separately for men and women, after adjustment for age and country (Model 1) and after additionally adjustment for other covariates (Model 2).

Table 5.8 shows the associations between OC at wave 1 and three drinking outcomes at wave 2 using three logistic regression analyses, respectively. In men, Model 2 reports that the odds of having heavy drinking, binge drinking and problem

drinking are 1.18, 1.72 and 1.64, respectively, for highest versus lowest tertile of OC. The adjusted OR changes for heavy drinking, binge drinking and problem drinking are 1.08 (p= 0.081), 1.31 and 1.28 (p < 0.05), respectively, by 1–tertile increase in OC.

In women, Model 2 reports that the odds of having binge drinking and problem drinking are 1.52 and 1.63 for highest versus lowest tertile of OC. The adjusted OR changes for binge drinking and problem drinking are 1.24 and 1.27 (p < 0.05) by 1–tertile increase in OC score; nevertheless, the association between OC and heavy drinking does not reach statistical significance (p= 0.281).

Model	OC score	Heavy drinking	Binge drinking	Problem drinking
	Tertile▲	OR (95% CI)	OR (95% CI)	OR (95% CI)
Men (n= 3782)				
Model 1*	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.06 (0.90 – 1.25)	1.23 (0.93 – 1.62)	1.17 (0.91 – 1.50)
	Tertile 3	1.21 (1.01 – 1.44)*	1.68 (1.29 – 2.20)*	1.50 (1.20 – 1.88)*
	P for trend	0.034	< 0.001	0.001
	OR by 1 tertile	1.10 (1.01 – 1.20)*	1.30 (1.13 – 1.48)*	1.21 (1.08 – 1.36)*
Model 2**	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.02 (0.87 – 1.21)	1.24 (0.96 – 1.60)	1.19 (0.94 – 1.50)
	Tertile 3	1.18 (0.99 – 1.40)	1.72 (1.32 – 2.24)*	1.64 (1.29 – 2.07)*
	P for trend	0.081	< 0.001	0.001
	OR by 1 tertile	1.08 (0.99 – 1.18)	1.31 (1.15 – 1.50)*	1.28 (1.13 – 1.44)*
Women (n= 3731)				
Model 1*	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.04 (0.85 – 1.29)	1.14 (0.79 – 1.64)	1.35 (0.86 – 2.09)
	Tertile 3	1.07 (0.88 – 1.32)	1.60 (1.11 – 2.28)*	1.75 (1.15 – 2.68)*
	P for trend	0.375	0.012	0.008
	OR by 1 tertile	1.03 (0.95 – 1.14)	1.27 (1.05 – 1.51)*	1.34 (1.08 – 1.65)*
Model 2**	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.09 (0.90 – 1.33)	1.09 (0.75 – 1.61)	1.39 (0.90 – 2.14)
	Tertile 3	1.11 (0.91 – 1.37)	1.52 (1.03 – 2.25)*	1.63 (1.05 – 2.52)*
	P for trend	0.281	0.036	0.028
	OR by 1 tertile	1.06 (0.96 – 1.17)	1.24 (1.01 – 1.50)*	1.27 (1.03 – 1.58)*

Table 5.8 Associations between OC at wave 1 and drinking outcomes at wave 2

* Model 1: adjusted for age and country. ** Model 2: additionally adjusted for other covariates such as education, occupation, marital status, deprivation, depression, social isolation, and self-rated health. \blacktriangle Gender–specific tertile of OC score: in men, tertile 1 (6–12), tertile 2 (12–15), and tertile 3 (15–24); in women, tertile 1 (6–12), tertile 2 (12–15), and tertile 3 (15–24). * P value < 0.05.

Previous analyses have found consistent and significant associations of exposure variables (ER ration and OC) with three drinking outcomes, respectively, with the same direction of causality but different magnitude of effect. Higher levels of ER ratio and OC, respectively, are associated with higher levels of drinking outcomes.

5.2.3 Antecedent or mediator role of OC in ERI-drinking relationship

To assess antecedent or mediator roles of OC in ERI–drinking relationship, the structural equation modelling (SEM) for categorical outcomes with an autoregressive and cross–lagged model is adopted and applied in Mplus 7.⁵⁸² The measurement model and the structural model for the SEM are specified in Figure 5.1.

Figure 5.1 Measurement model and structural model specified for the SEM for antecedent or mediator role of OC in ERI–drinking relation



In the measurement model, a latent factor (drinking outcome) underlies the relationship among 3 continuous latent response variables, which underlie 3 observed binary variables (binge drinking, heavy drinking, and problem drinking), respectively.

Justification for adopting a latent variable is summarized. The results of path analyses for 3 separate drinking outcomes are compared (see Appendix 4). Despite different magnitudes of effect, there are consistent directions of causality between ERI and 3 drinking outcomes; for example, path coefficients are all positive – 0.146 for binge drinking, 0.116 for problem drinking, and 0.069 for heavy drinking in women. Similarly, there are consistent directions of causality but diverse magnitudes of effect between OC and 3 drinking outcomes. Other paths between OC and ERI are quite similar across 3 drinking outcomes. The objective of path analyses focuses on antecedent or mediator role of OC in ERI–drinking relationship, and I summarize 3 drinking outcomes by a latent variable in order to find an overall trend in the relationships between OC, ERI and drinking in one model. A data reduction definition views a latent variable as a way to reduce complexity or dimensionality of a set of data; a latent variable is viewed as an emergent property that summarizes the indicators. This method assumes an overabundance of data regarding the variables of interest and the need to find a parsimonious means of using the data to test relationships between these variables.⁵⁸³

The measurement model is shown in Table 5.9. Factor loadings estimate direct effects of a latent factor on the latent response variable and are interpreted as regression coefficients. For example, in women, 1–unit increase in drinking outcome at wave 2 is associated with 1.000, 0.657 and 0.762 unit of increase in latent response variables for binge drinking, heavy drinking and problem drinking at wave 2, respectively. Additionally, standardized factor loadings are estimated correlation between the latent response variable and the latent factor; squared standardized factor loadings are proportions of explained variance (R^2). Thus, residual variance (1 – R^2) are proportion of unexplained variance. Finally, the non–linear relationship between each latent response variable and corresponding observed categorical variable is linked by the probit model.

The acceptability of the measurement model is evaluated by the interpretability, size, and statistical significance of the model's parameter estimates. For interpretability,

150

the parameter estimates do not take out-of-range values (e.g. negative error variance). For statistical significance, all factor loadings are shown statistically significant. For size, when a standardized factor loading is > 0.7, the factor would explain the majority of variance of the indicator (R^2 > 0.5). Researchers suggest that standardized factor loadings are considered poor (0.32–0.45), fair (0.45–0.55), good (0.55–0.63), very good (0.63–0.71), and excellent (> 0.71).⁵⁸⁴ In my measurement model, the lowest standardized factor loading is still considered to be good (> 0.55).

Table 5.9Measurement model of the SEM for antecedent or mediator role of OC inERI-drinking relation

	Fa	ctor loadings		
Parameter	Unstandardized	Standard	Standardized	Residual
Farameter	coefficient	Error	coefficient	variance
Men (n= 3782)				
Latent factor – Drinking outcome				
Binge drinking, wave 1	1.000	0.000	0.999	0.002
Heavy drinking, wave 1	0.863	0.038	0.862	0.257
Problem drinking, wave 1	0.620	0.029	0.619	0.616
Latent factor – Drinking outcome				
Binge drinking, wave 2	1.000	0.000	0.932	0.131
Heavy drinking, wave 2	0.730	0.036	0.680	0.537
Problem drinking, wave 2	0.615	0.029	0.573	0.671
Women (n= 3731)				
Latent factor – Drinking outcome				
Binge drinking, wave 1	1.000	0.000	0.957	0.084
Heavy drinking, wave 1	0.774	0.034	0.741	0.451
Problem drinking, wave 1	0.807	0.037	0.772	0.404
Latent factor – Drinking outcome				
Binge drinking, wave 2	1.000	0.000	0.974	0.051
Heavy drinking, wave 2	0.657	0.042	0.640	0.590
Problem drinking, wave 2	0.762	0.049	0.742	0.449

In the structural model, an autoregressive and cross–lagged model is adopted in this 2–wave cohort study (Figure 5.1). First, "autoregressive" means regressed on itself, so each variable is predicted by the same variable at an early wave. Second, the cross–lagged effects of "OC at wave 1 on ERI at wave 2" and "ERI at wave 1 on OC at wave 2" are measured, respectively, in order to identify causal directionality between OC and ERI. Bidirectional relationship between OC and ERI is possible based on my hypotheses. Third, as mediation is a causal chain involving at least two causal relations, these causal relations can be tested separately using two phases of data. The mediator role of ERI in OC–drinking relation is assessed by two cross–lagged effects: (1) OC at wave 1 on ERI at wave 2; (2) ERI at wave 1 on drinking at wave 2. The mediator role of OC in ERI–drinking relation is estimated by two cross–lagged effects: (1) ERI at wave 1 on OC at wave 2; (2) OC at wave 1 on drinking at wave 2. Partial mediation applies if both causal relations are confirmed; the product of two path coefficients (two cross–lagged effects) can estimate the strength of mediator effect.⁵⁸⁵ For a mediator effect, the effect size measure is the product of two path coefficients. The bootstrap method is used for significance testing of the mediator effect with 5000 bootstrap samples to yield valid estimates for the mediator effect by Mplus 7; this method is adopted due to complicated models (categorical outcomes and multiple mediators).⁵⁸⁶

In terms of predictors, the tertiles of ERI and OC are firstly transformed into a series of dummy variables (one dummy variable for each tertile) to compare between tertile groups. Next, ERI tertile and OC tertile are treated as continuous variables to estimate assumed linear trend between the predictor and drinking outcome. In terms of model fit, three indexes are used: Root Mean Square Error of Approximation (RMSEA), Comparative Fit Index (CFI), and Tucker–Lewis Index (TLI). RMSEA < 0.06, CFI > 0.95 or TLI > 0.95 indicate "good model fit".

Table 5.10 presents the structural model in men, with the results illustrated in Figure 5.2. First, the mediator effect of ERI in OC–drinking relationship is estimated by multiplying 2 cross–lagged effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 2 (unstandardized β = 0.142; standard error= 0.020); (2) higher ERI at wave 1 significantly associated with higher levels of drinking at wave 2 (unstandardized β = 0.152; standard error= 0.033). This mediator effect of ERI is significant (0.022= 0.142 x 0.152; standard error= 0.006 and p= 0.001 estimated by bootstrap method).

Second, the mediator effect of OC in ERI-drinking relationship is estimated by

multiplying 2 cross–lagged effects: (1) higher ERI at wave 1 significantly associated with higher OC at wave 2 (unstandardized β = 0.083; standard error= 0.020); (2) higher OC at wave 1 non-significantly associated with higher levels of drinking at wave 2 (unstandardized β = 0.050; standard error= 0.033; p= 0.104). This mediator effect of OC is not significant (0.004= 0.083 x 0.050; standard error= 0.003 and p= 0.155 by bootstrap method). Third, the fit indexes are considered good fit (RMSEA= 0.049 < 0.06) or close to cutoffs for good fit (CFI= 0.945; TLI= 0.923).

Table 5.10 Structural model of the SEM for antecedent or mediator role of OC in

ERI–drinking relation in men (n= 3782)

Parameter	Unstandardized coefficient	Standardized coefficient	P value
Drinking wave $1 \rightarrow$ Drinking wave 2	0.467	0.428	< 0.001
OC wave 1 \rightarrow OC wave 2			
OC tertile 2 vs tertile 1	0.320	0.174	< 0.001
OC tertile 3 vs tertile 1	0.681	0.335	< 0.001
1-tertile increase	0.346	0.316	< 0.001
ERI wave 1 → ERI wave 2			
ERI tertile 2 vs tertile 1	0.216	0.126	< 0.001
ERI tertile 3 vs tertile 1	0.473	0.277	< 0.001
1-tertile increase	0.242	0.238	< 0.001
$OC \rightarrow ERI \rightarrow Drinking$	0.022	0.017	0.001
OC wave 1 \rightarrow ERI wave 2			
OC tertile 2 vs tertile 1	0.113	0.066	0.001
OC tertile 3 vs tertile 1	0.279	0.148	< 0.001
1-tertile increase	0.142	0.132	< 0.001
ERI wave 1 \rightarrow Drinking wave 2			
ERI tertile 2 vs tertile 1	0.162	0.071	0.003
ERI tertile 3 vs tertile 1	0.306	0.135	< 0.001
1-tertile increase	0.152	0.117	< 0.001
$\underline{ERI} \rightarrow OC \rightarrow Drinking$	0.004	0.003	0.155
ERI wave 1 \rightarrow OC wave 2			
ERI tertile 2 vs tertile 1	0.095	0.051	0.011
ERI tertile 3 vs tertile 1	0.166	0.089	< 0.001
1-tertile increase	0.083	0.078	< 0.001
OC wave 1 \rightarrow Drinking wave 2			
OC tertile 2 vs tertile 1	0.049	0.021	0.298
OC tertile 3 vs tertile 1	0.096	0.037	0.145
1-tertile increase	0.050	0.035	0.104
Tests of model fit	RMSEA= 0.049	CFI= 0.945	TLI= 0.923

Figure 5.2 Results of the SEM for antecedent or mediator role of OC in ERI–drinking relation in men



Table 5.11 presents the structural model in women; the results are illustrated in Figure 5.3. First, the mediator effect of ERI in OC–drinking relationship is estimated by multiplying 2 cross–lagged effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 2 (unstandardized β = 0.148; standard error= 0.019); (2) higher ERI at wave 1 significantly associated with higher levels of drinking at wave 2 (unstandardized β = 0.148; standard error= 0.019); (2) higher ERI at wave 1 significantly associated with higher levels of drinking at wave 2 (unstandardized β = 0.138; standard error= 0.040). This mediator effect of ERI is significant (0.020= 0.148 x 0.138; standard error= 0.006 and p= 0.002).

Second, the mediator effect of OC in ERI-drinking relationship is estimated by

multiplying 2 cross–lagged effects: (1) higher ERI at wave 1 significantly associated with higher OC at wave 2 (unstandardized β = 0.077; standard error= 0.019); (2) higher OC at wave 1 non-significantly associated with higher levels of drinking at wave 2 (unstandardized β = 0.040; standard error= 0.038; p= 0.162). This mediator effect of OC is not significant (0.003; standard error= 0.003 and p= 0.308). Third, the fit indexes are considered good fit (RMSEA= 0.050 < 0.06) or close to cutoffs for good fit (CFI= 0.925; TLI= 0.908).

Table 5.11 Structural model of the SEM for antecedent or mediator role of OC in ERI–drinking relation in women (n=3731)

Parameter	Unstandardized coefficient	Standardized coefficient	P value
Drinking wave $1 \rightarrow$ Drinking wave 2	0.473	0.468	< 0.001
OC wave 1 \rightarrow OC wave 2			
OC tertile 2 vs tertile 1	0.292	0.170	< 0.001
OC tertile 3 vs tertile 1	0.730	0.400	< 0.001
1-tertile increase	0.360	0.358	< 0.001
ERI wave 1 → ERI wave 2			
ERI tertile 2 vs tertile 1	0.293	0.174	< 0.001
ERI tertile 3 vs tertile 1	0.484	0.290	< 0.001
1-tertile increase	0.252	0.242	< 0.001
$OC \rightarrow ERI \rightarrow Drinking$	0.020	0.016	0.002
OC wave 1 → ERI wave 2			
OC tertile 2 vs tertile 1	0.155	0.093	< 0.001
OC tertile 3 vs tertile 1	0.308	0.173	< 0.001
1-tertile increase	0.148	0.141	< 0.001
ERI wave 1 \rightarrow Drinking wave 2			
ERI tertile 2 vs tertile 1	0.142	0.063	0.068
ERI tertile 3 vs tertile 1	0.280	0.126	0.001
1-tertile increase	0.138	0.108	0.001
ERI \rightarrow OC \rightarrow Drinking	0.003	0.002	0.308
ERI wave 1 \rightarrow OC wave 2			
ERI tertile 2 vs tertile 1	0.054	0.031	0.123
ERI tertile 3 vs tertile 1	0.154	0.090	< 0.001
1-tertile increase	0.077	0.074	< 0.001
OC wave 1 \rightarrow Drinking wave 2			
OC tertile 2 vs tertile 1	0.034	0.016	0.568
OC tertile 3 vs tertile 1	0.075	0.033	0.268
1-tertile increase	0.040	0.032	0.162
Tests of model fit	RMSEA= 0.050	CFI= 0.925	TLI= 0.908

Figure 5.3 Results of the SEM for antecedent or mediating role of OC in ERI–drinking relation in women



Interpretation of path coefficient in the SEM is summarized. The effect of ERI at wave 1 on drinking outcome at wave 2 in women is taken for example (Table 5.11). Unstandardized path coefficient for this effect is 0.138, which means that 1-tertile increase in ER ratio results in 0.138 unit of increase in drinking outcome. Next, the measurement model is considered (Table 5.9). In women, 1-unit increase in drinking outcome at wave 2 is associated with 1.000, 0.657 and 0.762 unit of increase in latent response variables for binge drinking, heavy drinking and problem drinking at wave 2. Thus, 1-tertile increase in ER ratio results in 0.138, 0.091 (0.138 x 0.657) and 0.105

(0.138 x 0.762) unit of increase in latent response variables for binge drinking, heavy drinking and problem drinking at wave 2. The nonlinear relationship between each latent response variable and observed binary variable is linked by the probit model. The OR is the exponential (antilog) of estimated logistic coefficient, which is derived from probit coefficient multiplied by 1.8. Thus, 1–tertile increase in ER ratio results in OR changes of 1.28 (exponential function 0.138 x 1.8), 1.18 and 1.21 for observed binary variables of binge drinking, heavy drinking and problem drinking at wave 2.

5.2.4 Modifying role of OC in ERI–drinking relationship

To evaluate modifying effect of OC in ERI–drinking relationship, two approaches are adopted. In the first approach, the associations between ER ratio at wave 1 and drinking outcomes at wave 2 in different strata of OC tertile are assessed after adjustment for covariates. Next, logistic regression analyses are conducted for 3 drinking outcomes at wave 2, respectively, regressed by OC tertile, ER–ratio tertile, and their interaction term at wave 1 after adjustment for covariates. By comparing the log likelihoods for the model with this interaction term and the model without, likelihood–ratio (LR) test is adopted to test significance of this interaction term.

Approaches	Strata	Heavy drinking	Binge drinking	Problem drinking
Men		OR of outcome by 1-tertile increase in ER ratio (95% CI)		
1. ERI-drinking relation	OC tertile 1	1.06 (0.92 – 1.23)	1.39 (1.10 – 1.75)	1.21 (0.99 – 1.49)
in different strata of OC	OC tertile 2	1.23 (1.05 – 1.44)	1.51 (1.17 – 1.94)	1.22 (0.98 – 1.53)
tertile	OC tertile 3	1.17 (0.95 – 1.43)	1.45 (1.05 – 2.00)	1.46 (1.09 – 1.96)
2. Interaction OC x ERI	P-value	P= 0.362	P= 0.853	P= 0.196
Women		OR of outcome by 1	-tertile increase in ER	ratio (95% CI)
1. ERI-drinking relation	OC tertile 1	1.06 (0.90 – 1.24)	1.36 (0.95 – 1.93)	1.03 (0.68 – 1.57)
in different strata of OC	OC tertile 2	1.19 (0.99 – 1.42)	1.68 (1.09 – 2.57)	1.48 (0.97 – 2.26)
tertile	OC tertile 3	1.00 (0.82 – 1.22)	1.10 (0.75 – 1.63)	1.49 (0.91 – 2.44)
2. Interaction OC x ERI	P-value	P= 0.932	P= 0.851	P= 0.312

Table 5.12 Evaluation for modifying role of OC in ERI–drinking relationship

In Table 5.12, LR tests show that the interaction term between OC tertile and ER– ratio tertile is not significant for heavy drinking in men (p= 0.362) and women (p= 0.932), not significant for binge drinking in men (p= 0.853) and women (p= 0.851), and not significant for problem drinking in men (p= 0.196) and women (p= 0.312).

The second approach adopts a measurement model (Table 5.13) similar to previous SEM (Table 5.9), and a path analysis is conducted for the latent drinking outcome at wave 2 regressed by OC tertile, ER–ratio tertile, and the interaction term between OC tertile and ER–ratio tertile at wave 1 after adjustment for covariates. The significance of this interaction term is evaluated in the SEM by Mplus 7. This interaction term is not significant in men (p= 0.324; unstandardized β = 0.038; standard error= 0.039) and women (p= 0.282; unstandardized β = 0.044; standard error= 0.040). My results show that OC has no significantly modifying role in ERI–drinking relationship.

	Fa			
Parameter	Unstandardized coefficient	Standard Error	Standardized coefficient	Residual variance
Men (n= 3782)				
Latent variable – Drinking outcome				
Binge drinking, wave 1	1.000	0.000	0.997	0.006
Heavy drinking, wave 1	0.861	0.038	0.858	0.264
Problem drinking, wave 1	0.617	0.029	0.615	0.622
Latent variable – Drinking outcome				
Binge drinking, wave 2	1.000	0.000	0.934	0.128
Heavy drinking, wave 2	0.735	0.037	0.686	0.529
Problem drinking, wave 2	0.611	0.029	0.571	0.674
Women (n= 3731)				
Latent variable – Drinking outcome				
Binge drinking, wave 1	1.000	0.000	0.954	0.090
Heavy drinking, wave 1	0.782	0.033	0.746	0.443
Problem drinking, wave 1	0.806	0.035	0.769	0.409
Latent variable – Drinking outcome				
Binge drinking, wave 2	1.000	0.000	0.977	0.045
Heavy drinking, wave 2	0.642	0.042	0.627	0.607
Problem drinking, wave 2	0.768	0.052	0.750	0.438

 Table 5.13
 Measurement model of the SEM for modifying role of OC in ERI–

5.3 Potential Role of PC in Relation between OC, ERI, and Drinking

The focus of this section is, according to the second aim of the thesis, on the associations between PC and drinking outcomes, and on the potential role of PC in the relationship between ERI, OC, and drinking outcomes.

5.3.1 Associations between PC and drinking outcomes

The associations between PC at wave 1 and three drinking outcomes at wave 2 are assessed following the same steps as for ERI–drinking associations.

Model	Perceived	Heavy drinking	Binge drinking	Problem drinking
	control tertile▲	OR (95% CI)	OR (95% CI)	OR (95% CI)
Men (n= 3782)				
Model 1*	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.01 (0.82 – 1.22)	0.99 (0.77 – 1.27)	0.82 (0.66 – 1.02)
	Tertile 3	0.86 (0.71 – 1.04)	0.67 (0.51 – 0.89)*	0.67 (0.53 – 0.84)*
	P for trend	0.179	0.010	0.006
	OR by 1 tertile	0.94 (0.85 – 1.03)	0.84 (0.74 – 0.96)*	0.82 (0.73 – 0.92)*
Model 2**	Tertile 1	1.00	1.00	1.00
	Tertile 2	1.00 (0.83 – 1.20)	0.98 (0.76 – 1.27)	0.83 (0.66 – 1.05)
	Tertile 3	0.81 (0.67 – 0.97)*	0.63 (0.47 – 0.86)*	0.64 (0.49 – 0.85)*
	P for trend	0.033	0.006	0.004
	OR by 1 tertile	0.90 (0.83 – 0.99)*	0.82 (0.71 – 0.94)*	0.81 (0.71 – 0.93)*
Women (n= 3731)				
Model 1*	Tertile 1	1.00	1.00	1.00
	Tertile 2	0.93 (0.76 – 1.13)	0.84 (0.60 – 1.17)	0.88 (0.60 – 1.26)
	Tertile 3	0.87 (0.70 – 1.06)	0.62 (0.43 – 0.90)*	0.67 (0.44 – 1.04)
	P for trend	0.155	0.023	0.073
	OR by 1 tertile	0.93 (0.83 – 1.03)	0.80 (0.66 – 0.96)*	0.84 (0.70 – 1.01)
Model 2**	Tertile 1	1.00	1.00	1.00
	Tertile 2	0.90 (0.73 – 1.11)	0.88 (0.61 – 1.25)	0.81 (0.54 – 1.22)
	Tertile 3	0.83 (0.66 – 1.02)	0.61 (0.41 – 0.94)*	0.62 (0.40 – 1.00)*
	P for trend	0.083	0.035	0.046
	OR by 1 tertile	0.91 (0.81 – 1.01)	0.80 (0.65 - 0.98)*	0.80 (0.63 – 1.00)*

Table 5.14 Associations of PC at wave 1 with drinking outcomes at wave 2

* Model 1: adjusted for age and country. ** Model 2: additionally adjusted for other covariates. A Gender–specific tertile of PC: in men, tertile 1 (0–34), tertile 2 (34–41), and tertile 3 (41–55); in women, tertile 1 (0–33), tertile 2 (33–40), and tertile 3 (40–55). * P value < 0.05.

Table 5.14 shows the associations between PC at wave 1 and three drinking outcomes at wave 2 using logistic regression analyses, respectively. In men, Model 2 shows that the adjusted odds of having heavy drinking, binge drinking and problem drinking are 0.81, 0.63 and 0.64, respectively, for highest versus lowest tertile of PC. The adjusted OR changes for heavy drinking, binge drinking and problem drinking are 0.90, 0.82 and 0.81, respectively, by 1–tertile increase in PC (all p-values < 0.05).

In women, Model 2 shows that the adjusted odds of heavy drinking, binge drinking and problem drinking are 0.83, 0.61 and 0.62, respectively, for highest versus lowest tertile of PC. The adjusted OR changes for heavy drinking, binge drinking and problem drinking are 0.91, 0.80 and 0.80 by 1–tertile increase in PC (all p-values < 0.1).

5.3.2 Mediator roles of PC and ERI in OC-drinking relationship

To assess the potential role of PC in the relationship between OC, ERI and drinking, the SEM for categorical outcomes with an autoregressive and cross–lagged model is adopted and applied by Mplus 7. The measurement model and the structural model for the SEM are specified in Figure 5.4.

The measurement model is specified in a similar way as the previous SEM model (Section 5.2.3). A latent drinking outcome is denoted by 3 continuous latent response variables, which underlie 3 observed binary variables (binge drinking, heavy drinking, and problem drinking) at wave 1 and 2, respectively. The justification for adopting a latent variable is summarized. The results of path analyses for 3 separate drinking outcomes are compared (see Appendix 4). Despite different magnitudes of effect, there are consistent directions of causality between ERI and 3 drinking outcomes. Similarly, there are consistent directions of causality but diverse magnitudes of effect between PC and 3 drinking outcomes. Other paths are quite similar across 3 drinking outcomes. I summarize 3 drinking outcomes by a parsimonious means (latent variable) in order to find an overall trend in the relations between OC, ERI, PC and drinking in one model. A data reduction definition views a latent variable as a way to reduce complexity or dimensionality of a set of data.

Figure 5.4 Measurement model and structural model specified for the SEM for mediator roles of PC and ERI in OC–drinking relationship



The results of the measurement model are shown in Table 5.15. The acceptability of the measurement model is evaluated by interpretability, size, and statistical significance of the model's parameter estimates. For interpretability, the parameter estimates do not take out-of-range values (e.g. negative error variance). For statistical significance, all factor loadings are statistically significant. For size, standardized factor loadings are considered poor (0.32–0.45), fair (0.45–0.55), good (0.55–0.63), very good (0.63–0.71), and excellent (> 0.71). In my measurement model, the lowest standardized factor loading is still considered to be good (> 0.55).

Table 5.15 Measurement model of the SEM for mediator roles of PC and ERI in

	Fa	Factor loadings		
Parameter	Unstandardized	Standard	Standardized	Residual
Falameter	coefficient	Error	coefficient	variance
Men (n= 3782)				
Latent variable – Drinking outcome				
Binge drinking, wave 1	1.000	0.000	0.985	0.030
Heavy drinking, wave 1	0.844	0.039	0.831	0.309
Problem drinking, wave 1	0.616	0.031	0.607	0.631
Latent variable – Drinking outcome				
Binge drinking, wave 2	1.000	0.000	0.934	0.127
Heavy drinking, wave 2	0.738	0.036	0.689	0.525
Problem drinking, wave 2	0.612	0.030	0.572	0.673
Women (n= 3731)				
Latent variable – Drinking outcome				
Binge drinking, wave 1	1.000	0.000	0.955	0.088
Heavy drinking, wave 1	0.768	0.033	0.733	0.462
Problem drinking, wave 1	0.807	0.037	0.771	0.405
Latent variable – Drinking outcome				
Binge drinking, wave 2	1.000	0.000	0.985	0.030
Heavy drinking, wave 2	0.662	0.042	0.652	0.575
Problem drinking, wave 2	0.783	0.050	0.771	0.406

OC-drinking relation

The structural model is specified in a different way from the previous SEM model (Section 5.2.3). There are two potential mediators (PC and ERI) between the effects of OC at wave 1 on drinking outcomes at wave 2 (Figure 5.4). Because the HAPIEE study is limited by no measurement of PC at wave 2, the cross–sectional associations between OC, ERI, and PC at wave 1 are used for this analysis. Thus, PC and ERI are only specified to be correlated, although bi-directional relationship between PC and ERI has been hypothesized.

An autoregressive and cross–lagged model for drinking outcomes is adopted. First, drinking outcomes at wave 2 are predicted by drinking outcomes at wave 1. Second, the cross–lagged effects of OC, PC, and ERI at wave 1 on drinking outcomes at wave 2 are measured, respectively. Third, the mediator effects can only be assessed by a half–longitudinal design. For example, the mediator role of PC in OC– drinking relation is assessed by two effects: (1) the cross–sectional association of OC at wave 1 on PC at wave 1; (2) the cross–lagged effect of PC at wave 1 on drinking at wave 2. Partial mediation applies if both causal relations are confirmed; the product of two path coefficients might estimate the strength of mediator effect.⁵⁸⁷

Table 5.16 presents the structural model in men; the results are illustrated in Figure 5.5. First, the mediator effect of ERI in OC–drinking relation is estimated by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 1 (unstandardized β = 0.249; standard error= 0.020); (2) higher ERI at wave 1 significantly associated with higher levels of drinking at wave 2 (unstandardized β = 0.138; standard error= 0.032). This mediator effect of ERI is significant (0.034= 0.249 x 0.138; standard error= 0.008 and p < 0.001 by bootstrap method).

Second, the mediator effect of PC in OC–drinking relation in men is estimated by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with lower PC at wave 1 (unstandardized β = –0.097; standard error= 0.017); (2) lower PC at wave 1 significantly associated with higher levels of drinking at wave 2 (unstandardized β = – 0.098; standard error= 0.031). This mediator effect of PC is significant (0.010= -0.097 x -0.098; standard error= 0.003 and p= 0.006). Third, ERI at wave 1 is inversely associated with PC at wave 1 (unstandardized β = –0.047; p= 0.002). Finally, the fit indexes are considered good fit (RMSEA= 0.053 < 0.06) or close to cutoffs for good fit (CFI= 0.917; TLI= 0.888).

Table 5.16 Structural model of the SEM for mediator roles of PC and ERI in OC-

drinking relation in men (n= 3782)

Parameter	Unstandardized coefficient	Standardized coefficient	P value
Drinking wave 1 \rightarrow Drinking wave 2	0.465	0.458	< 0.001
OC wave 1 \rightarrow Drinking wave 2			
OC tertile 2 vs tertile 1	0.050	0.022	0.254
OC tertile 3 vs tertile 1	0.098	0.040	0.125
1-tertile increase	0.051	0.036	0.097
$OC \rightarrow ERI \rightarrow Drinking$	0.034	0.024	< 0.001
OC wave 1 → ERI wave 1			
OC tertile 2 vs tertile 1	0.253	0.151	< 0.001
OC tertile 3 vs tertile 1	0.494	0.270	< 0.001
1-tertile increase	0.249	0.231	< 0.001
ERI wave 1 → Drinking wave 2			
ERI tertile 2 vs tertile 1	0.165	0.070	0.005
ERI tertile 3 vs tertile 1	0.312	0.136	< 0.001
1-tertile increase	0.138	0.103	< 0.001
$\underline{OC \rightarrow PC \rightarrow Drinking}$	0.010	0.008	0.006
OC wave 1 \rightarrow PC wave 1			
OC tertile 2 vs tertile 1	- 0.012	- 0.007	0.724
OC tertile 3 vs tertile 1	- 0.198	- 0.109	< 0.001
1-tertile increase	- 0.097	- 0.095	< 0.001
PC wave 1 \rightarrow Drinking wave 2			
PC tertile 2 vs tertile 1	- 0.030	- 0.013	0.484
PC tertile 3 vs tertile 1	- 0.208	- 0.095	< 0.001
1-tertile increase	- 0.098	- 0.072	0.003
ERI correlates with PC	- 0.047	- 0.071	0.002
Tests of model fit	RMSEA= 0.053	CFI= 0.917	TLI= 0.888





Table 5.17 shows the structural model in women, with the results illustrated in Figure 5.6. First, the mediator effect of ERI in OC–drinking relation is estimated by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 1 (unstandardized β = 0.240; standard error= 0.019); (2) higher ERI at wave 1 significantly associated with higher levels of drinking at wave 2 (unstandardized β = 0.122; standard error= 0.039). This mediator effect of ERI is significant (0.029= 0.240 × 0.122; standard error= 0.009 and p= 0.002).

Second, the mediator effect of PC in OC-drinking relation is estimated by

multiplying 2 effects: (1) higher OC at wave 1 significantly associated with lower PC at wave 1 (unstandardized β = –0.089; standard error= 0.017); (2) lower PC at wave 1 significantly associated with higher levels of drinking at wave 2 (unstandardized β = – 0.096; standard error= 0.036). This mediator effect of PC is significant (0.009= -0.089 x -0.096; standard error= 0.004 and p= 0.017). Third, ERI at wave 1 is inversely associated with PC at wave 1 (unstandardized β = –0.050; p= 0.001). Finally, the fit indexes are considered good fit (RMSEA= 0.057 < 0.06) or close to cutoffs for good fit (CFI= 0.886; TLI= 0.875).

Table 5.17 Structural model of the SEM for mediator roles of PC and ERI in OC– drinking relation in women (n= 3731)

Parameter	Unstandardized	Standardized	P value
Farameter	coefficient	coefficient	r value
Drinking wave 1 \rightarrow Drinking wave 2	0.466	0.470	< 0.001
OC wave 1 → Drinking wave 2			
OC tertile 2 vs tertile 1	0.021	0.010	0.646
OC tertile 3 vs tertile 1	0.091	0.039	0.252
1-tertile increase	0.052	0.040	0.186
$OC \rightarrow ERI \rightarrow Drinking$	0.029	0.022	0.002
OC wave 1 → ERI wave 1			
OC tertile 2 vs tertile 1	0.248	0.146	< 0.001
OC tertile 3 vs tertile 1	0.479	0.268	< 0.001
1-tertile increase	0.240	0.231	< 0.001
ERI wave 1 → Drinking wave 2			
ERI tertile 2 vs tertile 1	0.147	0.065	0.052
ERI tertile 3 vs tertile 1	0.274	0.124	< 0.001
1-tertile increase	0.122	0.096	0.002
$\underline{OC \rightarrow PC \rightarrow Drinking}$	0.009	0.007	0.017
OC wave 1 \rightarrow PC wave 1			
OC tertile 2 vs tertile 1	- 0.016	- 0.010	0.610
OC tertile 3 vs tertile 1	- 0.180	- 0.106	< 0.001
1-tertile increase	- 0.089	- 0.090	< 0.001
PC wave 1 \rightarrow Drinking wave 2			
PC tertile 2 vs tertile 1	- 0.073	- 0.033	0.274
PC tertile 3 vs tertile 1	- 0.196	- 0.090	< 0.001
1-tertile increase	- 0.096	- 0.073	0.006
ERI correlates with PC	- 0.050	- 0.085	0.001
Tests of model fit	RMSEA= 0.057	CFI= 0.886	TLI= 0.875

Figure 5.6 Results of the SEM for mediator roles of PC and ERI in OC–drinking relation in women



5.3.3 Modifying role of PC in ERI-drinking relationship

Modifying effect of PC in ERI-drinking relationship is evaluated by two approaches. In the first approach, logistic regression analyses are conducted for 3 drinking outcomes at wave 2, respectively, regressed by PC tertile, ER-ratio tertile, and interaction term between PC tertile and ER-ratio tertile at wave 1 after adjustment for covariates. The log likelihoods for the model with this interaction term and the model without are compared, and LR tests show that this interaction term is not significant for heavy drinking in men (p= 0.182) and women (p= 0.443), not significant for binge drinking in men (p= 0.523) and women (p= 0.206), and not significant for problem drinking in men (p= 0.175) and women (p= 0.284) (Table 5.18).

 Table 5.18
 Evaluation for modifying role of perceived control in ERI–drinking

relationship

Approaches	Strata	Heavy drinking	Binge drinking	Problem drinking
Men		OR of outcome by 1	-tertile increase in EF	R ratio (95% CI)
1. ERI-drinking relation	PC tertile 1	1.10 (0.96 – 1.25)	1.40 (1.14 – 1.71)	1.20 (0.99 – 1.43)
in different strata of PC	PC tertile 2	1.10 (0.94 – 1.27)	1.47 (1.17 – 1.85)	1.22 (0.98 – 1.52)
tertile	PC tertile 3	1.27 (1.06 – 1.52)	1.45 (1.07 – 1.97)	1.55 (1.18 – 2.03)
2. Interaction PC x ERI	P-value	P= 0.182	P= 0.523	P= 0.175
Women		OR of outcome by 1	-tertile increase in EF	R ratio (95% CI)
1. ERI-drinking relation	PC tertile 1	1.09 (0.93 – 1.26)	1.51 (1.10 – 2.07)	1.14 (0.82 – 1.60)
in different strata of PC	PC tertile 2	1.14 (0.96 – 1.35)	1.34 (0.96 – 1.86)	1.44 (0.95 – 2.17)
tertile	PC tertile 3	0.97 (0.81 – 1.16)	1.04 (0.72 – 1.52)	1.49 (0.90 – 2.45)
2. Interaction PC x ERI	P-value	P= 0.443	P= 0.206	P= 0.284

Second, by a measurement model (Table 5.19) similar to previous SEM (Table 5.15), a path analysis is conducted for the latent drinking outcome at wave 2 regressed by PC tertile, ER-ratio tertile, and the interaction term between PC tertile and ER-ratio tertile at wave 1 after adjustment for covariates. In the SEM, this interaction term is not significant in men (unstandardized β = 0.049; standard error= 0.037 and p= 0.218) and in women (unstandardized β = 0.039; standard error= 0.040 and p= 0.315). My results based on the two approaches show no significantly modifying effect of PC in ERI-drinking relationship.

Table 5.19 Measurement model of the SEM for modifying role of PC in ERI-

	Factor loadings			
Parameter	Unstandardized	Standard	Standardized	Residual
Parameter	coefficient	Error	coefficient	variance
Men (n= 3782)				
Latent variable – Drinking outcome				
Binge drinking, wave 1	1.000	0.000	0.990	0.020
Heavy drinking, wave 1	0.858	0.039	0.849	0.279
Problem drinking, wave 1	0.617	0.030	0.611	0.627
Latent variable – Drinking outcome				
Binge drinking, wave 2	1.000	0.000	0.939	0.118
Heavy drinking, wave 2	0.738	0.037	0.693	0.520
Problem drinking, wave 2	0.615	0.030	0.577	0.667
Women (n= 3731)				
Latent variable – Drinking outcome				
Binge drinking, wave 1	1.000	0.000	0.964	0.071
Heavy drinking, wave 1	0.778	0.033	0.750	0.438
Problem drinking, wave 1	0.814	0.036	0.785	0.384
Latent variable – Drinking outcome				
Binge drinking, wave 2	1.000	0.000	0.975	0.049
Heavy drinking, wave 2	0.652	0.042	0.636	0.595
Problem drinking, wave 2	0.771	0.051	0.752	0.434

drinking relation

5.4 Main Findings for Drinking Outcomes

The analyses based on this 2–wave cohort study (3782 men and 3731 women aged 45–69) from the HAPIEE study report the following findings, which are in line with specific objectives and relevant hypotheses listed in Chapter 3.

In terms of the associations between ER ratio and drinking outcomes, Hypothesis 1 that higher ER ratio (wave 1) is associated with higher levels of alcohol drinking (wave 2) after adjustment for covariates is supported. In terms of the associations between OC and drinking outcomes, Hypothesis 4 that higher OC (wave 1) is associated with higher levels of alcohol drinking (wave 2) after adjustment for covariates is partially supported in binge drinking and problem drinking; however, the associations between OC and heavy drinking are marginally significant in men but non–significant in women.

With regards to the potential role of OC (antecedent, mediator, modifier, or direct

effect) in ERI-drinking relationship, Hypothesis 7 is supported. OC and ERI have bidirectional relationship, but the effect of OC on ERI is stronger than the other direction in the middle-aged and older populations. Antecedent role of OC in ERI-drinking relationship is found significant, but mediator role of OC is not significant. Direct effect of OC on drinking is not significant. Finally, modifying role of OC in ERI-drinking relation is not significant.

In terms of the associations between PC and drinking outcomes, Hypothesis 10 that lower PC (wave 1) is associated with higher levels of alcohol drinking (wave 2) after adjustment for covariates is supported.

With regards to the potential role of PC (mediator or modifier) in the relationship between ERI, OC and drinking, Hypothesis 13 is partially supported. PC and ERI partially mediate the effects of OC on alcohol drinking. In addition, PC and ERI may have bi–directional relationship. PC and ERI are negatively associated with each other in the cross–sectional analyses; bi–directional relationship between PC and ERI is possible, but causal directionality cannot be established in cross–sectional analyses. Finally, modifying role of PC in ERI–drinking relation is non–significant.

Note that the methodological issues and interpretation of the main findings for drinking outcomes will be addressed in detail in Chapter 8, and their implications for practice, policy and research in Chapter 9.

Chapter 6. Smoking Outcomes

The aims of analyses for smoking outcomes in Chapter 6, based on the aims listed in Chapter 3, are as follows: (1) to examine the potential role of OC in ERI–smoking relationship, including modifying, antecedent, mediator, or direct effects; (2) to investigate the potential role of PC in the relationship between OC, ERI and smoking outcomes. The analyses use data from a 2–wave cohort study (3782 men and 3731 women aged 45–69), part of the HAPIEE study, which has been described in detail in Chapter 4 Methodology.

The results are presented in the following three parts. First, descriptive statistics for covariates and smoking outcomes by country and by gender are reported. Second, the associations of ERI and OC at wave 1 with smoking outcomes at wave 2, respectively, are evaluated by binary or ordinal logistic regression. The path analysis with an autoregressive and cross–lagged model is applied to examine antecedent or mediator roles of OC in ERI–smoking relationship. Modifying roles of OC in ERI–smoking relations between PC at wave 1 and smoking outcomes at wave 2 are assessed by binary or ordinal logistic regression. The path analysis with an autoregressive and cross–lagged model is applied to examine antecedent or mediator roles of OC in ERI–smoking relationship. Modifying roles of OC in ERI–smoking relation are also examined. Third, the associations between PC at wave 1 and smoking outcomes at wave 2 are assessed by binary or ordinal logistic regression. The path analysis with an autoregressive and cross–lagged model is used to examine the relationship between OC, ERI, PC, and smoking outcomes. Modifying roles of PC in ERI–smoking relation are also tested.

6.1 Descriptive Statistics

6.1.1 Descriptive characteristics of study populations

In this sample of 7513 subjects (3782 men and 3731 women), the means of age assessed at wave 1 are 54.8 years in men (standard deviation= 6.0) and 53.2 years in women (standard deviation= 5.4). The average follow–up periods between wave 1 and wave 2 are 3.5 years in men (standard deviation= 0.7) and 3.6 years in women

171

(standard deviation= 0.6). Note that the descriptive statistics of covariates in this sample are the same as those described in Table 5.1 for drinking outcomes. Across the three countries, men generally have higher proportions in age group over 55, university–educated, manager/profession and manual workers, married/cohabiting, lower deprivation, less depression, and more social isolation than women.

6.1.2 Descriptive characteristics of smoking outcomes

Table 6.1 summarizes the descriptive statistics of smoking outcomes (smoking status in all subjects and smoking intensity in current smokers) by country and by gender at wave 1 and wave 2, respectively. There are gender differences across three countries. Men have dramatically higher proportions in current smokers and in heavy smokers than women.

There are country differences noted in Table 6.1. At wave 1, the percentages of current smokers are highest for Russian sample (46.4%) in men; in women, these percentages are highest for Polish sample (33.8%). Next, the percentages of heavy smokers (>= 20 cigarettes per day) in current smokers are highest for Polish men (59.2%) with averagely 19.1 (SD= 10.8) cigarettes smoked per day; these percentages are highest for Polish women (33.2%) with averagely 13.9 (SD= 8.0) cigarettes smoked per day. At wave 2, the percentages of current smokers are highest for Polish sample (30.5%). The percentages of heavy smokers are highest for Russian men (58.7%) with averagely 18.7 (SD= 9.0) cigarettes per day; these percentages are highest for Polish women (31.7%) with averagely 13.4 (SD= 7.4) cigarettes per day.

172

Table 6.1 Descriptive statistics of smoking outcomes by country and gender (N=

7513)

	Czech		Russia		Poland		
Smoking outcomes	Men	Women	Men	Women	Men	Women	
	(n= 1082)	(n= 1099)	(n= 1402)	(n= 1394)	(n= 1298)	(n=1238)	
Wave 1							
Smoking status, n (%)							
Lifetime non-smokers	403 (37.3)	548 (49.9)	421 (30.0)	1146 (82.2)	418 (32.2)	547 (44.2)	
Former smokers	354 (32.7)	244 (22.2)	331 (23.6)	71 (5.1)	437 (33.7)	272 (22.0)	
Current smokers	325 (30.0)	307 (27.9)	650 (46.4)	177 (12.7)	443 (34.1)	419 (33.8)	
Smoking intensity in current smo	oker, n (%)						
Light smoker (1–9 /day)	68 (21.1)	122 (39.8)	74 (11.4)	89 (50.2)	72 (16.2)	104 (24.8)	
Medium smoker (10–19/day)	118 (36.2)	128 (41.6)	203 (31.3)	66 (37.2)	109 (24.6)	176 (42.0)	
Heavy smoker (>= 20/day)	139 (42.7)	57 (18.6)	373 (57.3)	22 (12.6)	262 (59.2)	139 (33.2)	
Average cigarettes smoked	15.5 (8.3)	10.6 (6.6)	18.2 (9.0)	9.5 (6.3)	10 1 (10 9)	13.9 (8.0)	
per day, mean (s.d.)	15.5 (6.5)	10.6 (6.6)	16.2 (9.0)	9.5 (0.5)	19.1 (10.8)	13.9 (0.0)	
Wave 2							
Smoking status, n (%)							
Lifetime non-smokers	415 (38.4)	602 (54.8)	395 (28.2)	1154 (82.8)	410 (31.6)	545 (44.0)	
Former smokers	362 (33.5)	224 (20.4)	379 (27.0)	70 (5.0)	473 (36.4)	312 (25.2)	
Current smokers	304 (28.1)	273 (24.8)	628 (44.8)	170 (12.2)	415 (32.0)	381 (30.8)	
Smoking intensity in current smo	oker, n (%)						
Light smoker (1–9 /day)	65 (21.4)	102 (37.2)	68 (10.9)	77 (45.3)	62 (14.9)	99 (26.1)	
Medium smoker (10–19/day)	127 (41.9)	127 (46.6)	191 (30.4)	73 (42.9)	123 (29.7)	161 (42.2)	
Heavy smoker (>= 20/day)	112 (36.7)	44 (16.2)	369 (58.7)	20 (11.8)	230 (55.4)	121 (31.7)	
Average cigarettes smoked	1 = 1 (0 0)	107(60)	107(00)	$0 \in (C 1)$	19.6 (10.0)	10 4 (7 4)	
per day, mean (s.d.)	15.1 (8.0)	10.7 (6.2)	18.7 (9.0)	9.5 (6.1)	18.6 (10.2)	13.4 (7.4)	

The ways of pooling the data in subsequent analyses are described. First, men and women are analysed separately, as most studies on the relationships between psychosocial factors and health outcomes. Second, crude associations between exposure variables (ER ratio and OC) and smoking outcomes in country–specific strata are assessed. Binary and ordinal logistic regression analyses are conducted for two smoking outcomes (smoking status and smoking intensity) at wave 2, respectively, regressed by ER–ratio tertile, country, and interaction term between country and ER– ratio tertile at wave 1. By comparing the log likelihoods for the model with this interaction term and the model without, LR test is used to test significance of this interaction term (Table 6.2). Similarly, the interaction term between country and OC tertile is evaluated (Table 6.3).

There are country differences seen in Table 6.2. In both sexes, the associations between ER ratio and smoking intensity in Poland are the strongest of all countries. Generally, crude associations between ER ratio at wave 1 and smoking outcomes at wave 2 are not very different across country–specific strata (all p–values by LR test > 0.32); no significant interaction between country and ER ratio is reported.

Table 6.2 Crude associations between ER ratio and smoking outcomes in country– specific strata

Strata	ER ratio Tertile▲	Smoking status: current smoker vs non–smoker OR (95% CI)	Smoking intensity in current smokers OR (95% CI)
Men			
Czech	Tertile 1	1.00	1.00
	Tertile 2	1.02 (0.72 – 1.45)	1.50 (0.88 – 2.56)
	Tertile 3	1.26 (0.90 – 1.76)	1.67 (0.93 – 2.99)
	P for trend	0.128	0.197
	OR by 1 tertile	1.14 (0.96 – 1.35)	1.19 (0.91 – 1.55)
Russia	Tertile 1	1.00	1.00
	Tertile 2	1.31 (1.04 – 1.64)*	1.05 (0.75 – 1.47)
	Tertile 3	1.65 (1.26 – 2.16)*	1.34 (0.91 – 1.98)
	P for trend	< 0.001	0.166
	OR by 1 tertile	1.29 (1.12 – 1.47)*	1.14 (0.95 – 1.39)
Poland	Tertile 1	1.00	1.00
	Tertile 2	1.31 (0.97 – 1.77)	1.43 (0.88 – 2.32)
	Tertile 3	1.47 (1.11 – 1.94)*	2.05 (1.30 – 3.25)*
	P for trend	0.008	0.002
	OR by 1 tertile	1.21 (1.05 – 1.38)*	1.42 (1.13 – 1.79)*
Interaction	Country x ERI	P= 0.669	P= 0.322
Women			
Czech	Tertile 1	1.00	1.00
	Tertile 2	1.27 (0.89 – 1.80)	0.88 (0.51 – 1.54)
	Tertile 3	1.40 (1.00 – 1.96)*	1.03 (0.61 – 1.76)
	P for trend	0.057	0.934
	OR by 1 tertile	1.17 (0.99 – 1.39)	1.01 (0.77 – 1.33)
Russia	Tertile 1	1.00	1.00
	Tertile 2	1.37 (0.94 – 1.99)	0.82 (0.46 – 1.58)
	Tertile 3	1.84 (1.22 – 2.75)*	1.24 (0.60 – 2.56)
	P for trend	0.003	0.603
	OR by 1 tertile	1.35 (1.10 – 1.66)*	1.10 (0.77 – 1.58)
Poland	Tertile 1	1.00	1.00
	Tertile 2	1.22 (0.90 – 1.66)	1.43 (0.91 – 2.24)
	Tertile 3	1.46 (1.09 – 1.94)*	1.48 (0.92 – 2.42)
	P for trend	0.009	0.148
	OR by 1 tertile	1.21 (1.05 – 1.39)*	1.20 (0.94 – 1.50)
Interaction	Country x ERI	P= 0.866	P= 0.451

▲ Country– and gender–specific tertiles of ER ratio. * P value < 0.05.

There are country differences observed in Table 6.3. In men, the association between OC and smoking status in Poland is the strongest of all countries. Overall, crude associations between OC at wave 1 and smoking outcomes at wave 2 are not very different across country–specific strata (all p–values by LR test > 0.26); no significant interaction between country and OC is reported. Due to no significant interactions between country and exposure variables (ER ratio and OC), data for the three countries are pooled for further analyses.

 Table 6.3
 Crude associations between OC and smoking outcomes in country–

Strata	OC score Tertile▲	Smoking status: current smoker vs non–smoker OR (95% CI)	Smoking intensity in current smokers OR (95% CI)
Men			
Czech	Tertile 1	1.00	1.00
	Tertile 2	0.97 (0.74 – 1.27)	1.02 (0.62 – 1.70)
	Tertile 3	1.03 (0.75 – 1.43)	1.17 (0.69 – 1.99)
	P for trend	0.968	0.557
	OR by 1 tertile	1.00 (0.85 – 1.19)	1.08 (0.83 – 1.41)
Russia	Tertile 1	1.00	1.00
	Tertile 2	1.17 (0.89 – 1.53)	1.03 (0.72 – 1.47)
	Tertile 3	1.20 (0.94 – 1.55)	1.12 (0.74 – 1.68)
	P for trend	0.149	0.614
	OR by 1 tertile	1.10 (0.97 – 1.26)	1.05 (0.86 – 1.28)
Poland	Tertile 1	1.00	1.00
	Tertile 2	1.19 (0.91 – 1.55)	1.18 (0.77 – 1.80)
	Tertile 3	1.48 (1.11 – 1.98)*	1.59 (1.00 – 2.54)*
	P for trend	0.008	0.056
	OR by 1 tertile	1.22 (1.05 – 1.41)*	1.25 (0.99 – 1.58)
Interaction	Country x OC	P= 0.262	P= 0.825
Women			
Czech	Tertile 1	1.00	1.00
	Tertile 2	1.17 (0.84 – 1.61)	0.90 (0.51 – 1.50)
	Tertile 3	1.42 (1.02 – 1.98)*	0.93 (0.53 – 1.61)
	P for trend	0.040	0.792
	OR by 1 tertile	1.19 (1.01 – 1.41)*	0.97 (0.74 – 1.27)
Russia	Tertile 1	1.00	1.00
	Tertile 2	1.19 (0.79 – 1.78)	1.14 (0.55 – 2.36)
	Tertile 3	1.33 (0.91 – 1.96)	1.21 (0.61 – 2.42)
	P for trend	0.129	0.571
	OR by 1 tertile	1.16 (0.96 – 1.40)	1.10 (0.78 – 1.55)
Poland	Tertile 1	1.00	1.00
	Tertile 2	0.93 (0.71 – 1.23)	1.17 (0.76 – 1.79)
	Tertile 3	1.12 (0.83 – 1.50)	1.62 (1.03 – 2.56)*
	P for trend	0.533	0.041
	OR by 1 tertile	1.05 (0.90 – 1.21)	1.27 (1.01 – 1.59)*
Interaction	country x OC	P= 0.756	P= 0.606

specific strata

▲ Country– and gender–specific tertiles of OC score. * P value < 0.05.

Table 6.4 shows the bivariate analyses between covariates at wave 1 and smoking outcomes at wave 2 in men. Chi–squared tests and p–values for heterogeneity are obtained to examine the differences between categories of the variable. Smoking status as current smokers is significantly (p < 0.05) associated with younger age, less education, manual worker, divorce or widowed, higher deprivation, and poorer self-rated health. Besides, the association of higher smoking intensity with lower education is significant (p < 0.05), and the association of higher smoking intensity with manual workers is marginally significant (p < 0.1).

	Smoking statu	S	Smoking inten	sity in current si	nokers
	Non-	Current	Light	Medium	Heavy
Covariates	smokers (%)	smokers (%)	smokers (%)	smokers (%)	smokers (%)
	n= 2435	n= 1347	n= 195	n= 441	n= 711
Age: 45 – 49	58.4	41.6	13.9	31.8	54.3
50 – 54	63.4	36.6	14.4	32.6	53.0
55 – 59	67.7	32.3	13.0	34.2	52.8
60 – 69	74.0	26.0	16.3	33.0	50.7
P value	P < 0.001		P = 0.519		
Education: Primary/ less	44.4	55.6	3.4	32.6	64.0
Vocational	60.2	39.8	10.2	37.7	52.1
Secondary	61.7	38.3	17.2	30.6	52.2
University	74.9	25.1	17.3	30.5	52.2
P value	P < 0.001		P < 0.001		
Occupation class					
Manager/ profession	68.1	31.9	13.2	35.9	50.9
Non-manual worker	70.4	29.6	17.1	32.0	50.9
Manual worker	56.2	43.8	11.9	31.6	56.4
P value	P < 0.001		P = 0.066		
Marital status					
Married/ cohabiting	66.2	33.8	14.8	32.6	52.5
Single	69.4	30.6	8.8	32.4	58.8
Divorce/ widowed	56.7	43.3	10.0	34.6	55.4
P value	P = 0.003		P = 0.523		
Deprivation: Low (0-3.9)	68.0	32.0	14.5	33.7	51.8
High (4– 9)	56.1	43.9	13.7	30.1	56.3
P value	P < 0.001		P = 0.323		
Depression: CESD < 16	67.1	32.9	14.5	33.7	51.8
CESD >= 16	64.6	35.4	13.7	30.1	56.3
P value	P = 0.299		P = 0.313		
Social isolation					
No (>= once a month)	65.9	34.1	15.3	33.0	51.8
Yes (< once a month)	65.2	34.8	13.1	32.6	54.3
P value	P = 0.596		P = 0.434		
Self-rated health					
Very good– average	66.2	33.8	14.5	32.9	52.7
Poor- very poor	56.6	43.4	11.9	32.2	55.9
P value	P = 0.001		P = 0.691		

Table 6.4 Bivariate analyses between covariates and smoking outcomes in men

Table 6.5 shows the bivariate analyses between covariates at wave 1 and smoking outcomes at wave 2 in women. Being current smokers is significantly (p < 0.05) associated with younger age, less education, manual worker, more depression, and less social isolation. In current smokers, higher smoking intensity is significantly associated with being single, more depression, and more social isolation, respectively. Table 6.5 Bivariate analyses between covariates and smoking outcomes in women

	Smoking statu	S	Smoking inten	sity in current sr	nokers
	Non-	Current	Light	Medium	Heavy
Covariates	smokers (%)	smokers (%)	smokers (%)	smokers (%)	smokers (%)
	n= 2907	n= 824	n= 278	n= 361	n= 185
Age: 45 – 49	73.4	26.6	35.2	42.7	22.1
50 – 54	76.6	23.5	32.2	44.4	23.4
55 – 59	83.7	16.3	32.8	45.6	21.6
60 - 69	92.1	7.9	35.1	43.4	21.4
P value	P < 0.001		P = 0.686		
Education					
Primary or less	73.2	26.8	34.0	39.6	26.4
Vocational	76.9	23.1	33.4	44.0	22.6
Secondary	77.3	22.7	31.7	46.1	22.2
University	82.8	17.2	35.9	41.3	22.9
P value	P < 0.001		P = 0.697		
Occupation class					
Manager/ profession	80.1	19.9	37.2	42.1	20.7
Non-manual worker	79.3	20.7	33.3	42.9	23.8
Manual worker	74.8	25.2	32.3	50.0	17.7
P value	P = 0.027		P = 0.340		
Marital status					
Married/ cohabiting	79.7	20.3	34.1	46.7	19.3
Single	77.5	22.5	29.5	35.9	34.6
Divorce/ widowed	76.4	23.6	34.7	39.6	25.7
P value	P = 0.069		P = 0.005		
Deprivation					
Low (0 – 3.9)	78.5	21.5	35.6	43.6	20.8
High (4 – 9)	79.2	20.8	29.1	44.8	26.1
P value	P = 0.653		P = 0.098		
Depression					
CESD < 16	78.8	21.2	30.0	47.8	22.3
CESD >= 16	74.9	25.1	40.8	32.6	26.6
P value	P = 0.016		P = 0.001		
Social isolation					
No (>= once a month)	77.2	22.8	36.8	44.9	18.4
Yes (< once a month)	79.7	20.3	29.3	42.2	28.5
P value	P = 0.006		P < 0.001		
Self-rated health					
Very good- average	78.6	21.4	33.2	44.2	22.6
Poor- very poor	79.9	20.1	38.3	40.7	21.0
P value	P = 0.533		P = 0.656		

6.2 Potential Role of OC in ERI–Smoking Relationship

This section focuses on the associations between ERI and smoking outcomes, on the associations between OC and smoking outcomes, and on assessing whether OC has the potential role of antecedent, mediator, modifier or direct effect in the relationship between ERI and smoking. Data for three countries are pooled for further analyses due to no significant interaction between country and exposure variables.

6.2.1 Associations between ERI and smoking outcomes

The association between ER ratio at wave 1 and smoking status (binary outcome: current smokers versus non-smokers) at wave 2 is assessed using binary logistic regression. Additionally, the association between ER ratio at wave 1 and smoking intensity among current smokers (ordinal categorical outcome: light smoker, medium smoker, and heavy smoker) is evaluated using ordinal logistic regression. These associations are assessed after adjustment for age and country (Model 1) and after additionally adjustment for other covariates (Model 2).

Table 6.6 reports the associations between ER ratio at wave 1 and two smoking outcomes at wave 2. In men, the odds of being current smokers are 1.40 for highest versus lowest tertile of ER ratio in Model 2; the adjusted OR changes for being current smokers are 1.18 by 1–tertile increase in ER ratio (p=0.001). Among current smokers, the odds of a higher versus a lower outcome of smoking intensity (>= 20 versus < 20 cigarettes/day; >= 10 versus < 10 cigarettes/day) are 1.41 for highest versus lowest tertile of ER ratio in Model 2; the adjusted OR changes for a higher versus a lower outcome of smoking intensity (p=0.015).

In women, the odds of being current smokers are 1.48 for highest versus lowest tertile of ER ratio in Model 2; the adjusted OR changes for being current smokers are 1.21 by 1–tertile increase in ER ratio (p < 0.001). Among current smokers, the odds of a higher versus a lower outcome of smoking intensity are 1.33 for highest versus

178

lowest tertile of ER ratio in Model 2; the adjusted OR changes for a higher versus a lower outcome of smoking intensity are 1.16 by 1–tertile increase in ER ratio (p= 0.088). Table 6.6 Associations between ER ratio at wave 1 and smoking outcomes at wave

2

Model	ER ratio	Smoking status – current	Smoking intensity in
	Tertile▲	smokers vs non-smokers	current smokers
		OR (95% CI)	OR (95% CI)
Men		N = 3782	N = 1347
Model 1*	Tertile 1	1.00	1.00
	Tertile 2	1.26 (1.07 – 1.49)*	1.32 (1.02 – 1.70)*
	Tertile 3	1.39 (1.18 – 1.63)*	1.49 (1.14 – 1.93)*
	P for trend	0.001	0.002
	OR by 1 tertile	1.18 (1.09 – 1.28)*	1.22 (1.07 – 1.39)*
Model 2**	Tertile 1	1.00	1.00
	Tertile 2	1.33 (1.09 – 1.61)*	1.28 (0.97 – 1.69)
	Tertile 3	1.40 (1.16 – 1.68)*	1.41 (1.06 – 1.87)*
	P for trend	0.001	0.015
	OR by 1 tertile	1.18 (1.08 – 1.30)*	1.19 (1.03 – 1.37)*
Women		N = 3731	N = 824
Model 1*	Tertile 1	1.00	1.00
	Tertile 2	1.32 (1.09 – 1.61)*	1.10 (0.80 – 1.51)
	Tertile 3	1.44 (1.18 – 1.74)*	1.45 (1.06 – 1.97)*
	P for trend	0.001	0.042
	OR by 1 tertile	1.20 (1.08 – 1.31)*	1.18 (1.01 – 1.38)*
Model 2**	Tertile 1	1.00	1.00
	Tertile 2	1.18 (0.92 – 1.51)	1.06 (0.74 – 1.50)
	Tertile 3	1.48 (1.15 – 1.90)*	1.33 (0.95 – 1.88)*
	P for trend	< 0.001	0.088
	OR by 1 tertile	1.21 (1.08 – 1.36)*	1.16 (0.98 – 1.37)

* Model 1: adjusted for age and country. ** Model 2: additionally adjusted for other covariates such as education, occupation, marital status, deprivation, depression, social isolation, and self-rated health. A Gender–specific tertile of ER ratio: in men, tertile 1 (0.20–0.32), tertile 2 (0.32–0.47), and tertile 3 (>= 0.47); in women, tertile 1 (0.20–0.31), tertile 2 (0.31–0.46), and tertile 3 (>= 0.46). * P value < 0.05.

6.2.2 Associations between OC and smoking outcomes

The association between OC at wave 1 and smoking status at wave 2 is assessed using binary logistic regression. Additionally, the association between OC at wave 1 and smoking intensity among current smokers is evaluated using ordinal logistic regression. These associations are assessed after adjustment for age and country (Model 1) and after additionally adjustment for other covariates (Model 2).

Table 6.7 reports the associations between OC score at wave 1 and two smoking

outcomes at wave 2, respectively. In men, the odds of being current smokers are 1.33 for highest versus lowest tertile of OC in Model 2; the adjusted OR changes for being current smokers are 1.14 by 1–tertile increase in OC (p= 0.008). For current smokers, the odds of a higher versus a lower outcome of smoking intensity are 1.20 for highest versus lowest tertile of OC in Model 2; the adjusted OR changes for a higher versus a lower outcome of smoking intensity are 1.20 for highest versus lowest tertile of OC in Model 2; the adjusted OR changes for a higher versus a lower outcome of smoking intensity are 1.20 for highest versus a lower outcome of smoking intensity are 1.20 for highest versus lowest tertile of OC in Model 2; the adjusted OR changes for a higher versus a lower outcome of smoking intensity are 1.11 by 1–tertile increase in OC (p= 0.171).

In women, the odds of being current smokers are 1.32 for highest versus lowest tertile of OC in Model 2; the adjusted OR changes for being current smokers are 1.15 by 1–tertile increase in OC (p= 0.024). Among current smokers, the odds of a higher versus a lower outcome of smoking intensity are 1.29 for highest versus lowest tertile of OC in Model 2; the adjusted OR changes for a higher versus a lower outcome of smoking intensity are 1.29 for highest versus a lower outcome of smoking intensity are 1.29 for highest versus a lower outcome of smoking intensity are 1.29 for highest versus a lower outcome of smoking intensity are 1.29 for highest versus a lower outcome of smoking intensity are 1.14 by 1–tertile increase in OC (p= 0.145).

Model	OC score	Smoking status – current	Smoking intensity in
	Tertile	smokers vs non-smokers	current smokers
		OR (95% CI)	OR (95% CI)
Men		N = 3782	N = 1347
Model 1*	Tertile 1	1.00	1.00
	Tertile 2	1.07 (0.91 – 1.26)	1.21 (0.93 – 1.56)
	Tertile 3	1.20 (1.01 – 1.40)*	1.25 (0.96 – 1.63)
	P for trend	0.061	0.054
	OR by 1 tertile	1.08 (1.00 – 1.18)	1.13 (1.00 – 1.30)
Model 2**	Tertile 1	1.00	1.00
	Tertile 2	1.06 (0.89 – 1.26)	1.17 (0.90 – 1.52)
	Tertile 3	1.33 (1.07 – 1.60)*	1.20 (0.90 – 1.61)
	P for trend	0.008	0.171
	OR by 1 tertile	1.14 (1.03 – 1.28)*	1.11 (0.96 – 1.28)
Women		N = 3731	N = 824
Model 1*	Tertile 1	1.00	1.00
	Tertile 2	1.24 (1.04 – 1.49)*	1.13 (0.84 – 1.53)
	Tertile 3	1.35 (1.11 – 1.62)*	1.21 (0.89 – 1.65)
	P for trend	0.001	0.220
	OR by 1 tertile	1.16 (1.06 – 1.28)*	1.10 (0.94 – 1.28)
Model 2**	Tertile 1	1.00	1.00
	Tertile 2	1.15 (0.91 – 1.47)	1.09 (0.79 – 1.52)
	Tertile 3	1.32 (1.03 – 1.69)*	1.29 (0.92 – 1.82)
	P for trend	0.024	0.145
	OR by 1 tertile	1.15 (1.01 – 1.30)*	1.14 (0.96 – 1.35)

* Model 1: adjusted for age and country. ** Model 2: additionally adjusted for other covariates.

▲ Gender–specific tertile of OC score in men and women: tertile 1 (6–12), tertile 2 (12–15), and tertile 3 (15–24). * P value < 0.05.
Previous analyses have shown consistent associations of exposure variables (ER ratio and OC) with two smoking outcomes. Most associations between exposure variables and smoking outcomes are significant, but OC–smoking intensity associations in men and women do not reach statistical significance.

6.2.3 Antecedent or mediator role of OC in ERI-smoking relationship

To assess antecedent or mediator roles of OC in ERI–smoking relationship, the path analyses for binary categorical outcome (smoking status) and for ordinal categorical outcome (smoking intensity) are applied by Mplus 7. Each path coefficient is obtained by probit regression for an outcome on a predictor after adjustment for covariates. The odds ratio (OR) is the exponential (antilog) of estimated logistic coefficient, which is calculated from probit coefficient multiplied by 1.8.⁵⁸⁸ In terms of predictors, first, the tertiles of ERI and OC are transformed into a series of dummy variables to compare between tertile groups in each predictor. Second, ERI tertile and OC tertile are treated as continuous variables to estimate assumed linear trend between the exposure and odds of smoking outcomes.

Path analysis with an autoregressive and cross–lagged model is specified in this 2–wave cohort study (Figure 6.1). First, each variable is predicted by the same variable at an early wave. Second, the cross–lagged effects of "OC at wave 1 on ERI at wave 2" and "ERI at wave 1 on OC at wave 2" are assessed, respectively, to identify causal directionality between OC and ERI. Bidirectional relationship between OC and ERI is possible based on my hypotheses.

Third, as mediation is a causal chain involving at least two causal relations, these causal relations can be tested separately using two phases of data. The mediator role of ERI in OC–smoking relation is assessed by two cross–lagged effects: (1) OC at wave 1 on ERI at wave 2; (2) ERI at wave 1 on smoking at wave 2. The mediator role of OC in ERI–smoking relation is estimated by two cross–lagged effects: (1) ERI at

wave 1 on OC at wave 2; (2) OC at wave 1 on smoking at wave 2. Partial mediation applies if both causal relationships are confirmed; the product of two path coefficients can estimate the strength of mediator effect.

For a mediator effect, the effect size measure is the product of two path coefficients. The bootstrap method is used for significance testing of the mediator effect with 5000 bootstrap samples to yield valid estimates for the mediator effect by Mplus 7. This method is adopted due to complicated models (categorical outcomes and multiple mediators) in the SEM.

For tests of model fit, three indexes are adopted: Root Mean Square Error of Approximation (RMSEA), Comparative Fit Index (CFI), and Tucker–Lewis Index (TLI). RMSEA < 0.06, CFI > 0.95 or TLI > 0.95 indicate "good model fit".

Figure 6.1 Hypothetical model specified for the path analysis for antecedent or mediator role of OC in ERI–smoking relation



Table 6.8 reports the results of path analysis for the outcome of smoking status in men, with the results shown in Figure 6.2. First, the mediator effect of ERI in OC– smoking relation is estimated by multiplying 2 cross–lagged effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 2 (unstandardized β = 0.155; standard error= 0.020); (2) higher ERI at wave 1 significantly associated with being current smokers at wave 2 (unstandardized β = 0.075; standard error= 0.028). Thus, this mediator effect of ERI is significant (0.012= 0.155 x 0.075; standard error= 0.005 and p= 0.011 by bootstrap method).

Second, the mediator effect of OC in ERI–smoking relation is estimated by multiplying 2 cross–lagged effects: (1) higher ERI at wave 1 significantly associated with higher OC at wave 2 (unstandardized β = 0.084; standard error= 0.020); (2) higher OC at wave 1 non-significantly associated with being current smokers at wave 2 (unstandardized β = 0.028; standard error= 0.028; p= 0.321). This mediator effect of OC is not significant (0.002= 0.084 x 0.028; standard error= 0.002 and p= 0.330). Third, the fit indexes are considered good fit (RMSEA= 0.048 < 0.06; CFI= 0.953 > 0.95) or close to cutoffs for good fit (TLI= 0.843).

Table 6.8 Results of path analysis for antecedent or mediator role of OC in ERI-

smoking status relationship in men (N= 3782)

Parameter	Odds	Unstandardiz	Standardized	P value
Parameter	Ratio	ed coefficient	coefficient	P value
Autoregressive model				
Smoking status wave 1 \rightarrow wave 2	4.68	0.908	0.554	< 0.001
OC wave 1 \rightarrow OC wave 2				
OC tertile 2 vs tertile 1	1.66	0.290	0.157	< 0.001
OC tertile 3 vs tertile 1	3.14	0.674	0.323	< 0.001
OR change by 1 tertile	1.79	0.337	0.303	< 0.001
ERI wave 1 \rightarrow ERI wave 2				
ERI tertile 2 vs tertile 1	1.46	0.211	0.123	< 0.001
ERI tertile 3 vs tertile 1	2.28	0.484	0.283	< 0.001
OR change by 1 tertile	1.53	0.238	0.242	< 0.001
$OC \rightarrow ERI \rightarrow Smoking status$	1.02	0.012	0.009	0.011
OC wave 1 → ERI wave 2				
OC tertile 2 vs tertile 1	1.32	0.156	0.091	< 0.001
OC tertile 3 vs tertile 1	1.70	0.304	0.158	< 0.001
OR change by 1 tertile	1.32	0.155	0.151	< 0.001
ERI wave 1 → Smoking status wave 2				
ERI tertile 2 vs tertile 1	1.27	0.131	0.048	0.020
ERI tertile 3 vs tertile 1	1.35	0.168	0.062	0.008
OR change by 1 tertile	1.14	0.075	0.058	0.010
$\underline{ERI} \rightarrow OC \rightarrow Smoking status$	1.00	0.002	0.002	0.330
ERI wave 1 \rightarrow OC wave 2				
ERI tertile 2 vs tertile 1	1.26	0.128	0.070	< 0.001
ERI tertile 3 vs tertile 1	1.38	0.179	0.096	< 0.001
OR change by 1 tertile	1.16	0.084	0.078	< 0.001
OC wave 1 \rightarrow Smoking status wave 2				
OC tertile 2 vs tertile 1	1.02	0.010	0.003	0.725
OC tertile 3 vs tertile 1	1.17	0.088	0.029	0.132
OR change by 1 tertile	1.05	0.028	0.021	0.321
Tests of model fit	RMSEA=	0.048	CFI= 0.953	TLI= 0.843

Figure 6.2 Results of path analysis for antecedent or mediator role of OC in ERI– smoking status relationship in men



Table 6.9 shows the results of path analysis for the outcome of smoking status in women, with the results illustrated in Figure 6.3. First, the mediator effect of ERI in OC–smoking relation is estimated by multiplying 2 cross–lagged effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 2 (unstandardized β = 0.155; standard error= 0.019); (2) higher ERI at wave 1 significantly associated with being current smokers at wave 2 (unstandardized β = 0.085; standard error= 0.024). This mediator effect of ERI is significant (0.013= 0.155 x 0.085; standard error= 0.004 and p= 0.002 by bootstrap method).

Second, the mediator effect of OC in ERI–smoking relationship is evaluated by multiplying 2 cross–lagged effects: (1) higher ERI at wave 1 significantly associated with higher OC at wave 2 (unstandardized β = 0.075; standard error= 0.020); (2) higher OC at wave 1 non-significantly associated with being current smokers at wave 2 (unstandardized β = 0.037; standard error= 0.025; p= 0.113). This mediator effect of

OC is not significant (0.003; standard error= 0.002 and p= 0.168). Third, the fit indexes show good fit (RMSEA= 0.055 < 0.06) or close to good fit (CLI= 0.926; TLI= 0.825).

Table 6.9 Results of path analysis for antecedent or mediator role of OC in ERI-

Deremeter	Odds	Unstandardiz	Standardized	Dualua
Parameter	Ratio	ed coefficient	coefficient	P value
Autoregressive model				
Smoking status wave 1 \rightarrow wave 2	4.75	0.917	0.562	< 0.001
OC wave 1 \rightarrow OC wave 2				
OC tertile 2 vs tertile 1	1.74	0.326	0.175	< 0.001
OC tertile 3 vs tertile 1	3.27	0.698	0.357	< 0.001
OR change by 1 tertile	1.83	0.357	0.340	< 0.001
ERI wave 1 → ERI wave 2				
ERI tertile 2 vs tertile 1	1.62	0.275	0.163	< 0.001
ERI tertile 3 vs tertile 1	2.37	0.507	0.301	< 0.001
OR change by 1 tertile	1.56	0.252	0.262	< 0.001
$OC \rightarrow ERI \rightarrow Smoking status$	1.02	0.013	0.010	0.002
OC wave 1 \rightarrow ERI wave 2				
OC tertile 2 vs tertile 1	1.27	0.131	0.076	< 0.001
OC tertile 3 vs tertile 1	1.76	0.334	0.185	< 0.001
OR change by 1 tertile	1.32	0.155	0.149	< 0.001
ERI wave 1 \rightarrow Smoking status wave 2				
ERI tertile 2 vs tertile 1	1.12	0.063	0.022	0.395
ERI tertile 3 vs tertile 1	1.45	0.211	0.074	< 0.001
OR change by 1 tertile	1.17	0.085	0.066	0.002
$ERI \rightarrow OC \rightarrow Smoking status$	1.01	0.003	0.002	0.168
ERI wave 1 \rightarrow OC wave 2				
ERI tertile 2 vs tertile 1	1.16	0.083	0.046	0.022
ERI tertile 3 vs tertile 1	1.35	0.166	0.091	< 0.001
OR change by 1 tertile	1.07	0.075	0.071	< 0.001
OC wave 1 \rightarrow Smoking status wave 2				
OC tertile 2 vs tertile 1	1.07	0.037	0.012	0.569
OC tertile 3 vs tertile 1	1.13	0.066	0.021	0.364
OR change by 1 tertile	1.07	0.037	0.030	0.113
Tests of model fit	RMSEA=	0.055	CFI= 0.926	TLI= 0.825

Figure 6.3 Results of path analysis for antecedent or mediator role of OC in ERI– smoking status relationship in women



Table 6.10 reports the results of path analysis for the outcome of smoking intensity among current smokers in men, with the results illustrated in Figure 6.4. First, the mediator effect of ERI in OC–smoking relation is assessed by multiplying 2 cross– lagged effects: (1) higher OC at wave 1 is significantly associated with higher ERI at wave 2 (unstandardized β = 0.167; standard error= 0.030); (2) higher ERI at wave 1 is marginally significantly associated with higher levels of smoking intensity at wave 2 (unstandardized β = 0.084; standard error= 0.039; p= 0.080). This mediator effect of ERI appears significant (0.014= 0.167 x 0.084; standard error= 0.007 and p= 0.042).

Second, the mediator effect of OC in ERI–smoking relation is estimated by multiplying 2 cross–lagged effects: (1) higher ERI at wave 1 significantly associated with higher OC at wave 2 (unstandardized β = 0.091; standard error= 0.033); (2) higher OC at wave 1 non-significantly related to higher levels of smoking intensity at wave 2 (unstandardized β = 0.028; standard error= 0.040; p= 0.358). This mediator effect of

OC is not significant (0.003; standard error= 0.004 and p= 0.497). Third, the fit indexes show good fit (RMSEA= 0.044 < 0.06; CFI= 0.962 > 0.95) or close to cutoffs of good fit (TLI= 0.905).

Table 6.10 Results of path analysis for antecedent or mediator role of OC in ERI-

smoking intensity relationship in men (N= 1347)

Parameter	Odds Ratio	Unstandardiz ed coefficient	Standardized coefficient	P value
Autoregressive model				
Smoking intensity wave 1 \rightarrow wave 2	4.49	0.884	0.616	< 0.001
OC wave 1 \rightarrow OC wave 2				
OC tertile 2 vs tertile 1	1.58	0.260	0.138	< 0.001
OC tertile 3 vs tertile 1	3.28	0.699	0.342	< 0.001
OR change by 1 tertile	1.80	0.346	0.313	< 0.001
ERI wave 1 → ERI wave 2				
ERI tertile 2 vs tertile 1	1.52	0.235	0.140	< 0.001
ERI tertile 3 vs tertile 1	2.05	0.442	0.271	< 0.001
OR change by 1 tertile	1.47	0.223	0.235	< 0.001
$OC \rightarrow ERI \rightarrow Smoking intensity$	1.03	0.014	0.009	0.042
OC wave 1 \rightarrow ERI wave 2				
OC tertile 2 vs tertile 1	1.45	0.208	0.125	< 0.001
OC tertile 3 vs tertile 1	1.74	0.327	0.178	< 0.001
OR change by 1 tertile	1.35	0.167	0.170	< 0.001
ERI wave 1 \rightarrow Smoking intensity wave 2				
ERI tertile 2 vs tertile 1	1.26	0.129	0.046	0.185
ERI tertile 3 vs tertile 1	1.35	0.168	0.061	0.088
OR change by 1 tertile	1.16	0.084	0.054	0.080
<u>ERI \rightarrow OC \rightarrow Smoking intensity</u>	1.00	0.003	0.002	0.497
ERI wave 1 \rightarrow OC wave 2				
ERI tertile 2 vs tertile 1	1.19	0.097	0.051	0.129
ERI tertile 3 vs tertile 1	1.41	0.190	0.103	< 0.001
OR change by 1 tertile	1.18	0.091	0.086	0.006
OC wave 1 \rightarrow Smoking intensity wave 2				
OC tertile 2 vs tertile 1	1.06	0.035	0.013	0.620
OC tertile 3 vs tertile 1	1.14	0.075	0.028	0.253
OR change by 1 tertile	1.05	0.028	0.016	0.358
Tests of model fit	RMSEA=	0.044	CFI= 0.962	TLI= 0.905

Figure 6.4 Results of path analysis for antecedent or mediator role of OC in ERI– smoking intensity relationship in current smokers in men



Table 6.11 presents the results of path analysis for the outcome of smoking intensity among current smokers in women, with the results shown in Figure 6.5. First, the mediator effect of ERI in OC–smoking relation is assessed by multiplying 2 cross–lagged effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 2 (unstandardized β = 0.170; standard error= 0.039); (2) higher ERI at wave 1 non-significantly associated with higher levels of smoking intensity at wave 2 (unstandardized β = 0.068; standard error= 0.046; p= 0.221). This mediator effect of ERI is not significant (0.012= 0.170 x 0.068; standard error= 0.008 and p= 0.161).

Second, the mediator effect of OC in ERI–smoking relation is estimated by multiplying 2 cross–lagged effects: (1) higher ERI at wave 1 significantly associated with higher OC at wave 2 (unstandardized β = 0.080; standard error= 0.041); (2) higher OC at wave 1 non-significantly related to higher levels of smoking intensity at wave 2 (unstandardized β = 0.036; standard error= 0.045; p= 0.408). This mediator effect of

OC is not significant (0.003; standard error= 0.004 and p= 0.459). Third, the fit indexes show good fit (RMSEA= 0.057 < 0.06; CFI= 0.957 > 0.95) or close to cutoffs for good fit (TLI= 0.865).

Table 6.11 Results of path analysis for antecedent or mediator role of OC in ERI-

smoking intensity relationship in women (N= 824)

Parameter	Odds Ratio	Unstandardiz ed coefficient	Standardized coefficient	P value
Autoregressive model				
Smoking intensity wave 1 \rightarrow wave 2	4.37	0.867	0.605	< 0.001
OC wave 1 \rightarrow OC wave 2				
OC tertile 2 vs tertile 1	1.80	0.346	0.193	< 0.001
OC tertile 3 vs tertile 1	3.37	0.715	0.391	< 0.001
OR change by 1 tertile	1.83	0.357	0.352	< 0.001
ERI wave 1 → ERI wave 2				
ERI tertile 2 vs tertile 1	1.36	0.170	0.100	0.020
ERI tertile 3 vs tertile 1	2.40	0.516	0.313	< 0.001
OR change by 1 tertile	1.59	0.265	0.270	< 0.001
$OC \rightarrow ERI \rightarrow Smoking intensity$	1.02	0.012	0.007	0.161
OC wave 1 → ERI wave 2				
OC tertile 2 vs tertile 1	1.27	0.131	0.075	0.059
OC tertile 3 vs tertile 1	1.80	0.347	0.199	< 0.001
OR change by 1 tertile	1.36	0.170	0.176	< 0.001
ERI wave 1 \rightarrow Smoking intensity wave 2				
ERI tertile 2 vs tertile 1	1.03	0.022	0.010	0.867
ERI tertile 3 vs tertile 1	1.26	0.129	0.049	0.297
OR change by 1 tertile	1.13	0.068	0.044	0.221
ERI \rightarrow OC \rightarrow Smoking intensity	1.01	0.003	0.002	0.459
ERI wave 1 \rightarrow OC wave 2				
ERI tertile 2 vs tertile 1	1.25	0.126	0.072	0.087
ERI tertile 3 vs tertile 1	1.33	0.161	0.095	0.053
OR change by 1 tertile	1.15	0.080	0.078	0.053
OC wave 1 \rightarrow Smoking intensity wave 2				
OC tertile 2 vs tertile 1	1.05	0.029	0.011	0.764
OC tertile 3 vs tertile 1	1.15	0.078	0.030	0.431
OR change by 1 tertile	1.07	0.036	0.023	0.408
Tests of model fit	RMSEA=	0.057	CFI= 0.957	TLI= 0.865

Figure 6.5 Results of path analysis for antecedent or mediator role of OC in ERI– smoking intensity relationship in current smokers in women



6.2.4 Modifying role of OC in ERI-smoking relationship

For evaluating modifying role of OC in ERI–smoking relationship, the associations between ER ratio at wave 1 and smoking outcomes at wave 2 in different strata of OC tertile are assessed after adjustment for covariates. Next, binary and ordinal logistic regression analyses are conducted for two smoking outcomes (smoking status and smoking intensity) at wave 2, respectively, regressed by OC tertile, ER–ratio tertile, and the interaction term between OC tertile and ER–ratio tertile at wave 1 after adjustment for covariates. The log likelihoods for the model with this interaction term and the model without are compared, and LR test is used to test significance of the interaction term. In Table 6.12, LR tests show that this interaction term is not significant for smoking status in men (p= 0.371) and women (p= 0.874), and not significant for

smoking intensity in men (p= 0.243) and women (p= 0.988). My results show that OC has no significantly modifying role in ERI–smoking relation.

Approaches	Strata	Smoking status: current	Smoking intensity among
Approaches	Strata	smokers vs non-smokers	current smokers
Men		OR of outcome by 1-tertile	increase in ERI (95% CI)
1. ERI-smoking relation	OC tertile 1	1.24 (1.06 – 1.45)	1.15 (0.91 – 1.45)
in different strata of OC	OC tertile 2	1.11 (0.92 – 1.33)	1.09 (0.83 – 1.43)
tertile	OC tertile 3	1.04 (0.82 – 1.32)	0.88 (0.61 – 1.22)
2. Interaction OC x ERI			
Likelihood-ratio test	P-value	P= 0.371	P= 0.243
Women		OR of outcome by 1-tertile	increase in ERI (95% CI)
1. ERI-smoking relation	OC tertile 1	1.28 (1.04 – 1.59)	1.14 (0.86 – 1.52)
in different strata of OC	OC tertile 2	1.01 (0.80 – 1.26)	0.92 (0.65 – 1.30)
tertile	OC tertile 3	1.34 (0.97 – 1.84)	1.15 (0.78 – 1.69)
2. Interaction OC x ERI			
Likelihood-ratio test	P-value	P= 0.874	P= 0.988

Table 6.12 Evaluation for modifying role of OC in ERI–smoking relationship

6.3 Potential Role of PC in Relation between OC, ERI, and Smoking

The focus of this section is, according to the second aim of the thesis, on the associations between PC and smoking outcomes, and on the potential role of PC in the relationship between ERI, OC, and smoking outcomes.

6.3.1 Associations between PC and smoking outcomes

The associations between PC at wave 1 and two smoking outcomes at wave 2 are assessed following the same steps as for ERI–smoking associations.

Table 6.13 reports the associations between PC at wave 1 and two smoking outcomes at wave 2, respectively. In men, the odds of being current smokers are 0.66 for highest versus lowest tertile of PC in Model 2; the adjusted OR changes of being current smokers are 0.81 by 1–tertile increase in PC (p < 0.001). Among current smokers, the odds of a higher versus a lower outcome of smoking intensity are 0.63

for highest versus lowest tertile of PC in Model 2; the adjusted OR changes of a higher versus a lower outcome of smoking intensity are 0.79 by 1–tertile increase in PC.

In women, the odds of being current smokers are 0.65 for highest versus lowest tertile of PC in Model 2; the adjusted OR changes of being current smokers are 0.81 by 1–tertile increase in PC (p= 0.001). Among current smokers, the odds of a higher outcome versus a lower outcome of smoking intensity are 0.65 for highest versus lowest tertile of PC in Model 2; the adjusted OR changes of a higher outcome versus a lower outcome of smoking intensity are 0.65 for highest versus a lower outcome of smoking intensity are 0.65 for highest versus lowest tertile of PC in Model 2; the adjusted OR changes of a higher outcome versus a lower outcome of smoking intensity are 0.80 by 1–tertile increase in PC. Table 6.13 Associations between perceived control at wave 1 and smoking

outcomes	at	wave 2	

Model	Perceived	Smoking status – current	Smoking intensity in
	control	smokers vs non-smokers	current smokers
	Tertile▲	OR (95% CI)	OR (95% CI)
Men		N = 3782	N = 1347
Model 1*	Tertile 1	1.00	1.00
	Tertile 2	0.81 (0.68 – 0.97)*	0.74 (0.58 – 0.96)*
	Tertile 3	0.69 (0.59 – 0.84)*	0.68 (0.53 – 0.89)*
	P for trend	< 0.001	0.002
	OR by 1 tertile	0.83 (0.76 – 0.91)*	0.82 (0.72 – 0.93)*
Model 2**	Tertile 1	1.00	1.00
	Tertile 2	0.84 (0.70 – 1.00)	0.75 (0.57 – 0.98)*
	Tertile 3	0.66 (0.55 – 0.82)*	0.63 (0.48 – 0.83)*
	P for trend	< 0.001	0.001
	OR by 1 tertile	0.81 (0.74 – 0.90)*	0.79 (0.68 – 0.91)*
Women		N = 3731	N = 824
Model 1*	Tertile 1	1.00	1.00
	Tertile 2	0.87 (0.72 – 1.04)	0.72 (0.52 – 0.97)*
	Tertile 3	0.70 (0.58 – 0.86)*	0.68 (0.50 – 0.92)*
	P for trend	0.001	0.023
	OR by 1 tertile	0.84 (0.76 – 0.92)*	0.81 (0.69 – 0.95)*
Model 2**	Tertile 1	1.00	1.00
	Tertile 2	0.79 (0.62 – 1.01)	0.75 (0.54 – 1.04)
	Tertile 3	0.65 (0.50 - 0.84)*	0.65 (0.47 – 0.91)*
	P for trend	0.001	0.011
	OR by 1 tertile	0.81 (0.72 – 0.92)*	0.80 (0.67 – 0.96)*

* Model 1: adjusted for age and country. ** Model 2: additionally adjusted for other covariates such as education, occupation, marital status, deprivation, depression, social isolation, and self-rated health.

▲ Gender–specific tertile of PC: in men, tertile 1 (0–34), tertile 2 (34–41), and tertile 3 (41–55); in women, tertile 1 (0–33), tertile 2 (33–40), and tertile 3 (40–55).

* P value < 0.05.

6.3.2 Mediator roles of PC and ERI in OC-smoking relationship

To assess mediator roles of PC and ERI in OC–smoking relationship, path analyses for binary outcome (smoking status) and for ordinal categorical outcome (smoking intensity in current smokers) are applied in Mplus 7. Each path coefficient is obtained by probit regression for an outcome on a predictor after adjustment for covariates. There are two potential mediators (PC and ERI) between the effects of OC at wave 1 on smoking outcomes at wave 2. As the HAPIEE study is limited by no measurement of PC at wave 2, the cross–sectional associations between OC, ERI, and PC at wave 1 are analyzed. Thus, PC and ERI are only specified to be correlated, while bi-directional relationship between PC and ERI has been hypothesized.

Path analysis with an autoregressive and cross–lagged model for smoking outcomes is specified (Figure 6.6). First, smoking outcomes at wave 2 are predicted by corresponding smoking outcomes at wave 1. Second, the cross–lagged effects of OC, PC, and ERI at wave 1 on smoking outcomes at wave 2 are measured.

Third, the mediator effect can only be assessed by a half–longitudinal design. For example, the mediator role of PC in OC–smoking relation is assessed by two effects: (1) the cross–sectional association of OC at wave 1 on PC at wave 1; (2) the cross–lagged effect of PC at wave 1 on smoking at wave 2. Partial mediation applies if both causal relation are confirmed; the product of two path coefficients can estimate the strength of mediator effect.

Figure 6.6 Hypothetical model specified for the path analysis for mediator roles of PC and ERI in OC–smoking relation



Table 6.14 presents the results of path analysis for the outcome of smoking status in men; the results are illustrated in Figure 6.7. First, the mediator effect of ERI in the relationship between OC and smoking status is estimated by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 1 (unstandardized β = 0.253; standard error= 0.020); (2) higher ERI at wave 1 significantly associated with being current smokers at wave 2 (unstandardized β = 0.082; standard error= 0.028). The mediator effect of ERI is significant (0.021= 0.253 x 0.082; standard error= 0.007 and p= 0.004 by bootstrap method).

Second, the mediator effect of PC in the relationship between OC and smoking status is assessed by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with lower PC at wave 1 (unstandardized β = –0.110; standard error= 0.020); (2) lower PC at wave 1 significantly associated with being current smokers at wave 2 (unstandardized β = –0.112; standard error= 0.025). The mediator effect of PC is

significant (0.012= -0.110 x -0.112; standard error= 0.004 and p= 0.001). Third, ERI at wave 1 is significantly and inversely associated with PC at wave 1 (unstandardized β = -0.052; p< 0.001). Finally, the fit indexes are considered good fit (RMSEA= 0.052 < 0.06) or close to cutoffs for good fit (CFI= 0.921; TLI= 0.834).

Table 6.14 Results of path analysis for mediator roles of PC and ERI in OC-

smoking status relationship in men (N= 3782)

Parameter	Odds	Unstandardiz	Standardized	P value
	Ratio	ed coefficient	coefficient	
Smoking status wave 1 \rightarrow wave 2	4.70	0.911	0.556	< 0.001
OC wave 1 \rightarrow Smoking status wave 2				
OC tertile 2 vs tertile 1	1.02	0.012	0.004	0.696
OC tertile 3 vs tertile 1	1.17	0.085	0.028	0.179
OR change by 1 tertile	1.05	0.027	0.021	0.338
$OC \rightarrow ERI \rightarrow Smoking status$	1.04	0.021	0.016	0.004
OC wave 1 \rightarrow ERI wave 1				
OC tertile 2 vs tertile 1	1.55	0.250	0.146	< 0.001
OC tertile 3 vs tertile 1	2.28	0.485	0.256	< 0.001
OR change by 1 tertile	1.56	0.253	0.235	< 0.001
ERI wave 1 → Smoking status wave 2				
ERI tertile 2 vs tertile 1	1.25	0.129	0.060	0.002
ERI tertile 3 vs tertile 1	1.41	0.192	0.087	< 0.001
OR change by 1 tertile	1.16	0.082	0.066	0.004
$OC \rightarrow PC \rightarrow Smoking status$	1.02	0.012	0.009	0.001
OC wave 1 \rightarrow PC wave 1				
OC tertile 2 vs tertile 1	0.92	- 0.044	- 0.025	0.127
OC tertile 3 vs tertile 1	0.66	- 0.238	- 0.121	< 0.001
OR change by 1 tertile	0.82	- 0.110	- 0.105	< 0.001
PC wave 1 \rightarrow Smoking status wave 2				
PC tertile 2 vs tertile 1	0.82	- 0.116	- 0.054	0.004
PC tertile 3 vs tertile 1	0.67	- 0.227	- 0.100	< 0.001
OR change by 1 tertile	0.82	- 0.112	- 0.088	< 0.001
ERI correlates with PC		- 0.052	- 0.089	< 0.001
Tests of model fit	RMSEA =	0.052	CFI= 0.921	TLI= 0.834

Figure 6.7 Results of path analysis for mediator roles of PC and ERI in OC–smoking status relationship in men



Table 6.15 shows the results of path analysis for the outcome of smoking status in women, with the results shown in Figure 6.8. First, the mediator effect of ERI in the relationship between OC and smoking status is estimated by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 1 (unstandardized β = 0.244; standard error= 0.020); (2) higher ERI at wave 1 significantly associated with being current smokers at wave 2 (unstandardized β = 0.084; standard error= 0.031). This mediator effect of ERI is significant (0.020= 0.244 x 0.084; standard error= 0.008 and p= 0.008).

Second, the mediator effect of PC in the relationship between OC and smoking status is estimated by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with lower PC at wave 1 (unstandardized β = -0.122; standard error= 0.019); (2) lower PC at wave 1 significantly associated with being current smokers at wave 2 (unstandardized β = -0.116; standard error= 0.033). This mediator effect of PC

is significant (0.014; standard error= 0.005 and p= 0.002). Third, ERI at wave 1 is significantly and inversely associated with PC at wave 1 (unstandardized β = -0.056; p< 0.001). Finally, the fit indexes are considered good fit (RMSEA= 0.055 < 0.06) or close to cutoffs for good fit (CFI= 0.931; TLI= 0.819).

Table 6.15 Results of path analysis for mediator roles of PC and ERI in OC-

smoking status relationship in women (N= 3731)

Parameter	Odds Ratio	Unstandardiz ed coefficient	Standardized coefficient	P value
Smoking status wave $1 \rightarrow$ wave 2	4.77	0.920	0.563	< 0.001
OC wave 1 \rightarrow Smoking status wave 2				
OC tertile 2 vs tertile 1	1.08	0.043	0.015	0.520
OC tertile 3 vs tertile 1	1.12	0.062	0.020	0.407
OR change by 1 tertile	1.06	0.033	0.020	0.390
$OC \rightarrow ERI \rightarrow Smoking status$	1.04	0.020	0.015	0.008
OC wave 1 \rightarrow ERI wave 1				
OC tertile 2 vs tertile 1	1.58	0.262	0.147	< 0.001
OC tertile 3 vs tertile 1	2.25	0.476	0.255	< 0.001
OR change by 1 tertile	1.53	0.244	0.236	< 0.001
ERI wave 1 \rightarrow Smoking status wave 2				
ERI tertile 2 vs tertile 1	1.14	0.073	0.025	0.290
ERI tertile 3 vs tertile 1	1.36	0.170	0.059	0.022
OR change by 1 tertile	1.16	0.084	0.052	0.017
$\frac{OC \rightarrow PC \rightarrow Smoking status}{OC wave 1 \rightarrow PC wave 1}$	1.03	0.014	0.010	0.002
OC tertile 2 vs tertile 1	0.85	- 0.093	- 0.052	0.002
OC tertile 3 vs tertile 1	0.64	- 0.251	- 0.133	< 0.002
OR change by 1 tertile	0.80	- 0.122	- 0.133	< 0.001
PC wave 1 \rightarrow Smoking status wave 2	0.80	- 0.122	- 0.120	< 0.001
PC tertile 2 vs tertile 1	0.79	- 0.133	- 0.048	0.024
PC tertile 3 vs tertile 1	0.79	- 0.133	- 0.048	< 0.001
OR change by 1 tertile	0.81	- 0.224 - 0.116	- 0.077	< 0.001 0.001
ERI correlates with PC		- 0.056	- 0.094	< 0.001
Tests of model fit	RMSEA =	0.055	CFI= 0.931	TLI= 0.819

Figure 6.8 Results of path analysis for mediator roles of PC and ERI in OC-

smoking status relationship in women



Table 6.16 shows the results of path analysis for the outcome of smoking intensity among current smokers in men, with the results illustrated in Figure 6.9. First, the mediator effect of ERI in the relationship between OC and smoking intensity is estimated by multiplying 2 effects: (1) higher OC at wave 1 is significantly associated with higher ERI at wave 1 (unstandardized β = 0.260; standard error= 0.033); (2) higher ERI at wave 1 is marginally significantly associated with higher levels of smoking intensity at wave 2 (unstandardized β = 0.078; standard error= 0.038; p= 0.097). This mediator effect of ERI is significant (0.020; standard error= 0.010 and p= 0.042).

Second, the mediator effect of PC in the relationship between OC and smoking intensity is estimated by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with lower PC at wave 1 (unstandardized β = –0.114; standard error= 0.028); (2) lower PC at wave 1 significantly associated with higher levels of smoking intensity at wave 2 (unstandardized β = –0.125; standard error= 0.039). This mediator effect of

PC is significant (0.014; standard error= 0.006 and p= 0.017). Third, ERI at wave 1 is significantly and inversely associated with PC at wave 1 (unstandardized β = -0.055; p< 0.001). Finally, the fit indexes are considered good fit (RMSEA= 0.050 < 0.06; CFI= 0.962 > 0.95) or close to cutoffs for good fit (TLI= 0.916).

Table 6.16 Results of path analysis for mediator roles of PC and ERI in OC-

smoking intensity relationship in men (N= 1347)

Parameter	Odds	Unstandardize	Standardized	P value
Parameter	Ratio	d coefficient	coefficient	
Smoking intensity wave $1 \rightarrow$ wave 2	4.45	0.878	0.612	< 0.001
OC wave 1 \rightarrow Smoking intensity wave 2				
OC tertile 2 vs tertile 1	1.02	0.011	0.004	0.732
OC tertile 3 vs tertile 1	1.12	0.064	0.021	0.504
OR change by 1 tertile	1.04	0.024	0.015	0.611
$OC \rightarrow ERI \rightarrow Smoking intensity$	1.04	0.020	0.012	0.042
OC wave 1 \rightarrow ERI wave 1				
OC tertile 2 vs tertile 1	1.65	0.288	0.164	< 0.001
OC tertile 3 vs tertile 1	2.39	0.513	0.269	< 0.001
OR change by 1 tertile	1.58	0.260	0.254	< 0.001
ERI wave 1 → Smoking intensity wave 2				
ERI tertile 2 vs tertile 1	1.28	0.136	0.049	0.134
ERI tertile 3 vs tertile 1	1.35	0.166	0.059	0.057
OR change by 1 tertile	1.15	0.078	0.049	0.097
$OC \rightarrow PC \rightarrow Smoking intensity$	1.03	0.014	0.009	0.017
OC wave 1 \rightarrow PC wave 1				
OC tertile 2 vs tertile 1	0.85	- 0.090	- 0.051	0.077
OC tertile 3 vs tertile 1	0.66	- 0.234	- 0.123	< 0.001
OR change by 1 tertile	0.81	- 0.114	- 0.113	< 0.001
PC wave 1 \rightarrow Smoking intensity wave 2				
PC tertile 2 vs tertile 1	0.77	- 0.143	- 0.050	0.063
PC tertile 3 vs tertile 1	0.64	- 0.258	- 0.086	0.003
OR change by 1 tertile	0.80	- 0.125	- 0.078	0.001
ERI correlates with PC		- 0.055	- 0.091	0.001
Tests of model fit	RMSEA =	0.050	CFI = 0.962	TLI = 0.916

Figure 6.9 Results of path analysis for mediator roles of PC and ERI in OC-



smoking intensity relationship in current smokers in men

Table 6.17 reports the results of path analysis for the outcome of smoking intensity among current smokers in women, with the results illustrated in Figure 6.10. First, the mediator effect of ERI in the relationship between OC and smoking intensity is assessed by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with higher ERI at wave 1 (unstandardized β = 0.258; standard error= 0.038); (2) higher ERI at wave 1 non-significantly related to higher levels of smoking intensity at wave 2 (unstandardized β = 0.064; standard error= 0.041; p= 0.276). This mediator effect of ERI appears not significant (0.017= 0.258 x 0.064; standard error= 0.011 and p= 0.117).

Second, the mediator effect of PC in the relationship between OC and smoking intensity is estimated by multiplying 2 effects: (1) higher OC at wave 1 significantly associated with lower PC at wave 1 (unstandardized β = –0.118; standard error= 0.037); (2) lower PC at wave 1 significantly associated with higher levels of smoking intensity at wave 2 (unstandardized β = –0.121; standard error= 0.043). This mediator effect of

PC is significant (0.014; standard error= 0.007 and p= 0.040). Third, ERI at wave 1 is significantly and inversely associated with PC at wave 1 (unstandardized β = -0.062; p< 0.001). Finally, the fit indexes are considered good fit (RMSEA= 0.058 < 0.06) or close to cutoffs for good fit (CFI= 0.929; TLI= 0.815).

Table 6.17 Results of path analysis for mediator roles of PC and ERI in OC-

smoking intensity relationship in women (N= 824)

Parameter	Odds	Unstandardize	Standardized	P value
T didificter	Ratio	d coefficient	coefficient	
Smoking intensity wave 1 \rightarrow wave 2	4.35	0.865	0.603	< 0.001
OC wave 1 \rightarrow Smoking intensity wave 2				
OC tertile 2 vs tertile 1	1.05	0.027	0.011	0.774
OC tertile 3 vs tertile 1	1.14	0.072	0.028	0.455
OR change by 1 tertile	1.07	0.035	0.023	0.462
$OC \rightarrow ERI \rightarrow Smoking intensity$	1.03	0.017	0.010	0.117
OC wave 1 \rightarrow ERI wave 1				
OC tertile 2 vs tertile 1	1.41	0.190	0.109	< 0.001
OC tertile 3 vs tertile 1	2.43	0.522	0.292	< 0.001
OR change by 1 tertile	1.57	0.258	0.260	< 0.001
ERI wave 1 \rightarrow Smoking intensity wave 2				
ERI tertile 2 vs tertile 1	1.07	0.038	0.014	0.774
ERI tertile 3 vs tertile 1	1.23	0.113	0.042	0.297
OR change by 1 tertile	1.12	0.064	0.040	0.276
$OC \rightarrow PC \rightarrow Smoking intensity$	1.03	0.014	0.009	0.040
OC wave 1 \rightarrow PC wave 1				
OC tertile 2 vs tertile 1	0.94	- 0.035	- 0.020	0.601
OC tertile 3 vs tertile 1	0.65	- 0.247	- 0.140	< 0.001
OR change by 1 tertile	0.81	- 0.118	- 0.120	0.001
PC wave 1 \rightarrow Smoking intensity wave 2				
PC tertile 2 vs tertile 1	0.75	- 0.159	- 0.057	0.073
PC tertile 3 vs tertile 1	0.66	- 0.241	- 0.084	0.029
OR change by 1 tertile	0.80	- 0.121	- 0.077	0.018
ERI correlates with PC		- 0.062	- 0.103	0.002
Tests of model fit	RMSEA=	0.058	CFI = 0.929	TLI = 0.815

Figure 6.10 Results of path analysis for mediator roles of PC and ERI in OC-



smoking intensity relationship in current smokers in women

6.3.3 Modifying role of PC in ERI–smoking relationship

The modifying role of PC in ERI–smoking relationship is evaluated following the same steps as for the modifying role of OC in ERI–smoking relationship.

Table 6.18 shows the results of evaluation for modifying effect of PC in ERIsmoking relationship. Binary and ordinal logistic regression analyses are conducted for two smoking outcomes at wave 2, respectively, regressed by PC tertile, ER-ratio tertile, and interaction term between PC tertile and ER-ratio tertile at wave 1 after adjustment for covariates. LR tests reveal that this interaction term is not significant for smoking status in men (p= 0.363) and women (p= 0.780); this interaction term is not significant for smoking intensity in men (p= 0.221) and women (p= 0.961). My results show that PC has no significantly modifying role in ERI-smoking relationship.

Approaches	Strata	Smoking status: current	Smoking intensity among	
Approaches	Oliala	smokers vs non-smokers	current smokers	
Men		OR of outcome by 1-tertile increase in ERI (95% CI)		
1. ERI–smoking relation	PC tertile 1	1.04 (0.90 – 1.19)	0.97 (0.78 – 1.20)	
in different strata of PC	PC tertile 2	1.23 (1.04 – 1.47)	1.01 (0.77 – 1.31)	
tertile	PC tertile 3	1.25 (1.04 – 1.52)	1.18 (0.88 – 1.57)	
2. Interaction PC x ERI				
Likelihood-ratio test	P-value	P= 0.363	P= 0.221	
Women		OR of outcome by 1-tertile	increase in ERI (95% CI)	
1. ERI–smoking relation	PC tertile 1	1.17 (0.96 – 1.44)	1.18 (0.91 – 1.52)	
in different strata of PC	PC tertile 2	1.28 (1.02 – 1.61)	1.16 (0.85 – 1.59)	
tertile	PC tertile 3	1.09 (0.86 – 1.39)	1.20 (0.84 – 1.71)	
2. Interaction PC x ERI				
Likelihood-ratio test	P-value	P= 0.780	P= 0.961	

Table 6.18 Evaluation for modifying role of PC in ERI–smoking relationship

6.4 Main Findings for Smoking Outcomes

The analyses based on a 2–wave cohort study (3782 men and 3731 women aged 45–69) from the HAPIEE study show the following findings, which are in line with specific objectives and relevant hypotheses listed in Chapter 3.

In terms of the associations between ER ratio and smoking outcomes, Hypothesis 2 that higher ER ratio (wave 1) is associated with higher levels of smoking outcomes (wave 2) after adjustment for covariates is supported. In terms of the associations between OC and smoking outcomes, Hypothesis 5 that higher OC (wave 1) is associated with higher levels of smoking outcomes (wave 2) after adjustment for covariates is partially supported in smoking status, but the associations between OC and smoking intensity do not reach statistical significance in both sexes.

With regards to the potential role of OC (antecedent, mediator, modifier, or direct effect) in ERI–smoking relation, Hypothesis 8 is partially supported. OC and ERI have bi–directional relationship, but the effect of OC on ERI is stronger than the other direction in the middle-aged and older populations. Antecedent role of OC in ERI– smoking relationship is found significant among two smoking outcomes in both sexes (except smoking intensity in women), but mediator role of OC is not significant. Direct effect of OC on smoking is not significant. Finally, modifying role of OC in ERI–smoking relationship is non–significant.

In terms of the associations between PC and smoking outcomes, Hypothesis 11 that lower PC (wave 1) is associated with higher levels of smoking (wave 2) after adjustment for covariates is supported.

With regards to the potential role of PC (mediator or modifier) in the relationship between ERI, OC, and smoking, Hypothesis 14 is partially supported. PC and ERI partially mediate the effects of OC on all smoking outcomes in both sexes (except smoking intensity in women). In addition, PC and ERI may have bi–directional relationship. PC and ERI are negatively associated with each other in the cross– sectional analyses; bi–directional relationship between PC and ERI is possible, but causal directionality cannot be established in the cross–sectional analyses. Finally, modifying role of PC in ERI–smoking relationship is non–significant.

Note that the methodological issues and interpretation of the main findings for smoking outcomes will be addressed in detail in Chapter 8, and their implications for practice, policy, and research in Chapter 9.

Chapter 7. Dietary Outcomes

The aims of analyses for dietary outcomes are: (1) to examine the relationship between ERI, OC, and dietary outcomes; (2) to evaluate the potential role of PC in the relationship between ERI, OC and dietary outcomes. The analyses only use data from the cross–sectional study (wave 1) of the HAPIEE study, as PC and dietary outcomes were not collected at wave 2. The subsample consists of 11012 subjects (5735 men / 5277 women). Antecedent role of OC in ERI–diet relation was specified according to previous findings in drinking and smoking outcomes, as bidirectional relationship between OC and ERI (Hypothesis 9) cannot be tested in a cross–sectional study.

The results are presented in three parts. First, descriptive statistics for covariates and dietary outcomes by country and by gender are presented. Second, for assessing the associations between ERI, OC and dietary outcomes, the associations of exposure variables (ERI and OC) and dietary outcomes (HDI components and HDI) are assessed using binary and ordinal logistic regression, respectively. Third, to examine the potential role of PC in the relationship between ERI, OC and dietary outcomes, the path analysis with an ordinal categorical outcome (HDI outcome) is used.

7.1 Descriptive Statistics

7.1.1 Descriptive characteristics of study populations

In this sample of 11012 subjects (5735 men and 5277 women), the mean age is 55.0 years in men (standard deviation= 6.0) and 53.0 years in women (standard deviation= 5.3). Descriptive statistics for covariates are presented by country and gender in Table 7.1. First, gender differences are found across three countries; men have higher proportions in age group over 55, university–educated, manager and manual worker, married and cohabiting, lower deprivation, less depression, and more social isolation than women. Second, country differences are reported in both sexes.

Russian samples are older than Czech and Polish samples. University education in Czech samples is less common than other countries, and Czech samples have the largest gender inequality in education. The proportions of social isolation in Czech Republic are the lowest among all countries The proportions of high deprivation and poor to very poor self-rated health in Russia are the highest of all countries.

	Czech Repu	blic	Russia		Poland	
Variable	Men	Women	Men	Women	Men	Women
	(n= 1645)	(n= 1560)	(n= 2297)	(n= 2121)	(n= 1793)	(n= 1596)
Age, N (%)						
45 – 49	446 (27.1)	558 (35.8)	489 (21.3)	657 (31.0)	500 (27.9)	600 (37.6)
50 – 54	498 (30.3)	602 (38.6)	583 (25.4)	689 (32.5)	529 (29.5)	539 (33.8)
55 – 59	454 (27.6)	275 (17.6)	616 (26.8)	469 (22.1)	422 (23.5)	303 (19.0)
60 – 69	247 (15.0)	125 (8.0)	609 (26.5)	306 (14.4)	342 (19.1)	154 (9.6)
Education, N (%)						
Primary or less	54 (3.3)	137 (8.8)	158 (6.9)	83 (3.9)	70 (3.9)	73 (4.6)
Vocational	654 (39.7)	416 (26.7)	508 (22.1)	702 (33.1)	400 (22.3)	174 (10.9)
Secondary	549 (33.4)	769 (49.3)	797 (34.7)	619 (29.2)	554 (30.9)	669 (41.9)
University	388 (23.6)	237 (15.2)	836 (36.4)	717 (33.8)	769 (42.9)	680 (42.6)
Occupation class, N (%)						
Manager/profession	444 (27.0)	268 (17.2)	611 (26.6)	424 (20.0)	545 (30.4)	314 (19.7)
Non-manual worker	668 (40.6)	1011 (64.8)	781 (34.0)	1277 (60.2)	862 (48.1)	1071 (67.1)
Manual worker	533 (32.4)	281 (18.0)	905 (39.4)	420 (19.8)	386 (21.5)	211 (13.2)
Marital status, N (%)						
Married/cohabiting	1383 (84.1)	1129 (72.4)	2030 (88.4)	1332 (62.8)	1631 (91.0)	1116 (69.9)
Single	52 (3.2)	44 (2.8)	74 (3.2)	123 (5.8)	65 (3.6)	142 (8.9)
Divorce/widowed	209 (12.7)	387 (24.8)	193 (8.4)	666 (31.4)	97 (5.4)	337 (21.1)
Deprivation, N (%)						
Low (0 – 3.9)	1423 (86.5)	1279 (82.0)	1629 (70.9)	1126 (53.1)	1449 (80.8)	1171 (73.4)
High (4 – 9)	222 (13.5)	281 (18.0)	668 (29.1)	995 (46.9)	344 (19.2)	425 (26.6)
Depression, N (%)						
CESD < 16	1439 (87.5)	1237 (79.3)	1996 (86.9)	1525 (71.9)	1540 (85.9)	1285 (80.5)
CESD >= 16	206 (12.5)	323 (20.7)	301 (13.1)	596 (28.1)	253 (14.1)	311 (19.5)
Social isolation, N (%)						
No (>= once a month)	1076 (65.4)	1114 (71.4)	1015 (44.2)	982 (46.3)	809 (45.1)	776 (48.6)
Yes (< once a month)	569 (34.6)	446 (28.6)	1282 (55.8)	1139 (53.7)	984 (54.9)	820 (51.4)
Self-rated health, N (%)						
Very good – average	1538 (93.5)	1476 (94.6)	2099 (91.4)	1731 (81.6)	1678 (93.6)	1495 (93.7)
Poor – very poor	107 (6.5)	84 (5.4)	198 (8.6)	390 (18.4)	115 (6.4)	101 (6.3)

Table 7.1 Descriptive statistics of study sample by country and gender (N= 11012)

7.1.2 Descriptive characteristics of dietary outcomes

Dietary data are collected using the Food Frequency Questionnaire (FFQ) developed by Willett et al and adapted from the Whitehall II Study.⁵⁸⁹ Table 7.2 shows descriptive characteristics of dietary outcomes by country and by gender. Absolute nutrient/food intakes (quantity per day) are presented by means and medians; medians are shown as the distributions are often skewed. Nutrient density is presented by percentage of total energy intakes without energy provided by alcohol. Table 7.2 Descriptive statistics of dietary outcomes by country and gender

	Czech		Russia		Poland	
Dietary outcomes	Men	Women	Men	Women	Men	Women
	(n= 1645)	(n= 1560)	(n= 2297)	(n= 2121)	(n= 1793)	(n= 1596)
Saturated fat						
Mean (sd), g/day	32 (13)	29 (13)	48 (20)	40 (16)	40 (16)	35 (14)
Median, g/day	30	27	45	38	37	33
Nutrient density (%)	13.2	13.0	14.8	14.5	15.3	14.6
Polyunsaturated fat						
Mean (sd), g/day	15 (7)	14 (6)	26 (10)	25 (10)	13 (6)	12 (5)
Median, g/day	14	13	24	24	12	11
Nutrient density (%)	6.2	6.3	8.0	9.1	5.0	5.0
Total carbohydrate						
Mean (sd), g/day	240 (96)	238 (96)	287 (85)	253 (82)	267 (86)	262 (87)
Median, g/day	223	222	279	243	257	248
Nutrient density (%)	43.7	47.8	39.8	40.8	45.5	48.7
Free sugars						
Mean (sd), g/day	110 (57)	128 (67)	126 (49)	125 (48)	124 (54)	133 (59)
Median, g/day	101	115	118	118	116	123
Nutrient density (%)	20.2	25.5	17.3	20.1	21.0	24.7
Protein						
Mean (sd), g/day	96 (35)	87 (30)	125 (38)	107 (33)	106 (32)	95 (29)
Median, g/day	91	83	121	103	101	92
Nutrient density (%)	17.6	17.4	17.1	17.2	18.0	17.7
Cholesterol						
Mean (sd), mg/day	326 (141)	283 (123)	544 (253)	413 (165)	424 (195)	357 (144)
Median, mg/day	304	266	492	391	390	337
Sodium						
Mean (sd), mg/day	3013 (1123)	2562 (1022)	4020 (1358)	3379 (1171)	3756 (1306)	3256 (1155
Median, mg/day	2854	2416	3847	3262	3570	3078
Fruit and vegetable						
Mean (sd), g/day	452 (396)	678 (582)	379 (255)	450 (305)	456 (267)	559 (347)
Median, g/day	369	537	314	369	408	492
Non-starch polysaccharide						
Mean (sd), g/day	17 (9)	19 (10)	18 (6)	18 (6)	19 (7)	19 (8)
Median, g/day	15	17	18	17	18	18
Total energy intake						
Mean (sd), MJ/day	9.1 (3.1)	8.4 (3.0)	12.2 (3.6)	10.4 (3.2)	9.9 (3.0)	9.0 (2.7)
Median, MJ/day	8.7	7.9	11.7	10.0	9.5	8.6

The Healthy Diet Indicator (HDI) is constructed to reflect the adherence to dietary recommendations of WHO for the prevention of chronic diseases (2003).⁵⁹⁰ Nine selected nutrient/food intakes are: (1) nutrient density from saturated fat, polyunsaturated fat, total carbohydrate, free sugars, and protein; (2) nutrient intakes of non–starch polysaccharides (NSP), cholesterol, and sodium; (3) food intakes of fruit and vegetable. A dichotomous variable is generated for each nutrient/food intake; if one's intake is within the WHO recommended range this variable is coded as 1 (healthy intake), otherwise it is coded as 0 (unhealthy intake). The HDI score is the sum of 9 dichotomous variables, ranged from 0 to 9.⁵⁹¹

Table 7.3 reports the percentage of subjects who meet the WHO dietary recommendations. Across three countries, the proportions of women who meet the WHO recommended ranges are generally higher than those of men, except polyunsaturated fat (in Russia and Poland), free sugars, and NSP (in Russia).

Table 7.3Percentage of subjects who meet dietary recommendations of WHO forthe prevention of chronic diseases (2003)

Proportion meeting	WHO	Czech Rep	oublic	Russia		Poland	
WHO suggested	suggested	Men	Women	Men	Women	Men	Women
ranges (%)	ranges	(n= 1645)	(n= 1560)	(n= 2297)	(n= 2121)	(n= 1793)	(n= 1596)
Saturated fat	< 10 %	5.6	9.7	2.8	3.3	2.0	4.9
Polyunsaturated fat	6–10 %	56.7	58.8	70.8	62.2	17.2	15.8
Total carbohydrate	55–75 %	5.4	16.0	1.1	2.1	7.3	17.2
Free sugars	< 10 %	4.3	1.0	5.2	1.9	2.8	1.2
Protein	10–15 %	9.0	13.9	11.8	19.3	6.5	9.6
Cholesterol	< 300 mg	48.6	63.1	10.9	24.3	21.7	35.8
Sodium	< 2000 mg	19.0	36.2	4.4	12.8	5.5	12.6
Fruit & vegetable	> 400 g	55.8	75.1	51.7	64.6	65.2	75.4
NSP	> 20 g	19.1	28.8	22.2	18.7	27.4	31.5

Table 7.4 shows the means and medians of overall HDI score and the proportions of subjects meeting different HDI scores by gender and by country. There are gender differences across three countries; mean HDI scores in women are higher than those in men. Country differences are observed in both sexes; mean HDI scores in Czech Republic are the highest of all countries, but mean HDI scores in Poland are the lowest of all countries. Due to very low proportions of those having HDI 5 to 9, six categories of HDI outcome (HDI= 0, 1, 2, 3, 4, and 5–9) are adopted for further analyses. Table 7.4 Overall scores of Healthy Diet Indicator and proportions of subjects in

different scores

Healthy Diet	Czech Repul	olic	Russia		Poland	
Indicator	Men	Women	Men	Women	Men	Women
(HDI)	(n= 1645)	(n= 1560)	(n= 2297)	(n= 2121)	(n= 1793)	(n= 1596)
Different score	N (%)		N (%)		N (%)	
0	100 (6.1)	34 (2.2)	221 (9.6)	121 (5.7)	298 (16.6)	145 (9.1)
1	357 (21.7)	136 (8.7)	733 (31.9)	534 (25.2)	609 (34.0)	405 (25.4)
2	551 (33.5)	388 (24.9)	744 (32.4)	774 (36.5)	579 (32.3)	557 (34.9)
3	423 (25.7)	516 (33.1)	487 (21.2)	473 (22.3)	217 (12.1)	297 (18.6)
4	155 (9.4)	306 (19.6)	87 (3.8)	176 (8.3)	63 (3.5)	118 (7.4)
5–9	59 (3.6)	179 (11.5)	25 (1.1)	42 (2.0)	27 (1.5)	74 (4.6)
Overall score						
Mean (sd)	2.2 (1.2)	3.0 (1.3)	1.8 (1.1)	2.1 (1.1)	1.6 (1.1)	2.0 (1.2)

In terms of the way of pooling the data, crude associations between exposure variables (ER ratio and OC) and HDI outcome in country–specific strata are assessed by ordinal logistic regression, respectively (Table 7.5). By comparing the log likelihoods for the model with the interaction term (ERI x country; OC x country) and the model without, likelihood–ratio (LR) test is used to test significance of the interaction term.

There are small country differences in crude associations between exposure variables and HDI outcome. The associations between ER ratio and HDI outcome in Czech Republic are slightly stronger than those in other countries; a similar pattern of country differences is observed in OC–HDI associations. Overall, these associations between exposure variables and HDI outcome are not very different across country–specific strata (all p–values > 0.128). The interaction terms between country and exposure variables do not reach statistical significance. Thus, data for three countries are pooled for further analyses.

Table 7.5 Crude associations between exposure variables and HDI outcome in

Strata	Tertile of exposure	HDI outcome	HDI outcome
	variables▲	OR (95% CI)	OR (95% CI)
ERI–HDI relation	ER-ratio tertile	Men	Women
Czech	Tertile 1	1.00	1.00
	Tertile 2	0.83 (0.66 – 1.04)	0.71 (0.57 – 0.89)*
	Tertile 3	0.65 (0.53 – 0.81)*	0.64 (0.51 – 0.79)*
	P for trend	< 0.001	< 0.001
	OR by 1 tertile	0.81 (0.73 – 0.90)*	0.80 (0.72 - 0.89)*
Russia	Tertile 1	1.00	1.00
	Tertile 2	0.80 (0.68 – 0.94)*	0.96 (0.81 – 1.14)
	Tertile 3	0.68 (0.56 – 0.82)*	0.75 (0.62 - 0.90)*
	P for trend	0.001	0.005
	OR by 1 tertile	0.83 (0.76 – 0.91)*	0.87 (0.80 - 0.96)*
Poland	Tertile 1	1.00	1.00
	Tertile 2	0.93 (0.76 – 1.15)	0.96 (0.77 – 1.19)
	Tertile 3	0.88 (0.72 – 1.08)	0.90 (0.74 – 1.10)
	P for trend	0.337	0.309
	OR by 1 tertile	0.94 (0.85 – 1.04)	0.94 (0.85 – 1.05)
Interaction			
country x ERI	LR test	P= 0.152	P= 0.135
OC-HDI relation	OC tertile	Men	Women
Czech	Tertile 1	1.00	1.00
	Tertile 2	0.87 (0.70 – 1.06)	0.76 (0.62 - 0.94)*
	Tertile 3	0.67 (0.54 – 0.84)*	0.64 (0.51 – 0.80)*
	P for trend	0.001	< 0.001
	OR by 1 tertile	0.83 (0.74 – 0.92)*	0.81 (0.72 – 0.91)*
Russia	Tertile 1	1.00	1.00
	Tertile 2	0.94 (0.77 – 1.14)	1.01 (0.84 – 1.22)
	Tertile 3	0.78 (0.66 – 0.94)*	0.91 (0.76 – 1.09)
	P for trend	0.108	0.360
	OR by 1 tertile	0.91 (0.81 – 1.02)	0.95 (0.87 – 1.04)
Poland	Tertile 1	1.00	1.00
	Tertile 2	0.98 (0.82 – 1.18)	0.86 (0.70 – 1.06)
	Tertile 3	0.85 (0.69 – 1.04)	0.85 (0.69 – 1.04)
	P for trend	0.150	0.143
	OR by 1 tertile	0.93 (0.84 – 1.03)	0.92 (0.83 – 1.03)
Interaction			
country x OC	LR test	P= 0.203	P= 0.128

country-specific strata

▲ Country– and gender–specific tertiles of ER ratio or OC score. * P value < 0.05.

After the data of three countries are pooled, crude associations between exposure variables (ER ratio and OC) and HDI outcome in gender–specific strata are assessed by ordinal logistic regression, respectively (Table 7.6). By comparing the log likelihoods for the model with the interaction term (ERI x gender; OC x gender) and the model without, LR test is used to test significance of the interaction term. The crude associations between exposure variables and HDI outcome are not very different across gender–specific strata (all p–values > 0.17). The interaction terms between gender and exposure variables do not reach statistical significance. Although men and women are initially assessed separately as most literature on psychosocial factors and diet, data from both sexes will be pooled in final analyses. Table 7.6 Crude associations between exposure variables and HDI outcome in gender–specific strata

Strata	Tertile of exposure	HDI outcome
	variables	OR (95% CI)
ERI–HDI relationship	ER-ratio tertile	
Men	Tertile 1	1.00
	Tertile 2	0.87 (0.78 – 0.97)*
	Tertile 3	0.80 (0.72 - 0.89)*
	P for trend	0.001
	OR by 1 tertile	0.90 (0.85 - 0.95)*
Women	Tertile 1	1.00
	Tertile 2	0.97 (0.86 – 1.08)
	Tertile 3	0.93 (0.83 – 1.04)
	P for trend	0.231
	OR by 1 tertile	0.96 (0.91 – 1.02)
Interaction term: gender x ERI	LR test	P= 0.176
OC-HDI relationship	OC tertile	
Men	Tertile 1	1.00
	Tertile 2	0.95 (0.85 – 1.05)
	Tertile 3	0.85 (0.76 – 0.95)*
	P for trend	0.007
	OR by 1 tertile	0.93 (0.88 – 0.98)*
Women	Tertile 1	1.00
	Tertile 2	1.03 (0.93 – 1.16)
	Tertile 3	0.90 (0.81 – 1.01)
	P for trend	0.132
	OR by 1 tertile	0.96 (0.90 – 1.01)
Interaction term: gender x OC	LR test	P= 0.253

▲ Gender–specific tertiles of ER ratio or OC score. * P value < 0.05.

The associations between covariates and HDI outcome by ordinal logistic regression are summarized (Table 7.7). In men, age 55–59 (OR= 1.12, P= 0.088) and age 60–69 (OR= 1.27, P= 0.001) are associated with higher HDI; in contrast, high deprivation (OR= 0.80, P < 0.001), depression (OR= 0.88, P= 0.084) and social isolation (OR= 0.85, P= 0.001) are associated with lower HDI. In women, age 50–54 (OR= 1.18, P= 0.004), age 55–59 (OR= 1.29, P < 0.001), age 60–69 (OR= 1.18, P= 0.074), and university-educated (OR= 1.29, P= 0.048) are associated with higher HDI. High deprivation (OR= 0.75, P < 0.001) and social isolation (OR= 0.79, P <

0.001) are associated with lower HDI.

Table 7.7 Associations between HDI outcome and covariates by ordinal logistic

roa	ression
180	ression
109	10001011

Variables	Men (n = 5735)		Women (n = 5277)	Women (n = 5277)	
	OR (95% CI)	P-value	OR (95% CI)	P-value	
Age					
45 – 49	1.00		1.00		
50 – 54	1.06 (0.93 – 1.21)	0.355	1.18 (1.05 – 1.34)	0.004	
55 – 59	1.12 (0.98 – 1.28)	0.088	1.29 (1.12 – 1.49)	< 0.001	
60 – 69	1.27 (1.10 – 1.47)	0.001	1.18 (0.98 – 1.41)	0.074	
Education					
Primary or less	1.00		1.00		
Vocational	1.17 (0.88 – 1.56)	0.271	1.15 (0.89 – 1.29)	0.289	
Secondary	1.10 (0.83 – 1.46)	0.488	1.08 (0.84 – 1.38)	0.534	
University	1.22 (0.90 – 1.64)	0.194	1.29 (1.00 – 1.66)	0.048	
Occupation class					
Manager/ professional	1.00		1.00		
Non-manual worker	0.91 (0.81 – 1.03)	0.134	0.96 (0.85 – 1.09)	0.564	
Manual worker	0.94 (0.82 – 1.07)	0.370	0.95 (0.80 – 1.13)	0.557	
Marital status					
Married/ cohabiting	1.00		1.00		
Single	1.12 (0.84 – 1.49)	0.437	1.06 (0.92 – 1.23)	0.388	
Divorced/ widowed	1.01 (0.85 – 1.20)	0.833	0.90 (0.80 – 1.04)	0.118	
Self-rated health					
Very good/ average	1.00		1.00		
Very poor/ poor	0.96 (0.79 – 1.17)	0.747	0.92 (0.78 – 1.09)	0.356	
Deprivation					
low (0 – 3.9)	1.00		1.00		
High (4 – 9)	0.80 (0.70 – 0.90)	< 0.001	0.75 (0.66 – 0.84)	< 0.001	
Depression					
No	1.00		1.00		
Yes	0.88 (0.76 – 1.02)	0.084	0.92 (0.82 – 1.04)	0.185	
Social isolation					
No	1.00		1.00		
Yes	0.85 (0.77 – 0.94)	0.001	0.79 (0.71 – 0.87)	< 0.001	

7.2 Associations between ERI, OC, and Dietary Outcomes

7.2.1 Associations between ERI and dietary outcomes

The associations between ER-ratio tertile and dietary outcomes are evaluated by the following two steps. First, binary logistic regression is used to assess the associations between ER-ratio tertile and 9 dichotomous variables (nutrient/food intakes), respectively, after adjustment for age and country (Model 1) and after additionally adjustment for other covariates (Model 2). Second, ordinal logistic regression is used to assess the associations between ER–ratio tertile and HDI as an ordinal categorical variable. Due to low proportions of subjects in HDI 5 to 9, six categories of HDI outcome (HDI= 0, 1, 2, 3, 4, and 5–9) are used. The ERI–HDI associations are assessed after adjustment for age and country (Model 1) and after additionally adjustment for other covariates (Model 2). Ordinal logistic regression assumes that the coefficient for relationship between, for example, the lowest versus all higher categories of outcome variable is the same as that coefficient for relationship between the next lowest category and all higher categories (parallel regression assumption); this assumption is found not violated by Brant test (p > 0.05).

In Table 7.8, binary logistic regression analyses for men indicate that higher ER ratio is significantly (p < 0.05) associated with less healthy intakes of saturated fat (OR changes by 1–tertile increase in ER ratio= 0.81), free sugars (OR= 0.80), protein (OR= 0.88), cholesterol (OR= 0.91), sodium (OR= 0.83), and fruit/vegetable (OR= 0.93) in Model 2. For women, higher ER ratio is significantly associated with less healthy intakes of saturated fat (OR= 0.80), total carbohydrate (OR= 0.89), free sugars (OR= 0.74), cholesterol (OR= 0.92), sodium (OR= 0.88), fruit/vegetable (OR= 0.92), and NSP (OR= 0.91) in Model 2.

At the bottom of Table 7.8, ordinal logistic regression analyses demonstrate a consistent and significant (p < 0.001) association between higher ER ratio and lower HDI score. For 1–tertile increase in ER ratio, the adjusted odds of being in a higher HDI category (e.g. HDI > 0 versus <= 0, HDI > 1 versus <= 1, or HDI > 4 versus <= 4) are changed by 0.86 and 0.88 in men and women, respectively. For highest versus lowest tertile of ER ratio, the adjusted odds of being in a higher HDI category are 0.73 and 0.78 in men and women, respectively.

		Men (n= 5735)		Women (n= 5277)	
Outcome	ER ratio	Model 1 †	Model 2 ‡	Model 1 †	Model 2 ‡
Variables	tertile **	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
Saturated fat	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.75 (0.53–1.05)	0.74 (0.51–1.08)	0.75 (0.57–0.98)*	0.77 (0.58–1.02)
	Tertile 3	0.67 (0.47–0.96)*	0.66 (0.44–0.98)*	0.64 (0.49–0.88)*	0.63 (0.48–0.87)*
	OR by 1 tertile	0.82 (0.68–0.98)*	0.81 (0.66–0.99)*	0.80 (0.69–0.92)*	0.80 (0.69–0.93)*
	P for trend	0.029	0.039	0.002	0.003
Poly-	Tertile 1	1.00	1.00	1.00	1.00
unsaturated	Tertile 2	1.00 (0.87–1.15)	0.97 (0.83–1.13)	1.03 (0.89–1.19)	1.01 (0.86–1.20)
fat	Tertile 3	0.94 (0.81–1.08)	0.90 (0.77–1.06)	0.99 (0.85–1.15)	1.00 (0.84–1.19)
	OR by 1 tertile	0.97 (0.90–1.04)	0.95 (0.88–1.03)	1.01 (0.94–0.18)	1.00 (0.92–1.09)
	P for trend	0.382	0.207	0.699	0.881
Total carbo-	Tertile 1	1.00	1.00	1.00	1.00
hydrate	Tertile 2	0.90 (0.65–1.25)	0.92 (0.65–1.29)	0.99 (0.79–1.23)	0.98 (0.77-1.24)
	Tertile 3	0.88 (0.64–1.22)	0.90 (0.64–1.27)	0.83 (0.67–1.03)	0.81 (0.64–1.02)
	OR by 1 tertile	0.94 (0.81–1.11)	0.95 (0.80–1.13)	0.91 (0.82–1.01)	0.89 (0.79–1.00)*
	P for trend	0.469	0.584	0.077	0.048
Free sugars	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.71 (0.52–0.98)*	0.67 (0.48–0.93)*	0.72 (0.43–1.20)	0.74 (0.41–1.26)
	Tertile 3	0.69 (0.51–0.93)*	0.65 (0.46-0.92)*	0.54 (0.30-0.96)*	0.53 (0.28-1.00)*
	OR by 1 tertile	0.83 (0.71–0.98)*	0.80 (0.67–0.96)*	0.74 (0.56–0.98)*	0.74 (0.54–0.98)*
	P for trend	0.030	0.014	0.032	0.030
Protein	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.86 (0.70–1.06)	0.87 (0.70–1.10)	0.90 (0.75–1.07)	0.91 (0.75–1.10)
	Tertile 3	0.81 (0.65–1.01)	0.78 (0.61–0.99)*	0.88 (0.73–1.05)	0.84 (0.69–1.03)
	OR by 1 tertile	0.90 (0.81–1.01)	0.88 (0.78–0.99)*	0.93 (0.85–1.02)	0.92 (0.83–1.01)
	P for trend	0.062	0.044	0.148	0.108
Cholesterol	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.88 (0.75–1.03)	0.84 (0.71–0.99)*	0.93 (0.81–1.07)	0.89 (0.77–1.04)
	Tertile 3	0.87 (0.74–1.02)	0.82 (0.69–0.99)*	0.91 (0.79–1.04)	0.86 (0.74–1.00)*
	OR by 1 tertile	0.94 (0.87–1.02)	0.91 (0.84–0.99)*	0.95 (0.88–1.02)	0.92 (0.85–1.00)*
	P for trend	0.125	0.041	0.225	0.050
Sodium	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.75 (0.60–0.94)*	0.75 (0.59–0.96)*	0.91 (0.77–1.07)	0.90 (0.75–1.07)
	Tertile 3	0.71 (0.56–0.89)*	0.69 (0.54–0.88)*	0.84 (0.71–1.00)*	0.78 (0.64–0.94)*
	OR by 1 tertile	0.84 (0.75–0.94)*	0.83 (0.73–0.94)*	0.92 (0.84–1.00)*	0.88 (0.80–0.97)*
	P for trend	0.003	0.003	0.049	0.008
Fruit and	Tertile 1	1.00	1.00	1.00	1.00
vegetable	Tertile 2	0.99 (0.88–1.12)	0.96 (0.84–1.11)	0.95 (0.83–1.09)	0.96 (0.82–1.12)
	Tertile 3	0.86 (0.76–0.98)*	0.87 (0.75–1.00)*	0.84 (0.73–0.97)*	0.85 (0.72–1.00)
	OR by 1 tertile	0.93 (0.87–0.99)*	0.93 (0.86–1.00)*	0.92 (0.85–0.98)*	0.92 (0.85–1.00)*
	P for trend	0.029	0.050	0.016	0.048
NSP	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.91 (0.78–1.05)	0.90 (0.76–1.07)	0.94 (0.82–1.09)	0.94 (0.80–1.11)
	Tertile 3	0.87 (0.75–1.02)	0.89 (0.76–1.05)	0.81 (0.70–0.94)*	0.83 (0.70–0.98)*
	OR by 1 tertile	0.93 (0.87–1.01)	0.95 (0.87–1.03)	0.90 (0.84–0.97)*	0.91 (0.84–0.99)*
	P for trend	0.082	0.233	0.006	0.027
HDI score	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.84 (0.76–0.94)*	0.81 (0.72–0.92)*	0.91 (0.81–1.02)	0.91 (0.81–1.04)
	Tertile 3	0.75 (0.67–0.84)*	0.73 (0.65–0.83)*	0.80 (0.71–0.90)*	0.78 (0.69–0.90)*
	OR by 1 tertile	0.87 (0.81–0.92)*	0.86 (0.80–0.91)*	0.89 (0.84–0.95)*	0.88 (0.83–0.94)*
	P for trend	< 0.001	< 0.001	0.001	< 0.001

 Table 7.8
 Associations between ER ratio and dietary outcomes

† Model 1: adjusted for age and country.

‡ Model 2: additionally adjusted for other covariates such as education, occupation, marital status, deprivation, depression, social isolation, and self-rated health.

* P value < 0.05. ** Gender-specific tertile of ER ratio: in men, tertile 1 (0.20–0.32), tertile 2 (0.32–0.47), and tertile 3 (>= 0.47); in women, tertile 1 (0.20–0.31), tertile 2 (0.31–0.46), and tertile 3 (>= 0.46).

7.2.2 Associations between OC and dietary outcomes

The associations between OC and dietary outcomes are assessed following the same two steps as for ERI–diet associations.

In Table 7.9, binary logistic regression analyses for men indicate that higher OC is significantly (p < 0.05) associated with less healthy intakes of saturated fat (OR changes by 1–tertile increase in OC score= 0.78), polyunsaturated fat (OR= 0.91), free sugars (OR= 0.81), protein (OR= 0.89), and fruit/vegetable (OR= 0.92) in Model 2. For women, higher OC is significantly (p < 0.05) associated with less healthy intakes of saturated fat (OR= 0.86), polyunsaturated fat (OR= 0.91), protein (OR= 0.89), and NSP (OR= 0.90) in Model 2. The associations between high OC and less healthy intakes of sodium (OR= 0.92) and fruit/vegetable (OR= 0.93) reach marginal significance (p < 0.1).

At the bottom of Table 7.9, ordinal logistic regression analyses show a consistent and significant (p= 0.001) association between higher OC and lower HDI score. For 1-tertile increase in OC score, the adjusted odds of being in a higher HDI category are changed by 0.90 in both men and women. For highest versus lowest tertile of OC score, the adjusted odds of being in a higher HDI category are 0.81 and 0.80 in men and women, respectively.
		Men (n= 5735)		Women (n= 5277)	
Outcome Variables	OC score tertile **	Model 1 † OR (95% CI)	Model 2 ‡ OR (95% CI)	Model 1 † OR (95% CI)	Model 2 ‡ OR (95% CI)
Saturated fat	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.83 (0.60–1.16)	0.82 (0.57–1.17)	0.80 (0.62–1.04)	0.85 (0.64–1.11)
	Tertile 3	0.57 (0.39–0.84)*	0.60 (0.40–0.90)*	0.72 (0.54–0.96)*	0.75 (0.55–1.00)*
	OR by 1 tertile	0.76 (0.63–0.92)*	0.78 (0.63–0.96)*	0.85 (0.73–0.97)*	0.86 (0.73–1.00)*
	P for trend	0.006	0.018	0.019	0.045
Poly-	Tertile 1	1.00	1.00	1.00	1.00
unsaturated	Tertile 2	1.01 (0.88–1.16)	1.00 (0.87–1.16)	0.97 (0.84–1.11)	0.94 (0.81–1.09)
fat	Tertile 3	0.85 (0.74–0.99)*	0.82 (0.70–0.97)*	0.83 (0.72–0.95)*	0.82 (0.70–0.96)*
lat	OR by 1 tertile	0.93 (0.87–1.00)	0.91 (0.84–0.99)*	0.91 (0.85–0.98)*	0.91 (0.84–0.98)*
	P for trend	0.061	0.026	0.011	0.017
Total carbo-	Tertile 1	1.00	1.00	1.00	1.00
hydrates	Tertile 2				
nyulales	Tertile 3	0.83 (0.62–1.11)	0.83 (0.61–1.12)	0.94 (0.76–1.15)	0.96 (0.77–1.19) 0.90 (0.71–1.14)
		0.82 (0.60–1.13)	0.82 (0.58–1.16)	0.90 (0.72–1.13)	,
	OR by 1 tertile	0.90 (0.77–1.06)	0.90 (0.76–1.07)	0.95 (0.85–1.06)	0.95 (0.84–1.07)
F	P for trend	0.212	0.245	0.372	0.382
Free sugars	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.92 (0.69–1.22)	0.87 (0.64–1.18)	0.65 (0.37–1.13)	0.67 (0.35–1.24)
	Tertile 3	0.68 (0.48–0.96)*	0.65 (0.44–0.94)*	0.61 (0.35–1.05)	0.60 (0.32–1.10)
	OR by 1 tertile	0.84 (0.71–0.99)*	0.81 (0.68–0.98)*	0.78 (0.59–1.03)	0.79 (0.58–1.06)
	P for trend	0.033	0.026	0.084	0.106
Protein	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.80 (0.65–0.98)*	0.85 (0.67–1.07)	0.79 (0.66–0.94)*	0.84 (0.69–1.03)
	Tertile 3	0.80 (0.64–0.99)*	0.82 (0.65–1.03)	0.77 (0.64–0.92)*	0.79 (0.65–0.96)*
	OR by 1 tertile	0.88 (0.79–0.99)*	0.89 (0.79–1.00)*	0.87 (0.79–0.95)*	0.89 (0.80–0.99)*
	P for trend	0.032	0.048	0.003	0.045
Cholesterol	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	1.06 (0.91–1.23)	1.04 (0.89–1.22)	0.97 (0.84–1.11)	0.95 (0.82–1.10)
	Tertile 3	0.99 (0.84–1.16)	0.93 (0.79–1.11)	1.05 (0.91–1.21)	1.02 (0.87–1.19)
	OR by 1 tertile	1.00 (0.92–1.08)	0.97 (0.89–1.05)	1.02 (0.95–1.09)	1.01 (0.93–1.09)
	P for trend	0.970	0.499	0.511	0.840
Sodium	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	1.09 (0.88–1.34)	1.13 (0.90–1.41)	0.91 (0.77–1.06)	0.86 (0.72–1.02)
	Tertile 3	0.89 (0.70–1.12)	0.83 (0.64–1.07)	0.93 (0.78–1.09)	0.85 (0.70–1.03)
	OR by 1 tertile	0.95 (0.85–1.06)	0.92 (0.82–1.04)	0.96 (0.88–1.05)	0.92 (0.83–1.01)
	P for trend	0.385	0.209	0.423	0.091
Fruit and	Tertile 1	1.00	1.00	1.00	1.00
vegetable	Tertile 2	0.88 (0.77–1.00)	0.89 (0.77–1.02)	1.00 (0.87–1.15)	1.01 (0.87–1.18)
	Tertile 3	0.81 (0.71–0.91)*	0.81 (0.71–0.92)*	0.85 (0.74–0.98)*	0.86 (0.73-1.00)*
	OR by 1 tertile	0.92 (0.87-0.99)*	0.92 (0.86–1.00)*	0.93 (0.86–0.99)*	0.93 (0.86–1.01)
	P for trend	0.018	0.048	0.036	0.085
NSP	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.90 (0.78–1.03)	0.93 (0.79–1.10)	0.87 (0.75–1.01)	0.93 (0.80–1.08)
	Tertile 3	0.88 (0.75–1.03)	0.89 (0.76–1.04)	0.78 (0.67–0.91)*	0.81 (0.68–0.95)*
	OR by 1 tertile	0.93 (0.87–1.01)	0.96 (0.88–1.04)	0.88 (0.82–0.96)*	0.90 (0.83–0.98)*
	P for trend	0.084	0.330	0.002	0.013
HDI score	Tertile 1	1.00	1.00	1.00	1.00
···	Tertile 2	0.89 (0.80–0.99)*	0.88 (0.79–0.99)*	0.89 (0.80–1.00)*	0.91 (0.81–1.03)
	Tertile 3	0.81 (0.72–0.91)*	0.81 (0.71–0.92)*	0.82 (0.73–0.92)*	0.80 (0.70–0.91)*
	OR by 1 tertile	0.90 (0.85–0.95)*	0.90 (0.84–0.96)*	0.91 (0.86–0.97)*	0.90 (0.84–0.96)*
	P for trend	0.001	0.001	0.008	0.001

 Table 7.9
 Associations between OC and dietary outcomes

† Model 1: adjusted for age and country.

‡ Model 2: additionally adjusted for other covariates such as education, occupation, marital status, deprivation, depression, social isolation, and self-rated health.
* P value < 0.05. ** Gender- specific tertile of OC score in men and women: tertile 1 (6–11), tertile 2 (11–15), tertile 3 (15–24)

7.2.3 Modifying role of OC in ERI-diet relationship

To evaluate modifying role of OC in ERI-diet relationship, the associations between ER ratio and HDI outcome in different strata of OC tertile are assessed after adjustment for age, country and other covariates. Next, ordinal logistic regression is conducted for HDI outcome regressed by OC tertile, ER-ratio tertile, and interaction term between OC tertile and ER-ratio tertile after adjustment for covariates. By comparing the log likelihoods for the model with this interaction term and the model without, LR test is adopted to test the significance of this interaction term.

Table 7.10 shows the results for modifying effect of OC in ERI–diet relationship. LR tests show that the interaction term between OC tertile and ER–ratio tertile does not reach significance in men (p= 0.140) and women (p= 0.146), respectively. OC has no significantly modifying role in the relationship between ERI and diet.

Approaches	Strata	Healthy Diet Indicator (HDI) outcome
Men		OR of outcome by 1-tertile increase in ERI (95% CI)
1. ERI-diet relationship in	OC tertile 1	0.85 (0.76 – 0.93)
different strata of OC tertile	OC tertile 2	0.94 (0.84 – 1.04)
	OC tertile 3	0.97 (0.84 – 1.11)
2. Interaction term OC x ERI	P-value	P= 0.140
Women		OR of outcome by 1-tertile increase in ERI (95% CI)
1. ERI-diet relationship in	OC tertile 1	0.89 (0.80 – 1.00)
different strata of OC tertile	OC tertile 2	0.96 (0.86 – 1.08)
	OC tertile 3	1.06 (0.93 – 1.22)
2. Interaction term OC x ERI	P-value	P= 0.146

Table 7.10 Evaluation for modifying role of OC in ERI-diet relationship

7.3 Potential Role of PC in Relation between OC, ERI, and Diet

7.3.1 Associations between PC and dietary outcomes

The associations between PC and dietary outcomes are assessed following the same two steps as for ERI–diet associations.

In Table 7.11, binary logistic regression analyses for men indicate that higher PC is significantly (p < 0.05) associated with more healthy intakes of saturated fat (OR changes by 1–tertile increase in PC= 1.29), fruit/vegetable (OR= 1.17), and NSP (OR= 1.17) in Model 2; in addition, the associations of PC with total carbohydrate (OR= 1.15) and sodium (OR= 1.10) reach marginal significance (p < 0.1). For women, higher PC is significantly (p < 0.05) associated with more healthy intakes of saturated fat (OR changes by 1–tertile increase in PC= 1.32), total carbohydrate (OR= 1.16), fruit/vegetable (OR= 1.13), and NSP (OR= 1.10); the association of PC with cholesterol reaches marginal significance (p < 0.1).

At the bottom of Table 7.11, ordinal logistic regression analyses show a consistent and significant (p= 0.001) association between higher PC and higher levels of HDI outcome. For 1–tertile increase in PC, the adjusted odds of being in a higher HDI category are changed by 1.12 in both men and women. For highest verus lowest tertile of PC, the adjusted odds of being in a higher HDI category are 1.26 in both men and women, respectively.

	Perceived	Men (n= 5735)		Women (n= 5277)	
Outcome	Control	Model 1 †	Model 2 ‡	Model 1 †	Model 2 ‡
Variables	tertile **	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
Saturated fat	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	1.11 (0.77–1.59)	1.18 (0.79–1.67)	1.18 (0.90–1.56)	1.29 (0.95–1.74)
	Tertile 3	1.58 (1.12–2.21)*	1.67 (1.12–2.32)*	1.56 (1.18–2.05)*	1.74 (1.25–2.39)*
	OR by 1 tertile	1.27 (1.06–1.49)*	1.29 (1.06–1.52)*	1.25 (1.09–1.44)*	1.32 (1.13–1.54)*
	P for trend	0.008	0.007	0.002	< 0.001
Poly-	Tertile 1	1.00	1.00	1.00	1.00
unsaturated	Tertile 2	0.98 (0.86–1.12)	0.95 (0.82–1.10)	0.98 (0.86–1.13)	0.97 (0.83–1.13)
fat	Tertile 3	1.01 (0.88–1.15)	1.01 (0.86–1.17)	0.94 (0.82-1.09)	0.96 (0.82-1.14)
	OR by 1 tertile	1.00 (0.94–1.07)	1.00 (0.93–1.08)	0.97 (0.91–1.04)	0.98 (0.90-1.07)
	P for trend	0.923	0.963	0.364	0.661
Total carbo-	Tertile 1	1.00	1.00	1.00	1.00
hydrates	Tertile 2	1.14 (0.82–1.57)	1.10 (0.78–1.54)	1.01 (0.81–1.26)	1.00 (0.80 –1.27)
	Tertile 3	1.35 (0.99–1.84)	1.29 (0.91–1.79)	1.29 (1.04–1.60)*	1.33 (1.04–1.69)*
	OR by 1 tertile	1.17 (1.00–1.36)	1.15 (0.98–1.35)	1.14 (1.02–1.27)*	1.16 (1.03–1.32)*
	P for trend	0.056	0.098	0.017	0.017
Free sugars	Tertile 1	1.00	1.00	1.00	1.00
U	Tertile 2	0.87 (0.65–1.16)	0.90 (0.67–1.19)	0.79 (0.50–1.27)	0.88 (0.52–1.54)
	Tertile 3	1.01 (0.72–1.20)	1.02 (0.73–1.23)	0.94 (0.58–1.56)	1.23 (0.60–2.32)
	OR by 1 tertile	0.99 (0.84–1.17)	1.00 (0.84–1.19)	0.96 (0.73–1.26)	1.12 (0.80–1.55)
	P for trend	0.930	0.956	0.751	0.459
Protein	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	0.89 (0.73–1.09)	0.90 (0.72–1.12)	0.84 (0.71–0.99)*	0.86 (0.71–1.03)
	Tertile 3	0.87 (0.71–1.07)	0.87 (0.69–1.11)	0.97 (0.82–1.16)	1.04 (0.84–1.28)
	OR by 1 tertile	0.93 (0.84–1.03)	0.93 (0.83–1.05)	0.98 (0.89–1.07)	1.01 (0.91–1.12)
	P for trend	0.142	0.258	0.592	0.875
Cholesterol	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	1.03 (0.88–1.20)	1.00 (0.85–1.17)	0.98 (0.86–1.12)	1.03 (0.89–1.20)
	Tertile 3	1.14 (0.98–1.33)	1.10 (0.93–1.30)	1.07 (0.93–1.23)	1.15 (0.97–1.35)
	OR by 1 tertile	1.07 (0.99–1.15)	1.06 (0.97–1.15)	1.03 (0.96–1.11)	1.08 (0.99–1.17)
	P for trend	0.077	0.168	0.385	0.097
Sodium	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	1.01 (0.81–1.24)	1.04 (0.83–1.32)	0.98 (0.84–1.15)	1.03 (0.86–1.23)
	Tertile 3	1.11 (0.90–1.38)	1.19 (0.94–1.50)	1.03 (0.87–1.22)	1.07 (0.87–1.30)
	OR by 1 tertile	1.06 (0.95–1.18)	1.10 (0.99–1.24)	1.01 (0.93–1.10)	1.03 (0.93–1.14)
	P for trend	0.331	0.062	0.747	0.525
Fruit and	Tertile 1	1.00	1.00	1.00	1.00
vegetable	Tertile 2	1.35 (1.19–1.52)*	1.24 (1.08–1.41)*	1.27 (1.11–1.45)*	1.14 (0.98–1.32)
	Tertile 3	1.59 (1.40–1.79)*	1.34 (1.17–1.54)*	1.60 (1.38–1.82)*	1.28 (1.07–1.51)*
	OR by 1 tertile	1.26 (1.18–1.35)*	1.17 (1.08–1.25)*	1.26 (1.18–1.36)*	1.13 (1.04–1.23)*
	P for trend	< 0.001	< 0.001	< 0.001	0.006
NSP	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	1.33 (1.15–1.55)*	1.23 (1.04–1.44)*	1.10 (0.95–1.27)	1.05 (0.89–1.22)
	Tertile 3	1.61 (1.39–1.85)*	1.37 (1.16–1.60)*	1.33 (1.14–1.54)*	1.21 (1.02–1.43)*
	OR by 1 tertile	1.27 (1.18–1.36)*	1.17 (1.08–1.26)*	1.15 (1.07–1.24)*	1.10 (1.01–1.20)*
	P for trend	< 0.001	< 0.001	< 0.001	0.032
HDI score	Tertile 1	1.00	1.00	1.00	1.00
	Tertile 2	1.16 (1.04–1.29)*	1.07 (0.95–1.20)	1.03 (0.92–1.15)	1.03 (0.91–1.16)
	Tertile 3	1.40 (1.26–1.56)*	1.26 (1.12–1.43)*	1.30 (1.15–1.46)*	1.26 (1.10–1.44)*
	OR by 1 tertile	1.18 (1.12–1.24)	1.12 (1.06–1.20)*	1.14 (1.07–1.21)*	1.12 (1.05–1.20)*
	P for trend	< 0.001	0.001	< 0.001	0.001
		< 0.001	0.001	< 0.001	0.001

 Table 7.11
 Associations between perceived control and dietary outcomes

+ Model 1: adjusted for age and country.

Model 2: additionally adjusted for other covariates such as education, occupation, marital status, deprivation, depression, social isolation, and self-rated health.
* P value < 0.05. ** Gender– specific tertile of PC: in men, tertile 1 (0–34), tertile 2 (34–41), and tertile 3 (41–55); in women, tertile 1 (0–33), tertile 2 (33–40), and tertile 3 (40–55).

7.3.2 Modifying role of PC in ERI-diet relationship

For evaluating the modifying role of PC in ERI–diet relationship, ordinal logistic regression is conducted for HDI outcome regressed by PC tertile, ER–ratio tertile, and interaction term between PC tertile and ER–ratio tertile after adjustment for covariates. The log likelihoods for the model with this interaction term and the model without are compared. By comparing the log likelihoods for the model with this interaction term and this interaction term and the model without, LR test is used to test the significance of this interaction term.

As shown in Table 7.12, LR tests report that this interaction term does not reach significance in men (p= 0.960) and women (p= 0.714), respectively. PC has no significantly modifying role in the relationship between ERI and diet.

Table 7.12 Evaluation for modifying role of PC in ERI-diet relationship

Approaches Strata		Healthy Diet Indicator (HDI) outcome
Men		OR of outcome by 1-tertile increase in ERI (95% CI)
1. ERI-diet relationship in	PC tertile 1	0.91 (0.82 – 1.01)
different strata of PC tertile	PC tertile 2	0.88 (0.79 – 0.98)
	PC tertile 3	0.90 (0.81 – 1.00)
2. Interaction term PC x ERI	P-value	P= 0.960
Women		OR of outcome by 1-tertile increase in ERI (95% CI)
1. ERI-diet relationship in	PC tertile 1	0.95 (0.86 – 1.06)
different strata of PC tertile	PC tertile 2	0.96 (0.86 – 1.07)
	PC tertile 3	0.99 (0.88 – 1.11)
2. Interaction term PC x ERI	P-value	P= 0.714

7.3.3 Mediator roles of PC and ERI in OC-diet relationship

To assess mediator roles of PC and ERI in OC-diet relationship, path analysis for an ordinal categorical outcome (HDI outcome) is adopted in Mplus 7. Each path coefficient is obtained by probit regression for HDI outcome on a predictor after adjustment for covariates. The odds ratio (OR) is the antilog of estimated logistic coefficient, calculated from probit coefficient multiplied by 1.8. In terms of exposures, the tertiles of ERI and OC are transformed into a series of dummy variables to compare between tertile groups in each exposure. Next, ERI tertile and OC tertile are treated as continuous variables to estimate assumed linear trend between the exposure and odds of HDI outcome (OR changes by 1-tertile increase in the exposure).

The path model is specified in Figure 7.1. First, there are two potential mediators (PC and ERI) between the effects of OC on HDI outcome. As the HAPIEE study is limited by no measurement of PC and dietary outcomes at wave 2, the cross–sectional associations between OC, ERI, PC and dietary outcomes (all at wave 1) are analyzed. Thus, PC and ERI are only specified to be correlated, although bi-directional relationship between PC and ERI had been hypothesized. Second, antecedent roles of OC in ERI–diet and PC–diet associations are specified based on previous results in the cohort studies on drinking (Chapter 5) and smoking outcomes (Chapter 6).

Third, the mediator effect of PC in OC–diet relation is assessed by two effects: (1) the effect of OC on PC; (2) the effect of PC on diet. The partially mediator effect of ERI in OC–diet relation is estimated by two effects: (1) the effect of OC on ERI; (2) the effect of ERI on diet. Partial mediation applies if both causal relations are confirmed; the product of two path coefficients can estimate the strength of mediator effect.

The bootstrap method is used for significance testing of a mediator effect due to complicated models (an ordinal categorical outcome and two mediators), with 5000 bootstrap samples used to yield more valid estimates for indirect effects by Mplus 7.⁵⁹³ Finally, three indexes for model fit are used: Root Mean Square Error of Approximation

(RMSEA), Comparative Fit Index (CFI), and Tucker–Lewis Index (TLI). RMSEA < 0.06,

CFI > 0.95 or TLI > 0.95 indicate "good model fit".⁵⁹⁴

Figure 7.1 Hypothetical model specified for path analysis for mediator roles of PC and ERI in OC–diet relation



In the results of path analysis in men (Table 7.13; Figure 7.2), first, the mediator effect of ERI in the relationship between OC and HDI outcome is estimated by multiplying two effects: (1) higher OC significantly associated with higher ERI (unstandardized β = 0.286; standard error= 0.016); (2) higher ERI significantly associated with lower levels of HDI outcome (unstandardized β = -0.057; standard error= 0.019). This mediator effect of ERI is significant (-0.016= 0.286 x -0.057; standard error= 0.006 and p= 0.003 by bootstrap method).

Second, the mediator effect of PC in the relationship between OC and HDI outcome is estimated by multiplying two effects: (1) higher OC significantly associated with lower PC (unstandardized β = -0.064; standard error= 0.014); (2) lower PC significantly associated with lower levels of HDI outcome (unstandardized β = 0.077; standard error= 0.018). This mediator effect of PC is significant (-0.005; standard error= 0.002 and p= 0.004). Third, the inverse association between PC and ERI reaches statistical significance (unstandardized β = -0.039; p= 0.001). Finally, the fit

indexes (RMSEA= 0.066, CFI= 0.826, and TLI= 0.796) are not far from the cutoffs for good fit (the cutoffs are RMSEA < 0.06, CFI > 0.95, and TLI > 0.95).

 Table 7.13
 Results of path analysis for mediator roles of PC and ERI in OC-diet

relationship in men (N= 5735)

Parameter	Odds Ratio	Unstandardize d coefficient	Standardized coefficient	P value
$OC \rightarrow HDI$ outcome				
OC tertile 2 vs tertile 1	0.97	- 0.016	- 0.007	0.646
OC tertile 3 vs tertile 1	0.92	- 0.048	- 0.020	0.223
OR change by 1 tertile	0.96	- 0.023	- 0.018	0.232
$\frac{OC \rightarrow ERI \rightarrow HDI \text{ outcome}}{OC \rightarrow ERI}$	0.97	- 0.016	- 0.013	0.003
OC tertile 2 vs tertile 1	1.74	0.316	0.187	< 0.001
OC tertile 3 vs tertile 1	2.62	0.567	0.308	< 0.001
OR change by 1 tertile	1.65	0.286	0.288	< 0.001
ERI → HDI outcome				
ERI tertile 2 vs tertile 1	0.85	- 0.093	- 0.042	0.006
ERI tertile 3 vs tertile 1	0.81	- 0.114	- 0.053	0.003
OR change by 1 tertile	0.90	- 0.057	- 0.044	0.003
$\frac{\text{OC} \rightarrow \text{PC} \rightarrow \text{HDI outcome}}{\text{OC} \rightarrow \text{PC}}$	0.99	- 0.005	- 0.004	0.004
OC tertile 2 vs tertile 1	0.98	- 0.012	- 0.007	0.684
OC tertile 3 vs tertile 1	0.80	- 0.144	- 0.081	< 0.001
OR change by 1 tertile	0.91	- 0.064	- 0.066	< 0.001
$PC \rightarrow HDI$ outcome				
PC tertile 2 vs tertile 1	1.13	0.067	0.030	0.060
PC tertile 3 vs tertile 1	1.33	0.159	0.072	< 0.001
OR change by 1 tertile	1.15	0.077	0.058	< 0.001
ERI correlates with PC		- 0.039	- 0.065	0.001
Tests of model fit	RMSEA=	0.066	CFI= 0.826	TLI= 0.796

Figure 7.2 Results of path analysis for mediator roles of PC and ERI in OC–diet relationship in men



In the results of path analysis in women (Table 7.14; Figure 7.3), first, the mediator effect of ERI in the relationship between OC and HDI outcome is estimated by multiplying two effects: (1) higher OC significantly associated with higher ERI (unstandardized β = 0.284; standard error= 0.016); (2) higher ERI significantly associated with lower levels of HDI outcome (unstandardized β = -0.042; standard error= 0.020). This mediator effect of ERI is significant (-0.012= 0.284 x -0.042; standard error= 0.006 and p= 0.004).

Second, the mediator effect of PC in the relationship between OC and HDI outcome is assessed by multiplying two effects: (1) higher OC significantly associated with lower PC (unstandardized β = -0.066; standard error= 0.014); (2) lower PC significantly associated with lower levels of HDI outcome (unstandardized β = 0.085; standard error= 0.020). The mediator effect of PC is significant (-0.006; standard error= 0.002 and p= 0.001). Third, the inverse association between PC and ERI reaches statistical significance (unstandardized β = -0.051; p < 0.001). Finally, the fit indexes (RMSEA= 0.068, CFI= 0.811, and TLI= 0.782) are not far from the cutoffs for good fit (the cutoffs are RMSEA < 0.06, CFI > 0.95, and TLI > 0.95).

Table 7.14 Results of path analysis for mediator roles of PC and ERI in OC-diet

relationship in	women (N=	5277)
-----------------	-----------	-------

Parameter	Odds Ratio	Unstandardize d coefficient	Standardized coefficient	P value
$OC \rightarrow HDI$ outcome				
OC tertile 2 vs tertile 1	0.98	- 0.011	- 0.006	0.750
OC tertile 3 vs tertile 1	0.96	- 0.025	- 0.012	0.426
OR change by 1 tertile	0.98	- 0.015	- 0.008	0.412
$\frac{\text{OC} \rightarrow \text{ERI} \rightarrow \text{HDI outcome}}{\text{CO} \rightarrow \text{ERI}}$	0.98	- 0.012	- 0.009	0.004
OC → ERI	4.00	0.004	0.474	0.004
OC tertile 2 vs tertile 1	1.66	0.291	0.171	< 0.001
OC tertile 3 vs tertile 1	2.62	0.567	0.317	< 0.001
OR change by 1 tertile	1.64	0.284	0.288	< 0.001
ERI → HDI outcome				
ERI tertile 2 vs tertile 1	0.90	- 0.050	- 0.024	0.224
ERI tertile 3 vs tertile 1	0.86	- 0.088	- 0.039	0.054
OR change by 1 tertile	0.93	- 0.042	- 0.032	0.045
$\frac{\text{OC} \rightarrow \text{PC} \rightarrow \text{HDI outcome}}{\text{OC} \rightarrow \text{PC}}$	0.99	- 0.006	- 0.004	0.001
OC tertile 2 vs tertile 1	0.99	- 0.008	- 0.005	0.777
OC tertile 3 vs tertile 1	0.81	- 0.140	- 0.085	< 0.001
OR change by 1 tertile	0.90	- 0.066	- 0.073	< 0.001
$PC \rightarrow HDI outcome$				
PC tertile 2 vs tertile 1	1.01	0.004	0.002	0.906
PC tertile 3 vs tertile 1	1.37	0.174	0.075	< 0.001
OR change by 1 tertile	1.17	0.085	0.060	< 0.001
ERI correlates with PC		- 0.051	- 0.093	< 0.001
Tests of model fit	RMSEA=	0.068	CFI= 0.811	TLI= 0.782

Figure 7.3 Results of path analysis for mediator roles of PC and ERI in OC-diet



relationship in women

As mentioned earlier, the interactions between gender and exposure variables (ERI and OC) are not significant. Thus, the data from men and women are pooled for final path analysis for mediator roles of PC and ERI in OC–diet relationship (Table 7.15; Figure 7.4). First, the mediator effect of ERI in the relationship between OC and HDI outcome is estimated by multiplying two effects: (1) higher OC significantly associated with higher ERI (unstandardized β = 0.279; standard error= 0.012); (2) higher ERI significantly associated with lower levels of HDI outcome (unstandardized β = -0.046; standard error= 0.014). The mediator effect of ERI is significant (–0.013; standard error= 0.004 and p= 0.001 by bootstrap method).

Second, the mediator effect of PC in the relationship between OC and HDI outcome is estimated by multiplying two effects: (1) higher OC significantly associated with lower PC (unstandardized β = -0.063; standard error= 0.011); (2) lower PC significantly associated with lower levels of HDI outcome (unstandardized β = 0.079; standard error= 0.013). The mediator effect of PC is significant (-0.005; standard error= 0.001 and p < 0.001). Third, the inverse association between PC and ERI reaches statistical significance (unstandardized β = -0.043; p < 0.001). Finally, the fit indexes (RMSEA= 0.066, CFI= 0.819, and TLI= 0.790) are not far from the cutoffs for good fit (RMSEA < 0.06, CFI > 0.95 or TLI > 0.95).

Table 7.15 Results of path analysis for mediator roles of PC and ERI in OC-diet

Parameter	Odds Ratio	Unstandardize d coefficient	Standardized coefficient	P value
OC → HDI outcome	Rallo	u coemcient	coemcient	
OC tertile 2 vs tertile 1	0.99	- 0.006	- 0.003	0.814
OC tertile 3 vs tertile 1	0.99	- 0.008	- 0.003	0.193
OR change by 1 tertile	0.97	- 0.017	- 0.013	0.237
$OC \rightarrow ERI \rightarrow HDI outcome$	0.98	- 0.013	- 0.010	0.001
OC → ERI				
OC tertile 2 vs tertile 1	1.68	0.295	0.174	< 0.001
OC tertile 3 vs tertile 1	2.56	0.554	0.306	< 0.001
OR change by 1 tertile	1.63	0.279	0.282	< 0.001
ERI → HDI outcome				
ERI tertile 2 vs tertile 1	0.88	- 0.068	- 0.031	0.009
ERI tertile 3 vs tertile 1	0.85	- 0.091	- 0.042	0.001
OR change by 1 tertile	0.92	- 0.046	- 0.035	0.001
$\frac{OC \rightarrow PC \rightarrow HDI \text{ outcome}}{OC \rightarrow PC}$	0.99	- 0.005	- 0.004	< 0.001
$OC \rightarrow PC$	0.00	0.010	0.000	0 705
OC tertile 2 vs tertile 1	0.98	- 0.010	- 0.006	0.735
OC tertile 3 vs tertile 1	0.78	- 0.136	- 0.078	< 0.001
OR change by 1 tertile	0.91	- 0.063	- 0.066	< 0.001
$PC \rightarrow HDI outcome$				
PC tertile 2 vs tertile 1	1.08	0.041	0.018	0.108
PC tertile 3 vs tertile 1	1.34	0.163	0.072	< 0.001
OR change by 1 tertile	1.15	0.079	0.059	< 0.001
ERI correlates with PC		- 0.043	- 0.075	< 0.001
Tests of model fit	RMSEA=	0.066	CFI= 0.819	TLI= 0.790

relationship in pooled data of men and women (N= 11012)

Figure 7.4 Results of path analysis for mediator roles of PC and ERI in OC-diet

relationship in pooled data of men and women



7.4 Main Findings for Dietary Outcomes

The analyses are based on the cross–sectional study of 11012 subjects (5735 men and 5277 women) aged 45–69 from the wave 1 of HAPIEE study. The aims of analyses are: (1) to examine the relationship between ERI, OC, and dietary outcomes; (2) to evaluate the potential role of PC in the relationship between ERI, OC and dietary outcomes. Antecedent role of OC in ERI–diet association was specified according to previous findings in drinking and smoking outcomes, as potentially bidirectional relationship between OC and ERI (Hypothesis 9) cannot be disentangled in the cross–sectional study. The following findings are reported according to specific objectives and hypotheses listed in Chapter 3.

In terms of the associations between ERI, OC and dietary outcomes, Hypothesis 3 that higher ER ratio is associated with less healthy diet (lower levels of HDI) after adjustment for covariates is supported. Hypothesis 6 that higher OC is associated with less healthy diet after adjustment for covariates is supported. Although there are inconsistent associations between exposure variables and individual HDI components, the overall effects of exposure variables on HDI (the sum of 9 components) appear more robust. This finding implies that exposure variables are associated with overall diet quality that is linked to the risks of chronic diseases. In addition, modifying role of OC in ERI–diet relationship is found non–significant.

With regards to the potential role of PC in the relationship between ERI, OC and dietary outcomes, Hypothesis 12 that lower PC is associated with less healthy diet after adjustment for covariates is supported.

Hypothesis 15 is partially supported. PC and ERI partially mediate the effects of OC on dietary outcomes. Additionally, PC and ERI may have bi-directional relationship. PC and ERI are negatively associated with each other; bi-directional relationship between PC and ERI is possible, but causal directionality cannot be established in the cross-sectional study. Finally, modifying role of PC in ERI-directional

relation is non-significant.

Note that the methodological issues and interpretation of the main findings for dietary outcomes will be addressed in detail in Chapter 8, and their implications for practice, policy, and research in Chapter 9. The analyses for the associations between OC, ERI and dietary outcomes have been peer reviewed and published in the British Journal of Nutrition (please see Appendix 5).

Chapter 8. General Discussion

In this chapter, I will firstly summarize the results in terms of all objectives and hypotheses listed in Chapter 3. This part will be followed by the discussion of various methodological issues of the thesis. Finally, in the last part of Chapter 8, I will compare my findings with those from other existing studies.

8.1 Summary for Results

This section provides the summary of results for drinking, smoking, and dietary outcomes, respectively, according to objectives and hypotheses listed in Chapter 3.

Objective 1

To assess crude and adjusted associations between ERI and three health behaviours – alcohol drinking, smoking, and diet, respectively.

According to Hypothesis 1, higher ER ratio is associated with higher levels of alcohol drinking after adjustment for covariates. In men, the OR changes for heavy drinking, binge drinking and problem drinking were 1.15, 1.49 and 1.35, respectively, by 1–tertile increase in ER ratio. In women, these OR changes were 1.13, 1.42 and 1.36, respectively. All p–values were significant (p < 0.05). My findings suggest that Hypothesis 1 is supported.

Based on Hypothesis 2, higher ER ratio is associated with higher levels of smoking after adjustment for covariates. The OR changes for being current smokers were 1.18 and 1.21 by 1–tertile increase in ER ratio in men and women, respectively. In current smokers, the OR changes for being in a higher level of smoking intensity were 1.19 and 1.16 (p= 0.088) by 1–tertile increase in ER ratio in men and women, respectively. All p–values were significant (p < 0.05) except that identified as marginally significant. My findings suggest that Hypothesis 2 is supported.

According to Hypothesis 3, higher ER ratio is associated with less healthy diet

(lower levels of HDI) after adjustment for covariates. The OR changes of being in a higher HDI category were 0.86 and 0.88 by 1–tertile increase in ER ratio in men and women, respectively. Both p–values were significant (p < 0.05). My findings indicate that Hypothesis 3 is supported.

Objective 2

To assess crude and adjusted associations between OC and three health behaviours.

Based on Hypothesis 4, higher OC is associated with higher levels of alcohol drinking after adjustment for covariates. In men, the OR changes for heavy drinking, binge drinking and problem drinking were 1.08 (p= 0.081), 1.31 and 1.28, respectively, by 1–tertile increase in OC; all p–values were significant (p < 0.05) except that identified as marginally significant. In women, these OR changes were 1.06 (p= 0.281), 1.24 and 1.27, respectively; all p–values were significant except that identified as non–significant. My findings suggest that Hypothesis 4 is partially supported.

According to Hypothesis 5, higher OC is associated with higher levels of smoking after adjustment for covariates. The OR changes for being current smokers were 1.14 and 1.15 by 1–tertile increase in OC in men and women, respectively; both p–values were significant (p < 0.05). In current smokers, the OR changes for being in a higher level of smoking intensity were 1.11 (p= 0.171) and 1.14 (p= 0.145) by 1–tertile increase in OC in men and women, respectively; these associations did not reach statistical significance. My findings indicate that Hypothesis 5 is partially supported.

Based on Hypothesis 6, higher OC is associated with less healthy diet (lower levels of HDI) after adjustment for covariates. The OR changes of being in a higher HDI category were 0.90 by 1-tertile increase in OC in both sexes; both p-values were significant (p < 0.05). My findings suggest that Hypothesis 6 is supported.

Objective 3

To evaluate the potential role of OC (antecedent, mediator, modifier, or direct effect) in

the relationship between ERI and health behaviours.

As predicted by Hypothesis 7, OC and ERI had bi–directional relationship, but the effect of OC on ERI was stronger than the other direction in the middle-aged and older populations. Antecedent role of OC in ERI–drinking relationship was found significant, but mediator role of OC was not significant. Direct effect of OC on drinking was not significant. Finally, modifying role of OC in ERI–drinking relation was non–significant. My findings show that Hypothesis 7 is supported.

As suggested by Hypothesis 8, OC and ERI had bi–directional relationship, but the effect of OC on ERI was stronger than the other direction in the middle-aged and older populations. Antecedent role of OC in ERI–smoking relationship was found significant among all smoking outcomes in both sexes (except smoking intensity in women), but mediator role of OC was not significant. Direct effect of OC on smoking was not significant. Finally, modifying role of OC in ERI–smoking relation was non– significant. My findings suggest that Hypothesis 8 is partially supported.

Hypothesis 9 suggested that OC and ERI had bi–directional relationship; however, bidirectional relationship between OC and ERI cannot be disentangled in this cross–sectional study (diet was only available at wave 1 whereas drinking and smoking were available at wave 1 and wave 2). Antecedent role of OC in ERI–diet relationship was significant, but mediator role of OC was not tested. Direct effect of OC on diet was not significant. Finally, modifying role of OC in ERI–diet relationship was non–significant. Hypothesis 9 is partially supported.

Objective 4

To assess crude and adjusted associations between PC and three health behaviours.

According to Hypothesis 10, lower PC is associated with higher levels of alcohol drinking after adjustment for covariates. In men, the OR changes for heavy drinking, binge drinking and problem drinking were 0.90, 0.82 and 0.81, respectively, by 1–tertile increase in PC. In women, these OR changes were 0.91 (p= 0.083), 0.80 and 0.80,

respectively. All p-values were significant (p < 0.05) except that identified as marginally significant. My findings suggest that Hypothesis 10 is supported.

Based on Hypothesis 11, lower PC is associated with higher levels of smoking after adjustment for covariates. The OR changes for being current smokers were 0.81 by 1-tertile increase in PC in both sexes. In current smokers, the OR changes for being in a higher level of smoking intensity were 0.79 and 0.80 by 1-tertile increase in PC in men and women, respectively. All p-values were significant (p < 0.05). My findings indicate that Hypothesis 11 is supported.

According to Hypothesis 12, lower PC is associated with less healthy diet after adjustment for covariates. The OR changes for being in a higher HDI category were 1.12 by 1-tertile increase in PC in both sexes; both p-values were significant (p < 0.05). My findings show that Hypothesis 12 is supported.

Objective 5

To examine the potential role of PC (mediator or modifier) in the relationship between ERI, OC, and health behaviours.

As predicted by Hypothesis 13, ERI and PC partially mediated the effects of OC on alcohol drinking. Additionally, PC and ERI were negatively associated with each other; bi–directional relationship between PC and ERI is possible, but causal directionality cannot be established in the cross–sectional analyses. Finally, modifying role of PC in ERI–drinking relation was non–significant. My findings show that Hypothesis 13 is partially supported.

As suggested by Hypothesis 14, ERI and PC partially mediated the effects of OC on all smoking outcomes in both sexes (except smoking intensity in women). Additionally, PC and ERI were negatively associated with each other; bi–directional relationship between PC and ERI is possible, but causal directionality cannot be established in the cross–sectional analyses. Finally, modifying role of PC in ERI– smoking relation was non–significant. My findings indicate that Hypothesis 14 is

partially supported.

As predicted by Hypothesis 15, ERI and PC partially mediated the effects of OC on diet. In addition, PC and ERI were negatively associated with each other; bi– directional relationship between PC and ERI is possible, but causal directionality cannot be established in the cross–sectional study. Finally, modifying role of PC in ERI–diet relation was non–significant. Hypothesis 15 is partially supported.

8.2 Methodological Issues

Before comparing my findings summarized above with those from previous studies, it is essential to focus on several methodological issues and to highlight potential strengths and limitations of the current project. The strengths of this thesis are a population–based approach with random community samples from three CEE countries, a large sample size with strong statistical power, a 2–wave cohort study design, a central protocol across all study centres (such as questionnaires or training), data collected on diverse aspects of health behaviours, and application of advanced statistical methods (such as SEM and path analysis). However, the results described in this thesis should be interpreted carefully, because as with all studies, there are several methodological features of study population and of study design whose limitations need to be taken into account. These methodological issues are discussed in detail in the following sections.

8.2.1 Representativeness of study sample and selection bias

External validity refers to the extent to which the results of a study can be extrapolated or generalized to the reference population; this concept is related to the representativeness of the study, and thus depends on the procedure of selection of participants. Selection bias is systematic error that results from procedures used to

select subjects and from factors that influence study participation.595

The participants in the HAPIEE study were randomly selected from population registers in Novosibirsk, Krakow and six towns in the Czech Republic, and they are considered representative of urban populations. While the selected urban centers cannot be entirely representative for the whole countries, available indicators of socioeconomic characteristics, health behaviours, and mortality suggest that these selected urban populations approximately represented their national populations in the WHO systematic reviews and reports.^{596,597}

Overall response rate in the HAPIEE study was 60%, which was typical for contemporary cohort studies, such as the Survey of Health, Ageing and Retirement in Europe (SHARE) and the English Longitudinal Study of Ageing (ELSA).^{598,599} In all three countries, short questionnaires from subsample of those who refused participation were analyzed. As a small proportion of non–respondents had died or moved away after the sample was selected but before being invited to the study, they were ineligible for inclusion. Extrapolating from the proportion of incorrect addresses identified in home visits and from the evaluation of accuracy of the population register, real response rates were estimated to be 71% in Novosibirsk, 68% in Krakow, and 60% in Czech Republic. The comparison of respondents and non–respondents in the HAPIEE study was conducted; non–respondents were more likely to be younger age, male, with lower levels of education, with higher prevalence of smoking, and with poorer self–rated health.⁶⁰⁰

The approach of handling missing values in this thesis is "complete case analysis"; only subjects with complete data on exposures and outcomes were analyzed. In some cases, a complete case analysis can provide unbiased estimates; for example, fitting the regression model to complete cases would be unbiased if missingness is independent of outcome, after adjusting for predictors. This assumption may not be entirely true in this project, particularly for drinking outcome, but should not influence results substantially. A disadvantage of this approach is that excluding observations

with missing values may reduce sample size of analytical sample.⁶⁰¹ Notably, there are alternative methods such as imputation methods – which predict missing values based on observed data and missing–data pattern. This approach assumes that missing is dependent on observed data but not on unobserved data (missing at random; MAR). However, multiple imputation might make some statistical analyses (e.g. SEM) too complex and MAR assumption might result in biased results.

As a form of sensitivity analysis, bivariate analyses were conducted among study samples and excluded subjects due to missing values in exposures, outcomes and covariates (for more details, see Section 4.2). In the subsample for drinking/smoking outcomes, excluded subjects (n= 5758) were more likely to be male, with lower educational level, with higher alcohol consumption, with more current smokers, and with poorer self–rated health than the study sample. In the subsample for dietary outcomes, excluded subjects (n= 2259) were more likely to be older and male, with lower educational level, with more current smokers, and with poorer self–rated health than the study sample. In the subsample for dietary outcomes, excluded subjects (n= 2259) were more likely to be older and male, with lower educational level, with more current smokers, and with poorer self–rated health than the study sample. There were no major systemic differences between complete cases and incomplete cases. Those with worse profiles of health behaviours were more likely to drop out from follow–up or have missing values; the levels of risky health behaviours were probably underestimated in this thesis, but this bias should not substantially affect the associations between exposures and health behaviours.

8.2.2 Information bias

Information bias is systematic error in estimating an effect caused by measurement error in the needed information. Differential misclassification is dependent on exposure or outcome status. Non–differential misclassification is independent from exposure or outcome status. Non–differential misclassification in exposures occurs when the proportion of subjects misclassified on exposure does not depend on outcome status of subjects; non–differential misclassification in outcomes

occurs when the proportion of subjects misclassified on outcome does not depend on exposure status of subjects.⁶⁰²

Common scenarios of differential misclassification were not found in this thesis, because original aims of the HAPIEE study were to investigate psychosocial risk factors for chronic diseases in CEE – different from the aims of this thesis.⁶⁰³ However, analytical procedures such as collapsing continuous or categorical exposures into fewer categories can change non–differential error to differential misclassification, thereby exaggerating or underestimating an effect.⁶⁰⁴

In terms of non–differential misclassification in exposures, ER ratio, OC and PC were evaluated by self–reported measurements that were potentially subjective to recall bias. In addition, in CEE countries, people may tend to answer questions in psychosocial measures around the middle of the scale, rather than using the extreme points of this scale, thereby leading to overestimation of the effect.⁶⁰⁵

Non-differential misclassifications in outcomes (drinking, smoking, and diet) are described in the following paragraphs. For drinking outcomes, self-reported measures of alcohol drinking typically underestimate actual consumption.⁶⁰⁶ In particular, social stigma associated with alcohol affects women more than men, and systematic underreporting of alcohol is probably greater in women than in men.⁶⁰⁷ Nevertheless, GFQ method appears less prone to underreporting among available alcohol measures. In the HAPIEE study, both GFQ-based variables and problem drinking were strongly associated with separately taken measures of alcohol consumption and serum gamma-glutamyl transferase, indicating acceptable validity of drinking outcomes.⁶⁰⁸

For smoking outcomes, self–reported measures of smoking are often subject to underreporting. As smoking is considered more socially acceptable in men than in women, smoking status and smoking intensity may be underestimated particularly in women. In the HAPIEE study, the validity of smoking outcomes has not been tested by estimating correlation with plasma biomarkers.⁶⁰⁹

For dietary outcomes, FFQ is the primary method to gather dietary information

from large population samples, as it is inexpensive and representative for average long-term diet. However, the following limitations should be considered. First, the FFQ method tends to be semi-quantitative, rather than fully quantitative, probably resulting in overestimation or underestimation of dietary intakes.⁶¹⁰ Assigning HDI scores may introduce some misclassification, but the ranking of subjects in terms of HDI should be unbiased. Second, local but internationally comparable food composition tables are unavailable for these CEE countries. The inclusion of local food tables to capture country-specific foods may introduce misclassification. Besides, the FFQ components differed slightly in three CEE countries, thereby leading to imprecision in comparing dietary intakes between these countries. However, added items of country-specific foods were approved by local nutritionists to ensure that diet was assessed properly in each country. Third, the validity of the FFQ regarding fruit, vegetable and micronutrient intakes was found acceptable by estimating correlation with plasma biomarkers in a random HAPIEE subsample; these correlations were similar to other large studies. However, other HDI components have not been tested for validity.⁶¹¹ Fourth, the HDI was constructed by Huijbregts' original approach consisting of dietary components coded as dichotomous variables. Note that Jankovic et al proposed a new HDI approach which applied continuous scoring to obtain greater variation between individuals, providing more precise estimation of diet quality.⁶¹²

8.2.3 Confounding

Confounding means that the apparent effect of the exposure of interest is distorted because the effect of extraneous factor (confounding factor) is mistaken for or mixed with the actual exposure effect, leading to overestimation or underestimation of an effect. For a variable to be a confounding factor, it should meet three necessary criteria: it must be a risk factor for the outcome; it must be associated with the exposure in the source population of the study; it cannot be a mediator in the causal path between the

exposure and the outcome.

In this thesis, potential confounding factors were selected from the HAPIEE data and then adjusted as covariates in regression analyses, including age, country, social position (education and occupation), material factors (deprivation), psychosocial factors (marital status, depression, and social isolation), and self–rated health. Although possible confounding factors were controlled in the analyses, there may be residual confounding factors not taken into account. For example, chronic stressors outside workplace (e.g. daily hassles), psychological constructs (e.g. attitude and subjective norm in Theory of Planned Behaviour), and community–level factors (e.g. access to neighborhood resources) are known risk factors for health behaviours but unavailable in the HAPIEE study. These confounding factors may lead to underestimation or overestimation of exposure–outcome relationships, depending on direction of associations of the confounding factor with the exposure and outcome.⁶¹³

8.2.4 Limitations of two–wave cohort study designs

The ideal condition for mediation analyses is a 3–wave cohort study design, in which the exposure variable (wave 1) precedes the mediator (wave 2) in time and the mediator precedes the outcome (wave 3) in time. Although a 3–wave cohort study provides the best estimation for mediation, a 2–wave cohort study still can offer indication for the presence of partial mediation, but not full mediation, thereby yielding better evidence than a cross–sectional study or a half–longitudinal design (one of the associations is cross–sectional).⁶¹⁴

As three waves of relevant measurements for this thesis were unavailable in the HAPIEE study, a 2–wave cohort design (an autoregressive and cross–lagged model) was adopted in path analyses for the potential role of OC in ERI–drinking relation and ERI–smoking relation, respectively. Note that only partial mediation (e.g. ERI partially mediated the effects of OC on drinking), rather than full mediation, can be examined

in these analyses. Moreover, the effect size measure for a mediated effect (the proportion of total effect that is mediated) cannot be measured as precisely as that in a 3-wave cohort study; the mediated effect, if not strong, would tend to be underestimated in a 2-wave cohort study.⁶¹⁵

8.2.5 Limitations of cross–sectional study designs

As PC and dietary outcomes were unavailable at wave 2 in the HAPIEE study, two parts of the analyses were actually in cross-sectional design: (1) the relationship between OC, ERI and PC at wave 1 in the analyses of drinking or smoking outcomes at wave 2 (a half-longitudinal design); (2) the relationship between exposure variables at wave 1 and dietary outcomes at wave 1. A cross-sectional study often has difficulty in determining the time order between the exposure and the outcome, unless the exposure is defined prior to recruitment and measurement of the outcome.

In the cross–sectional analysis for the associations between OC, ERI and PC, the parsimonious hypothesis that OC causes ERI and PC, rather than the other causal direction, has been supported by theoretical explanation (e.g. Transactional Model of Stress or personality psychology) and empirical studies (including the 2–wave cohort analysis in this thesis). However, potentially bidirectional relationship between OC, ERI and PC still cannot be disentangled in the cross–sectional analysis.

In the cross-sectional analysis for exposure-diet associations, the difficulty in determining time sequence between events would introduce bias into the analysis. For example, those with unhealthy diet may tend to have more mental and physical problems, thereby causing them to perceive or encounter higher levels of work stress. Thus, the reverse causality that unhealthy diet causes high levels of work stress cannot be ruled out by the cross-sectional design.

8.2.6 Random error

Random error (chance or random variation) is the divergence, due to chance alone, of an observation on a sample from the true population value, leading to imprecise measurement of an association. Random error is often induced by the process of selecting study subjects and the unpredictable inaccuracies in occurrence measures. A common way to reduce random error in an epidemiologic study is to enlarge the size of the study, which is planned based on statistical sample–size formulas. In addition, significance testing in epidemiology focuses on deciding whether chance or random error could be solely responsible for an observed association.

In this thesis, the subsamples were generated from the HAPIEE study and the data was stratified by gender in the analyses, it is not at risk of being underpowered to detect small differences. In the subsample on dietary outcomes (sample size= 6000), power calculation shows that statistical power is over 99% for odds ratio larger than 1.3 in baseline probability of 0.05. In the subsample on drinking/smoking outcomes (sample size= 4000), power calculation reports that statistical power is over 95% for odds ratio larger than 1.3 in baseline probability of 0.05. However, the study was sometimes overpowered to investigate the proposed research questions, and small effects may have been detected as statistically significant. Thus, the interpretation of results should not rely entirely on statistical significance of an effect, but the magnitude of an effect estimate and previous work should also be considered in order to make careful interpretation of statistics.

8.3 Discussion of Results

Taking all the methodological issues of the thesis into account, it is possible to compare the findings of this thesis with previous literature. This discussion of results covers the following topics: (1) work stress and health behaviours; (2) OC personality and health behaviours; (3) the potential role of OC in the relationship between ERI and

health behaviours; (4) PC and health behaviours; (5) the potential role of PC in the relationship between OC, ERI and health behaviours.

8.3.1 Work stress and health behaviours

In terms of work stress and drinking outcomes, this thesis found that higher ER ratio at wave 1 was associated with higher levels of drinking outcomes (heavy drinking, binge drinking and problem drinking) at wave 2 after adjustment for covariates in both sexes. This finding is in line with previous literature on work stress and alcohol drinking. Several reviews and prospective studies have supported the associations between the DC model and alcohol drinking (for more details, see Section 2.3.3).^{616,617,618} Moreover, existing studies have showed promising results to support the links between the ERI model and alcohol drinking. For example, Head et al reported that high ER ratio predicted alcohol dependence in a British cohort study (n= 7,372); this association was stronger in men than in women.⁶¹⁹ In the pilot HAPIEE study (n= 694) in the same CEE populations as this thesis, Bobak et al found that high ER ratio was associated with binge drinking (OR= 1.36), problem drinking (OR= 1.37), high annual intake of alcohol (OR= 1.29), and high annual number of drinking sessions (OR= 1.34) in the cross–sectional analyses.⁶²⁰ Thus, my finding provides evidence for consistency of longitudinal associations between ERI and drinking outcomes in the CEE populations.

With regard to work stress and smoking outcomes, this thesis reported that higher ER ratio at wave 1 was associated with higher levels of smoking outcomes (smoking status and smoking intensity) at wave 2 after adjustment for covariates in both sexes. This finding is consistent with previous reviews on the associations of the DC model with smoking status and smoking intensity (for more details, see Section 2.3.3).^{621,622} Additionally, several cross–sectional studies have supported the links between the ERI model and smoking outcomes. For example, Kouvonen et al reported that high ER ratio was associated with being current smokers (OR= 1.28) in a Finnish cross–

sectional study (n= 46,190); among current smokers, high ER ratio was associated with high smoking intensity (OR= 1.19).⁶²³ Radi et al showed that higher ER ratio was associated with higher smoking intensity among current smokers in women, but not in men in an Australian cross–sectional study (n= 1,101).⁶²⁴ However, Ota et al found that ER ratio did not significantly predict smoking cessation at 2–year follow-up in 1,423 middle-aged men in Japan.⁶²⁵ Thus, my finding provides further evidence for longitudinal associations of ERI with smoking outcomes in a population–based study.

In terms of work stress and dietary outcomes, this thesis found that higher ER ratio was cross-sectionally associated with less healthy diet (HDI) after adjustment for covariates in both sexes. While there are inconsistent associations between ER ratio and individual HDI components (9 nutrient/food intakes), which may reflect sex or individual differences in dietary responses to work stress, the overall effects of ER ratio on HDI (the sum of 9 nutrient/food intakes) appeared more robust. This finding implies that high ER ratio is associated with poor diet quality linked to the risks of chronic diseases. My finding is generally in line with previous literature on work stress and diet. Many studies have supported the links between the DC model and dietary outcomes (see Section 2.3.3).^{626,627,628} To my best knowledge, this cross-sectional analysis is the first study to demonstrate the links between the ERI model and dietary outcomes. Moreover, the measurements of dietary outcomes varied considerably between previous studies on the DC model and diet. The method of diet quality takes into account intakes of various foods and nutrients, thereby providing more accurate pictures of diet than single food/nutrient intake.629 Thus, this thesis adopting the outcome of diet quality can offer more solid evidence for the associations between work stress and diet.

8.3.2 Overcommitment personality and health behaviours

The associations of OC personality with three health behaviours were examined

in this thesis. Firstly, higher OC at wave 1 was associated with higher levels of drinking outcomes (heavy drinking, binge drinking and problem drinking) at wave 2 after adjustment for covariates in both sexes, except non-significant association between OC and heavy drinking in women. Secondly, higher OC at wave 1 was associated with higher levels of smoking outcomes (smoking status) at wave 2 after adjustment for covariates, but the association between OC and smoking intensity did not reach statistical significance in both sexes. Thirdly, higher OC was cross–sectionally associated with less healthy diet (HDI) after adjustment for covariates in both sexes. The associations of OC with individual HDI components were inconsistent, but the overall effects of OC on HDI appeared remarkable.

There has been very limited literature regarding the effect of or the potential role of OC (main or modifying effect) on health behaviours. Importantly, two studies from Japan and Australia have reported negative findings on main effects of OC on smoking outcomes; however, modifying effect of OC was not examined.^{630,631} Although this thesis is probably the first to show the effect of OC on health behaviours, there have been many studies supporting the effects of Type A behaviour and related personality constructs (Neuroticism and Hostility) on health behaviours (for more details, see Section 2.4.2),^{632,633} together with numerous studies supporting the effects of PC on health behaviours (see Section 2.4.4).^{634,635} As the origins of OC are traced to Type A behaviour and PC, the aforementioned studies can be used to partially support the links between OC and health behaviours identified in this thesis.

8.3.3 Potential role of OC in relationship between ERI and health behaviours

The potential role of OC in ERI–outcome relationship was examined in the thesis, including modifying, antecedent, mediator or direct effects. The potential role of OC in ERI–outcome relationship was originally suggested as main or modifying effect by Siegrist.⁶³⁶ The review by Van Vegchel et al found that main effect of OC was

supported in 17 of 27 studies (63%), but modifying effect was only supported in 3 of 12 studies (25%).⁶³⁷ The role of OC in ERI–outcome relationship remains inconclusive in existing literature, and originally assumed role of OC (main or modifying effect) appears relatively simple compared to accumulated research on diverse roles of personality in stress processes. For instance, the Michigan model proposed that objective work environments influence subjective perceptions of work stress, which affect short–term psycho–biological responses, leading to long–term health problems; personality can operate at several points in the stress process, including modifying, antecedent, mediator, or direct effects.⁶³⁸

Antecedent role of OC (OC predicts ERI which subsequently affects outcome) has been supported by theories and empirical studies. Personality can influence work stress via theoretical mechanisms: selection, stressor creation, and perception.⁶³⁹ Several studies have supported antecedent roles of OC and related personality constructs (e.g. Type A behaviour and Neuroticism) in the relationship between work stress and outcomes (for more details, see Section 2.5.2).^{640,641} Siegrist also implied the possibility of antecedent role of OC in ERI–outcome relationship; individuals with high OC might expose themselves more often to high efforts at work, or they exaggerate their efforts beyond what is formally needed, thereby resulting in continued imbalance between high effort and low reward.⁶⁴²

Mediator role of OC (ERI predicts OC which then affects outcome) appears not impossible based on theories and empirical studies. The meta-analysis of 92 longitudinal studies found that personality continues to change throughout adulthood but only modestly after age 50.⁶⁴³ Several studies supported mediator roles of OC and related personality in the relation between work stress and outcomes (For more details, see Section 2.5.3).^{644,645} Notably, there may be bidirectional relationship between personality and work environment across life course; antecedent and mediator roles of OC might coexist (see Section 2.5.4).⁶⁴⁶

In the path analyses in drinking and smoking outcomes, I found that OC and ERI

may have bi-directional relationship; the effect of OC on ERI is stronger than the other causal direction in the middle-aged and older populations. Antecedent role of OC in the relation between ERI and health behaviours is significant, but mediator role of OC is not. Finally, modifying role of OC in ERI-outcome relationship is not significant.

In my analyses, the traditional approach (logistic regression) initially found "main effects" of OC on health behaviours. Similarly, the review by Van Vegchel et al supported "main effects" of OC in 63% of studies. However, previous evidence may not really support main effects in a more critical appraisal, as possibilities of antecedent and mediator roles had not been tested simultaneously. Confirmation of OC–outcome relationship is just a first step for either OC–ERI–outcome (OC as antecedent) or ERI–OC–outcome (OC as mediator) causal chains in mediation analysis.⁶⁴⁷ It is possible that previously reported "main effect" of OC actually represents the "snapshot" of dynamic relationship between OC, ERI and outcomes.

Bidirectional relationship between OC and ERI was found in this thesis; OC at wave 1 predicted ER ratio at wave 2, and ER ratio at wave 1 predicted OC at wave 2. To my best knowledge, this cohort study is the first to demonstrate the bidirectional relationship between OC personality and work stress in the ERI model; this finding corresponds to increased recognition for bidirectional relationship between personality and environment across life course. ⁶⁴⁸ Social environments in childhood and adulthood (e.g. workplace) may alter an individual's personality; conversely, personality influences an individual to select, create and perceive environmental stressors.^{649,650} High OC personality may perceive, select and create high levels of ERI work stress, which further aggravate their vulnerable personality, resulting in a vicious circle. Thus, intervention should focus on both work environment and person in order to disrupt the cumulated effects in the reciprocal relationship.

This thesis also found that the effect of OC personality on work stress is much stronger than the other causal direction in this sample aged 45–69. As noted previously, the meta-analysis reported that personality changes only modestly after age 50.651

The effect of work stress on personality appeared significant but modest in the middleaged and older populations, but this effect is expected to be stronger if younger populations would be examined. Although bidirectional relationship between OC and ERI may exist, only antecedent role of OC, not mediator role, reached statistical significance; this finding that OC acts mainly as an antecedent in ERI–outcome relationship is crucial for the implications discussed later.

8.3.4 Perceived control and health behaviours

The associations of PC with three health behaviours were examined in this thesis. Firstly, higher PC at wave 1 was associated with lower levels of drinking outcomes (heavy drinking, binge drinking and problem drinking) at wave 2 after adjustment for covariates in both sexes. Secondly, higher PC at wave 1 was associated with lower levels of smoking outcomes (smoking status and smoking intensity) at wave 2 after adjustment for covariates in both sexes. Thirdly, higher PC was cross–sectionally associated with more healthy diet (HDI) after adjustment for covariates in both sexes.

The findings in this thesis are consistent with previous studies, which have extensively supported that PC and its components (e.g. self-efficacy or PBC in TPB) can predict health behaviours – drinking, smoking, poor diet and physical inactivity (for more details, see Section 2.4.4).^{652,653,654} In fact, self-efficacy is the common construct across all Social Cognitive Models – the most commonly used theories in predicting health behaviours at intrapersonal level.⁶⁵⁵ Moreover, PC is suggested to influence various health outcomes directly by psychobiological processes and indirectly via health behaviours (for more details, see Section 2.4.3).^{656,657}

8.3.5 Potential role of PC in relationship between OC, ERI and health behaviours By the Transactional Model of Stress, the potential role of PC is considered in the

relationship between ERI, OC and health behaviours. When a person faces a stressor, one would evaluate potential threat (primary appraisal – perceived work stress ERI) and one's ability to alter the situation and manage negative emotion (secondary appraisal – PC). Both cognitive appraisals affect one's coping efforts (problem management and emotional regulation) and then influence health behaviours.⁶⁵⁸ This model has been enriched by accumulating literature on personality psychology; personality traits (e.g. OC) are suggested to influence the transactional process at several points – primary appraisal, secondary appraisal and coping efforts (for more details, see Section 2.6.1).⁶⁵⁹

In Transactional Model of Stress, Lazarus suggested that primary appraisal and secondary appraisal engage in a dynamic and reciprocal relationship with each other.⁶⁶⁰ The potentially bidirectional relationship between perceived work stress (primary appraisal) and PC (secondary appraisal) has been supported by two types of empirical evidence: (1) PC acts as a mediator in the effects of work stress on outcomes; (2) work stress acts as a mediator in the effects of PC on outcomes (for more details, see Section 2.6.2).^{661,662} On the other hand, the interaction between work stress and PC has sometimes been supported in previous studies, and this possibility was tested in this thesis (for more details, see Section 2.6.3).^{663,664}

In this thesis, path analyses were conducted for potential relationship between OC, ERI, PC and health behaviours. The results showed that both ERI and PC partially mediated the effects of OC on health behaviours. Additionally, ERI and PC were negatively associated with each other; ERI and PC may have bi–directional relationship, but causal directionality cannot be established in the cross–sectional study. Finally, PC had no significantly modifying role in the relationship between ERI and health behaviours.

Previous analyses for the potential role of OC in ERI–outcome relationship found significant antecedent role of OC; high OC persons tend to select, create, and perceive high levels of ERI works stress. Transactional Model of Stress provides another

explanation for the effect of OC personality on health behaviours; high OC persons tend to perceive high levels of ERI work stress (primary appraisal) and feel low levels of PC to alter the situation (secondary appraisal); both cognitive appraisals affect their use of emotion–focused coping (e.g. engaging in risky health behaviours). Moreover, high PC might decrease ERI work stress, and vice versa. Cognitive appraisal appears essential to understand the dynamic stress processes.⁶⁶⁵ Whether the influences come from environment (workplace) or person (personality), primary and secondary appraisals are main cognitive processes associated with subsequent coping outcomes. Thus, both cognitive appraisals can serve as modifiable targets by treatments such as cognitive–behavioral therapy.⁶⁶⁶

Work stress research has a long and rich history of identifying work factors potentially causing stress, but individual differences have not been paid enough attention. The ERI model addressed individual differences by OC personality, but the mechanisms via which OC can influence stress processes remained unclear. Transactional Model of Stress provided mediating pathways linking environment and personality to health, thereby recognizing possible mechanisms of individual differences (personality, cognitive appraisal, and coping). This thesis attempts to integrate the ERI model with Transactional Model of Stress in order to enrich work stress research by psychological literature and to identify potential causal pathways in guiding effective interventions for work stress in the future.

Chapter 9. Implications for Research, Practice and Policy

Implications for research, practice and policy are proposed based on the three main findings in the thesis: (1) associations between work stress (ERI) and health behaviours; (2) potential role of OC in the relation between ERI and health behaviours; (3) potential role of PC in the relation between OC, ERI and health behaviours.

Published evidence of organisation interventions to reduce work stress has yielded mixed findings so far; most interventions focused on changing objective work characteristics, rather than personality factors.⁶⁶⁷ My findings showed that personality constructs (OC and PC) can play active and crucial roles in the relationship between work stress and health behaviours, thereby providing potential targets for interventions. Thus, I propose several types of interventions for work stress, personality and health behaviours, including: organisational interventions for health behaviours and the ERI model, individual intervention targeting mechanisms via which OC influences work stress, and stress management intervention for the ERI model. Finally, a multiple–level and integrated approach combining organisational intervention for work stress and individual intervention for vulnerable personality is recommended to improve health behaviours at workplace.

9.1 Associations between Work Stress and Health Behaviours

In this thesis using prospective data, I found that high work stress defined by the ERI model was associated with worse profiles of health behaviours. Based on the following literature, I propose that work stress should be a main target for organisational intervention designed to improve health behaviours; organisational intervention based on the ERI model appears a promising approach to reduce employees' work stress.

9.1.1 Organisational intervention for health behaviours

During the 1980s and 1990s, most interventions for health behaviours were grounded in psychological theories for behaviour change, including perceived control. However, such individual interventions did not address the upstream social contexts influencing health behaviours. Increased recognition that prevention requires efforts beyond the individual level resulted in the development of community interventions at workplace or school. ⁶⁶⁸ Workplace has emerged as an important medium for delivering behaviour change interventions. Health behaviours such as smoking, diet and physical activity have been targeted through organisational interventions. ^{669,670,671} For instance, workplace may offer smoke–free office buildings, smoking cessation classes, facilities for physical activity, and healthy foods in cafeterias.

In the earlier organisational interventions, employees' health behaviours were treated as factors unrelated to occupational hazards. However, evidence showed that occupational hazards influenced employees to adopt and maintain risky health behaviours.⁶⁷² Thus, Sorenson et al tested an integrated intervention to reduce exposure to occupational hazards and to improve health behaviours as opposed to another intervention that only focused on health behaviours; the rate of smoking cessation in the integrated program was twice as high as that in another program.⁶⁷³ In fact, this integrated model addressing both occupational hazards and health behaviours is now the prevailing approach for workplace health promotion.⁶⁷⁴

Since the impacts of works stress (defined by the DC model and the ERI model) on health behaviours have been repeatedly reported in empirical studies and confirmed in this thesis, organisational interventions for health behaviours should also address the potential occupational exposures – work stressors – in order to increase the opportunity of successful changes in health behaviours.
9.1.2 Organisational intervention for the ERI model

Based on the ERI model, restoring the balance between effort and reward at work is considered the best intervention strategy. Tsutsumi and Kawakami proposed an approach of organisational intervention for the ERI model. In terms of extrinsic effort, interventions can focus on reduction of overtime work and long working time, even distribution of workload and responsibility among employees, and provision of holidays and sufficient rest time. In terms of reward, social skill training can improve supervisors' leadership behaviours in praising employees' good performance and providing support for employees, resulting in increased esteem reward. Introduction of additional benefits for employees such as welfare facilities or recreational facilities can increase non–monetary reward. Provision of vocational training and steps for promotion might ensure employees' sense of job security.⁶⁷⁵

Meta-analyses of organisational interventions to reduce work stress have yielded mixed findings so far.^{676,677} Montano et al reviewed 39 studies of organisational interventions based on various work stress models. The interventions frequently reporting significant effects included: reduction of workloads and rotation schedules, improvement of communication between workers and supervisors, introduction of employee training, or improvement in material conditions; these strategies were quite similar to Tsutsumi and Kawakami's approach based on the ERI model, and they can be adopted to design organisational interventions.⁶⁷⁸ Importantly, there have been several studies reporting that the organisational interventions based on the ERI model

9.2 Potential Role of OC in Relationship between ERI and Health Behaviours

This thesis found that OC and ERI have bi-directional relationship; antecedent role of OC in ERI-outcome relation is significant, but mediator role of OC is not. Antecedent role of OC personality in the relation between work stress and health behaviours appears crucial for clinical practice, but personality is often neglected in interventions to reduce work stress. Based on the following literature, I propose individual intervention targeting cognitive-behaviour mechanisms via which OC may influence work stress.

9.2.1 Cognitive behaviour therapy and personality

Cognitive behaviour therapy (CBT) is one form of psychotherapy that has been empirically tested in many clinical trials for different psychiatric diseases such as depression, anxiety or personality disorders.⁶⁸² The cognitive model describes how people's perceptions or thoughts about situations influence their emotional and behavioural reactions. CBT uses a wide range of techniques to help individuals change cognitive appraisal of stressors and their coping responses. Cognitive restructuring encourages individuals to become aware of negative thoughts or irrational beliefs, to recognize distortion and irrationality in thought processes, and to substitute positive thoughts or rational beliefs. In addition, behavioural techniques are used to challenge specific dysfunctional beliefs or to change coping responses, including behavioural rehearsal, modelling, relaxation training, and time management.

In cognitive psychology, personality is conceptualized as a relatively stable organisation composed of schemas (responsible for the sequence from selecting and synthesizing stimulus to a behavioural response) and modes (network of cognitive, affective, and motivational components that organize response patterns).⁶⁸³ The goal of CBT for personality disorder is to decrease valence of dysfunctional schemas (e.g. irrational beliefs) and to strengthen availability of benevolent schemas (e.g. rational beliefs); the patient gradually reinterprets schemas and modes in a more functional way. However, considerably more time and effort are required to produce changes in personality than depression or anxiety.⁶⁸⁴

Despite extensive support on CBT for personality disorders, there has been no

literature on CBT targeting OC personality itself. Nevertheless, several intervention studies have targeted "cognitive-behaviour mechanisms" via which OC and related personality (Hostility or Type A behaviour) can influence work stress and health (see Section 9.2.2). Although it is not easy to induce strong changes in personality itself by an intervention, it appears rather practical to change individual's specific tendency in cognition and behaviour.

9.2.2 Individual intervention targeting cognitive–behaviour mechanisms via which personality influences work stress

A meta-analysis from 36 intervention studies to reduce work stress found that CBT consistently produced larger effects than other individual interventions.⁶⁸⁵ Several intervention studies have targeted "cognitive–behaviour mechanisms" via which OC and related personality can influence work stress and health. For example, Aust, Peter, and Siegrist conducted a 12–week intervention program in bus drivers; this program included 90–min sessions of self–observation for perception of arousal, relaxation training, management of conflict with supervisors, and coping with anger. The mean OC levels significantly reduced in the intervention group, and the effects persisted after 3 months.⁶⁸⁶ William and William reported that CBT reduced adverse effects of Hostility on stress; hostile people tend to interpret neutral situations as threatening and become angry easily. The program included early recognition of angry feelings, cognitive restructuring for negative thoughts, relaxation training, and communication skill training.⁶⁸⁷ Furthermore, several interventions have targeted similar cognitive–behaviour mechanisms to change negative effects of Type A behaviour on health.^{688,689}

This thesis is probably the first to support antecedent role of OC personality in the relation between work stress (ERI) and health behaviours based on a 2–wave cohort study, thereby providing fundamental basis and further support for practical applications of CBT targeting cognitive–behaviour mechanisms. Based on theories,

future interventions are suggested to target the following "cognitive–behaviour mechanisms" via which OC may influence work stress: (1) Perception: individual's cognitive appraisals of stressful situation – mismatch between effort and reward can be changed by cognitive restructuring. (2) Selection: high OC persons may select themselves into stressful jobs or tasks; unrealistic high goal can be changed by cognitive restructuring, and time management would help. (3) Stressor creation: high OC persons may create real work stressors for themselves by conflicting with colleagues and by anxiety on time pressure; social skill training and relaxation training would help. (4) Reaction: high OC persons might react exaggeratedly to work stressors in their psycho–biological processes and health behaviours; behavioural therapy targeted at coping efforts would be beneficial.^{690,691} To match individual need, the intervention can be designed by evaluating one's personality traits and specific cognitive–behaviour mechanisms before the CBT.

9.3 Potential Role of PC in Relation between OC, ERI, and Health Behaviours

This thesis attempted to link the ERI model with Transactional Model of Stress by demonstrating the relationship between OC (personality), ERI (primary appraisal), PC (secondary appraisal), and health behaviours (coping effort and outcome). By the following review, I suggest that stress management intervention can be applied to the ERI model by their common theoretical platform – Transactional Model of Stress.

9.3.1 Stress management intervention and Transactional Model of Stress

A stress management intervention (SMI) is a program initiated by an organisation that focuses on reducing work stressors or on assisting individuals to minimize negative outcomes of exposure to work stressors.⁶⁹² A SMI can be implemented in the form of individual intervention or organisational intervention. Despite extensive

application in organisational settings, SMIs were criticized by lack of theoretical basis.⁶⁹³ Since the 1990s, SMIs have been highly influenced by Transactional Model of Stress, which provided a theoretical platform for the design of interventions.⁶⁹⁴ Thus, contemporary SMI programs often target three points in stress processes: the intensity of work stressors, the cognitive appraisal of stressful situations, and the ability to cope with stressful situations.⁶⁹⁵

The components of SMI encompass a broad array of treatments. Cognitive behaviour therapy (CBT) – the most effective component in SMI – is intended to change individuals' cognitive appraisal of stressful situations and coping responses. Meditation and relaxation interventions – the most popular components adopted in 69% of SMIs – are designed to reduce employees' adverse reactions to stress by inducing psychological and physiological status opposite to stress reactions.⁶⁹⁶ Time management interventions are designed to help employees manage time when working on multiple tasks. Time management provides skills training in defining one's goals to achieve in a specified time period, prioritizing tasks to ensure that important ones receive attention, self–monitoring, and problem solving.⁶⁹⁷

9.3.2 Stress management intervention for the ERI model

This thesis is probably the first study showing that the ERI model can be integrated well with Transactional Model of Stress; the path analyses demonstrated the relationship between OC (personality), ERI (primary appraisal), PC (secondary appraisal), and health behaviours (coping effort and outcome). Thus, it is plausible that SMI can be applied to design interventions for the ERI model by their common theoretical platform – Transactional Model of Stress.

In fact, one intervention study have adopted the SMI based on the ERI model. Limm et al conducted a randomized controlled trial to test the effectiveness of a SMI based on the ERI model in 174 German managers.⁶⁹⁸ The SMI was conducted by a

group–orientated prevention program, including: (1) to foster awareness of and insight into stress situations at workplace – high effort and low reward; (2) to provide tools to cope with stressful situations such as work overload, social conflicts, negative emotion, or failure at work; (3) individual resources were promoted in group processes. The SMI was found to reduce perceived stress reactivity, sympathetic activation, and ER ratio; these effects persisted for 1 year. Although this SMI targeted primary appraisal and coping effort, other potential pathways in Transactional Model of Stress (e.g. secondary appraisal or personality) were not incorporated. To change potential pathways in Transactional Model of Stress in a more significant way, future research can incorporate more treatment components into SMIs.

Note that the interventions to enhance PC (secondary appraisal) have often been adopted in individual interventions for health behaviours rather than SMIs. These interventions were proposed by Social Cognitive Models, with techniques such as changing existing beliefs, introducing new beliefs, role modelling, or enactive mastery experience.^{699,700} As my findings implied that PC might change causal pathways in Transactional Model of Stress, the intervention to enhance PC can be adopted as a SMI component in future research.⁷⁰¹

9.4 A Multi–Level and Integrated Perspective for Psychosocial Factors at Work and Health Behaviours

I propose several types of organisational and individual interventions for work stress, personality and health behaviours in future research and hopefully in practice if results are favourable. My findings of bidirectional relationship between ERI and OC imply that interventions should focus on both work environments and individuals in order to disrupt the cumulated effects in the reciprocal relationship.

Okechukwu et al in Harvard School of Public Health proposed that it is critical to address the "dualism" of individual versus organisational approaches to intervention

design and delivery; they argued that focusing on impact and reach is more useful than putting individual and organisational approaches against each other.⁷⁰² Individual interventions and organisational interventions are complementary; individual interventions are effective at individual–level outcomes such as health behaviours and health outcomes, but organisational interventions have favorable impacts at organisational–level outcomes such as reducing exposure in working conditions. Thus, LaMontagne et al suggested that superior results would be expected from combining individual and organisational interventions over a single type.⁷⁰³

Mellor et al in the Health and Safety Laboratory UK suggested that tackling the impacts of psychosocial factors at work on health should be considered from a multi– level perspective (interplay between work factors and individual differences); multi– level interventions combining organisational and individual interventions had the strongest effects on health. Given that many employees' diseases are often linked to health behaviours, management of psychosocial factors at work needs to be integrated with health promotion to improve health behaviours; this integration has been recommended by public health authorities like the World Health Organisation.⁷⁰⁴

Based on my evidence, the best strategy for addressing psychosocial factors at work and health behaviours is a multi–level and integrated perspective combining organisational and individual interventions. My opinions for the multi–level interventions are that, first, organisational interventions for work stress and health behaviours can be implemented if organisational resources are available; second, for identified individuals with personality vulnerable to work stress, individual interventions targeting cognitive–behaviour mechanisms or stress management interventions for the ERI model can be adopted according to individual needs.

Chapter 10. Conclusions

This thesis examines the relationship between OC, ERI and health behaviours, together with the relationship between OC, ERI, PC and health behaviours, through a two–wave cohort study for drinking and smoking outcomes (n= 7513) and a cross–sectional study for dietary outcomes (n= 11012) conducted in the middle–aged and older populations in Central and Eastern Europe.

The results of this thesis are summarized as follows. First, higher ER ratio (work stress) was associated with higher levels of drinking (heavy drinking, binge drinking and problem drinking), higher levels of smoking (smoking status and smoking intensity), and less healthy diet (diet quality) after adjustment for covariates. Second, higher OC score was generally associated with higher levels of drinking, higher levels of smoking, and less healthy diet after adjustment for covariates. However, these associations of OC with health behaviours existed but did not reach statistical significance in heavy drinking in women and in smoking intensity in both sexes.

Third, the potential role of OC in the relationship between ERI and health behaviours was examined, including modifying, antecedent, mediator, and direct effect of OC. I found that OC and ERI may have bi–directional relationship, but the effect of OC on ERI was stronger than the other direction in the middle-aged and older populations. Thus, antecedent role of OC in the relationship between ERI and health behaviours was found significant, but mediator role of OC was not. Direct effect of OC on health behaviours was not significant. Finally, OC had no significantly modifying effect in the relationship between ERI and health behaviours.

Fourth, lower PC was associated with higher levels of drinking, higher levels of smoking, and less healthy diet after adjustment for covariates. Fifth, the potential role of PC in the relationship between OC, ERI and health behaviours was assessed based on Transactional Model of Stress. I found that both ERI and PC partially mediated the effects of OC on health behaviours. ERI and PC were negatively associated with each

other in the cross-sectional analyses; it is possible that ERI and PC have bi-directional relationship. Finally, PC had no significantly modifying effect in the relationship between ERI and health behaviours.

Work stress research has a long and rich history of identifying those work factors potentially causing stress, but individual differences have not been paid enough attention. This thesis contributes to deeper understanding of intersecting pathways by which work stress (ERI) and personality constructs (OC and PC) jointly influence health behaviours. The ERI model can be integrated well with Transactional Model of Stress, which provides potential mechanisms to explain individual differences (personality, cognitive appraisal, and coping). Investigating psychosocial mechanisms may help to identify a broad set of intervention opportunities; I propose that the next steps are to develop, implement and evaluate several types of interventions for work stress, personality and health behaviours in order to translate my findings into practice. To further clarify the relationship between work stress, personality and health behaviours, future research should recruit samples from wider cultural bases and younger populations with at least three waves of data.

Reference

¹ Yusuf S, Hawken S, Ounpuu S, et al. Effect of potentially modifiable risk factors associated with myocardial infarction in 52 countries (the INTERHEART study): case-control study. Lancet. 2004; 364: 937-952.

² Hemingway H, Marmot M. Psychosocial factors in the etiology and prognosis of coronary heart disease: systematic review of prospective cohort studies. BMJ. 1999; 318: 1460-1467.
 ³ Steptoe A, O'Donnell K, Marmot M, Wardle J. Positive affect and psychosocial processes related to health. Br J Psychol. 2008; 99: 211-227.

⁴ Lazarus RS, Cohen JB. Environmental stress. In: Altman I, Wohlwill JF, editors. Human behavior and environment. New York: Plenum; 1977.

⁵ Siegrist J, Rödel A. Work stress and health risk behavior. Scand J Work Environ Health. 2006; 32(6): 473-481.

⁶ Van Vegchel N, De Jonge J, Bosma H, Schaufeli W. Reviewing the effort-reward imbalance model: Drawing up the balance of 45 empirical studies. Soc Sci Med. 2005; 60: 1117-1131.

⁷ Siegrist J, Peter R, Junge A, Cremer P, Seidel D. Low status control, high effort at work and ischemic heart disease: prospective evidence from blue-collar men. Soc Sci Med. 1990; 31: 1127-1134.

⁸ Carver CS, Scheier MF. Perspectives on Personality, 6th ed. Boston: Pearson; 2008.

⁹ Rhodes RE, Smith NEI. Personality correlates of physical activity: a review and metaanalysis. Br J Sports Med. 2006; 40: 958-965.

¹⁰ Bogg T, Roberts BW. Conscientiousness and health-related behaviours: A meta-analysis of the leading behavioural contributors to mortality. Psychol Bull. 2004; 130(6): 887-919.

¹¹ Rosenman RH. Current and past history of type A behavior pattern. In: Schmidt TH, Dembroski T, Blumchen G, editors. Biological and psychological factors in cardiovascular disease. Berlin: Springer Verlag; 1986.

¹² Connor-Smith JK, Flachsbart C. Relation between personality and coping: a metaanalysis. J Pers Soc Psychol. 2007; 93(6): 1080–1107.

¹³ Siegler IC, Peterson BL, Barefoot JC, Williams RB. Hostility during late adolescence predicts coronary risk factors at mid-life. Am J Epidemiol. 1992; 136: 146-154.

¹⁴ Munafò MR, Black S. Personality and smoking status: a longitudinal analysis. Nicotine Tob Res. 2007; 9(3): 397-404.

¹⁵ Siegrist J. Effort-reward imbalance at work and health. In: Perrewe PL and Ganster DC, editors. Historical and current perspectives on stress and health. Amsterdam: Elsevier; 2002. p. 261-291.

¹⁶ Israel BA, Baker EA, Goldenhar LM, Heaney CA, Schurman SJ. Occupational stress, safety, and health: conceptual framework and principles for effective prevention interventions. J Occup Health Psychol. 1996; 1: 261-286.

¹⁷ Greenberger DB, Strasser S. Development and application of a model of personal control in organisations. Acad Manage Rev. 1986; 11: 164-177.

¹⁸ Smith TW, MacKenzie J. Personality and risk of physical illness. Annu Rev Clin Psychol. 2006; 2: 435-467.

¹⁹ Berkman LF, Syme SL. Social networks, host resistance, and mortality: a nine-year followup study of Alameda County residents. Am J Epidemiol. 1979; 109: 186-204.

²⁰ Marmot M, Smith GD, Stansfeld S, et al. Health Inequalities among British Civil Servants: The Whitehall II Study. Lancet. 1991; 337: 1387-1393.

²¹ Conner M, Norman P. Predicting health behaviour. Maidenhead: Open University Press;1996.

²² Gochman DS. Handbook of health behavior research I: personal and social determinants. New York: Plenum Press; 1997.

²³ Cockerham WC. Medical sociology. 7th ed. Upper Saddle River, NJ: Prentice Hall; 1998.

²⁴ Baum A, Posluszny DM. Health psychology: mapping bio–behavioral contributions to health and illness. Annu Rev Psychol. 1999; 50: 137-163.

²⁵ Britton A, Marmot M. Different measures of alcohol consumption and risk of coronary heart disease and all-cause mortality: 11-year follow-up of the Whitehall II Cohort Study. Addiction. 2004; 99(1): 109-116.

²⁶ Yach D, Hawkes C, Gould CL, Hoffman KJ. The global burden of chronic diseases: overcoming impediments to prevention and control. JAMA. 2004; 291(21): 2616-2622.

²⁷ Ferreira MP, Willoughby D. Alcohol consumption: the good, the bad, and the indifferent. Appl Physiol Nutr Metab. 2008; 33:12-20.

²⁸ Rehm J, Mathers C, Popova S, Thavorncharoensap M, Teerawattananon Y, Patra J. Global burden of disease and injury and economic cost attributable to alcohol use and alcohol-use disorders. Lancet. 2009; 373: 2223-2233.

²⁹ Miller P, Plant M, Plant M. Spreading out or concentrating weekly consumption: alcohol problems and other consequences within a UK population sample. Alcohol Alcohol. 2005; 40: 461-468.

³⁰ Rehm J, Sempos CT. Alcohol consumption and all–cause mortality. Addiction. 1995; 90: 471-480.

³¹ Vineis P, Alavanja M, Buffler P, Fontham E, Franceschi S. Tobacco and cancer: recent epidemiological evidence. J Natl Cancer Inst. 2004; 96: 99-106.

³² Forey BA, Thornton AJ, Lee PN. Systematic review with meta-analysis of the epidemiological evidence relating smoking to COPD, chronic bronchitis and emphysema.
 BMC Pulm Med. 2011; 11: 36.

³³ Doll R, Peto R, Boreham J, Sutherland I. Mortality in relation to smoking: 50 years' observations on male British doctors. BMJ. 2004; 328: 15-19.

³⁴ Peto R, Lopez AD, Boreham J, Thun M, Heath C, Doll R. Mortality from smoking worldwide. Br Med Bull. 1996; 52: 12-21.

³⁵ Jankovic N, Geelen A, Streppel MT, et al. Adherence to a healthy diet according to the World Health Organization guidelines and all-cause mortality in elderly adults from Europe and the United States. Am J Epidemiol. 2014; 180: 978-988. ³⁶ Stefler D, Pikhart H, Jankovic N, et al. Healthy diet indicator and mortality in Eastern
European populations: prospective evidence from the HAPIEE cohort. Eur J Clin Nutr. 2014;
68: 1346-1352.

³⁷ Ness AR, Powles JW. Fruit and vegetables, and cardiovascular disease: a review. Int J Epidemiol. 1997; 26: 1-13.

³⁸ Luepker RV, Lakshminarayan K. Cardiovascular and cerebrovascular diseases. In: Detels R, Beaglehole R, Lansang MA, Gulliford, editors. Oxford textbook of public health. Oxford: Oxford University Press; 2009. p. 971-996.

³⁹ Green LW, Hiatt RA. Behavioural determinants of health and disease. In: Detels R, Beaglehole R, Lansang MA, Gulliford, editors. Oxford textbook of public health. Oxford: Oxford University Press; 2009. p. 120-136.

⁴⁰ World Health Organization. Global recommendations on physical activity for health. Geneva, Switzerland: WHO Press, 2010.

⁴¹ James WP. The epidemiology of obesity: the size of the problem. J Intern Med. 2008; 263: 336-352.

⁴² Warburton DE, Katzmarzyk PT, Rhodes RE, Shephard RJ. Evidence informed physical activity guidelines for Canadian adults. Can J Public Health. 2007; 98: 16–68.

⁴³ Marmot M, Bobak M. International comparators and poverty and health in Europe. BMJ.2000; 321: 1124–1128.

⁴⁴ World Health Organisation (WHO) Regional Office for Europe. European Health for all Database. Copenhagen: WHO Regional Office for Europe; 2011.

⁴⁵ Bobak M, Marmot M. East-west mortality divide and its potential explanations: proposed research agenda. BMJ. 1996; 312: 421-425.

⁴⁶ Wilkinson RG. Unhealthy societies: the afflictions of inequality. London: Routledge; 1996.
 ⁴⁷ Shkolnikov V, Leon DA, Adamets S, Andreev E, Deev A. Educational level and adult

mortality in Russia: an analysis of routine data 1979-94. Soc Sci Med. 1998; 47: 357-369.

⁴⁸ Marmot MG, Davey Smith G, Stansfeld SA, et al. Health inequalities among British civil servants: the Whitehall II study. Lancet. 1991; 337: 1387-1393.

⁴⁹ McKee M, Shkolnikov V. Understanding the toll of premature death among men in eastern Europe. BMJ. 2001; 323: 1051-1055.

⁵⁰ Mackenbach JP, Stirbu I, Roskam AR, et al. Socioeconomic Inequalities in Health in 22 European Countries. N Engl J Med. 2008; 358: 2468-2481.

⁵¹ Eberstadt N. Health and mortality in Eastern Europe, 1965-1985. Communist Econ. 1990;2: 349-365.

⁵² Boniol M, Autier P. Prevalence of main cancer lifestyle risk factors in Europe in 2000. Eur J Cancer. 2010; 46: 2534-2544.

⁵³ Perlman F, Bobak M, Gilmore A, McKee M. Trends in the prevalence of smoking in Russia during the transition to a market economy. Tob Control. 2007; 16(5): 299-305.

⁵⁴ Nemtsov AV. Alcohol-related human losses in Russia in the 1980s and 1990s. Addiction. 2002; 97: 1413-1425.

⁵⁵ Leon DA, Shkolnikov VM. Social stress and the Russian mortality crisis. JAMA. 1998; 279: 790-791.

⁵⁶ Malyutina S, Bobak M, Kurilovitch S, et al. Binge drinking and cardiovascular mortality: a prospective cohort study. Lancet. 2000; 360: 1448-1454.

 ⁵⁷ Leon DA, Saburova L, Tomkins S, et al. Hazardous alcohol drinking and premature mortality in Russia: a population based case-control study. Lancet. 2007; 369: 2001-2009.
 ⁵⁸ Bobak M, Room R, Pikhart H, et al. Contribution of drinking patterns to differences in rates of alcohol related problems between three urban populations. J Epidemiol Community Health. 2004; 58: 238-242.

⁵⁹ Varvasovsky Z, Bain C, McKee M. Deaths from cirrhosis in Poland and Hungary: the impact of different alcohol policies during the 1990s. J Epidemiol Community Health. 1996; 51: 167-171.

⁶⁰ Kubicka L, Csemy L, Duplinsky J, Kozeny J. Czech men's drinking in changing politic climate 1983-93: a three-wave longitudinal study. Addiction. 1998; 93: 1219-1230.

⁶¹ Peto R, Lopez AD, Boreham J, et al. Mortality from smoking in developed countries 1950– 2000. Oxford: Oxford University Press; 1994.

⁶² Kubik AK, Parkin DM, Plesko I, et al. Patterns of cigarette sales and lung cancer mortality in some Central and Eastern European countries, 1960-1989. Cancer. 1995; 75: 2452-2560.
⁶³ Brennan P, Crispo A, Zaridze D, et al. High cumulative risk of lung cancer death among smokers and non-smokers in Central and Eastern Europe. Am J Epidemiol. 2006; 164(12):

1233-1241.

⁶⁴ Bobak M, Gilmore A, McKee M, Rose R, Marmot M. Changes in smoking prevalence in Russia, 1996– 2004. Tob Control. 2006; 15: 131-135.

⁶⁵ Perlman F, Bobak M. Socioeconomic and behavioral determinants of mortality in posttransition Russia: A prospective population study. Ann Epidemiol. 2008; 18: 92–100.

⁶⁶ Skodova ZR, Poledne PL, Berka Z, et al. Decline of the cardiovascular mortality in Czech Republic in 1984-1993 and its possible causes. Casopis Lekaru Ceskych. 1997; 136: 373-379.

⁶⁷ Forey B, Hamling J, Lee P, et al. International smoking statistics, 2nd ed. Oxford: Oxford University Press, 2002.

⁶⁸ Robertson A, Brunner E, Sheiham A. Food is a political issue. In: Marmot M, Wilkinson RG, editors. Social determinants of health. 2nd ed. New York: Oxford University Press; 2006.
⁶⁹ Boylan S, Welch A, Pikhart H, et al. Dietary habits in three Central and Eastern European countries: the HAPIEE study. BMC Public Health. 2009; 9: 439-452.

⁷⁰ Boylan S, Lallukka T, Lahelma E, et al. Socio-economic circumstances and food habits in Eastern, Central and Western European populations. Public Health Nutr. 2011; 14(4): 678-687.

⁷¹ Shapiro J. The Russian mortality crisis and its causes. In: Aslund A, editor. Russian economic reform at risk. London: Printer; 1995. p. 149-178.

⁷² Zatonski WA, McMichael AJ, Powles JW. Ecological study of reasons for sharp decline in

mortality from ischaemic heart disease in Poland since 1991. BMJ. 1998; 316: 1047-1051. ⁷³ Bobak M, Skodova Z, Pisa Z, Poledne R, Marmot M. Political changes and trends in cardiovascular risk factors in the Czech Republic, 1985–92. J Epidemiol Community Health. 1997; 51: 272-277.

⁷⁴ Steptoe A, Wardle J. Health behaviour, risk awareness and emotional well-being in students from Eastern Europe and Western Europe. Soc Sci Med. 2001; 53: 1621-1630.
 ⁷⁵ Palosuo H. Health-related lifestyles and alienation in Moscow and Helsinki. Soc Sci Med. 2000; 51: 1325-1341.

⁷⁶ Field MG. The health crisis in the Former Soviet Union: A report from the post-war zone. Soc Sci Med. 1995; 41: 1469-1478.

⁷⁷ National Center for Health System Management. Health care system in transition, profile of Poland. Warsaw: Ministry of Health and Social Welfare; 1996.

⁷⁸ Institute of Health Information and Statistics of the Czech Republic. Sample survey of the health status of the Czech population. Praque: Institute of Health Information and Statistics; 1995.

⁷⁹ Cockerham WC. Health and social changes in Russia and Eastern Europe. UK: Routledge; 1999.

⁸⁰ Weber M. Economy and society. Berkeley: University of California Press; 1978.

⁸¹ Bourdieu P. The logic of practice. Stanford: Stanford University Press; 1990.

⁸² Hurt RD. Smoking in Russia: What do Stalin and Western tobacco companies have in common? Mayo Clin Proc. 1995; 70: 1007-1011.

⁸³ Cockerham WC. Health lifestyles in Russia. Soc Sci Med. 2000; 51: 1313-1324.

⁸⁴ Hemingway H, Marmot M. Psychosocial factors in the etiology and prognosis of coronary heart disease: systematic review of prospective cohort studies. BMJ. 1999; 318: 1460–1467.
 ⁸⁵ Martikainen P, Bartley M, Lahelma E. Psychosocial determinant of health in social epidemiology. Int J Epidemiol. 2002; 31: 1091-1093.

⁸⁶ Kuper H, Marmot M, Hemingway H. Systematic review of prospective cohort studies of psycho- social factors in the aetiology and prognosis of coronary heart disease. In: Elliott P, Marmot M, editors. Coronary heart disease epidemiology. 2nd ed. Oxford: Oxford University Press; 2005. p. 363-413.

⁸⁷ Marmot M, Smith GD, Stansfeld S, et al. Health inequalities among British civil servants: the Whitehall II study. Lancet. 1991; 337: 1387-1393.

⁸⁸ Adler NE, Marmot M, McEwen BS, Stewart J. Socioeconomic status and health in industrial nations: social, psychological and biological pathways. New York: New York Academy of Sciences; 1999.

⁸⁹ Bartley M. Health Inequality: an introduction to theories, concepts and methods. Cambridge: Polity Press; 2005.

⁹⁰ World Health Organisation (WHO) Commission on Social Determinants of Health. Closing the gap in a generation: health equity through action on the social determinants of health. Geneva: World Health Organisation; 2008.

⁹¹ Steptoe A. Psychobiological processes linking socio-economic position with health. In: Siegrist J, Marmot M, editors. Social inequalities in health. New York: Oxford University Press; 2006. p. 101-126.

⁹² Kristenson M. Socio-economic differences in health: the role of coping. In: Siegrist J, Marmot M, editors. Social inequalities in health. New York: Oxford University Press; 2006. p. 153-166.

⁹³ Steptoe A, O'Donnell K, Marmot M, Wardle J. Positive affect and psychosocial processes related to health. Br J Psychol. 2008; 99: 211-227.

⁹⁴ Feldman PJ, Steptoe A. How neighborhoods and physical functioning are related: the roles of neighborhood socioeconomic status, perceived neighborhood strain, and individual health risk factors. Ann Behav Med. 2004; 27: 91-99.

⁹⁵ Vitaliano PP, Zhang J, Scanlan JM. Is caregiving hazardous to one's physical health? A meta-analysis. Psychol Bull. 2003; 129: 946-972.

⁹⁶ Leserman J, Petitto JM, Golden RN, et al. Impact of stressful life events, depression, social support, coping, and cortisol on progression to AIDS. Am J Psychiatry. 2000; 157: 1221-1228.

⁹⁷ Kivimäki M, Nyberg ST, Batty GD, et al. Job strain as a risk factor for coronary heart disease: a collaborative meta-analysis of individual participant data. Lancet. 2012; 380: 1491-1497.

⁹⁸ Siegrist J. Effort-reward imbalance at work and cardiovascular diseases. Int J Occup Med Environ Health. 2010; 23(3): 1-7.

⁹⁹ Smith TW, MacKenzie J. Personality and risk of physical illness. Annu Rev Clin Psychol. 2006; 2: 435-467.

 ¹⁰⁰ Lahey BB. Public health significance of neuroticism. Am Psychol. 2009; 64: 241-256.
 ¹⁰¹ Everson-Rose SA, Lewis TT. Psychosocial factors and cardiovascular diseases. Annu Rev Public Health. 2005; 26: 469-500.

¹⁰² Pressman SD, Cohen S. Does positive affect influence health? Psychol Bull. 2005; 131: 925-971.

¹⁰³ Kawachi I, Kennedy BP, Lochner K, Prothwow SD. Social capital, income inequality, and mortality. Am J Public Health. 1997; 87(9): 1491-1498.

¹⁰⁴ Berkman LF, Glass TA. Social integration, social networks, social support and health. In: Berkman LF, Kawachi I, editors. Social epidemiology. New York: Oxford University Press; 2000.

¹⁰⁵ Everson-Rose SA, Clark CJ. Assessment of psychosocial factors in population studies. In: Steptoe A, editor. Handbook of behavioral medicine: methods and applications. New York: Springer; 2010. p. 291-306.

¹⁰⁶ Bobak M, Marmot M. East-west mortality divide and its potential explanations: proposed research agenda. BMJ. 1996; 312: 421-425.

¹⁰⁷ McKee M, Shkolnikov V. Understanding the toll of premature death among men in eastern Europe. BMJ. 2001; 323: 1051-1055.

¹⁰⁸ Kopp MS, Skrabski A, Szedmak S. Psychological risk factors, inequality and self-rated morbidity in a changing society. Soc Sci Med. 2000; 51: 1350-1361.

¹⁰⁹ Marmot M, Bobak M. Psychosocial and biological mechanisms behind the recent mortality crisis in Central and Eastern Europe. In: Cornia AG, Panicci R, editors. The mortality crisis in transitional economies. Oxford: Oxford University Press; 2001.

¹¹⁰ Pollert A. Transformation at work in the new market economies of Central Eastern Europe. London: Sage publications; 1999.

¹¹¹ Kopp MS, Stauder A, Purebl G, Janszky I, Skrabski A. Work stress and mental health in a changing society. Eur J Pub Health. 2007; 18(3): 238–244.

¹¹² Havlik P. Structural change, productivity and employment in the New EU Member States. Vienna: Vienna Institute for International Economic Studies; 2005.

¹¹³ Benach J, Muntaner C, Chung H, et al. Reducing the health inequalities associated with employment conditions. BMJ. 2010; 26: 1392-1395.

¹¹⁴ Marin D. A new international division of labor in Europe: outsourcing and offshoring to Eastern Europe. J Eur Econ Assoc. 2006; 2(3): 612-622.

¹¹⁵ Chen SW, Wang PC, Hsin PL, et al. Job stress models, depressive disorders and work performance of engineers in microelectronics industry. Int Arch Occup Environ Health. 2011;
84: 91-103.

¹¹⁶ Bormann S, Plank L. Under pressure: working conditions and economic development in ICT production in Central and Eastern Europe. Berlin: World Economy, Ecology and Development; 2010.

¹¹⁷ Weidner G, Cain VS. The gender gap in heart disease: lessons from Eastern Europe. Am J Public Health. 2003; 93: 768-770.

¹¹⁸ Bobak M, Hertzman C, Skodova Z, et al. Association between psychosocial factors at work and non-fatal myocardial infarction in a population based case-control study in Czech men. Epidemiology. 1998; 9: 43-47.

¹¹⁹ Kopp MS, Skrabski A, Szanto Z, Siegrist J. Psychosocial determinants of premature cardiovascular mortality differences within Hungary. J Epidemiol Commun Health. 2006; 60: 782-788.

¹²⁰ Pikhart H, Bobak M, Siegrist J, et al. Psychosocial work characteristics and self-rated health in four post-communist countries. J Epidemiol Commun Health. 2001; 55: 624-630.
 ¹²¹ Laszlo KD, Gyorffy Z, Adam S, et al. Work-related stress factors and menstrual pain: a

nation-wide representative survey. J Psychosom Obstet Gynaecol. 2008; 29: 133-138.

¹²² Pikhart H, Bobak M, Pajak A, et al. Psychosocial factors at work and depression in three countries of Central and Eastern Europe. Soc Sci Med. 2004; 58: 1475-1482.

¹²³ Bobak, Pikhart, Kubinova, et al. The association between psychosocial characteristics at work and problem drinking: a cross- sectional study of men in three Eastern European urban populations. Occup Environ Med. 2005; 62: 546-550.

¹²⁴ Salavecz G, Chandola T, Pikhart H, et al. Work stress and health in Western European and post-communst countries: an East-West comparison study. J Epidemiol Community Health. 2010; 64: 57-62.

¹²⁵ Laszlo KD, Pikhart H, Kopp MS, et al. Job insecurity and health: a study of 16 European countries. Soc Sci Med. 2010; 70: 867-874.

¹²⁶ Cockerham WC. Health lifestyles in Russia. Soc Sci Med. 2000; 51: 1313-1324.

¹²⁷ Skinner EA, Chapman M, Baltes PB. Control, means-ends, and agency beliefs: A new conceptualization and its measurement during childhood. J Pers Soc Psychol. 1988; 54: 117-133.

¹²⁸ Rose R. Understanding post-communist transformation: a bottom up approach. New York: Routledge; 2009.

¹²⁹ Bobak M, Pikhart H, Rose R, Hertzman C, Marmot M. Socioeconomic factors, material inequalities, and perceived control in self-rated health: cross-sectional data from seven post-communist countries. Soc Sci Med. 2000; 51: 1343-1350.

¹³⁰ Lundberg J , Bobak M , Malyutina S , Kristenson M, Pikhart H. Adverse health effects of low levels of perceived control in Swedish and Russian community samples. BMC Public Health. 2007; 7: 314.

¹³¹ Carlson P. Self-perceived health in East and West Europe. Another European health divide. Soc Sci Med. 1998; 46: 1355-1366.

¹³² Pikhart H. Social and psychosocial determinants of self-rated health in Central and Eastern Europe. Boston: Kluwer Academic Publishers; 2002.

¹³³ Institute of Medicine. Promoting health: intervention strategies from social and behavioral research. Washington: National Academy Press; 2000.

¹³⁴ Bandura A. Social foundations of thought and action: a social cognitive theory. Englewood Cliffs, NJ: Prentice Hall; 1986.

¹³⁵ Sorensen G, Emmons K, Hunt MK, et al. Model for incorporating social context in health behavior interventions: applications for cancer prevention for working-class, multiethnic populations. Prev Med. 2003; 37: 188-197.

¹³⁶ US Department of Health and Human Services. Healthy people 2010. Washington DC: US Department of Health and Human Services; 2000.

¹³⁷ World Health Organisation. Global strategy on diet, physical activity, and health. Geneva: World Health Organisation; 2004.

¹³⁸ McLeroy KR, Bibeau D, Steckler A, Glanz K. An ecological perspective on health promotion programs. Health Educ Quart. 1988; 15: 351-377.

¹³⁹ Glanz K, Rimer BK, Viswanath K. Health behavior and health education: theory, research, and practice, 4th ed. San Francisco: John Wiley Sons; 2008.

¹⁴⁰ Brunner E, Marmot M. Social organisation, stress, and health. In: Marmot M, Wilkinson RG, editors. Social Determinants of Health. New York: Oxford University Press; 1999. p. 17-43.

¹⁴¹ Conner M, Norman P. Predicting health behaviour: research and practice with social cognition models. 2nd ed. Maidenhead: Open University Press; 2005.

¹⁴² Carver CS, Scheier MF. Perspectives on Personality, 6th ed. Boston: Pearson; 2008.

¹⁴³ Bradley. Work-induced changes in feelings of mastery. J Psychol. 2010; 144(2): 97-119.
¹⁴⁴ Bailis DS, Segall A, Mahon MJ, Chipperfield JG, Dunn EM. Perceived control in relation to socioeconomic and behavioral resources for health. Soc Sci Med. 2001; 52: 1661-1676.
¹⁴⁵ Smith TW, MacKenzie J. Personality and risk of physical illness. Annu Rev Clin Psychol. 2006; 2:435-467.

¹⁴⁶ McNeill LH. Individual, social, environmental, and physical environmental influences on physical activity among black and white adults: a structural equation analysis. Ann Behav Med. 2006; 31: 36-44.

¹⁴⁷ Siegrist J, Theorell T. Socio-economic position and health: the role of work and employment. In: Siegrist J, Marmot M, editors. Social inequalities in health. New York: Oxford University Press; 2006.

¹⁴⁸ Wainwright D, Calnan M. Work stress: the making of a modern epidemic. Milton Keynes: Open University Press; 2002.

¹⁴⁹ Siegrist J. Effort-reward imbalance at work and cardiovascular diseases. Int J Occup Med Environ Health. 2010; 23(3): 1-7.

¹⁵⁰ Karasek RA, Theorell T. Healthy work: Stress, productivity and the reconstruction of working life. New York: Basic Books; 1990.

¹⁵¹ Selye H. The stress of life. New York: McGraw-Hill; 1936.

¹⁵² Hinkle LE, Whitney LH, Lehman EW, et al. Occupation, education, and coronary heart disease. Science. 1968; 161: 238-248.

¹⁵³ Kahn R. Conflict, ambiguity, and overload: three elements in job stress. In: McLean A, editor. Occupational stress. Springfield, Charles Thomas; 1974.

¹⁵⁴ Johnson J, Hall E. Job strain, workplace social support, and cardiovascular disease: a cross sectional study of a random sample of the Swedish working population. Am J Public Health. 1988; 78: 1336-1342.

¹⁵⁵ Siegrist J, Peter R, Junge A, Cremer P, Seidel D. Low status control, high effort at work and ischemic heart disease: prospective evidence from blue-collar men. Soc Sci Med. 1990; 31(10): 1127-1134.

¹⁵⁶ Siegrist J. Place, social exchange and health: proposed sociological framework. Soc Sci Med. 2000; 51: 1283-1293.

¹⁵⁷ Siegrist J. Adverse health effects of high effort–low reward conditions at work. J Occup Health Psychol. 1996; 1:27-43.

¹⁵⁸ Siegrist J. Effort–reward Imbalance at work and health. In: Perrewe P, Ganster D, editors. Research in occupational stress and well-being. New York: Elsevier; 2002. p. 261-291.

¹⁵⁹ Johnson J. Globalization, workers' power and the psychosocial work environment—is the demand-control-support model still useful in a neoliberal era? Scand J Work Environ Health. 2008; 6: 15-21.

¹⁶⁰ Wainwright D, Calnan M. Work stress: the making of a modern epidemic. Milton Keynes: Open University Press; 2002.

¹⁶¹ De Jonge J, Bosma H, Peter R, Siegrist J. Job strain, effort-reward imbalance and

employee well-being: a large scale cross-sectional study. Soc Sci Med. 2000; 50: 1317-1327. ¹⁶² Calnan M, Wadsworth E, May M, Smith A, Wainwright D. Job strain, effort–reward imbalance, and stress at work: competing or complementary models? Scand J Public Health. 2004; 32(2): 84-93.

¹⁶³ Ostry AS, Kelly S, Demers PA, Mustard C, Hertzman C. A comparison between the effort-reward imbalance and demand control models. BMC Public Health. 2003; 3: 10-19.
 ¹⁶⁴ Stansfeld SA, Clark C, Caldwell T, Rodgers B, Power C. Psychosocial work characteristics and anxiety and depressive disorders in midlife: the effects of prior psychological distress. Occup Environ Med. 2008; 65(9): 634-642.

¹⁶⁵ Head J, Kivimäki M, Siegrist J, et al. Effort-reward imbalance and relational injustice at work predicts sickness absence: the Whitehall II Study. J Psychosom Res. 2007; 63: 433-440.

¹⁶⁶ Kuper H, Singh-Manoux A, Siegrist J, Marmot M. When reciprocity fails: Effort-reward imbalance in relation to coronary heart disease and health functioning within the Whitehall II study. Occup Environ Med. 2002; 59: 777-784.

¹⁶⁷ Kivimäki M, Nyberg ST, Batty GD, et al. Job strain as a risk factor for coronary heart disease: a collaborative meta-analysis of individual participant data. Lancet. 2012; 380: 1491-1497.

¹⁶⁸ Stansfeld S, Candy B. Psychosocial work environment and mental health – A metaanalytic review. Scand J Work Environ Health. 2006; 32: 443-462.

¹⁶⁹ Van Vegchel N, De Jonge J, Bosma H, Schaufeli W. Reviewing the effort-reward imbalance model: Drawing up the balance of 45 empirical studies. Soc Sci Med. 2005; 60: 1117-1131.

¹⁷⁰ Siegrist J, Rödel A. Work stress and health risk behavior. Scand J Work Environ Health. 2006; 32(6): 473-481.

¹⁷¹ Heikkilä K, Fransson EI, Nyberg ST, et al. Job strain and health-related lifestyle: findings from an individual-participant meta-analysis of 118 000 working adults. Am J Public Health. 2013; 103(11): 2090-2097.

¹⁷² Kouvonen A, Kivimaki M, Vaananen A. Job strain and adverse health behaviors: The Finnish Public Sector Study. J Occup Environ Med. 2007; 49: 68-74.

¹⁷³ Kouvonen A, Kivimäki M, Virtanen M, et al. Effort-reward imbalance at work and the cooccurrence of lifestyle risk factors: cross-sectional survey in a sample of 36,127 public sector employees. BMC Public Health. 2006; 6: 24-35.

¹⁷⁴ Blum K, Cull JG, Braverman ER, et al. Reward deficiency syndrome: addictive, impulsive and compulsive disorders may have a common genetic basis. American Scientist. 1996; 84: 132-145.

¹⁷⁵ Newman E, O'Connor DB, Conner M. Daily hassles and eating behaviour: the role of cortisol reactivity status. Psychoneuroendocrinology. 2007; 32: 125-132.

¹⁷⁶ Chen MJ, Cunradi C. Job stress, burnout and substance use among urban transit operators: The potential mediating role of coping behaviour. Work Stress. 2008; 22(4): 327-

¹⁷⁷ Heikkilä K, Nyberg ST, Fransson EI, et al. Job strain and alcohol intake: a collaborative meta-analysis of individual–participant data from 140 000 men and women. PLoS ONE. 2012; 7(7): e40101.

¹⁷⁸ Siegrist J, Rödel A. Work stress and health risk behavior. Scand J Work Environ Health. 2006; 32(6): 473-481.

¹⁷⁹ Crum RM, Muntaner C, Eaton WW, Anthony JC. Occupational stress and the risk of alcohol abuse and dependence. Alcohol Clin Exp Res. 1995; 19: 647-655.

¹⁸⁰ Marchand A, Blanc ME. Occupation, work organisation conditions, and alcohol misuse in Canada: an 8-year longitudinal study. Subst Use Misuse. 2011; 46: 1003-1014.

¹⁸¹ Gimeno D, Amick BC, Barrientos-Gutiérrez T. Work organisation and drinking: an epidemiological comparison of two psychosocial work exposure models. Int Arch Occup Environ Health. 2009; 82: 305-317.

¹⁸² Niedhammer I, Goldberg M, Leclerc A, David S, Bugel I, Landre MF. Psychosocial work environment and cardiovascular risk factors in an occupational cohort in France. J Epidemiol Community Health. 1998; 52: 93-100.

¹⁸³ Head J, Stansfeld SA, Siegrist J. The psychosocial work environment and alcohol dependence: a prospective study. Occup Environ Med. 2004; 61: 219-224.

¹⁸⁴ Puls W, Winold H, Blank T. The influence of effort-reward imbalance in the workplace on the consumption of alcohol: a written survey carried out in metal-working companies. Sucht. 1998; 44: 183-199.

¹⁸⁵ Bobak, Pikhart, Kubinova, et al. The association between psychosocial characteristics at work and problem drinking: a cross- sectional study of men in three Eastern European urban populations. Occup Environ Med. 2005; 62: 546-550.

¹⁸⁶ Albertsen K, Borg V, Oldenburg B. A systematic review of the impact of work environment on smoking cessation, relapse and amount smoked. Prev Med. 2006; 43: 291-305.

¹⁸⁷ Heikkilä K, Nyberg ST, Fransson EI, et al. Job strain and tobacco smoking: An individualparticipant data meta-Analysis of 166 130 adults in 15 European studies. PLoS ONE. 2012; 7(7): e35463.

¹⁸⁸ Kouvonen A, Kivimäki M, Virtanen M, Pentti J, Vahtera J. Work stress, smoking status, and smoking intensity: an observational study of 46 190 employees. J Epidemiol Community Health. 2005; 59: 63-69.

¹⁸⁹ Peter R, Siegrist J, Stork J, Mann H, Labrot B. Cigarette smoking and psychosocial work stress in middle-management employees. Soz Präv Med. 1991; 36: 315-321.

¹⁹⁰ Radi S, Ostry A, LaMontagne AD. Job stress and other working conditions: relationships with smoking behaviors in a representative sample of working Australians. Am J Ind Med. 2007; 50: 584-596.

¹⁹¹ Ota A, Masue T, Yasuda N, et al. Psychosocial job characteristics and smoking cessation: a prospective cohort study using the Demand-Control-Support and Effort-Reward Imbalance job stress models. Nicotine Tob Res. 2010; 12(3): 287-293.

^{340.}

¹⁹² Mcintosh WA, Shifflett PA, Picou JS. Social support, stressful events, strain, dietary intake, and the elderly. Med Care. 1989; 27: 140-153.

¹⁹³ Pollard TM, Steptoe A, Canaan L, Davies GJ, Wardle J. Effects of academic examination stress on eating behavior and blood lipid levels. Int J Behav Med. 1995; 2: 299-320.
 ¹⁹⁴ Hellerstedt WL, Jeffery RW. The association of job strain and health behaviours in men and women. Int J Epidemiol. 1997; 26: 575-583.

¹⁹⁵ Lallukka T, Lahelma E, Rahkonen O, et al. Associations of job strain and working overtime with adverse health behaviors and obesity: Evidence from the Whitehall II Study, Helsinki Health Study, and the Japanese Civil Servants Study. Soc Sci Med. 2008; 66: 1681-1698.
¹⁹⁶ Lallukka T, Sarlio-Lähteenkorva S, Roos E, et al. Working conditions and health behaviours among employed women and men: the Helsinki Health Study. Prev Med. 2004; 38(1): 48-56.

¹⁹⁷ Raulio S, Roos E, Mukala K, Prättälä R. Can working conditions explain differences in eating patterns during working hours? Public Health Nutr. 2008; 11(3): 258-270.

¹⁹⁸ Kawakami N, Tsutsumi A, Haratani T, et al. Job strain, worksite support, and nutrient intake among employed Japanese men and women. J Epidemiol. 2006; 16(2): 79-89.
¹⁹⁹ Tsutsumi A, Kayaba K, Yoshimura M, et al. Association between job characteristics and health behaviours in Japanese rural workers. Int J Behav Med. 2003; 10: 125-142.
²⁰⁰ Nomura K, Nakao M, Tsurugano S, et al. Job stress and healthy behavior among male Japanese office workers. Am J Ind Med. 2010; 53: 1128-1134.

²⁰¹ Van Loon AJM, Tijhuis M, Sturtees PG, Ormel J. Lifestyle risk factors for cancer: the relationship with psychosocial work environment. Int J Epidemiol. 2000; 29: 785-792.

²⁰² Jankovic N, Geelen A, Streppel MT, et al. Adherence to a healthy diet according to the World Health Organisation guidelines and all-cause mortality in elderly adults from Europe and the United States. Am J Epidemiol. 2014; 180(10): 978-988.

²⁰³ Fransson EL, Heikkilä K, Nyberg ST, et al. Job strain as a risk factor for leisure-time physical inactivity: An individual-participant meta-analysis of up to 170,000 men and women. Am J Epidemiol. 2012; 176(12): 1078-1089.

²⁰⁴ Kirk MA, Rhodes RE. Occupation correlates of adults' participation in leisure-time physical activity: a systematic review. Am J Prev Med. 2011; 40(4): 476-485.

²⁰⁵ Lallukka T, Lahelma E, Rahkonen O, et al. Associations of job strain and working overtime with adverse health behaviors and obesity: Evidence from the Whitehall II Study, Helsinki Health Study, and the Japanese Civil Servants Study. Soc Sci Med. 2008; 66: 1681-1698.
 ²⁰⁶ Gimeno D, Elovainio M, Jokela M, et al. Association between passive jobs and low levels of leisure-time physical activity: the Whitehall II cohort study. Occup Environ Med. 2009; 66: 772-776.

²⁰⁷ Kouvonen A, Kivimaki M, Elovainio M, et al. Effort-reward imbalance and sedentary
lifestyle: an observational study in a large occupational cohort. Occup Environ Med. 2006; 63:
422-427.

²⁰⁸ Kuper H, Singh-Manoux A, Siegrist J, et al. When reciprocity fails: effort-reward imbalance

in relation to coronary heart disease and health functioning within the Whitehall II study. Occup Environ Med. 2002; 59: 777-784.

²⁰⁹ Plotnikoff RC, Mayhew A, Birkett N, et al. Age, gender, and urban-rural differences in the correlates of physical activity. Prev Med. 2004; 39: 1115-1125.

²¹⁰ Grunberg L, Moore S, Greenberg ES. Work stress and problem alcohol behavior: a test of the spillover model. J Organ Behav. 1998; 19(5): 487-502.

²¹¹ Siegrist J, Starke D, Chandola T, et al. The measurement of Effort-Reward Imbalance at work: European Comparisons. Soc Sci Med. 2004; 58: 1483-1499.

²¹² Matschinger H, Siegrist J, Siegrist K, Dittmann KH. Type A as a coping career—toward a conceptual and methodological redefinition. In: Schmidt TH, Dembroski TM, Blumchen G, editors. Biological and psychological factors in cardiovascular disease. Berlin: Springer Verlag; 1986.

²¹³ Rosenman RH. Current and past history of type A behavior pattern. In: Schmidt TH, Dembroski TM, Blumchen G, editors. Biological and psychological factors in cardiovascular disease. Berlin: Springer Verlag; 1986.

²¹⁴ Siegrist J, Peter R, Junge A, Cremer P, Seidel D. Low status control, high effort at work and ischemic heart disease: Prospective evidence from blue-collar men. Soc Sci Med. 1990;
31: 1127-1134.

²¹⁵ Hanson EKS, Schaufeli W, Vrijkotte T, Plomp NH, Godaert GLR. The validity and reliability of the Dutch effort-reward imbalance questionnaire. J Occup Health Psychol. 2000; 5(1): 142-155.

²¹⁶ Niedhammer I, Siegrist J, Landre MF, Goldberg M, Leclerc A. Psychometric properties of the French version of the Effort–reward Imbalance model. Revue d'epidemiol et de sante publique. 2000; 48(5): 419-437.

²¹⁷ Siegrist J, Starke D, Chandola T, et al. The measurement of effort-reward imbalance at work: European comparisons. Soc Sci Med. 2004; 58: 1483-1499.

²¹⁸ Vearing A, Mak AS. Big five personality and effort-reward imbalance factors in employees' depressive symptoms. Pers Individ Dif. 2007; 43: 1744-1755.

²¹⁹ Lubinski D, Dawis RV. Aptitudes, skills and proficiencies. In: Dunnette MD, Hough LM, editors. Handbook of industrial and organisational psychology. 2nd ed. Palo Alto: Consulting Psychologists Press; 1992. p. 1-59.

²²⁰ Allisey A, Rodwell J, Noblet A. Personality and the effort-reward imbalance model of stress: Individual differences in reward sensitivity. Work Stress. 2012; 26(3): 230-251.

²²¹ Joksimovic L, Starke D, Knesebeck OVD, Siegrist J. Perceived work stress,

overcommitment, and self-reported musculoskeletal pain: A cross-sectional investigation. Int J Behav Med. 2002; 9: 122-138.

²²² Vearing A, Mak AS. Big five personality and effort-reward imbalance factors in employees' depressive symptoms. Pers Individ Dif. 2007; 43: 1744-1755.

²²³ Felsten G. Five-factor analysis of Buss-Durkee hostility inventory neurotic hostility and expressive hostility factors: Implications for health psychology. J Pers Assess. 1996; 67: 179-

194.

²²⁴ Smith TW, O'Keeffe JL, Allred KD. Neuroticism, symptom reports, and Type A Behavior: interpretive cautions for the Framingham Scale. J Behav Med. 1989; 12: 1-11.

²²⁵ Friedman M, Rosenman RH. Association of specific overt behavior pattern with blood and cardiovascular findings. JAMA. 1959; 169: 1286-1296.

²²⁶ Rosenman RH, Brand RJ, Jenkins CD, Friedman M, Straus R, Wurm M. Coronary heart disease in the Western Collaborative Group Study: final follow–up experience if 8.5 years. JAMA. 1975; 233: 872-877.

²²⁷ Matthews KA, Haynes SG. Type A behaviour pattern and coronary disease risk: update and critical evaluation. Am J Epidemiol. 1986; 123: 923-960.

²²⁸ Ragland DR, Brand RJ. Coronary heart disease mortality in the Western Collaborative Group Study: follow-up experience of 22 years. Am J Epidemiol. 1988; 127: 462-475.

²²⁹ Eaker ED, Abbott RD, Kannel WB. Frequency of uncomplicated angina pectoris in Type A compared with Type B persons (The Framingham Study). Am J Cardiology. 1989; 63: 1042-1045.

²³⁰ Johnston DW. The current status of the coronary-prone behaviour pattern. J R Soc Med. 1993; 86: 406-409.

²³¹ Eckhardt C, Norlander B, Deffenbacher J. The assessment of anger and hostility: a critical review. Aggress Violent Beh. 2004; 9: 17-43.

²³² Buss SH, Durkee A. An inventory for assessing different kinds of hostility. J Consult Psychol. 1957; 21: 343-349.

²³³ Cook W, Medley D. Proposed hostility and pharisaic-virtue scales for the MMPI. J Appl Psychol. 1954; 38: 414-418.

²³⁴ Everson-Rose SA, Lewis TT. Psychosocial factors and cardiovascular diseases. Annu Rev Public Health. 2005; 26: 469-500.

²³⁵ Miller TQ, Smith TW, Turner CW, Guijarro ML, Hallet AJ. A meta-analytic review of research on hostility and physical health. Psychol Bull. 1996; 119: 322-348.

²³⁶ Chida Y, Steptoe A. The association of anger and hostility with future coronary heart disease. A meta-analytic review of prospective evidence. J Am Coll Cardiol. 2009; 53: 936-946.

²³⁷ Whiteman MC. Personality, cardiovascular disease and public health. In: Vollrath ME, editor. Handbook of personality and health. Chichester: John Wiley Sons; 2006.

²³⁸ Smith TW, Christensen AJ. Hostility, health and social contexts. In: Friedman HS, editor. Hostility, coping and health. Washington DC: American Psychological Association; 1992.

²³⁹ Scherwitz LW, Perkins LL, Chesney MA, Hughes GH, Sidney S, Manolio TA. Hostility and health behaviours in young adults: The CARDIA study. Am J Epidemiol. 1992; 136: 136-145.

²⁴⁰ Pulkki L, Kivimaki M, Elovainio M, Vikari J, Keltikangas-Jarvinen L. Contribution of socioeconomic status to the association between hostility and cardiovascular risk behaviours:
 A prospective cohort study. Am J Epidemiol. 2003; 158: 736-742.

²⁴¹ Fowkes FGR, Housley E, Cawood EH, Macintyre CA, Ruckley CV, Prescott RJ.

Edinburgh Artery Study: Prevalence of asymptomatic and symptomatic peripheral arterial disease in the general population. Int J Epidemiol. 1991; 20: 384-392.

²⁴² Smith TW. Hostility and health: current status of a psychosomatic hypothesis. Health Psychol. 1992; 11: 139-150.

²⁴³ Felsten G. Five-factor analysis of Buss-Durkee hostility inventory neurotic hostility and expressive hostility factors: Implications for health psychology. J Pers Assess. 1996; 67: 179-194.

²⁴⁴ Eysenck HJ, Eysenck SBG. Manual for the Eysenck Personality Inventory. London:University Press; 1964.

²⁴⁵ Watson D, Clark LA. Negative affectivity: the disposition to experience aversive emotional states. Psychol Bull. 1984; 96: 465-490.

²⁴⁶ Costa PT, McCrae RR. Revised NEO Personality Inventory (NEO-PI-R) and NEO Five-Factor Inventory (NEO-FFI) manual. Odessa, FL: Psychological Assessment Resources; 1992.

²⁴⁷ Eysenck SBG, Eysenck HJ, Barrett P. A revised version of the psychoticism scale. Pers Individ Dif. 1985; 6: 21-29.

²⁴⁸ Clark LA, Watson D, Mineka S. Temperament, personality, and the mood and anxiety disorders. J Abnorm Psychol. 1994; 103: 103-116.

²⁴⁹ Suls J, Bunde J. Anger, anxiety, and depression as risk factors for cardiovascular disease:
the problems and implications of overlapping affective dispositions. Psychol Bull. 2005; 131:
260-300.

²⁵⁰ Denollet J, De Vries J. Positive and negative affect within the realm of depression, stress and fatigue: The two-factor distress model of the Global Mood Scale. J Affect Disord. 2006;
91: 171-180.

²⁵¹ Kaplan S, Bradley JC, Luchman JN, Haynes D. On the role of positive and negative affectivity in job performance: a meta-analytic investigation. J Appl Psychol. 2009; 94: 162-176.

²⁵² Lahey BB. Public health significance of neuroticism. Am Psychol. 2009; 64: 241-256.

²⁵³ Shipley BA, Weiss A, Der G, Taylor MD, Deary IJ. Neuroticism, extraversion, and mortality in the UK health and lifestyle survey: A 21-year prospective cohort study. Psychosom Med. 2007; 69: 923-931.

²⁵⁴ Penninx BW, Beekman AT, Honig A, et al. Depression and cardiac mortality: results from a community based study. Arch Gen Psychiatry. 2001; 58: 221-227.

²⁵⁵ Haas DC, Davidson KW, Schwartz DJ, et al. Depressive symptoms are independently predictive of carotid atherosclerosis. Am J Cardiol. 2005; 95: 547-550.

²⁵⁶ Golden SH, Williams JE, Ford DE, et al. Depressive symptoms and the risk of type 2
 diabetes: the Atherosclerosis Risk in Communities study. Diabetes Care. 2004; 27: 429-435.
 ²⁵⁷ Gump BB, Matthews KA, Eberly LE, Change YF, MRFIT Research Group. Depressive symptoms and mortality in men: results from the Multiple Risk Factor Intervention Trial.
 Stroke. 2005; 36: 98-102.

²⁵⁸ Stamatakis KA, Lynch J, Everson SA, Raghunathan T, Salonen JT, Kaplan GA. Selfesteem and mortality: prospective evidence from a population-based study. Ann Epidemiol. 2004; 14: 58-65.

²⁵⁹ Van Vegchel N, De Jonge J, Bosma H & Schaufeli W. Reviewing the effort-reward imbalance model: Drawing up the balance of 45 empirical studies. Soc Sci Med. 2005; 60: 1117-1131.

²⁶⁰ Rosenman RH, Friedman M. Association of specific behavior pattern in women with blood and cardiovascular findings. Circulation. 1961; 24: 1173-1184.

²⁶¹ Koskenvuo M, Kaprio J, Longinvaninio H, Romo M, Sarna S. Psychosocial and environmental correlates of coronary-prone behavior in Finland. J Chron Dis. 1981; 34: 331-340.

²⁶² Folsom AR, Hughes JR, Buehler JF, Mittelmark MB, Jacobs DR, Grimm RH. Do Type A men drink more frequently than Type B men? Findings in the Multiple Risk Factor Intervention Trail. J Behav Med. 1985; 8: 227-235.

²⁶³ Glynn RJ, De Labry LO, Hou DM. Alcohol consumption, Type A behavior, and demographic variables. Results from the normative aging study. Am J Epidemiol. 1988; 127(2): 310-320.

²⁶⁴ Everson SA, Kauhanen J, Kaplan GA, et al. Hostility and increased risk of mortality and acute myocardial infarction: the mediating role of behavioral risk factors. Am J Epidemiol. 1997; 146: 142-152.

²⁶⁵ Pulkki L, Kivimäki M, Elovainio M, Viikari J, Keltikangas-Järvinen L. Contribution of socioeconomic status to the association between hostility and cardiovascular risk behaviors: a prospective cohort study. Am J Epidemiol. 2003; 158: 736-742.

²⁶⁶ Siegler IC, Peterson BL, Barefoot JC, Williams RB. Hostility during late adolescence predicts coronary risk factors at mid-life. Am J Epidemiol. 1992; 136: 146-154.

 ²⁶⁷ Scherwitz LW, Perkins LL, Chesney MA, Hughes GH, Sidney S, Manolio TA. Hostility and health behaviours in young adults: The CARDIA study. Am J Epidemiol. 1992; 136: 136-145.
 ²⁶⁸ Whiteman MC, Fowkes FGR, Deary IJ, Lee AJ. Hostility, cigarette smoking and alcohol consumption in the general population. Soc Sci Med. 1997; 44: 1089-1096.

²⁶⁹ Kuntsche E, von Fischer M, Gmel G. Personality factors and alcohol use: A mediator analysis of drinking motives. Pers Individ Dif. 2008; 45(8): 796-800.

²⁷⁰ Almada SJ, Zonderman AB, Shekelle RB, Dyer AR, Daviglus ML, Costa PT. Neuroticism, cynicism and risk of death in middle-aged men: the western electric study. Psychosom Med. 1991; 53: 165-175.

²⁷¹ Flory K, Lynam D, Milich R, Leukefeld C, Clayton R. The relation among personality, symptoms of alcohol and marijuana abuse, and symptoms of comorbid psychopathology: results from a community sample. Exp Clin Psychopharm. 2002; 10(4): 425-434.

²⁷² Otonari J, Nagano J, Morita M, et al. Neuroticism and extraversion personality traits, health behaviours, and subjective well-being: the Fukuoka Study. Qual Life Res. 2012; 21(10): 1847-1855.

²⁷³ Connor-Smith JK, Flachsbart C. Relation between personality and coping: a metaanalysis. J Pers Soc Psychol. 2007; 93(6): 1080–1107.

²⁷⁴ Stewart SH, Loughlin HL, Rhyno E. Internal drinking motives mediate personality domain
— drinking relation in young adults. Pers Individ Dif. 2001; 30: 271-286.

²⁷⁵ Jenkins CD, Zyzanski SJ, Rosenman RH. Biological, psychological and social characteristics of men with different smoking habits. Health Serv Rep. 1973; 88: 834-843.
 ²⁷⁶ Shekelle RB, Schoenberger JA, Stamler J. Correlates of the JAS type A behavior pattern score. J Chron Dis. 1976; 29(6): 381-394.

²⁷⁷ Pulkki L, Kivimäki M, Keltikangas-Jarvinen L, Elovainio M, Leino M, Viikari J. Contribution of adolescent and early adult personality to the inverse association between education and cardiovascular risk behaviours: prospective population-based cohort study. Int J Epidemiol. 2003; 32(6): 968-975.

²⁷⁸ Siegler IC, Peterson BL, Barefoot JC, Williams RB. Hostility during late adolescence predicts coronary risk factors at mid-life. Am J Epidemiol. 1992; 136: 146-54.

²⁷⁹ Pulkki L, Kivimäki M, Elovainio M, Viikari J, Keltikangas-Järvinen L. Contribution of socioeconomic status to the association between hostility and cardiovascular risk behaviors: a prospective cohort study. Am J Epidemiol. 2003; 158(8): 736-742.

²⁸⁰ Everson SA, Kauhanen J, Kaplan GA, et al. Hostility and increased risk of mortality and acute myocardial infarction: the mediating role of behavioral risk factors. Am J Epidemiol. 1997; 146: 142-152.

²⁸¹ Scherwitz LW, Perkins LL, Chesney MA, Hughes GH, Sidney S, Manolio TA. Hostility and health behaviours in young adults: The CARDIA study. Am J Epidemiol. 1992; 136: 136-145.
 ²⁸² Schrijvers CTM, Bosma H, Mackenbach JP. Hostility and the educational gradient in health: The mediating role of health-related behaviours. Eur J Public Health. 2002; 12: 110-116.

²⁸³ Whiteman MC, Fowkes FGR, Deary IJ, Lee AJ. Hostility, cigarette smoking and alcohol consumption in the general population. Soc Sci Med. 1997; 44: 1089-1096.

²⁸⁴ Munafò MR, Black S. Personality and smoking status: a longitudinal analysis. Nicotine Tob Res. 2007; 9(3): 397-404.

²⁸⁵ Goodwin R, Hamilton SP. Cigarette smoking and panic: the role of neuroticism. Am J Psychiatry. 2002; 159: 1208-1213.

²⁸⁶ Kirk KM, Whitfield JB, Pang D, Heath AC, Martin NG. Genetic covariation of neuroticism with monoamine oxidase activity and smoking. Am J Med Genet. 2001; 105: 700-706.

²⁸⁷ Munafò MR, Zetteler JI, Clark TG. Personality and smoking status: a meta-analysis. Nicotine Tob Res. 2007; 9(3): 405-413.

²⁸⁸ Barker ME, Thompson KA, McClean SI. Do Type As eat differently? A comparison of men and women. Appetite. 1996; 26: 277-286.

²⁸⁹ Gallacher JE, Fehily AM, Yarnell JWG, Butland BK. Type A behaviour, eating pattern and nutrient intake: The Caerphilly Study. Appetite. 1988; 11: 129-136.

²⁹⁰ Appleton KM, Woodside JV, Yarnell JWG, et al. Type A behaviour and consumption of an

atherogenic diet: No association in the PRIME study. Appetite. 2007; 49(3): 554-560.
²⁹¹ Iribarren C, Markovitz JH, Jacobs DR, et al. Dietary intake of n-3, n-6 fatty acids and fish: relationship with hostility in young adults: the CARDIA study. Eur J Clin Nutr. 2004; 58: 24-31.
²⁹² Scherwitz LW, Perkins LL, Chesney MA, Hughes GH, Sidney S, Manolio TA. Hostility and health behaviours in young adults: The CARDIA study. Am J Epidemiol. 1992; 136: 136-145.
²⁹³ van den Bree MB, Przybeck TR, Cloninger CR. Diet and personality: associations in a population-based sample. Appetite. 2006; 46(2): 177-88.

²⁹⁴ Siegler IC, Peterson BL, Barefoot JC, Williams RB. Hostility during late adolescence predicts coronary risk factors at mid-life. Am J Epidemiol. 1992; 136: 146-154.

²⁹⁵ Tiainen AMK, Männistö S, Lahti M, et al. Personality and dietary Intake – findings in the Helsinki Birth Cohort Study. PLoS ONE. 2013; 8(7): e68284.

²⁹⁶ De Bruijn G, Brug J, Van Lenthe FJ. Neuroticism, conscientiousness and fruit consumption: exploring mediator and moderator effects in the theory of planned behaviour. Psychol health. 2009; 24(9): 1051-1069.

²⁹⁷ Vollrath ME, Hampson SE, Júlíusson PB. Children and eating: personality and gender are associated with obesogenic food consumption and overweight in 6- to 12-year-olds. Appetite. 2012; 58(3): 1113-1117.

²⁹⁸ Pulkki L, Elovainio M, Kivimäki M, Raitakari OT, Keltikangas-Järvinen L. Temperament in childhood predicts body mass in adulthood: the Cardiovascular Risk in Young Finns Study. Health Psychol. 2005; 24(3): 307-315.

²⁹⁹ Otonari J, Nagano J, Morita M, et al. Neuroticism and extraversion personality traits, health behaviours, and subjective well-being: the Fukuoka Study. Qual Life Res. 2012; 21(10): 1847-1855.

³⁰⁰ Pulkki L, Kivimäki M, Keltikangas-Jarvinen L, Elovainio M, Leino M, Viikari J. Contribution of adolescent and early adult personality to the inverse association between education and cardiovascular risk behaviours: prospective population-based cohort study. Int J Epidemiol. 2003; 32(6): 968-975.

³⁰¹ Yang X, Telama R, Hirvensalo M, et al. Leadership component of Type A behavior predicts physical activity in early midlife. Int J Behav Med. 2012; 19(1): 48-55.

³⁰² Abbott AV, Peters RK, Vogel ME. Type A behavior and physical activity: A follow-up study of coronary patients. J Psychosom Res. 1990; 34(2): 153-162.

³⁰³ Schrijvers CTM, Bosma H, Mackenbach JP. Hostility and the educational gradient in health: The mediating role of health-related behaviours. Eur J Public Health. 2002; 12: 110-116.

³⁰⁴ Maier KJ, James AE. Hostility and social support explain physical activity beyond negative affect among young men, but not women, in college. Behav Med. 2014; 40(1): 34-41.

³⁰⁵ Everson SA, Kauhanen J, Kaplan GA, et al. Hostility and increased risk of mortality and acute myocardial infarction: The mediating role of behavioral risk factors. Am J Epidemiol. 1997; 146(2): 142-152.

³⁰⁶ Wong JM, Na B, Regan MC, Whooley MA. Hostility, health behaviors, and risk of

recurrent events in patients with stable coronary heart disease: findings from the Heart and Soul Study. J Am Heart Assoc. 2013; 2: e000052.

³⁰⁷ Houston BK, Vavak CR. Cynical hostility: developmental factors, psychosocial correlates, and health behaviors. Health Psychol. 1991; 10: 9-17.

³⁰⁸ Bunde J, Suls J. A quantitative analysis of the relationship between the Cook-Medley
Hostility Scale and traditional coronary artery disease risk factors. Health Psychol. 2006; 25:
493-500.

³⁰⁹ Brunes A, Augestad LB, Gudmundsdottir SL. Personality, physical activity, and symptoms of anxiety and depression: the HUNT study. Soc Psychiatry Psychiatr Epidemiol. 2013; 48(5): 745-756.

³¹⁰ DeMoor MHM, Beem AL, Stubbe JH, Boomsma DI, De Geus EJC. Regular physical activity, anxiety, depression and personality: A population-based study, Prev Med. 2006; 42(4): 273-279.

³¹¹ Droomers M, Schrijvers CT, van de Mheen H, Mackenbach JP. Educational differences in leisure-time physical inactivity: a descriptive and explanatory study. Soc Sci Med. 1998; 47(11): 1665-1676.

³¹² Tolea MI, Terracciano A, Milaneschi Y, Metter EJ, Ferrucci L. Personality typology in relation to muscle strength. Int J Behav Med. 2012; 19(3): 382-390.

³¹³ Rhodes RE, Smith NEI. Personality correlates of physical activity: a review and metaanalysis. Br J Sports Med. 2006; 40: 958-965.

³¹⁴ Bogg T, Roberts BW. Conscientiousness and health-related behaviours: A meta-analysis of the leading behavioural contributors to mortality. Psychol Bull. 2004; 130(6): 887-919.
 ³¹⁵ Peterson C, Stunkard A. Personal control and health promotion. Soc Sci Med. 1989; 28(8): 819-828.

³¹⁶ Seeman M. Alienation and anomie. In: Robinson JP, Shaver PR, Wrightsman LS, editors. Measures of personality and social psychological attitudes. San Diego, CA: Academic Press; 1991. p. 291-372.

³¹⁷ Hartmann H. Ego psychology and the problem of adaptation. New York: International Universities Press; 1939.

³¹⁸ White RW. Motivation reconsidered: the concept of competence. Psychol Rev. 1959; 66: 297-333.

³¹⁹ Lewin K. Principles of topological psychology. New York: McGraw-Hill; 1936.

³²⁰ Deci EL, Ryan RM. Intrinsic motivation and self-determination in human behaviour. New York: Plenum; 1985.

³²¹ Gardner H. The mind's new science: a history of the cognitive revolution. New York: Basic Books; 1985.

³²² Carver CS, Scheier MF. Perspectives on personality. 4th ed. Boston: Allyn and Bacon; 2000.

³²³ Bandura A. Self-efficacy: towards a unifying theory of behavior change. Psychol Rev. 1977; 84: 191-215.

³²⁴ Skinner EA, Chapman M, Baltes PB. Control, means-ends, and agency beliefs: A new conceptualization and its measurement during childhood. J Pers Soc Psychol. 1988; 54: 117-133.

³²⁵ Skinner EA. A guide to constructs of control. J Pers Soc Psychol. 1996; 71: 549-570.
 ³²⁶ Peterson C, Stunkard AJ. Cognates of personal control: locus of control, self-efficacy, and explanatory style. Appl Prev Psychol. 1992; 1: 111-117.

³²⁷ Thompson SC, Spacapan S. Perceptions of control in vulnerable populations. J Social Issues. 1991; 41: 1-21.

³²⁸ Rotter JB. Generalized expectancies of internal versus external control of reinforcements. Psychol Monogr. 1966; 80: 1-28.

³²⁹ Pearlin LI, Schooler C. The structure of coping. J Health Soc Behav. 1978; 19: 2-21.

³³⁰ Overmeier JB, Seligman MEP. Effect of inescapable shock upon subsequent escape and avoidance responding. J Comp Physiol Psychol. 1967; 63: 28-33.

³³¹ Antonovsky A. The structure and properties of the sense of coherence scale. Soc Sci Med. 1993; 6: 725-733.

³³² Rothbaum F, Weisz JR, Snyder SS. Changing the world and changing the self: a two process model of perceived control. J Pers Soc Psychol. 1982; 42: 5-37.

³³³ Sherer M, Adams CH. Construct validation of the Self-Efficacy Scale. Psychol Rep. 1983;53: 899-902.

³³⁴ Paulhus DL. Sphere-specific measures of perceived control. J Per Soc Psychol. 1983; 44:1253-1265.

 ³³⁵ Pearlin LI, Schooler C. The structure of coping. J Health Soc Behav. 1978; 18: 2-21.
 ³³⁶ Lachman ME, Weaver SL. Sociodemographic variations in the sense of control by domain: findings from the MacArthur Studies of Midlife. Psychol Aging. 1998; 13(4): 553-562.
 ³³⁷ Bobak M, Pikhart H, Rose R, Hertzman C, Marmot M. Socioeconomic factors, material inequalities, and perceived control in self-rated health: cross-sectional data from seven post-communist countries. Soc Sci Med. 2000; 51: 1343-1350.

³³⁸ Lachman ME, Weaver SL. The sense of control as a moderator of social class differences in health and well-being. J Pers Soc Psychol. 1998; 74(3): 763-773.

³³⁹ Cotter KA, Lachman ME. No strain, no gain: psychosocial predictors of physical activity across the adult lifespan. J Phys Act Health. 2010; 7(5): 584-594.

³⁴⁰ Gerstorf D, Röcke C, Lachman ME. Antecedent-consequent relation of perceived control to health and social support: longitudinal evidence for between-domain associations across adulthood. J Gerontol B Psychol Sci. 2011; 66: 61-71.

³⁴¹ Syme SL. Control and health: a personal perspective. In: Steptoe A, Appels A, editors. Stress, personal control and health. UK: John Wiley & Sons press; 1989. p. 3-18.

³⁴² Steptoe A, Appels A. Stress, personal control and health. UK: John Wiley & Sons press; 1989.

³⁴³ Skinner EA. Perceived control, motivation, and coping. Thousand Oaks, CA: Sage; 1995.

³⁴⁴ Skinner EA, Wellborn JG. Coping during childhood and adolescence: a motivational

perspective. In: Featherman D, Lerner R, Perlmutter M, editors. Life-span development and behavior. Hillsdale, NJ: Erlbaum; 1994.

³⁴⁵ Bandura A. Self-efficacy: the exercise of control. New York: Freeman and Company press; 1997.

³⁴⁶ Glanz K, Rimer BK, Viswanath K. Health behavior and health education: theory, research, and practice. 4th ed. San Francisco, CA: John Wiley & Sons; 2008.

³⁴⁷ Conner M, Norman P. Predicting health behaviour: research and practice with social cognition models. 2nd ed. Maidenhead: Open University Press; 2005.

³⁴⁸ Bandura A. Social foundations of thought and action: a social cognitive theory. Englewood Cliffs NJ: Prentice Hall; 1986.

³⁴⁹ Rosenstock IM, Strecher VJ, Becker MH. Social learning theory and the health belief model. Health Educ Q. 1988; 15: 175-183.

³⁵⁰ Prochaska JO, DiClemente CC. The transtheoretical approach: crossing traditional boundaries of therapy. Homewood: Dow Jones Irwin; 1984.

³⁵¹ Ajzen I. Perceived behavioral control, self-efficacy, locus of control, and the Theory of Planned Behavior. J Appl Soc Psychol. 2002; 32(4): 665-683.

³⁵² Armitage CJ, Conner M. Efficacy of the theory of planned behaviour: a meta-analytic review. Br J Soc Psychol. 2001; 40: 471-499.

³⁵³ Perlman F, Bobak M, Steptoe A, Rose R, Marmot M. Do health control beliefs predict behaviour in Russians? Prev Med. 2003; 37: 73-81.

³⁵⁴ Troein M, Rastam L, Selander S. Health beliefs and heart disease risk among middleaged Swedish men. Results from screening in an urban primary care district. Scand J Prim Health Care. 1997; 15(4): 198-202.

³⁵⁵ Connor JP, Young RM, Williams RJ, Ricciardelli LA. Drinking restraint versus alcohol expectancies: Which is the better indicator of alcohol problems? J Stud Alcohol. 2000; 61: 352-359.

³⁵⁶ Hoeppner BB, Kelly JF, Urbanoski KA, et al. Comparative utility of a single-item versus multiple-item measure of self-efficacy in predicting relapse among young adults. J Subst Abuse Treat. 2011; 41(3): 305-312.

³⁵⁷ Norman P, Conner M. The theory of planned behaviour and binge drinking: assessing the moderating role of past behaviour within the theory of planned behaviour. Br J Health Psychol. 2006; 11: 55-70.

³⁵⁸ Grembowski D, Patrick D, Diehr P, et al. Self-efficacy and health behavior among older adults. J Health Soc Behav. 1993; 34(2): 89-104.

³⁵⁹ Conner M, Warren R, Close S, Sparks P. Alcohol consumption and the Theory of Planned Behavior: an examination of the cognitive mediation of past behaviour. J Appl Soc Psychol. 1999; 29: 1676-1704.

³⁶⁰ Johnston KL, White KM. Binge drinking: a test of the role of group norms in the theory of planned behaviour. Psychol Health. 2003; 18: 63-77.

³⁶¹ Hagger MS, Lonsdale A, Koda A, et al. An intervention to reduce alcohol consumption in

undergraduate students using implementation intentions and mental simulations. Int J Behav Med. 2012; 19: 82-96.

³⁶² Sigrun A, Fjolvar Darri R. Perceived control in adolescent substance use: concurrent and longitudinal analyses. Psychol Addict Behav. 2001; 15(1): 25-32.

³⁶³ Devogli R, Santinello. Unemployment and smoking: does psychosocial stress matter? Tob Control. 2005; 14: 389-395.

³⁶⁴ Troein M, Rastam L, Selander S. Health beliefs and heart disease risk among middleaged Swedish men. Results from screening in an urban primary care district. Scand J Prim Health Care. 1997; 15(4): 198-202.

³⁶⁵ Diclemente CC, Prochaska JO, Gibertini M. Self-efficacy and the stages of self-change of smoking. Cognit Ther Res. 1985; 9: 181-200.

³⁶⁶ Wangberg SC, Nilsen O, Antypas K, Gram IT. Effect of tailoring in an internet-based intervention for smoking cessation: randomized controlled trial. J Med Internet Res. 2011; 15(13): e121.

³⁶⁷ Schnoll RA, Martinez E, Tatum KL, et al. Increased self-efficacy to quit and perceived control over withdrawal symptoms predict smoking cessation following nicotine dependence treatment. Addict Behav. 2011; 36: 144-147.

³⁶⁸ Kim Y. Adolescents' health behaviours and its association with psychological variables. Cent Eur J Public Health. 2011; 19(4): 205-209.

³⁶⁹ Bennett P, Norman P, Moore L, Murphy S, Tudor-Smith C. Health locus of control and value for health in smokers and nonsmokers. Health Psychol. 1997; 16: 179-182.

³⁷⁰ Van De Ven MO, Engels RC, Otten R, Van Den Eijnden RJ. A longitudinal test of the theory of planned behavior predicting smoking onset among asthmatic and non-asthmatic adolescents. J Behav Med. 2007; 30(5): 435-445.

³⁷¹ Godin G, Valois P, Lepage L, Desharnais R. Predictors of smoking behaviour: an application of Ajzen's theory of planned behaviour. Br J Addict. 1992; 87(9): 1335-1343.
³⁷² Guo Q, Johnson CA, Unger JB, et al. Utility of the theory of reasoned action and theory of planned behaviour for predicting Chinese adolescent smoking. Addict Behav. 2007; 32(5):

1066-1081.

³⁷³ Moan IS, Rise J. Quitting smoking: Applying an extended version of the theory of planned behaviour. J Appl Biobehav Res. 2005; 10(1): 39-68.

³⁷⁴ Barker M, Lawrence W, Crozier S, et al. Educational attainment, perceived control and the quality of women's diets. Appetite. 2009; 52(3): 631-636.

³⁷⁵ Shaikh AR, Yaroch AL, Nebeling L, Yeh MC, Resnicow K. Psychosocial predictors of fruit and vegetable consumption in adults: a review of the literature. Am J Prev Med. 2008; 34(6): 535-543.

³⁷⁶ Grembowski D, Patrick D, Diehr P, et al. Self-efficacy and health behavior among older adults. J Health Soc Behav. 1993; 34(2): 89-104.

³⁷⁷ Steptoe A, Wardle J. Locus of control and health behaviour revisited: A multivariate analysis of young adults from 18 countries. Br J Psychol. 2001; 92: 659-672.

³⁷⁸ Bennett P, Moore L, Smith A, Murphy S, Smith C. Health locus of control and value for health as predictors of dietary behaviour. Psychol Health. 1994; 10(1): 41-54.

³⁷⁹ Blanchard CM, Kupperman J, Sparling PB, et al. Do ethnicity and gender matter when using the theory of planned behavior to understand fruit and vegetable consumption? Appetite. 2009; 52(1): 15-20.

³⁸⁰ Armitage CJ, Conner M. Efficacy of the theory of planned behaviour: a meta-analytic review. Br J Soc Psychol. 2001; 40: 471-499.

³⁸¹ Cotter KA, Lachman ME. No strain, no gain: psychosocial predictors of physical activity across the adult lifespan. J Phys Act Health. 2010; 7(5): 584-594.

³⁸² Lachman ME, Firth KM. The adaptive value of feeling in control during midlife. In: Brim OG, Ryff CD, Kessler RC, editors. How healthy are we? A national study of well-being at midlife. Chicago: University of Chicago Press; 2004.

³⁸³ Hagger MS, Chatzisarantis NLD, Biddle SJH. A meta-analytic review of the theories of reasoned action and planned behavior in physical activity: predictive validity and the contribution of additional variables. J Sport Exerc Psychol. 2002; 24(1): 3-32.

³⁸⁴ Dzewaltowski DA, Noble JM, Shaw JM. Physical activity participation: social cognitive theory versus the theories of reasoned action and planned behavior. J Sport Exerc Psychol. 1990; 12: 388-405.

 ³⁸⁵ Hill AM, Hoffmann T, McPhail S, et al. Factors associated with older patients' engagement in physical activity after hospital discharge. Arch Phys Med Rehabil. 2011; 92(9): 1395-1403.
 ³⁸⁶ Renner B, Hankonen N, Ghisletta P, Absetz P. Dynamic psychological and behavioral changes in the adoption and maintenance of physical activity. Health Psychol. 2012; 31(3): 306-315.

³⁸⁷ Steptoe A, Wardle J. Locus of control and health behaviour revisited: a multivariate analysis of young adults from 18 countries. Br J Psychol. 2001; 92: 659-672.

³⁸⁸ Rhodes RE, Courneya KS. Investigating multiple components of attitude, subjective norm, and perceived control: an examination of the theory of planned behaviour in the physical activity domain. Br J Soc Psychol. 2003; 42: 129-146.

³⁸⁹ Bailis DS, Segall A, Mahon MJ, Chipperfield JG, Dunn EM. Perceived control in relation to socioeconomic and behavioral resources for health. Soc Sci Med. 2001; 52: 1661-1676.
 ³⁹⁰ Norman P, Bennett P, Smith C, Murphy S. Health locus of control and leisure-time physical activity. Pers Individ Dif. 1997; 23: 769-774.

³⁹¹ Godin G, Kok G. The theory of planned behavior: A review of its applications to healthrelated behaviors. Am J Health Promot. 1996; 11: 87-98.

³⁹² Perlman F, Bobak M, Steptoe A, Rose R, Marmot M. Do health control beliefs predict behaviour in Russians? Prev Med. 2003; 37: 73-81.

³⁹³ Winter DG, John OP, Stewart AJ, et al. Traits and motives: toward an integration of two traditions in personality research. Psychol Bull. 1998; 105: 230-250.

³⁹⁴ Murray HA. Explorations in personality. New York: Oxford University Press; 1938.

³⁹⁵ Skinner EA. Perceived control, motivation and coping. Thousand Oakes, CA: Sage; 1995.

³⁹⁶ White RW. Motivation reconsidered: the concept of competence. Psychol Rev. 1959; 66: 297-333.

³⁹⁷ Piaget J. The grasp of consciousness: action and concept in the young child. Cambridge, MA: Harvard University Press; 1976.

³⁹⁸ DeCharms R. Personal causation. New York: Academic Press; 1968.

³⁹⁹ Deci EL, Ryan RM. Intrinsic motivation and self-determination in human behavior. New York: Plenum; 1985.

⁴⁰⁰ Wortman C, Brehm JW. Responses to uncontrollable outcomes: an integration of reactance theory and the learned helplessness model. In: Berkowitz L, editor. Advances in experimental social psychology. New York: Academic Press; 1975. p. 278-336.

⁴⁰¹ Rothbaum FM, Weisz JR, Snyder SS. Changing the world and changing the self: A twoprocess model of perceived control. J Pers Soc Psychol. 1982; 42: 5-37.

⁴⁰² Greenberger DB, Strasser S. Development and application of a model of personal control in organisations. Acad Manage Rev. 1986; 11: 164-177.

⁴⁰³ Connor M, Abraham C. Conscientiousness and the Theory of Planned Behavior: toward a more complete model of the antecedents of intentions and behavior. Pers Soc Psychol Bull. 2001; 27: 1547-1561.

⁴⁰⁴ De Bruijn GJ, Brug J, Van Lenthe FJ. Neuroticism, conscientiousness and fruit consumption: Exploring mediator and moderator effects in the theory of planned behaviour. Psychol Health. 2009; 24: 1051-1069.

⁴⁰⁵ Courneya KS, Bobick TM, Schinke RJ. Does the Theory of Planned Behavior mediate the relation between personality and physical activity behavior? Basic Appl Soc Psych. 1999; 21(4): 317-324.

⁴⁰⁶ McEachan RC, Sutton S, Myers L. Mediation of personality influences on physical activity within the Theory of Planned Behaviour. J Health Psychol. 2010; 15: 1170-1180.

⁴⁰⁷ Smith TW, MacKenzie J. Personality and risk of physical illness. Annu Rev Clin Psychol. 2006; 2: 435-467.

⁴⁰⁸ Glanz K, Rimer BK, Viswanath K, editors. Health behavior and health education: theory, research, and practice. 4th ed. San Francisco: John Wiley & Sons; 2008.

⁴⁰⁹ Wiebe DJ, Fortenberry KT. Mechanisms relating personality and health. In: Vollrath ME, editor. Handbook of personality and health. Chichester: John Wiley Sons; 2006. p. 137-156.
 ⁴¹⁰ Rosenman RH. Current and past history of type A behavior pattern. In: Schmidt TH,

Dembroski T, Blumchen G, editors. Biological and psychological factors in cardiovascular disease. Berlin: Springer Verlag; 1986.

⁴¹¹ Siegrist J. Effort-reward Imbalance at work and health. In: Perrewe P, Ganster D, editors. Research in occupational stress and well-being. New York: Elsevier; 2002. p. 261-291.

⁴¹² Van Vegchel N, De Jonge J, Bosma H, et al. Reviewing the effort-reward imbalance model: drawing up the balance of 45 empirical studies. Soc Sci Med. 2005; 60: 1117-1131.

⁴¹³ Parkes K. Personality, psychosocial risks at work, and health. Oxford: Department of Experimental Psychology, University of Oxford; 2010.

⁴¹⁴ Niedhammer I, Tek ML, Starke D, Siegrist J. Effort-reward imbalance model and selfreported health: Cross-sectional and prospective findings from the GAZEL cohort. Soc Sci Med. 2004; 58: 1531-1541.

⁴¹⁵ Ota A, Masue T, Yasuda N, et al. Psychosocial job characteristics and insomnia: a prospective cohort study using the Demand-Control-Support and Effort-Reward Imbalance job stress models. Sleep Med. 2009; 10: 1112-1117.

⁴¹⁶ Kuper H, Singh-Manoux A, Siegrist J, Marmot M. When reciprocity fails: Effort-reward imbalance in relation to coronary heart disease and health functioning within the Whitehall II study. Occup Environ Med. 2002; 59: 777-784.

⁴¹⁷ Buddeberg-Fischer B, Klaghofer R, Stamm M, et al. Work stress and reduced health in young physicians: Prospective evidence from Swiss residents. Int Arch Occup Environ Health. 2008; 82: 31-38.

⁴¹⁸ Vahtera J, Kivimäki M, Uutela A, Pentti J. Hostility and ill health: role of psychosocial resources in two contexts of working life. J Psychosom Res. 2000; 48: 89-98.

⁴¹⁹ Elovainio M, Kivimäki M, Vahtera J, et al. Personality as a moderator in the relation between perceptions of organisational justice and sickness absence. J Vocat Behav. 2003;
63: 379-395.

⁴²⁰ Paterniti S, Niedhammer I, Lang T, Consoli SM. Psychosocial factors at work, personality traits and depressive symptoms longitudinal results from the GAZEL Study. Br J Psychiatry. 2002; 181: 111-117.

⁴²¹ Baron RM, Kenny DA. The moderator-mediator variable distinction in social psychological research: conceptual, strategic, and statistical considerations. J Pers Soc Psychol. 1986; 51: 1173-1182.

⁴²² Baker E, Israel B, Schurman S. Role of control and support in occupational stress: an integrated model. Soc Sci Med. 1996; 43: 1145-1159.

⁴²³ Israel BA, Baker EA, Goldenhar LM, Heaney CA & Schurman SJ. Occupational stress, safety, and health: conceptual framework and principles for effective prevention interventions.
J Occup Health Psychol. 1996; 1: 261-286.

⁴²⁴ Siegrist J, Starke D, Chandola T, et al. The measurement of Effort-Reward Imbalance at work: European comparisons. Soc Sci Med. 2004; 58: 1483-1499.

⁴²⁵ Van Vegchel N, De Jonge J, Bosma H, et al. Reviewing the effort-reward imbalance model: drawing up the balance of 45 empirical studies. Soc Sci Med. 2005; 60: 1117-1131.

⁴²⁶ Spector PE, Zapf D, Chen PY, et al. Why negative affectivity should not be controlled in job stress research: Don't throw out the baby with the bath water. J Organ Behav. 2000; 21: 79-95.

⁴²⁷ Semmer NK. Personality, stress, and coping. In: Vollrath ME, editor. Handbook of personality and health. Chichester: John Wiley Sons; 2006. p. 73-114.

⁴²⁸ Hintsa T, Hintsanen M, Jokela M, Pulkki-Raback L, Keltikangas-Jarvinen L. Divergent influence of different type A dimensions on job strain and effort-reward imbalance. J Occup Environ Med. 2010; 52: 1-7.

⁴²⁹ Hintsanen M, Hintsa T, Widell A, Kivimaki M, Raitakari OT, Keltkangas-Jarvinen L. Negative emotionality, activity, and sociability temperaments predicting long-term job strain and effort–reward imbalance: A 15-year prospective follow-up study. J Psychosom Res. 2011; 71: 90-96.

⁴³⁰ Törnroos M, Hintsanen M, Hintsa T, et al. Associations between Five-Factor Model traits and perceived job strain: a population-based study. J Occup Health Psychol. 2013; 18(4): 492-500.

⁴³¹ Allisey A, Rodwell J, Noblet A. Personality and the effort-reward imbalance model of stress: Individual differences in reward sensitivity. Work Stress. 2012; 26(3): 230-251.
⁴³² Rennesund AB, Saksvik PO. Work performance norms and organisational efficacy as cross-level effects on the relationship between individual perceptions of self-efficacy, overcommitment, and work-related stress. Eur J Work Organ Psychol. 2010; 19(6): 629-653.
⁴³³ Hintsanen M, Puttonen S, Jarvinen P, et al. Cardiac stress reactivity and recovery of novelty seekers. Int J Behav Med. 2009; 16: 236-240.

⁴³⁴ Bolger N, Schilling EA. Personality and the problems of everyday life: the role of neuroticism in exposure and reactivity to daily stressors. J Personal. 1991; 59: 355-386.
⁴³⁵ McCrae RR, Costa PT Jr, Ostendorf F, et al. Nature over nurture: temperament, personality, and life span development. J Pers Soc Psychol. 2000; 78: 173-186.
⁴³⁶ Carver CS, Scheier MF. Perspectives on personality. 4th ed. Boston: Allyn and Bacon; 2000.

⁴³⁷ Roberts BW, DelVecchio WF. The rank-order consistency of personality traits from
childhood to old age: a quantitative review of longitudinal studies. Psychol Bull. 2000; 126: 325.

⁴³⁸ Jokela M, Hakulinen C, Singh-Manoux A, Kivimäki M. Personality change associated with chronic diseases: pooled analysis of four prospective cohort studies. Psychol Med. 2014; 44: 2629-2640.

⁴³⁹ Van Aken MAG, Denissen JJA, Branje SJT, Dubas JS, Goossens L. Midlife concerns and short-term personality change in middle adulthood. Eur J Pers. 2006; 20: 497-513.

⁴⁴⁰ Scollon CN, Diener E. Love, work, and changes in extraversion and neuroticism over time. J Pers Soc Psychol. 2006; 91: 1152-1165.

⁴⁴¹ Roberts BW, Walton K, Bogg T, Caspi A. De-investment in work and non-normative personality trait change in young adulthood. Eur J Pers. 2006; 20: 461-474.

⁴⁴² De Jonge J, VanDer Linden S, Schaufeli W, et al. Factorial invariance and stability of the effort-reward imbalance scales: A longitudinal analysis of two samples with different time lags. Int J Behav Med. 2008; 15: 62-72.

⁴⁴³ Tsutsumi A, Nagami M, Morimoto K, et al. Responsiveness of measures in the effortreward imbalance questionnaire to organisational changes: a validation study. J Psychosom Res. 2002; 52: 249-256.

Segerstrom SC, O'Connor DB. Stress, health and illness: four challenges for the future.Psychol Health. 2012; 27: 128-140.

⁴⁴⁵ Friedman HS. The multiple linkages of personality and disease. Brain Behav Immun.2008; 22: 668-675.

⁴⁴⁶ Caspi A, Roberts BW, Shiner RL. Personality development: stability and change. Annu Rev Psychol. 2005; 56: 453-484.

⁴⁴⁷ Judge TA, Higgins CA, Thoreson CJ, Barrick MR. The Big Five personality traits, general mental ability, and career success across the life span. Pers Psychol. 1999; 52: 621-652.
 ⁴⁴⁸ Roberts BW, Caspi A, Moffitt TE. Work experiences and personality development in young

adulthood. J Pers Soc Psychol. 2003; 84: 582-593.

⁴⁴⁹ Sutin AR, Costa Jr PT, Miech R, Eaton WW. Personality and career success: Concurrent and longitudinal relation. Eur J Pers. 2009; 23: 71-84.

⁴⁵⁰ Sutin AR, Costa PT. Reciprocal influences of personality and job characteristics across middle adulthood. J Pers. 2010; 78(1): 257-288.

⁴⁵¹ Roberts BW, DelVecchio WF. The rank-order consistency of personality traits from childhood to old age: a quantitative review of longitudinal studies. Psychol Bull. 2000; 126: 3-25.

⁴⁵² Watson D, Pennebaker JW, Folger R. Beyond negative affectivity: measuring stress and satisfaction in the workplace. In: Ivancevich JM, editor. Job stress: from theory to suggestion. New York: Haworth Press; 1987. p. 141-157.

⁴⁵³ Lazarus RS, Folkman S. Stress, appraisal and coping. New York: Springer; 1984.
⁴⁵⁴ Wiebe DJ, Fortenberry KT. Mechanisms relating personality and health. In: Vollrath ME, editor. Handbook of personality and health. Chichester: John Wiley Sons; 2006. p. 137-156.
⁴⁵⁵ Lazarus RS, Cohen JB. Environmental stress. In: Altman I, Wohlwill JF, editors. Human behavior and environment. New York: Plenum; 1977.

⁴⁵⁶ Schwartz M, Lerman C, Miller SM, Daly M, Masny A. Coping disposition, perceived risk, and psychological distress among women at increased risk for ovarian cancer. Health Psychol. 1995; 14(3): 232-235.

⁴⁵⁷ Cohen F. Coping. In: Matarazzo JD, Weiss SM, Herd JA, Miller NE, editors. Behavioral health: a handbook of health enhancement and disease prevention. New York: Wiley; 1984.
⁴⁵⁸ Folkman S. Personal control and stress and coping processes: a theoretical analysis. J Pers Soc Psychol. 1984; 48(6): 839-852.

⁴⁵⁹ Park CL, Armeli S, Tennen H. Appraisal–coping goodness of fit: a daily internet study.Pers Soc Psychol Bull. 2004; 30: 558-569.

⁴⁶⁰ Tennen H, Affleck G, Armeli S, Carney MA. A daily process approach to coping. Linking theory, research, and practice. Am Psychol. 2000; 55: 626-636.

⁴⁶¹ Glanz K, Schwartz MD. Stress, coping, and health behaviour. In: Glanz K, Rimer BK,
Viswanath K, editors. Health behavior and health education: theory, research, and practice.
4th ed. San Francisco: John Wiley Sons; 2008.

⁴⁶² Chen MJ, Cunradi C. Job stress, burnout and substance use among urban transit operators: The potential mediating role of coping behaviour. Work Stress. 2008; 22(4): 327-340.
⁴⁶³ Glanz K, Schwartz MD. Stress, coping, and health behaviour. In: Glanz K, Rimer BK,
Viswanath K, editors. Health behavior and health education: theory, research, and practice.
4th ed. San Francisco: John Wiley Sons; 2008.

⁴⁶⁴ Suls J, David JP, Harvey JH. Personality and coping: three generations of research. J Pers. 1996; 64(4): 711-735.

⁴⁶⁵ Lazarus RS. Coping theory and research: past, present, and future. Psychosom Med.1993; 55: 234-247.

⁴⁶⁶ Scheier MF, Carver CS. Effects of optimism on psychological and physical well-being: theoretical overview and empirical update. Cognit Ther Res. 1992; 16: 201-228.

⁴⁶⁷ Carver CS, Pozo C, Harris SD, et al. How coping mediates the effect of optimism on distress: a study of women with early stage breast cancer. J Pers Soc Psychol. 1993; 65(2): 375-390.

⁴⁶⁸ Taylor SE, Kemeny ME, Aspinwall LG, Schneider SG, Rodriguez R, Herbert M. Optimism, coping, psychological distress, and high-risk sexual behavior among men at risk for acquired immunodeficiency syndrome. J Pers Soc Psychol. 1992; 63: 460-473.

⁴⁶⁹ Courneya KS, Bobick TM, Schinke RJ. Does the Theory of Planned Behavior mediate the relation between personality and exercise behavior? Basic Appl Soc Psychol. 1999; 21(4): 317-324.

⁴⁷⁰ Wiebe DJ, Fortenberry KT. Mechanisms relating personality and health. In: Vollrath ME, editor. Handbook of personality and health. Chichester: John Wiley Sons; 2006. p. 137-156.
 ⁴⁷¹ Lazarus RS, Folkman S. Stress, appraisal and coping. New York: Springer; 1984.

⁴⁷² Connor M, Abraham C. Conscientiousness and the Theory of Planned Behavior: toward a more complete model of the antecedents of intentions and behavior. Pers Soc Psychol Bull. 2001; 27: 1547-1561.

⁴⁷³ De Bruijn GJ, Brug J, Van Lenthe FJ. Neuroticism, conscientiousness and fruit consumption: exploring mediator and moderator effects in the theory of planned behaviour.
Psychol Health. 2009; 24(9): 1051-1069.

⁴⁷⁴ Mceachan RC, Sutton S, Myers L. Mediation of Personality Influences on Physical Activity within the Theory of Planned Behaviour. J Health Psychol. 2010; 15: 1170-1180.

⁴⁷⁵ Smith TW, MacKenzie J. Personality and risk of physical illness. Annu Rev Clin Psychol. 2006; 2: 435-467.

⁴⁷⁶ Bradley. Work-induced changes in feelings of mastery. J Psychol. 2010; 144(2): 97-119.

⁴⁷⁷ Ranchor AV, Wardle J, Steptoe A, et al. The adaptive role of perceived control before and after cancer diagnosis: A prospective study. Soc Sci Med. 2010; 70(11): 1825-1831.

⁴⁷⁸ Steptoe A. The significance of personal control in health and disease. In: Steptoe A,
Appels A, editors. Stress, personal control and health. Chichester: John Wiley Sons; 1989. p.
309-319.

⁴⁷⁹ Lazarus RS. Stress and emotion: a new synthesis. London: Free Association Books; 1999.

⁴⁸⁰ Dewe PJ, O'Driscoll, Cooper CL. Coping with work stress: a review and critique. Oxford:

Wiley Blackwell; 2010.

⁴⁸¹ Kohn ML, Schooler C. Work and personality: an inquiry into the impact of social stratification. New Jersey: Ablex Publishing Corp; 1983.

⁴⁸² Pearlin LI, Lieberman MA, Menaghan EG, Mullan JT. The stress process. J Health Soc Behav. 1981; 22: 337-356.

⁴⁸³ Bandura A. Social cognitive theory: an agentic perspective. Annu Rev Psychol. 2001; 52:1-26.

⁴⁸⁴ Siegrist J. Place, social exchange and health: proposed sociological framework. Soc Sci Med. 2000; 51: 1283-1293.

⁴⁸⁵ Wickrama KAS, Surjadi FF, Lorenz FO, Elder Jr GH. The influence of work control trajectories on men's mental and physical health during the middle years: mediational role of personal control. J Gerontol B Soc Sci. 2008; 63: 135-145.

⁴⁸⁶ Payne N, Jones F, Harris PR. The impact of working life on health behavior: the effect of job strain on the cognitive predictors of exercise. J Occup Health Psychol. 2002; 7(4): 342-353.

⁴⁸⁷ Payne N, Jones F, Harris PR. The impact of job strain on the predictive validity of the theory of planned behaviour: an investigation of exercise and healthy eating. Br J Health Psychol. 2005; 10: 115-131.

⁴⁸⁸ Plotnikoff RC, Pickering MA, Flaman LM, Spence JC. The role of self-efficacy on the relationship between the workplace environment and physical activity: a longitudinal mediation analysis. Health Educ Behav. 2010; 37: 170-185.

⁴⁸⁹ Vander Elst T, De Cuyper N, De Witte H. The role of perceived control in the relationship between job insecurity and psychosocial outcomes: moderator or mediator? Stress Health. 2011; 27: 215-227.

⁴⁹⁰ Bordia P, Hunt E, Paulsen N, Tourish D, DiFonzo N. Uncertainty during organisational change: Is it all about control? Eur J Work Organ Psychol. 2004; 13: 345-365.

⁴⁹¹ Paulsen N, Callan VJ, Grice TA, et al. Job uncertainty and personal control during downsizing: A comparison of survivors and victims. Hum Relat. 2005; 58: 463-496.

⁴⁹² Ito JK, Brotheridge CM. An examination of the roles of career uncertainty, flexibility, and control in predicting emotional exhaustion. J Vocat Behav. 2001; 59: 406-424.

⁴⁹³ Schwarzer R. Self-efficacy: thought control of action. Washington DC: Hemisphere; 1992.

⁴⁹⁴ Bandura A. Self-efficacy: the exercise of control. New York: Freeman and Company; 1997.

⁴⁹⁵ Spector PE. Perceived control by employees: a meta-analysis of studies concerning autonomy and participation at work. Hum Relat. 1986; 39: 1005-1016.

⁴⁹⁶ Schwarzer R, Hallum S. Perceived teacher self-efficacy as a predictor of job stress and burnout: mediation analyses. Appl Psychol. 2008; 57: 152-171.

⁴⁹⁷ Hoge T, Bussing A. The impact of sense of coherence and negative affectivity on the work stressor-strain relationship. J Occup Health Psychol. 2004; 9: 195-205.

⁴⁹⁸ Spreitzer G, Mishra A. To stay or to go: voluntary survivor turnover following an

organisational downsizing. J Organ Behav. 2002; 23(6): 707-729.

⁴⁹⁹ Xanthopoulou D, Bakker A, Demerouti E, Schaufeli WB. The role of personal resources in the job demands-resources model. Int J Stress Manag. 2007; 14: 121-141.

⁵⁰⁰ Rennesund AB, Saksvik PO. Work performance norms and organisational efficacy as cross-level effects on the relationship between individual perceptions of self-efficacy, overcommitment, and work-related stress. Eur J Work Organ Psychol. 2010; 19(6): 629-653.
 ⁵⁰¹ Judge TA, Bono JE, Locke EA. Personality and job satisfaction: The mediating role of job characteristics. J Appl Psychol. 2000; 85: 237-249.

⁵⁰² Vander Elst T, Van den Broeck A, De Cuyper N, De Witte H. On the reciprocal relationship between job insecurity and employee well-being: mediation by perceived control? J Occup Organ Psychol. 2014; 87(4): 671-693.

⁵⁰³ Lazarus RS. Stress and emotion: a new synthesis. London: Free Association Books; 1999.

⁵⁰⁴ Bandura A. Self-efficacy: the exercise of control. New York: Freeman and Company; 1997.

⁵⁰⁵ Meier LL, Semmer NK, Elfering A, Jacobshagen N. The double meaning of control: threeway interactions between internal resources, job control, and stressors at work. J Occup Health Psychol. 2008; 13: 244-258.

⁵⁰⁶ Olsson G, Hemström O, Fritzell J. Identifying factors associated with good health and ill health - not just opposite sides of the same coin. Int J Behav Med. 2009; 16: 323-330.
 ⁵⁰⁷ Rodriguez I, Bravo MJ, Peiro JM, Schaufeli W. The Demands-Control-Support model, locus of control and job dissatisfaction: a longitudinal study. Work Stress. 2001; 15: 97-114.
 ⁵⁰⁸ Parkes KR. Locus of control as moderator: an explanation for additive versus interactive findings in the demand- discretion model of work stress? Br J Psychol. 1991; 82: 291-312.
 ⁵⁰⁹ Jimmieson NL, Terry DJ, Callan VJ. A longitudinal study of employee adaptation to organisational change: the role of change-related information and change-related self-efficacy. J Occup Health Psychol. 2004; 9: 11-27.

⁵¹⁰ Lu L, Kao SF, Cooper CL, Spector PE. Managerial stress, locus of control, and job strain in Taiwan and UK: A comparative study. Int J Stress Manag. 2000; 7: 209-226.

⁵¹¹ Brockner J, Spreitzer G, Mishra A, Hochwarter W, Pepper L, Weinberg J. Perceived control as an antidote to the negative effects of layoffs on survivors' organisational commitment and job performance. Admin Sci Q. 2004; 49(1): 76-100.

⁵¹² Rau R, Georgiades A, Fredrikson M. Psychosocial work characteristics and perceived control in relation to cardiovascular rewind at night. J Occup Health Psychol. 2001; 6: 171-181.

⁵¹³ Bethge M, Radoschewski FM. Physical and psychosocial work stressors, health-related control beliefs and work ability: cross-sectional findings from the German Sociomedical Panel of Employees. Int Arch Occup Environ Health. 2010; 83: 241-250.

⁵¹⁴ Schreurs B, van Emmerik H, Notelaers G, De Witte H. Job insecurity and employee health: the buffering potential of job control and job self-efficacy. Work Stress. 2010; 24: 5672.

⁵¹⁵ Xanthopoulou D, Bakker A, Demerouti E, Schaufeli WB. The role of personal resources in the job demands-resources model. Int J Stress Manag. 2007; 14: 121-141.

⁵¹⁶ Marchand A, Demers A, Durand P. Do occupation and work conditions really matter? A longitudinal analysis of psychological distress experiences among Canadian workers. Sociol Health III. 2005; 27: 602-627.

⁵¹⁷ Vander Elst T, De Cuyper N, De Witte H. The role of perceived control in the relationship between job insecurity and psychosocial outcomes: moderator or mediator? Stress Health. 2011; 27: 215-227.

⁵¹⁸ Folkman S. Personal control and stress and coping processes: a theoretical analysis. J Pers Soc Psychol. 1984; 48(6): 839-852.

⁵¹⁹ Siegrist J. Adverse health effects of high-effort/low-reward conditions. J Occup Health Psychol. 1996; 1(1): 27-41.

⁵²⁰ Everson–Rose SA, Clark CJ. Assessment of psychosocial factors in population studies. Steptoe A, editor. Handbook of behavioral medicine: methods and applications. New York: Springer; 2010. p. 291-306.

⁵²¹ Peter R, Siegrist J, Stork J, Mann H, Labrot B. Cigarette smoking and psychosocial work stress in middle managers. Sozial und Präventivmedizin. 1991; 36: 315-321.

⁵²² Puls W, Wienold H, Blank T. The influence of Effort-reward Imbalance in the workplace on the consumption of alcohol: a written survey carried out in metal-working companies. Sucht. 1998; 44: 183-199.

⁵²³ Ota A, Masue T, Yasuda N, et al. Psychosocial job characteristics and smoking cessation: a prospective cohort study using the Demand-Control-Support and Effort-Reward Imbalance job stress models. Nicotine Tob Res. 2010; 12(3): 287-293.

⁵²⁴ Radi S, Ostry A, LaMontagne AD. Job stress and other working conditions: relationships with smoking behaviors in a representative sample of working Australians. Am J Ind Med. 2007; 50: 584-596.

⁵²⁵ Peasey A, Bobak M, Kubinova R, et al. Determinants of cardiovascular disease and other non- communicable diseases in Central and Eastern Europe: rationale and design of the HAPIEE study. BMC Public Health. 2006; 6: 255-265.

⁵²⁶ Rehm J. Measururing quantity, frequency, and volume of drinking. Alcohol Clin Exp Res. 1998; 22: 4-14.

⁵²⁷ Hubacek JA, Pikhart H, Peasey A, Kubinova R, Bobak M. ADH1B polymorphism, alcohol consumption, and binge drinking in Slavic Caucasians: results from the Czech HAPIEE Study. Alcohol Clin Exp Res. 2012; 36(5): 900-905.

⁵²⁸ Rush AJ, Pincus HA, First MB, et al. Handbook of psychiatric measures. Washington DC: American Psychiatric Association; 2009.

⁵²⁹ Hu Y, Pikhart H, Malyutina S, et al. Alcohol consumption and physical functioning among middle-aged and older adults in Central and Eastern Europe: results from the HAPIEE study. Age Ageing. 2015; 44(1): 84-89.

⁵³⁰ Otten F, Bosma H, Swinkels H. Job stress and smoking in the Dutch labour force. Eur J Public Health. 1999; 9: 58-61.

⁵³¹ Kouvonen A, Kivimaki M, Virtanen M, Pentti J, Vahtera J. Work stress, smoking status, and smoking intensity: an observational study of 46190 employees. J Epidemiol Community Health. 2005; 59: 63-69.

⁵³² Willett W, Sampson L, Stampfer MJ, et al. Reproducibility and validity of a semi– quantitative food frequency questionnaire. Am J Epidemiol. 1985; 122: 51-65.

⁵³³ Brunner E, Stallone D, Juneja M, et al. Dietary assessment in Whitehall II: comparison of 7 d diet diary and food-frequency questionnaire and validity against biomarkers. Br J Nutr. 2001; 86: 405-414.

⁵³⁴ McCance R, Widdowson E. McCance & Widdowson's The Composition of Foods. Cambridge: Royal Society of Chemistry; 2002.

⁵³⁵ Boylan S, Welch A, Pikhart H, et al. Dietary habits in three Central and Eastern European countries: the HAPIEE study. BMC Public Health. 2009; 9: 439-452.

⁵³⁶ Jankovic N, Geelen A, Streppel MT, et al. Adherence to a healthy diet according to the World Health Organisation guidelines and all-cause mortality in elderly adults from Europe and the United States. Am J Epidemiol. 2014; 180(10): 978-988.

⁵³⁷ World Health Organisation. Population nutrient intake goals for preventing diet-related chronic diseases. In: Diet, Nutrition and the Prevention of Chronic Diseases. Geneva: World Health Organisation; 2003. p. 54-60.

⁵³⁸ Huijbregts P, Feskens E, Rasanen L, et al. Dietary pattern and 20 year mortality in elderly men in Finland, Italy, and Netherlands: longitudinal cohort study. BMJ. 1997; 315: 13-17.

⁵³⁹ Siegrist J, Starke D, Chandola T, et al. The measurement of effort-reward imbalance at work: European comparisons. Soc Sci Med. 2004; 58: 1483-1499.

⁵⁴⁰ Pikhart H, Bobak M, Pajak A, et al. Psychosocial factors at work and depression in three countries of Central and Eastern Europe. Soc Sci Med. 2004; 58: 1475-1482.

⁵⁴¹ Kivimäki M, Päivi LA, Luukkonen R, et al. Work stress and risk of cardiovascular mortality: prospective cohort study of industrial employees. BMJ. 2002; 325: 857-860.

⁵⁴² Lachman ME, Boone James J. Multiple paths of midlife development. Chicago: University of Chicago; 1997.

⁵⁴³ Lachman ME, Weaver SL. The sense of control as a moderator of social class differences in health and well-being. J Pers Soc Psychol. 1998; 74: 763-773.

⁵⁴⁴ Bobak M, Pikhart H, Rose R, Hertzman C, Marmot M. Socioeconomic factors, material inequalities, and perceived control in self-rated health: cross-sectional data from seven post-communist countries. Soc Sci Med. 2000; 51: 1343-1350.

⁵⁴⁵ Rose R. New Baltic Barometer III: a survey study. Centre for the Study of Public Policy:Studies in Public Policy, No. 284. Glasgow: University of Strathclyde; 1997.

⁵⁴⁶ Radloff LS. The CES-D scale: a self-report depression scale for research in the general population. Appl Psychol Meas. 1977; 1: 385-401.

⁵⁴⁷ Beekman AT, Deeg DJ, Van Limbeek J, et al. Criterion validity of the Center for

Epidemiologic Studies Depression scale (CES-D): results from a community-based sample of older subjects in The Netherlands. Psychol Med. 1997; 27: 231-235.

⁵⁴⁸ Ware JE, Sherbourne CD. The MOS 36-item short-form health survey (SF-36):

conceptual framework and item selection. Med Care. 1992; 30: 473-483.

⁵⁴⁹ Hsieh FY, Block DA, Larsen MD. A simple method of sample size calculation for linear and logistic regression. Stat Med. 1998; 17: 1623-1634.

⁵⁵⁰ Spearman C. General intelligence, objectively determined and measured. Am J Psychol. 1904; 15: 201-293.

⁵⁵¹ Wright S. Correlation and causation. J Agric Res. 1921; 20: 557-585.

⁵⁵² Joreskog KG. A general method for analysis of covariance structures. Biometrika. 1970; 57: 239-251.

⁵⁵³ Muthen BO. A general structural equation model with dichotomous, ordered categorical and continuous latent indicators. Psychometrika. 1984; 49: 115-132.

⁵⁵⁴ MacKinnon DP, Fairchild AJ. Current directions in mediation analysis. Curr Dir Psychol Sci. 2009; 18: 16-20.

⁵⁵⁵ MacKinnon DP, Dwyer JH. Estimation of mediated effects in prevention studies. Eval Rev. 1993; 17: 144-158.

⁵⁵⁶ Sobel ME. Asymptotic confidence intervals for indirect effects in structural equation models. Social Methodol. 1982; 13: 290-312.

⁵⁵⁷ Stone CA, Sobel ME. The robustness of estimates of total indirect effects in covariance structure models estimated by maximum likelihood. Psychometrika. 1990; 55: 337-352.
 ⁵⁵⁸ Springer MD The algebra of random variables. New York: Wiley; 1979.

⁵⁵⁹ MacKinnon DP, Lockwood CM, Hoffman JM, West SG, Sheets V. A comparison of methods to test mediation and other intervening variable effects. Psychol Methods. 2002; 7: 83-104.

⁵⁶⁰ MacKinnon DP, Fairchild AJ, Fritz MS. Mediation analysis. Annu Rev Psychol. 2007; 58: 593-614.

⁵⁶¹ William J, Mackinnon DP. Resampling and distribution of the product methods for testing indirect effects in complex models. Struct Equ Modeling. 2008; 15(1): 23-51.

⁵⁶² Preacher K, Hayes A. Asymptotic and resampling strategies for assessing and comparing indirect effects in multiple mediator models. Behav Res Methods. 2008; 40(3): 879-891.

⁵⁶³ Kline RB. Principles and practice of structural equation modelling. 3rd ed. New York: Guilford Press; 2011.

⁵⁶⁴ Schreiber JB, Nora A, Stage FK, et al. Reporting structural equation modelling and confirmatory factor analysis results: a review. J Education Res. 2006; 99: 323-337.
 ⁵⁶⁵ West SG, Taylor AB, Wu W. Model fit and model selection in structural equation

modelling. In: Hoyle RH, editor. Handbook of structural equation modelling. New York: Guilford Press; 2012.

⁵⁶⁶ Hu L, Bentler PM. Cutoff criteria for fit indexes in covariance structural analysis: conventional criteria versus new alternatives. Struct Equ Modeling. 1999; 6: 1-55.

⁵⁶⁷ Geiser C. Data analysis with Mplus. New York: Guilford Press; 2013.

⁵⁶⁸ MacKinnon DP. Introduction to statistical mediation analysis. New York: Taylor Francis; 2008.

⁵⁶⁹ Baron RM, Kenny DA. The moderator-mediator variable distinction in social psychological research: conceptual, strategic, and statistical considerations. J Pers Soc Psychol. 1986; 51(1): 173-182.

⁵⁷⁰ Taris TW, Kompier MAJ. Games researchers play extreme-groups analysis and mediation analysis in longitudinal occupational health research. Scand J Work Environ Health. 2006; 32(6): 463-472.

⁵⁷¹ Cole DA, Maxwell SE. Testing mediational models with longitudinal data: questions and tips in the use of structural equation modelling. J Abnorm Psychol. 2004; 112: 558-577.

⁵⁷² Taris TW. A primer in longitudinal data analysis. London: Sage; 2000.

⁵⁷³ Hoyle RH. Handbook of structural equation modelling. New York: Guilford Press; 2012.

⁵⁷⁴ Muthen BO. A general structural equation model with dichotomous, ordered categorical and continuous latent indicators. Psychometrika. 1984; 49: 115-132.

⁵⁷⁵ Muthen LK, Muthen BO. Mplus user's guide. Los Angeles: Muthen Muthen; 2004.

⁵⁷⁶ Muthen BO, Asparouhov T. Latent variable analysis with categorical outcomes: multiplegroup and growth modeling in Mplus. Mplus Web Notes. 2002; 4(5): 1-22.

⁵⁷⁷ Maddala GS. Limited-dependent and qualitative variables in econometrics. Cambridge: Cambridge University Press; 1983.

⁵⁷⁸ Pampel FC. Logistic regression: a primer. Sage university papers series on quantitative applications in the social sciences. Thousand Oaks, CA: Sage; 2000.

⁵⁷⁹ MacKinnon DP, Dwyer JH. Estimating mediated effects in prevention studies. Eval Rev. 1993; 17: 144-158.

⁵⁸⁰ MacKinnon DP. Introduction to statistical mediation analysis. New York: Taylor Francis; 2008.

⁵⁸¹ Kline RB. Principles and practice of structural equation modelling. 3rd ed. New York: Guilford Press; 2011.

⁵⁸² Muthen BO. A general structural equation model with dichotomous, ordered categorical and continuous latent indicators. Psychometrika. 1984; 49: 115-132.

⁵⁸³ Hoyle RH. Handbook of structural equation modelling. New York: Guilford Press; 2012.

⁵⁸⁴ Tabachnick BG, Fidell LS. Using multivariate statistics. 5th Edition. Pearson Education;2007.

⁵⁸⁵ Taris TW, Kompier MAJ. Games researchers play extreme-groups analysis and mediation analysis in longitudinal occupational health research. Scand J Work Environ Health. 2006; 32(6): 463-472.

⁵⁸⁶ MacKinnon DP, Lockwood CM, Hoffman JM, West SG, Sheets V. A comparison of methods to test mediation and other intervening variable effects. Psychol Methods. 2002; 7: 83-104.

⁵⁸⁷ Taris TW, Kompier MAJ. Games researchers play extreme-groups analysis and mediation

analysis in longitudinal occupational health research. Scand J Work Environ Health. 2006; 32(6): 463-472.

⁵⁸⁸ Pampel FC. Logistic regression: a primer. Sage University papers series on quantitative applications in the social sciences. Thousand Oaks, CA: Sage; 2000.

⁵⁸⁹ Willett W, Sampson L, Stampfer MJ, et al. Reproducibility and validity of a semi-quantitative food frequency questionnaire. Am J Epidemiol. 1985; 122: 51-65.

⁵⁹⁰ World Health Organisation. Population nutrient intake goals for preventing diet-related chronic diseases. In: Diet, Nutrition and the Prevention of Chronic Diseases. Geneva: World Health Organisation; 2003. p. 54-60.

⁵⁹¹ Huijbregts P, Feskens E, Rasanen L, et al. Dietary pattern and 20 year mortality in elderly men in Finland, Italy, and Netherlands: longitudinal cohort study. BMJ. 1997; 315: 13-17.
 ⁵⁹² World Health Organisation. Population nutrient intake goals for preventing diet-related chronic diseases. In: Diet, Nutrition and the Prevention of Chronic Diseases. Geneva: World Health Organisation; 2003. p. 54-60.

⁵⁹³ MacKinnon DP, Lockwood CM, Hoffman JM, West SG, Sheets V. A comparison of methods to test mediation and other intervening variable effects. Psychol Methods. 2002; 7: 83-104.

⁵⁹⁴ Hoyle RH. Handbook of structural equation modelling. New York: Guilford Press; 2012.
 ⁵⁹⁵ Rothman KJ, Greenland S, Lash TL. Modern epidemiology. Philadelphia: Lippincott
 Williams Wilkins; 2008.

⁵⁹⁶ World Health Organisation. Global status report on alcohol and health. Geneva: World Health Organisation; 2011.

⁵⁹⁷ World Health Organisation. Global InfoBase. Geneva: World Health Organisation; 2011. ⁵⁹⁸ Börsch-Supan A, Brugiavini A, Jürges H, et al. First results from the Survey of Health, Ageing and Retirement in Europe (2004-2007). Starting the longitudinal dimension.

Mannheim: Mannheim Research Institute for the Economics of Aging; 2008.

⁵⁹⁹ Banks J, Lessof C, Nazroo J, et al. Financial Circumstances, Health and Well-Being of the Older Population in England: The 2008 English Longitudinal Study of Ageing (Wave 4). London: The Institute for Fiscal Studies; 2010.

⁶⁰⁰ Peasey A, Bobak M, Kubinova R, et al. Determinants of cardiovascular disease and other non- communicable diseases in Central and Eastern Europe: rationale and design of the HAPIEE study. BMC Public Health. 2006; 6: 255-265.

Little RJA, Rubin DB. Statistical analysis with missing data, 2nd ed. New York: Wiley. 2002.
 Rothman KJ, Greenland S, Lash TL. Modern epidemiology. Philadelphia: Lippincott
 Williams Wilkins; 2008.

⁶⁰³ Susser E, Schwartz S, Morabia A, Bromet EJ. Psychiatric epidemiology: searching for the causes of mental disorders. New York: Oxford University Press; 2006.

⁶⁰⁴ Flegal KM, Keyl PM, Nieto FJ. Differential misclassification arising from non-differential errors in exposure measurement. Am J Epidemiol. 1991; 134: 1233-1244.

⁶⁰⁵ Pikhart H. Social and psychosocial determinants of self-rated health in Central and

Eastern Europe. Boston: Kluwer Academic Publishers; 2002.

⁶⁰⁶ Stockwell T, Donath S, Cooper-Stanbury M, Chikritzhs T, Catalano P, Mateo C. Underreporting of alcohol consumption in household surveys: a comparison of quantity-frequency, graduated-frequency and recent recall. Addiction. 2004; 99: 1024-1033.

⁶⁰⁷ Bobrova N, West R, Malyutina D, Malyutina S, Bobak M. Gender differences in drinking practices in middle aged and older Russians. Alcohol Alcohol. 2010; 45: 573-580.

⁶⁰⁸ Hu Y, Pikhart H, Malyutina S, et al. Alcohol consumption and physical functioning among middle-aged and older adults in Central and Eastern Europe: results from the HAPIEE study. Age Ageing. 2015; 44(1): 84-89.

⁶⁰⁹ Bobak M, Gilmore A, McKee M, Rose R, Marmot M. Changes in smoking prevalence in Russia, 1996–2004. Tob Control. 2006; 15: 131-135.

⁶¹⁰ Cade J, Thompson R, Burley V, Warm D. Development, validation and utilization of foodfrequency questionnaires—a review. Public Health Nutr. 2002; 5: 567-587.

⁶¹¹ Stefler D, Pikhart H, Jankovic N, et al. Healthy diet indicator and mortality in Eastern
European populations: prospective evidence from the HAPIEE cohort. Eur J Clin Nutr. 2014;
68: 1346-1352.

⁶¹² Jankovic N, Geelen A, Streppel MT, et al. Adherence to a healthy diet according to the World Health Organisation guidelines and all-cause mortality in elderly adults from Europe and the United States. Am J Epidemiol. 2014; 180(10): 978-988.

⁶¹³ Rothman KJ, Greenland S, Lash TL. Modern epidemiology. Philadelphia: Lippincott Williams Wilkins; 2008.

⁶¹⁴ Cole DA, Maxwell SE. Testing mediational models with longitudinal data: questions and tips in the use of structural equation modelling. J Abnorm Psychol. 2004; 112: 558-577.
 ⁶¹⁵ Taris TW, Kompier MAJ. Games researchers play extreme-groups analysis and mediation analysis in longitudinal occupational health research. Scand J Work Environ Health. 2006; 32(6): 463-472.

⁶¹⁶ Heikkilä K, Nyberg ST, Fransson EI, et al. Job strain and alcohol intake: a collaborative meta-analysis of individual–participant data from 140 000 men and women. PLoS ONE. 2012; 7(7): e40101.

⁶¹⁷ Crum RM, Muntaner C, Eaton WW, Anthony JC. Occupational stress and the risk of alcohol abuse and dependence. Alcohol Clin Exp Res. 1995; 19: 647-655.

⁶¹⁸ Marchand A, Blanc ME. Occupation, work organisation conditions, and alcohol misuse in Canada: an 8-year longitudinal study. Subst Use Misuse. 2011; 46: 1003-1014.

⁶¹⁹ Head J, Stansfeld SA, Siegrist J. The psychosocial work environment and alcohol dependence: a prospective study. Occup Environ Med. 2004; 61: 219-224.

⁶²⁰ Bobak, Pikhart, Kubinova, et al. The association between psychosocial characteristics at work and problem drinking: a cross-sectional study of men in three Eastern European urban populations. Occup Environ Med. 2005; 62: 546-550.

⁶²¹ Albertsen K, Borg V, Oldenburg B. A systematic review of the impact of work environment on smoking cessation, relapse and amount smoked. Prev Med. 2006; 43: 291-305. ⁶²² Heikkilä K, Nyberg ST, Fransson EI, et al. Job strain and tobacco smoking: An individualparticipant data meta-Analysis of 166 130 adults in 15 European studies. PLoS ONE. 2012; 7(7): e35463.

⁶²³ Kouvonen A, Kivimäki M, Virtanen M, Pentti J, Vahtera J. Work stress, smoking status, and smoking intensity: an observational study of 46190 employees. J Epidemiol Community Health. 2005; 59: 63-69.

⁶²⁴ Radi S, Ostry A, LaMontagne AD. Job stress and other working conditions: relationships with smoking behaviors in a representative sample of working Australians. Am J Ind Med. 2007; 50: 584-596.

⁶²⁵ Ota A, Masue T, Yasuda N, et al. Psychosocial job characteristics and smoking cessation: a prospective cohort study using the Demand-Control-Support and Effort-Reward Imbalance job stress models. Nicotine Tob Res. 2010; 12(3): 287-293.

⁶²⁶ Lallukka T, Sarlio-Lähteenkorva S, Roos E, et al. Working conditions and health behaviours among employed women and men: the Helsinki Health Study. Prev Med. 2004; 38(1): 48-56.

⁶²⁷ Hellerstedt WL, Jeffery RW. The association of job strain and health behaviours in men and women. Int J Epidemiol. 1997; 26: 575-583.

⁶²⁸ Kawakami N, Tsutsumi A, Haratani T, et al. Job strain, worksite support, and nutrient intake among employed Japanese men and women. J Epidemiol. 2006; 16(2): 79-89.
⁶²⁹ Jankovic N, Geelen A, Streppel MT, et al. Adherence to a healthy diet according to the World Health Organisation guidelines and all-cause mortality in elderly adults from Europe

and the United States. Am J Epidemiol. 2014; 180(10): 978-988.

⁶³⁰ Ota A, Masue T, Yasuda N, et al. Psychosocial job characteristics and smoking cessation: a prospective cohort study using the Demand-Control-Support and Effort-Reward Imbalance job stress models. Nicotine Tob Res. 2010; 12(3): 287-293.

⁶³¹ Radi S, Ostry A, LaMontagne AD. Job stress and other working conditions: relationships with smoking behaviors in a representative sample of working Australians. Am J Ind Med. 2007; 50: 584-596.

⁶³² Siegler IC, Peterson BL, Barefoot JC, Williams RB. Hostility during late adolescence predicts coronary risk factors at mid-life. Am J Epidemiol. 1992; 136: 146-154.

⁶³³ Munafò MR, Black S. Personality and smoking status: a longitudinal analysis. Nicotine Tob Res. 2007; 9(3): 397-404.

⁶³⁴ Armitage CJ, Conner M. Efficacy of the theory of planned behaviour: a meta-analytic review. Br J Soc Psychol. 2001; 40: 471-499.

⁶³⁵ Grembowski D, Patrick D, Diehr P, et al. Self-efficacy and health behavior among older adults. J Health Soc Behav. 1993; 34(2): 89-104.

⁶³⁶ Siegrist J. Effort-reward imbalance at work and health. In: Perrewe PL, Ganster DC, editors. Historical and current perspectives on stress and health. Amsterdam: Elsevier; 2002.
p. 261-291.

⁶³⁷ Van Vegchel N, De Jonge J, Bosma H, Schaufeli W. Reviewing the effort-reward

imbalance model: drawing up the balance of 45 empirical studies. Soc Sci Med. 2005; 60: 1117-1131.

⁶³⁸ Israel BA, Baker EA, Goldenhar LM, Heaney CA, Schurman SJ. Occupational stress, safety, and health: conceptual framework and principles for effective prevention interventions.
J Occup Health Psychol. 1996; 1: 261-286.

⁶³⁹ Spector PE, Zapf D, Chen PY, Frese M. Why negative affectivity should not be controlled in job stress research: Don't throw out the baby with the bath water. J Organ Behav. 2000; 21: 79-95.

⁶⁴⁰ Hintsa T, Hintsanen M, Jokela M, Pulkki-Raback L, Keltikangas-Jarvinen L. Divergent influence of different type A dimensions on job strain and effort-reward imbalance. J Occup Environ Med. 2010; 52: 1-7.

⁶⁴¹ Hintsanen M, Hintsa T, Widell A, Kivimaki M, Raitakari OT, Keltkangas-Jarvinen L. Negative emotionality, activity, and sociability temperaments predicting long-term job strain and effort–reward imbalance: A 15-year prospective follow-up study. J Psychosom Res. 2011; 71: 90-96.

⁶⁴² Siegrist J, Starke D, Chandola T, et al. The measurement of Effort-Reward Imbalance at work: European comparisons. Soc Sci Med. 2004; 58: 1483-1499.

⁶⁴³ Roberts BW, Del Vecchio WF. The rank-order consistency of personality traits from
childhood to old age: a quantitative review of longitudinal studies. Psychol Bull. 2000; 126: 325.

⁶⁴⁴ Van Aken MAG, Denissen JJA, Branje SJT, Dubas JS, Goossens L. Midlife concerns and short-term personality change in middle adulthood. Eur J Pers. 2006; 20: 497-513.

⁶⁴⁵ Scollon CN, Diener E. Love, work, and changes in extraversion and neuroticism over time. J Pers Soc Psychol. 2006; 91: 1152-1165.

⁶⁴⁶ Caspi A, Roberts BW, Shiner RL. Personality development: stability and change. Annu Rev Psychol. 2005; 56: 453-484.

⁶⁴⁷ Baron RM, Kenny DA. The moderator-mediator variable distinction in social psychological research: conceptual, strategic, and statistical considerations. J Pers Soc Psychol. 1986; 51: 1173-1182.

⁶⁴⁸ Kendler KS, Gardner CO, Prescott CA. Personality and the experience of environmental adversity. Psychol Med. 2003; 33: 1193-1202.

⁶⁴⁹ Caspi A, Roberts BW, Shiner RL. Personality development: stability and change. Annu Rev Psychol. 2005; 56: 453-484.

⁶⁵⁰ Stansfeld S, Pike C, McManus S, et al. Occupations, work characteristics and common mental disorder. Psychol Med. 2013; 43: 961-973.

⁶⁵¹ Roberts BW, Del Vecchio WF. The rank-order consistency of personality traits from
childhood to old age: a quantitative review of longitudinal studies. Psychol Bull. 2000; 126: 325.

⁶⁵² Perlman F, Bobak M, Steptoe A, Rose R, Marmot M. Do health control beliefs predict behaviour in Russians? Prev Med. 2003; 37: 73-81.

⁶⁵³ Steptoe A, Wardle J. Locus of control and health behaviour revisited: A multivariate analysis of young adults from 18 countries. Br J Psychol. 2001; 92: 659-672.

⁶⁵⁴ Bennett P, Moore L, Smith A, Murphy S, Smith C. Health locus of control and value for health as predictors of dietary behaviour. Psychol Health. 1994; 10(1): 41-54.

⁶⁵⁵ Glanz K, Rimer BK, Viswanath K. Health behavior and health education: theory, research, and practice. 4th ed. San Francisco: John Wiley Sons; 2008.

⁶⁵⁶ Steptoe A, Appels A. Stress, personal control and health. UK: John Wiley & Sons press; 1989.

⁶⁵⁷ Pikhart H. Social and psychosocial determinants of self-rated health in Central and Eastern Europe. Boston: Kluwer Academic Publishers; 2002.

⁶⁵⁸ Wiebe DJ, Fortenberry KT. Mechanisms relating personality and health. In: Vollrath ME, editor. Handbook of personality and health. Chichester: John Wiley Sons; 2006. p. 137-156.
⁶⁵⁹ Lazarus RS. Coping theory and research: past, present, and future. Psychosom Med. 1993; 55: 234-247.

⁶⁶⁰ Lazarus RS. Stress and emotion: a new synthesis. London: Free Association Books;1999.

⁶⁶¹ Payne N, Jones F, Harris PR. The impact of working life on health behavior: the effect of job strain on the cognitive predictors of exercise. J Occup Health Psychol. 2002; 7(4): 342-353.

⁶⁶² Rennesund AB, Saksvik PO. Work performance norms and organisational efficacy as cross-level effects on the relationship between individual perceptions of self-efficacy, overcommitment, and work-related stress. Eur J Work Organ Psychol. 2010; 19(6): 629-653.
⁶⁶³ Rau R, Georgiades A, Fredrikson M. Psychosocial work characteristics and perceived control in relation to cardiovascular rewind at night. J Occup Health Psychol. 2001; 6: 171-181.

⁶⁶⁴ Parkes KR. Locus of control as moderator: an explanation for additive versus interactive findings in the demand- discretion model of work stress? Br J Psychol. 1991; 82: 291-312.
 ⁶⁶⁵ Perrewe PL, Zellars KL. An examination of attributions and emotions in the transactional approach to the stress process. J Organiz Behav. 1999; 20: 739-752.

⁶⁶⁶ Dewe PJ, O'Driscoll MP, Cooper CL. Coping with work stress: a review and critique. Chichester: John Wiley Sons; 2010.

⁶⁶⁷ Daniels K. Stress and well-being are still issues and something still needs to be done: Why agency and interpretation are important for policy and practice. In: Hodgkinson GP, Ford JK, editors. International review of industrial and organisational psychology. Chichester: John Wiley Sons; 2011.

⁶⁶⁸ Merzel C, D'afflitti J. Reconsidering community-based health promotion: promise, performance, and potential. Am J Public Health. 2003; 93(4): 557-574.

⁶⁶⁹ Cahill K, Moher M, Lancaster T. Workplace intervention for smoking cessation. Cochrane Databases Syst Rev. 2008; 8(4): CD003440.

⁶⁷⁰ Sorenson G, Linnan L, Hunt MK. Worksite–based research and initiatives to increase fruit

and vegetable consumption. Prev Med. 2004; 39: 94-100.

⁶⁷¹ Conn VS, Hafdahl AR, Cooper RS, Brown LM, Lusk SL. Meta-analysis of workplace physical activity interventions. Am J Prev Med. 2009; 37(4): 330-339.

⁶⁷² Albertsen K, Hannerz H, Borg V, Burr H. Work environment and smoking cessation over a5–year period. Scand J Public Health. 2004; 32(3): 164-171.

⁶⁷³ Sorensen G, Stoddard AM, LaMontagne AD, et al. A comprehensive worksite cancer prevention intervention: behavior change results from a randomized controlled trial. Cancer Cause Control. 2002; 13(6): 493-502.

⁶⁷⁴ Sorenson G, McLellan D, Dennerlein JT, et al. Integration of health protection and health promotion: rationale, indicators, and metrics. J Occup Environ Med. 2013; 55(12): 12-18.
⁶⁷⁵ Tsutsumi A, Kawakami N. A review of empirical studies on the model of effort–reward imbalance at work: reducing occupational stress by implementing a new theory. Soc Sci Med. 2004; 59: 2335-2359.

⁶⁷⁶ Bambra C, Gibson M, Sowden A, Wright K, Whitehead M, Petticrew M. Working for health? Evidence from systematic reviews on the effects on health and health inequalities of organisational changes to the psychosocial work environment. Prev Med. 2009; 48: 454-461.
⁶⁷⁷ Bhui KS, Dinos S, Stansfeld SA, White PD. A synthesis of the evidence for managing stress at work: a Review of the reviews reporting on anxiety, depression, and absenteeism. J Environ Public Health. 2012; Article ID 515874.

⁶⁷⁸ Montano D, Hoven H, Siegrist J. Effects of organisational-level interventions at work on employees' health: a systematic review. BMC Public Health. 2014; 14: 135-144.

⁶⁷⁹ Aust B, Peter R, Siegrist J. Stress management in bus drivers: a pilot study based on the model of effort–reward imbalance. Int J Stress Manag. 1997; 4: 297-305.

⁶⁸⁰ Bourbonnais R, Jauvin N, Dussault J, Vezina M. Evaluation of an intervention to prevent mental health problems among correctional officers. In: Biron C, Karanika-Murray M, Copper CL, editors. Improving organisational interventions for stress and well-being. East Sussex: Routledge; 2012. p.187-215.

⁶⁸¹ Bourbonnais R, Brisson C, Vinet A, Vezina M, Lower A. Development and implementation of a participative intervention to improve the psychosocial work environment and mental health in an acute care hospital. Occup Environ Med. 2006; 63: 326-334.

⁶⁸² Lieb K, Zanarini MC, Schmahl C, Linehan MM, Bohus M. Borderline personality disorder. Lancet. 2004; 364: 453-461.

⁶⁸³ Beck AT, Haigh E. Advances in cognitive theory and therapy: the generic cognitive model. Annu Rev Clin Psychol. 2014; 10: 1-24.

⁶⁸⁴ Beck AT, Davis DD, Freeman A. Cognitive therapy of personality disorders. 3rd ed. New York: Guilford Press; 2015.

⁶⁸⁵ Richardson KM, Rothstein HR. Effects of occupational stress management intervention programs: a meta-analysis. J Occup Health Psychol. 2008; 13: 69-93.

⁶⁸⁶ Aust B, Peter R, Siegrist J. Stress management in bus drivers: a pilot study based on the model of effort–reward imbalance. Int J Stress Manag. 1997; 4: 297-305.

⁶⁸⁷ Williams RB, Williams VP. The prevention and treatment of hostility. In: Vollrath ME, editor.
Handbook of personality and health. Chichester: John Wiley Sons; 2006. p. 259-276.
⁶⁸⁸ Friedman M, Powell LH, Thoresen CE, Ulmer D, Price V, Gill JJ. Effect of discontinuance of type A behavioural counselling on type A behaviour and cardiac recurrence rate of post myocardial infarction patients. Am Heart J. 1987; 114(3): 483-490.

⁶⁸⁹ Powell LH, Shaker LA, Jones BA, Vaccarino LV, Thoresen CE, Pattillo JR. Psychosocial predictors of mortality in 83 women with premature myocardial infarction. Psychosom Med. 1993; 55(5): 426-433.

⁶⁹⁰ Spector PE, Zapf D, Chen PY, Frese M. Why negative affectivity should not be controlled in job stress research: don't throw out the baby with the bath water J Organiz Behav. 2000; 21: 79-95.

⁶⁹¹ Semmer NK. Personality, stress, and coping. In: Vollrath ME, editor. Handbook of personality and health. Chichester: John Wiley Sons; 2006. p. 73-114.

⁶⁹² Ivancevich JM, Matteson MT, Freedman SM, Phillips JS. Worksite stress management interventions. Am Psychol. 1990; 45: 252-261.

⁶⁹³ Hardy S, Carson J, Thomas B. Occupational stress: personal and professional approaches. Cheltenham: Stanley Thornes; 1998.

⁶⁹⁴ Dewe PJ, O'Driscoll, Cooper CL. Coping with work stress: a review and critique. Chichester: John Wiley Sons; 2010.

⁶⁹⁵ Giga SI, Noblet AJ, Faragher B, Cooper CL. The UK perspective: a review of research on organisational stress management interventions. Aust Psychol. 2003; 38: 158-164.

⁶⁹⁶ Richardson KM, Rothstein HR. Effects of occupational stress management intervention programs: a meta-analysis. J Occup Health Psychol. 2008; 13(1): 69-93.

⁶⁹⁷ Jex SM, Elacqua TC. Time management as a moderator of relation between stressors and employee strain. Work Stress. 1999; 13(2): 182-191.

⁶⁹⁸ Limm H, Gundel H, Heinmuller M, Marten-Mittag B, Nater UM, Siegrist J, Angerer P. Stress management interventions in the workplace improve stress reactivity: a randomised controlled trial. Occup Environ Med. 2011; 68: 126-133.

⁶⁹⁹ Ashford S, Edmunds J, French DP. What is the best way to change self-efficacy to promote lifestyle and recreational physical activity? A systematic review with meta-analysis. Br J Health Psychol. 2010; 15: 265-288.

⁷⁰⁰ Ajzen I. Behavioral interventions: design and evaluation guided by the theory of planned behaviour. In: Mark MM, Donaldson SI, Campbell BC, editors. Social psychology for program and policy evaluation. New York: Guilford; 2011. p. 74-100.

⁷⁰¹ Walach H, Nord E, Zier C, Dietz-Waschkowski B. Mindfulness–based stress reduction as a method for personnel development: a pilot evaluation. Int J Stress Manag. 2007; 14(2): 188-198.

⁷⁰² Okechukwu C, Davison K, Emmons K. Changing health behaviours in a social context. In: Berkman LF, Kawachi I, Glymour MM, editors. Social epidemiology. New York: Oxford University Press; 2014. ⁷⁰⁴ Mellor N, Karanika-Murray M, Waite E. Taking a multi-faceted, multi-level, and integrated perspective for addressing psychosocial issues at the workplace. In: Biron C, Karanika-Murray M, Copper CL, editors. Improving organisational interventions for stress and wellbeing. East Sussex: Routledge; 2012. p. 39-58.

⁷⁰³ LaMontagne AD, Keegel T, Louie A, Ostry A, Landsbergis P. A systematic review of the job-stress intervention evaluation literature, 1990–2005. Int J Occup Environ Health. 2007;
13: 268-280.

Appendix 1.

Personal questionnaire in the HAPIEE study wave 1

The health consequences of the profound social and economic changes in our country that have started in 1989 are not well understood. The present study has been set up to investigate and monitor the health impacts of these changes.

The study has two principal components:

- a) An assessment of your present health and your other characteristics that do or can influence health; this will be done by a questionnaire and a short medical examination, including a blood sample.
- b) Monitoring the changes in your health in the future; this will be done by a short annual postal questionnaire and by using data available in national health statistics registers.

We would like to assure you that all your personal data will be kept confidential and will not be available to any individuals or institutions except the core investigators of this study. Any published results will not identify individuals.

If you agree to participate in the study, you can change your mind and leave the study at any time in the future without giving any reason for your withdrawal. You will be asked for another specific consent before we take a blood sample.

Do you agree to participate in the study and complete the questionnaires? Please circle Yes or No.

Yes

No

No

Do you give your consent for us to use data on major changes in your health that are contained in national health statistics? *Please circle Yes or No*.

Yes

To be able to use national data on major changes in your health, we need your ID number. Please enter your ID number. _____/____/

Please date and sign this form below.

Signature of participant :	
----------------------------	--

Print name:

Date:

Thank you for your time.

1. Place of birth (region):

2. Sex:

- 1. Male
- 2. Female

3. What is your highest completed level of education?

- 1. Incomplete primary or no formal education
- 2. Primary
- 3. Vocational (apprenticeship)
- 4. Secondary
- 5. University (degree)

4. What is your marital status?

- 1. Single
- 2. Married
- 3. Cohabiting
- 4. Divorced
- 5. Widowed

About your health

5. What is your height in cm?		•	
6. What is your weight in kg?		•	

7. Over the last 12 months, would you say your health has been:

- 1. Very good
- 2. Good
- 3. Average
- 4. Poor
- 5. Very poor

8. Here is a list of activities that you might do during a typical day. Does your health now limit your ability in these activities? If so, how much?

	Yes, limited a lot	Yes, limited a little	No, not limited at all
Vigorous activities, such as running, lifting heavy objects, participating in strenuous sports	1	2	3
Moderate activities, such as moving a table, pushing a vacuum cleaner		2	3
Lifting or carrying bag of groceries		2	3
Climbing several flights of stairs		2	3
Climbing one flight of stairs			3
Bending, kneeling or stooping		\square_2	3

	Yes, limited a lot	Yes, limited a little	No, not limited at all
Walking two kilometres	1	2	3
Walking one kilometre	1	2	3
Walking one hundred metres	1	2	3
Bathing and dressing yourself	1	2	3

9. Do you have any long-term health problems for which medical treatment has been sought over last 12 months?

1. Yes

2. No

10. Have any of the following diseases ever been diagnosed in you <u>by a doctor</u> and have you ever been <u>hospitalised</u> for this disease?

	Yes, diagnosed and hospitalised	Yes, diagnosed, never hospitalised	No or do not know
heart attack / acute myocardial infarction			3
angina / ischaemic heart disease			
Stroke			3
chronic respiratory disease		2	3
Cancer		2	3
stomach ulcer			3
gallbladder disease			3
kidney stones			3
Asthma		2	3
atopic eczema	1	2	3
other allergy	1	2	3
hay fever	1	2	3
disease of spine or joints	1	2	3

11. Do you usually cough on most days for as much as 3 months each year?

- 1. Yes
- 2. No

12. Do you usually bring up any **phlegm** from your chest first thing in the morning for as much as 3 months each year?

- Yes 1.
- 2. No

Injuries and accidents

13. In the past 12 months have you been injured or have you had an accident serious enough to contact a doctor?

- 1. Yes
- 2. No, please go to the question 20

14. How many different times in the past 12 months were you injured or have you had an accident serious enough to contact a doctor?

Please would you tell us about the MOST SERIOUS INJURY OR ACCIDENT you have had in last 12 months.

15. Place: Where were you when you were injured or had your accident?

- 1. Home (yours or someone else's home)
- 2. Work
- 3. Road
- 4. Other
- 5. Unknown

16. Mechanism: How were you hurt or how was the injury inflicted?

- Traffic injury 2. Fall 1. 3. 4. Other blunt force Stab or cut 5. 6. Fire or hot subject or substance (e.g. scald) Firearm 8. 7. Chocking or hanging Drowning or submersion
 - 10. Poisoning
 - Struck by or against 12.
 - 14. Unknown

17. Intent: Was this accident:

Other

Suffocation

11. Machinery related

9.

13.

- 1. 2. Unintentional
- 3. Intentional
- 5. Unknown

- Self-harm
- 4. Other

18. What w	as the nature of your i	njury?		
1.	Fracture	2.	Sprain or strain	
3.	Cut, bite or open wour	nd 4.	Bruise	
5.	Burn	6.	Concussion	
7.	Organs system injury	8.	Other – please sp	pecify
9.	Unknown			
19. Did you	u require medical treat n	nent as a result of	your injury or accide	ent?
1.	No treatment required	2.	Treated as outpa	tient, discharged
3.	Admitted to hospital	4.	Other – please s	pecify
5.	Unknown			
20 . Have ye 1. 2.		octor that you have S, have you been ta lood pressure in th	aking drugs for	re? 1. Yes 2. No 3. Don't know
21 . Have y	ou every been told by a o	doctor that you hav	ve diabetes?	
1. 2.	Yes If YE No	S , how are you trea	ated?	 Only by diet By diet and insulin By diet and tablets By diet, tablets and insulin No treatment
22. Have y	ou ever been told by a de	octor that you have	e high blood choles	terol?
1. 2.	Yes If YE S No	S , how are you trea	ated?	 Only by diet By diet and tablets Tablets only No treatment

23. Are you under **long-term treatment or medical care** for any medical condition, except for high blood pressure, high cholesterol or diabetes?

- 1. Yes If **YES**, please give details:
- 2. No

24. Do you take any **vitamins or mineral supplements**?

- 1. Yes (regularly, at least 3 times per week)
- 2. Yes (irregularly, less than 3 times per week)
- 2. No

25. If YES, do these supplements contain vitamin C?

- 1. Yes
- 2. No

26. Can you seek medical advice when you need it?

- 1. Anytime I want to and without any difficulty
- 2. Usually, but it can be complicated e.g. difficult to get to doctor, doctor busy, or can't
- 3. Not usually, too complicated and often I do not bother
- 4. No, it is either too difficult to get to the doctor, the doctor is too busy, or it is too expensive

27. Where do you go, when you want medical advice and it is not an emergency?

- 1. State funded general practitioner
- 2. State funded specialist
- 3. State hospital
- 4. Private general practitioner
- 5. Private funded specialist
- 6. Private hospital
- 7. Other

28. Do you have to **pay** to see the doctor?

- 1. Yes
- 2. No

29. At any time in the last 6 months, have you been prescribed a medicine and not been able to buy it?

- 1. No, I can always obtain the medicines that I need
- 2. Yes, it was unavailable
- 3. Yes, it was too expensive
- 4. No, I have not been prescribed any medicines

30. How many times in the last 12 months did you seek medical advice?

31. Did any of your **parents or siblings suffer** from any of the **following diseases?**

	Did parents or siblings suffer from disease?		IF YES , did a parent or sibling have onset before the age of 60°		
	Yes	No	Yes	No	
Heart disease (infarction, angina)	1		1	2	
Stroke	1	2	1	2	
Diabetes	1			2	
Neoplasms	1		1	2	
Allergy		2			

32. Have you ever had any pain or discomfort in your chest?

- 1. Yes
- 2. No

If no, please, women proceed to Question 39, men proceed to Question 45.

33. Do you get it when you walk uphill or hurry or do physically demanding work?

- 1. Yes
- 2. No
- 3. Never hurries or walks uphill or does physically demanding work

34. Do you get it when you walk at an ordinary pace on the level?

- 1. Yes
- 2. No

35. What do you do if you get it while you are walking?

- 1. Stop or slow down
- 2. Carry on at the same pace
- 3. Take nitroglycerine

36. If you stand still, what happens to it?

- 1. Relieved
- 2. Not relieved

37. If relieved, how soon?

- 1. 10 minutes or less
- 2. More than 10 minutes

38. Can you specify where such pain or discomfort appeared? (Please choose all appropriate options)

- 1. Sternum (upper or middle)
- 2. Sternum (lower)
- 3. Left anterior chest
- 4. Left arm
- 5. Neck
- 6. Other Please specify:

Only for women

39. Do you still have periods?

- 1. Yes, regularly
- 2. Yes, irregularly **If YES**, go to question 42.
- 3. No

40. How old were you when the periods stopped?

Years

41. What was the cause of the menopause?

- 1. Natural
- 2. Surgical (operation)

42. Have you ever used hormonal contraception?

- 1. No, never
- 2. Yes, but I no longer use it
- 3. Yes and I still use it

43. Have you ever had hormonal replacement therapy?

- 1. Yes
- 2. No

44. If YES, are you still taking hormonal replacement therapy?

- 1. Yes
- 2. No

Health behaviours

45. How many hours during a typical week, <u>except when at work</u> , do you engage in physically demanding activities, such as housework, gardening, maintenance of the house (DIY) etc?	
46. How many hours during a typical week do you engage in sports, games or hiking?	
47. Do you smoke cigarettes?	
 Yes, regularly, at least one cigarette a day on average Yes, occasionally, less than one cigarette a day No, I smoked in the past but I stopped No, I have never smoked 	
48. For current and past smokers: How many cigarettes a day do you smoke now (or you used to smoke, if you stopped)?	
49. For current and past smokers: How old were you when you started smoking?	Years
50. For past smokers: How old were you when you stopped smoking?	Years
51. For past smokers: When did you stop smoking?	Calendar year

52. The next few questions are about how much wine, beer and spirits you may have had during the last 12 months. When we say one drink, we mean 0.5 litre of beer, 2 dl glass of wine, or 5 cl of spirits. Please answer each question below - ie. cross a square in each row - to indicate how often you had that amount of alcohol during one day.

Here is an example how to calculate correct amount of alcohol on a single occasion: if you had 0.7 l bottle of wine AND two 5cl measures of spirit in a single occasion you had 3.5 drinks of wine and 2 drinks of spirit which is a total of 5.5 drinks. Then you need to choose correct column to indicate how often in the last year you had such amount of alcohol.

ojien in the tast year you	Every day or almost every day	3-4 per week	1-2 per week	2-3 per month	About once a month	6-11 in past year	3-5 in past year	1-2 in past year	Never in past year
1. How often in the last ye	ar did you	have 10 c	drinks or n	nore durin	ig one day	?			
10 drinks or more 5 1 (10 x 0.5 1) of beer or 2 1 (10 x 2 dl) of wine or 0.5 1 (10 x 5 cl) of spirits									
2. How often in the last ye	ar did you	have 7-9	drinks du	ring one d	ay?				
7-9 drinks (7-9 x 0.5 l of beer or 7-9 x 2 dl of wine or 7-9 x 5 cl of spirits)									
3. How often in the last ye	ar did you	have 5-6	drinks du	ring one d	ay?				
5-6 drinks (5-6 x 0.5 l of beer or 5-6 x 2 dl of wine or 5-6 x 5 cl of spirits)									
4. How often in the last ye	ar did you	have 3-4	drinks du	ring one d	ay?				
3-4 drinks (3-4 x 0.5 l of beer or 3-4 x 2 dl of wine or 3-4 x 5 cl of spirits)									
5. How often in the last ye	ar did you	have 1-2	drinks du	ring one d	ay?				
1-2 drinks (1-2 x 0.5 l of beer or 1-2 x 2 dl of wine or 1-2 x 5 cl of spirits)									
6. How often in the last year did you have about half drink during one day?									
About half drink									
	Every day or almost every day	3-4 per week	1-2 per week	2-3 per month	About once a month	6-11 in past year	3-5 in past year	1-2 in past year	Never in past year

53. How much beer (litres) do you usually drink during one week?

54. How much wine (decilitres) do you usually drink during one week?

55. How much spirits (decilitres) do you usually drink during one week?

56. What was the largest amount of alcohol you had on a single occasion during the last 4 weeks?

0.5 L bottles or glasses of beer AND2 dl glasses of wine AND

5 cl glasses of spirits (double shots)

57. During the **last 12 months**, how often did you drink enough to feel drunk?

- 1. every day or at least 5 times a week
- 2. about 1-4 times a week
- 3. about 1-3 times a month
- 4. 3-11 times a year
- 5. once or twice a year
- 6. never in the past year

58. In last 12 months, did your drinking cause you difficulties with the following aspects of your life?

Please cross appropriate box in each row:	Yes	No
marriage/partner or home life		
friendships and social life	1	2
your work	1	2
Police or other authorities		2
your physical health		
any injury or accident		2
your psychological or mental health	1	2
your financial circumstances		

59. In the last 12 months, did you have any of the following experiences?

Please cross appropriate box in each row:	Yes	No
Have you ever felt you should cut down on your drinking?		
Have people ever annoyed you by criticising your drinking?		2
Have you ever felt bad or guilty about your drinking?		
Have you ever had a drink first thing in the morning to steady your nerves or get rid of a hangover?		2



Please cross appropriate box in each row:	Improve		No	Make I	t Worse
	Strongly	Slightly	effect	Slightly	Strongly
Eating meat			3-4	5	6
Eating fruit and vegetables			3-4	5	6
Lack of physical activity		2	3-4	5	6
Obesity			3-4	5	6
Smoking			3-4	5	6
Drinking alcohol		2	3-4	5	6
Passive smoking			3-4	5	6
Environmental pollution			3-4	5	6
Lack of money			3-4	5	6
Stress			3-4	5	6
Exercise	1	2	3-4	5	6

60. How do the following factors influence human health?

61. Below is a list of the ways you might have felt or behaved during **the last week**. For each of the following statements, please indicate how often you felt that way:

During the past week:	Less than one day	1-2 days	3-4 days	5-7 days
a) I was bothered by things that usually do not bother me	1	2	3	4
b) I did not feel like eating, my appetite was poor		2	3	4
c) I felt that I could not shake off the blues even with help from my family and friends	1	2	3	4
d) I felt that I was just as good as other people	1	2	3	4
e) I had trouble keeping my mind on what I was doing		2		4
f) I felt depressed.		\square_2		
g) I felt that everything I did was an effort		$\square 2$		
h) I felt hopeful about the future				4
i) I thought my life had been a failure	1	2	3	4
j) I felt fearful	1	2	3	4
k) My sleep was restless	1	2	3	4

During the past week:	Less than one day	1-2 days	3-4 days	5-7 days
l) I was happy	1	2	3	4
m)I talked less than usual	1	2	3	4
n) I felt lonely	1	2	3	4
o) People were unfriendly			3	4
p) I enjoyed life	1	2	3	4
q) I had crying spells	1	2	3	4
r) I felt sad	1	2	3	4
s) I felt people dislike me	1	2	3	4
t) I could not get going			3	4

62. Are you a member of club or organisation (sports club, church, political party)?

- Yes If **YES**, how often do you take
 No part in common activities?
- 1. Several times a week
- 2. Several times a month
- 3. About once a month
- 4. Several times a year
- 5. Never or almost never

63. On whom do you rely first of all when having problems?

- 1. friends
- 2. relatives
- 3. employer
- 4. state
- 5. private / commercial companies
- 6. public organisations such as trade unions
- 7. charities, church
- 8. no one
- 9. other, please give details:

64. Are you regularly in contact with your relatives who do not live in your household?

- 1. several times a week
- 2. about once a week
- 3. several times a month
- 4. about once a month
- 5. less than once a month
- 6. I do not have relatives / no relatives outside my household

65. How many relatives who do not live in your household do you see at least once a week?

- 1. none
- 2.1 or 2
- 3. 3 to 5
- 4. more than 5
- 5. I do not have relatives / no relatives outside my household

66. How often do you visit **friends**?

- 1. several times a week
- 2. about once a week
- 3. several times a month
- 4. about once a month
- 5. less than once a month
- 6. I do not have friends

67. How many friends do you see at least once a week?

- 1. none
- 2. 1 or 2
- 3. 3 to 5
- 4. more than 5
- 5. I do not have friends

68. We would like to ask about your area of residence and other people:

	Always	Mostly	Some- times	Rarely	Never
Do you feel safe in the area of your residence during the day?	1	2	3	4	5
Do you feel safe in the area of your residence at night?	1	2	3	4	5
Would your neighbours help you if you need it?		2	3	4	5
Is there trust among people in your area of residence?		2	□ ₃	4	5
Do you think that you can trust people?		2	3	4	5

69. Have the changes since 1989 been good or bad for you:

	Very good	Good	No change	Bad	Very bad
Occupational position	1	2	3	4	5
Income	1	2	3	4	5
Material circumstances			3	4	5
General social position	 1	<u>2</u>	3	4	5

70. How much do you agree or disagree with the following statements?

	DI	SAGR	EE	AGREE			
Measurement for perceived control		MODERATELY	SLIGHTLY	SLIGHTLY	MODERATELY	STRONGLY	
a) At home, I feel I have control over what happens in most situations	1	2	3	4	5	6	
b) Keeping healthy depends on things that I can do	1	2	3	4	5	6	
c) There are certain things I can do for myself to reduce the risk of a heart attack	1	2	3	4	5	6	
d) There are certain things I can do for myself to reduce the risk of getting cancer	 1	2	3	4	5	6	
e) I feel that what happens in my life is often determined by factors beyond my control	1	2	3	4	5	6	
f) Over the next 5-10 years I expect to have many more positive than negative experiences	 1	2	3	4	5	6	
g) I often have the feeling that I am being treated unfairly	1	□ ₂	3	 4	5	6	
h) In the past ten years my life has been full of changes without my knowing what will happen next	1	2	3	4	5	6	
i) I very often have the feeling that there's little meaning in the things I do in my daily life	 1	2	3	4	5	6	
j) I sometimes feel as if I've done all there is to do in life	1	2 ²	3	4	5	6	
 k) I gave up trying to make big improvements or changes in my life a long time ago 	1	2	3	4	5	6	

Social and economic conditions

71. How often does it happen that you do not have enough **money for food** which you and your family need? And how often did this happen before 1990?



72. How often does it happen that you do not have enough money for clothing which you and your family need? And how often did this happen before 1990?



73. Do you have difficulties with paying bills (for housing, electricity, heating etc)? And what was the situation before 1990?



74. Are you in receipt of any of the following benefits at the moment? Choose all that apply.

- 1. Child benefit
- 2. Unemployment benefit
- Care allowance (care for invalid) 3.
- 4. Widow(er)'s pension
- Social assistance (e.g. with food, fuel, clothes or medication) 5.
- Others please 6.
- specify:
- 7. Do not receive any state benefits

75. How many rooms does your house/flat have (excluding kitchen and bathrooms)?	

- 76. How many adults (18 years or older) live in your house/flat?
- 77. How many children (under 18 years old) live in your house/flat?

78. What was the highest completed level of education of your parents? Your mother:

- Your father:
- Incomplete primary or no formal education 1.
- 2. Primary
- Vocational (apprenticeship) 3.
- 4. Secondary
- 5. University (degree)

Primary Vocational (apprenticeship) 3.

Incomplete primary or no formal education

Secondary 4.

1.

2.

5. University (degree)

79. Did you have any of the following items in your house when you were a child (about 10 years old)?

Cold tap water	1.	Yes	2.	No	3.	I don't remember
Hot tap water	1.	Yes	2.	No	3.	I don't remember
Radio	1.	Yes	2.	No	3.	I don't remember
Fridge	1.	Yes	2.	No	3.	I don't remember
Own kitchen	1.	Yes	2.	No	3.	I don't remember
Own toilet	1.	Yes	2.	No	3.	I don't remember

80. What is your current economic activity?

- 1. Employed
- 2. Entrepreneur (owner of a company)
- 3. Self-employed / freelance
- 4. Housewife
- 5. Farmer
- 6. Pensioner, still employed
- 7. Pensioner, not employed. At what age did you years old retire ?
- 8. Unemployed

81. What was your main life-time occupation?

82. Have you ever experienced unemployment?

- 1. No
- 2. Yes, for up to 3 months in total
- 3. Yes, for 3 months to 1 year
- 4. Yes, for more than one year

83. If you are out of work, do you look for a job?

- 1. Yes
- 2. No, no hope
- 3. No, I choose not to work
- 4. No, I am too ill to work
- 5. No, I am retired
- 6. No, other reason: please specify

84. Now, would you tell us about your household? Below is a list of various items, which of the following do you have in your household?

	Yes	No, I do not want it	No, I can not afford
			it
Microwave			3
Video recorder			3
Television (colour)			3
Washing machine		2	3
Dishwasher			3
Car			3
Freezer		2	3
Cottage (for holidays / weekends etc.)		2	3
Video camera / camcorder		2	3
Satellite / cable TV		2	3
Telephone			3
Mobile phone		2	3

Appendix 2.

Questionnaire for working individuals in the HAPIEE study wave 1

- 1. What is your current occupation?
- 2. How many hours do you spend at work in a typical week? _____
- 3. Is part of it overtime work?
 - 1. Yes
 - 2. No
- 4. If yes, how many hours do you spend at overtime work in a typical week ? _____

5. What is your position at your main job?

- 1. higher managerial post or director
- 2. manager / supervisor / foreman, more than 25 inferiors
- 3. manager / supervisor / foreman, 5-25 inferiors
- 4. manager / supervisor / foreman, less than 5 inferiors
- 5. employee, without inferiors
- 6. self-employed (25+ employees)
- 7. self-employed (1-24 employees)
- 8. self-employed (no employees)

6. Which of these best describes your work in your main job? *Please choose one answer only*

- 1. <u>Sedentary occupation</u>: You spend most of your time sitting (such as in an office)
- 2. <u>Standing occupation</u>: You spend most of your time standing or walking. However the way you spend your time does not require intense physical effort (e.g. shop assistant, hairdresser, security guard etc.).
- 3. <u>Physical work</u>: This involves some physical effort including handling of heavy objects and use of tools (e.g. plumber, cleaner, nurse, sports instructor, electrician, carpenter etc.)
- 4. <u>Manual</u>: This involves very vigorous physical activity including handling of very heavy objects (e.g., miner, bricklayer, construction worker etc.)

7. What is size of firm you work at?

- 1. Working alone
- 2. 1-5 other people
- 3. 6-24 other people
- 4. 25-49 other people
- 5. 50-499 other people
- 6. 500+ other people

8. Here are some statements about possible strenuous aspects of your current work situation. *Please cross the answer that best describes your job:*

	strongly disagree	disagree	agree	strongly agree
I get easily overwhelmed by time pressures at work	1	2	3	4
As soon as I get up in the morning I start thinking about work problems		2	3	4
When I get home, I can easily relax and 'switch off' work	1	2	3	4
People close to me say I sacrifice too much for my job				4
Work rarely lets me go, it is still on my mind when I go to bed				4
If I postpone something that I was supposed to do today I'll have trouble sleeping at night		2	3	4

- strongly strongly disagree agree disagree agree My job requires that I learn new things \square_4 \Box_2 3 ____1 My job requires a high level of skill 1 2 3 My job requires me to be creative 73 \Box_2 1 4 I get to do a variety of different things on my job \square_2 \square_4 I have an opportunity to develop my own special abilities 3 4 \square_2 My job allows me to make a lot of decisions on my own]3 \square_4 \square_1 \square_2 On my job, I have very little freedom to decide how I do my work]3 \square_4 2 I have a lot of say about what happens in my job \square_1 \square_2 4 3 My job requires working very fast 3 \square_2 1 4 My job requires working very hard \square_2 I have enough time to get the job done 4 $\left[\right]_{2}$ 3 I am not asked to do an excessive amount of work 4 ____2 3 My job requires long periods of intense concentration on the task \square_2 My job is very hectic 2 3 4 My tasks are often interrupted before they can be completed, requiring \square_2 \square_4 attention at a later time I am free from conflicting demands that others make \square_2 Waiting on work from other people or departments often slows me 1 \square_2 3 4 down on my job My work puts me in emotionally disturbing situations \square_2 1 4 3 My work is emotionally demanding \square_2 I get emotionally involved in my work \square_1 4 \square_2 3 My supervisor is concerned about the welfare of those under him \square_4 \square_2 3 My supervisor pays attention to what I am saying \square_4 My supervisor is helpful in getting the job done 1 4 2 3 My supervisor is successful in getting people to work together \square_3 1 2 People I work with are competent in doing their jobs \square_1 \square_2 People I work with take a personal interest in me \square_2 3 4 People I work with are friendly 2 3 4 People I work with are helpful in getting the job done \square_2 3 4
- **9.** Here are some statements about possible strenuous aspects of your current work situation. *Please cross the answer that best describes your job:*

10. Here are some questions about your current work situation. Please cross the answer that best describes your job.

			IF YES			
	Yes	No	Not at all distressed	Somewhat distressed	Rather distressed	Very distressed
There is constant time pressure in my job due to a heavy workload	□ 1		1		3	4
There are many interruptions and disturbances in my job				2	3	4
I have a lot of responsibility in my job						4
There is pressure in my job to work overtime						4
My job is physically demanding				2	3	4
Over the past few years, my job has become more and more demanding	 1		□ ı		3	4
Are you treated unfairly at work?				2	3	4
Are the promotion prospects in your job poor?						4
Have you experienced or do you expect to experience an undesirable change in your work situation?					3	4
Have job redundancies recently affected your work colleagues?	1				3	4
Is your own job security poor?		2		2	3	4

				IF NO				
	Yes	No	Not at all distressed	Somewhat distressed	Rather distressed	Very distressed		
Do you receive the respect you deserve from your work colleagues?						4		
Do you receive the respect you deserve from your supervisors?						4		
Do you experience adequate support in difficult situations?						4		
Does your current job adequately reflect your knowledge, skills and training?	□ 1	\square_2		\square_2	3	4		
Does your salary/income adequately reflect all your past efforts and achievements?	□ ₁	\square_2	\Box_1	\square_2		4		
Considering all your efforts and achievements, do you receive the respect and prestige you deserve at work?		\square_2		\square_2		4		
Considering all your efforts and achievements, are your work prospects adequate?		\square_2		\square_2		4		

Appendix 3.

Dietary questionnaire in the HAPIEE study wave 1

We would like to ask you to estimate your average food use. Please cross the appropriate square in each row of the tables below a number indicating how often, <u>on average</u>, you have eaten the specified amount <u>during the last 3 months</u>.

	Amount	6+ per day	4-5 per day	2-3 per day	1 per day	5-6 per week	2-4 per week	1 per week	1-3 per month	
Bread and cereals										
White bread, rolls	Medium slice, 1 roll						□_ ₆	□ ₇	□ 8	□ ₉
Dark bread, rolls	Medium slice, 1 roll						□_ ₆	□ ₇	□ 8	□ ₉
Cereals	Medium bowl						□_ ₆			□ ₉
Potatoes, rice, pasta, du	mplings									
Potatoes boiled or mashed	Medium serving (about 100 g)						□_ ₆	□ ₇	□ 8	□ ₉
Potatoes fried (chips) or roasted	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Rice	Medium serving (about 100 g)				\square_4			□ ₇		□ ₉
Pasta (spaghetti, noodles)	Medium serving (about 100 g)						□_ ₆	□ ₇	□ ₈	9
Pizza	Medium slice				\square_4		□_ ₆	□ ₇		, o
Roll-dumplings	4 slices				\square_4		□_ ₆	□ ₇		□ ₉
Potato-dumplings	4 slices				\square_4		□_ ₆			9
Groats	Medium serving	\Box_1			\square_4		□_ ₆	□ ₇	□ 8	□ ₉
Dairy products and fats										
Cream, sour cream	50 ml						□ ₆	□ ₇	□ 8	9
White yoghurt	1 carton (100-150 ml)				\square_4				□ 8	9
Fruit yoghurt	1 carton (100- 150ml)						□_ ₆	□ ₇	□ 8	9
Milk desserts	1 carton (100-150 ml)						□_ ₆	□ ₇		□ ₉
Soft cottage cheese	Medium serving (about 30 g)						□_ ₆	□ ₇	□ 8	□ ₉
Hard cottage cheese	Medium serving (about 30 g)				\square_4		□_ ₆	□ ₇	□ 8	□ ₉
Low fat soft cheese	Medium serving (about 30 g)						□_ ₆	□ ₇	□ 8	9
High fat soft cheese	Medium serving (about 30g)						□_ ₆	□ ₇		9
Hard cheese, processed cheese	Medium serving (about 30 g)				□_ ₄		□_ ₆	□ ₇		9

	Amount	6+ per day	4-5 per day	2-3 per day	1 per day	5-6 per week	2-4 per week	1 per week	1-3 per month	Never or less than 1 per month
Eggs	1 egg						□_ ₆	□ ₇		□ ₉
Margarine (on bread)	1 teaspoon						□_ ₆			9
Margarine (in food)	1 teaspoon				\square_4		□_ ₆	□ ₇		□ ₉
Butter (on bread)	1 teaspoon						□ ₆	□ ₇		9
Butter (in food)	1 teaspoon						□_ ₆	□ ₇	□ 8	9
Mixture of margarine and butter (on bread)	1 teaspoon						□_ ₆	□ ₇		9
Mixture of margarine and butter (in food)	1 teaspoon						6	□ ₇		9
Vegetable oil	1 tablespoon						□_ ₆	□ ₇		□ ₉
Lard (on bread)	1 teaspoon				□_ ₄		6			□ ₉
Lard (in food)	1 teaspoon				□_ ₄		□ ₆	□ ₇		9
Mayonnaise	1 tablespoon				□_ ₄		□_ ₆	□ ₇		9
Soups, sauces and spread	ds									
Borsch, shiee, vegetable soup	Medium serving (about 250 ml)						□_ ₆	□ ₇		9
Bouillon	Medium serving (about 250 ml)						□_ ₆	□ ₇		□ ₉
Beetroot soup, white borsch	(about 250 ml)			□ ₃			□_ ₆	□ ₇	□ 8	□ ₉
Cabbage soup	Medium serving (about 250 ml)						□_ ₆	□ ₇		9
Other soups	Medium serving (about 250 ml)						□_ ₆	□ ₇		9
Ketchup	1 tablespoon						□_ ₆	□ ₇		□ ₉
Sauces with meat, pasta, groats (such as gravy or white sauces)	Medium serving						6	□ ₇	□ ₈	9
Marmalade, jam, honey	1 teaspoon						□_ ₆	□ ₇	□ 8	9
Sweets and snacks										
Biscuits	1 medium				\square_4		□_ ₆			□ ₉
Cakes, pies (sweet)	medium slice						□_ ₆	□ ₇		□ ₉
Buns, pastries, doughnuts, muffins	1 piece						6		□ 8	9
Sweets	1 bonbon						□_ ₆			□ ₉
Chocolate	1 bar				\square_4		□_ ₆			□ ₉
Ice cream	one scoop						□_ ₆	□ ₇		9
Milk pudding	medium serving						□ ₆	□ ₇		9

	Amount	6+ per day	4-5 per day	2-3 per day	1 per day	5-6 per week	2-4 per week	1 per week	1-3 per month	Never or less than 1 per month
Sweet rice	medium serving						□ ₆	□ ₇		□ ₉
Pancakes	1 pancake	\square_1					□ ₆	□ ₇	□ 8	9
Sweet (fruit) dumplings	4 pieces						□ ₆	□ ₇		9
Crisps, crackers and other packet-snacks	1 small packet (25 g)					□ ₅	□ ₆	□ ₇		9
Peanuts and other nuts	1 small packet (50 g)						□_ ₆	□ ₇	□ 8	□ ₉
Sugar into coffee, tea	1 teaspoon				\square_4		□_ ₆	□ ₇	□ 8	9
Sweetener into coffee, tea	1 capsule, 1 tablet				\square_4					9
Drinks										
Milk	2 dl				\square_4			□_ ₇		□ ₉
Cocoa	2 dl				\square_4			□_ ₇		□ ₉
Fruit juice	2 dl						□_ ₆	□ ₇		9
Fizzy drinks (lemonade, coke, fanta)	2 dl					□ ₅	6	□ ₇		9
Diet/low calorie drinks	2 dl	\square_1			\square_4		□_ ₆	□ ₇		9
Squash	one tablespoon	\square_1			\square_4		□_ ₆	□ ₇		9
Coffee	2 dl	\square_1					□_ ₆	□ ₇		9
Tea	2 dl						□_ ₆	□ ₇		9
Wine	1 dl						□_ ₆	□ ₇	□ 8	9
Beer	0.25 1				\square_4		□ ₆	□ ₇		9
Port, sherry, vermouth	1 dl				\square_4		□ ₆	□ ₇		9
Liqueurs	0.5 dl						6	□ ₇		□ ₉
Spirits	0.25 dl						□ ₆	□ ₇		9
Meat and fish										
Beef : roast, steak, mince, stew or casserole	Medium serving (about 100 g)						□_ ₆	□ ₇	□ 8	9
Lamb: roast, chops or stew	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Pork: roast, chops or stew	Medium serving (about 100 g)						6	□ ₇		□ ₉
Poultry	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Rabbit	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Offals (heart, kidney, liver)	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Soft sausages	Medium serving (about 100 g)						□_ ₆	□ ₇	□ ₈	9

	Amount	6+ per day	4-5 per day	2-3 per day	1 per day	5-6 per week	2-4 per week	1 per week	1-3 per month	Never or less than 1 per month
Hard sausages	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Soft salami	50 g						□ ₆	□ ₇		9
Hard salami	50 g						□_ ₆	□ ₇		9
Ham	about 50 g						□ 6	□ ₇		9
Bacon	2 slices						□_ ₆	□ ₇	□ 8	9
Pate	50 g						□_ ₆	□ ₇	□ 8	9
Meat pie	Medium serving							□ ₇		9
Luncheon meat	50 g						□_ ₆	□ ₇	□ 8	9
Canned meat	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Meat ravioli	Serving (10 pieces)						□_ ₆	□ ₇		9
Fish – fresh, frozen or canned (not in oil)										
Fresh water fish (e.g. carp, pike)	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Salt water white fish (e.g. cod of haddock)	Medium serving (about 100 g)						□_ ₆			□ ₉
Oily fish (e.g. mackerel, tuna, salmon, sardines, herring, kippers)	Medium serving (about 100 g)				□ 4	□ ₅	6	□ ₇	□ ₈	9
Other fish										
Fish canned in oil	Medium serving (about 100 g)						□_ ₆	□ ₇		□_ ₉
Fish fingers, fish Afilé	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Salted fish	25 g						□ ₆			9
Crab, prawns, mussels (sea food)	Medium serving						6	□ ₇		9
Fresh fruit										
Apples	1 medium				\square_4		□_ ₆	□ ₇	□ 8	□ ₉
Pears	1 medium				\square_4		□ ₆		□ 8	9
Oranges	1 medium				\square_4		□ ₆		□ 8	9
Grapefruit	¹ / ₂ medium				\square_4		□_ ₆	\Box_7		□ ₉
Mandarins	1 medium				\square_4		□ ₆	\Box_7		□ ₉
Lemons	¹ / ₂ medium				\square_4		□_ ₆			9
Peaches	1 medium						□_ ₆	\Box_7	□ 8	□ ₉
Apricots	1 medium						□_ ₆	□ ₇	□ 8	9
Plums	about 100 g						□_ ₆	□ ₇		9

	Amount	6+ per day	4-5 per day	2-3 per day	1 per day	5-6 per week	2-4 per week	1 per week	1-3 per month	Never or less than 1 per month
Cherries	about 100 g						□_ ₆	□ ₇		□ ₉
Strawberries	Medium serving (about 100 g)						6	□ ₇		9
Raspberries	Medium serving (about 100 g)							□ ₇		□ ₉
Red currant	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Black currant	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Blueberries	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Gooseberry	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Kiwi	1 medium						□_ ₆	□ ₇		□ ₉
Melon	Medium serving (about 100 g)							□ ₇		□ ₉
Pineapple	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Bananas	1 medium				\square_4		□_ ₆	□ ₇		□ ₉
Grapes	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Tinned or bottled fruit	medium serving (about 100g)						□_ ₆	□ ₇		□ ₉
Dried fruit (e.g. raisins, apricots, apples)	medium serving (about 50g)							□ ₇		□ ₉
Vegetables										
Green salad (lettuce)	Medium serving		\square_2				□_ ₆			□ ₉
Spinach	Medium serving				\square_4		6	□ ₇		9
Brussels sprouts	5 sprouts				\square_4		□ ₆	□ ₇		9
Cabbage	Medium serving				\square_4		□_ ₆	□ ₇		9
Beans	Medium serving (about 100 g)						6	□ ₇	□ 8	9
Lentils	Medium serving (about 100 g)						□_ ₆			9
Dried peas	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Green beans	Medium serving (about 100 g)						6	□ ₇		9
Green peas	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Turnips, swedes, parsnips	Medium serving (about 100 g)						□_ ₆	□ ₇		□ ₉
Radish	4 radishes				\square_4		□_ ₆	□ ₇		□ ₉
Celeriac	50 g	\square_1					□_ ₆	□ ₇		9

	Amount	6+ per day	4-5 per day	2-3 per day	1 per day	5-6 per week	2-4 per week	1 per week	1-3 per month	Never or less than 1 per month
Parsley	1 medium							□ ₇		9
Cauliflower	Medium serving (about 100 g)						□ ₆	□ ₇	□ 8	9
Broccoli	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Carrots	1 medium							□ ₇		□ ₉
Onion	¹ / ₂ medium							□ ₇		9
Leeks	¹ ₂ medium				\square_4			□ ₇	□ 8	□ ₉
Garlic	1 clove				\square_4			□ ₇		□ ₉
Peppers	1 medium				\square_4			□ ₇		□ ₉
Tomatoes	1 medium							□ ₇		□ ₉
Cucumbers	Medium serving (about 100 g)						□_ ₆	□ ₇	□ 8	9
Aubergine	Medium serving (about 100 g)						□_ ₆	□_ ₇		9
Courgette/marrow	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Corn	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Beet-root cooked Russian salad (RU)	Medium serving (about 100 g)						□_ ₆	□ ₇		9
Sauerkraut	Medium serving (about 100 g)							□ ₇		9
Pickled vegetables, gherkins	Medium serving (about 50 g)						□_ ₆	□ ₇		9
Mushrooms	Medium serving						□ ₆	□ ₇		9
Soya meat	Medium serving (about 100g)						□_ ₆	□ ₇		9
Mixed frozen vegetables	Medium serving (about 100 g)						□_ ₆	□ ₇	□ 8	□ ₉

- **1.** Are the foods and drinks listed in the previous table representative of the foods and drinks that you consumed in the last 3 months?
 - 1. Yes
 - 2. No
- 2. Are there any other foods, which you ate more than once a week?
 - 1. Yes
 - 2. No

3. If yes, please list below

Food name	Usual serving size	Number of times eaten each week

- **4.** What type of milk did you most often use?
 - 1. Full cream (3% of fat and more)
 - 2. Semi- skimmed (2% of fat)
 - 3. Skimmed (about 0.5% of fat)
 - 4. Soya milk
 - 5. Cream into coffee, tea
 - 6. I do not use milk
 - 7. I do not know
- 5. How much milk do you drink each day, including milk with tea, coffee, cereals etc.?
 - 1. None
 - 2. Less than 250 ml
 - 3. More than 250, less than 500 ml
 - 4. More than 500 ml, less than 1000 ml
 - 5. More than 1000 ml
- 6. How often do you have coffee or deserts with added cream?
 - 1. Daily
 - 2. 4-6 times a week
 - 3. 1-3 times a week
 - 4. 1-3 times a month
 - 5. Less than once a month
 - 6. I do not have coffee or deserts with cream
 - 7. I do not have coffee or deserts
 - 8. I do not know
- 7. How often do you eat soups with added cream?
 - 1. Daily
 - 2. 4-6 times a week
 - 3. 1-3 times a week
 - 4. 1-3 times a month
 - 5. Less than once a month
 - 6. I do not eat soup with cream
 - 7. I do not eat soup
 - 8. I do not know
- **8.** How often do you use sour cream including when added to the food?
 - 1. Daily
 - 2. 4-6 times a week
 - 3. 1-3 times a week
 - 4. 1-3 times a month
 - 5. Less than once a month
 - 6. I do not use sour cream
 - 7. I do not know

- 9. What kind of fat do you use most often for frying, roasting, grilling, baking etc?
 - 1. vegetable oil
 - 2. olive oil
 - 3. butter
 - 4. margarine, solid vegetable fat
 - 5. lard
 - 6. none
 - 7. I do not know

10. How often do you eat fried food ?

- 1. Daily
- 2. 4-6 times a week
- 3. 1-3 times a week
- 4. 1-3 times a month
- 5. Less than once a month
- 6. I do not eat fried food
- 7. I do not know

11. What do you do with the visible fat on your meat?

- 1. I do not eat meat
- 2. I eat as little as possible of the fat
- 3. I eat some of the fat
- 4. I eat most of the fat

12. How often do you or your spouse add salt to food during cooking?

- 1. Never
- 2. Rarely
- 3. Sometimes
- 4. Usually
- 5. Always
- 6. I do not cook
- 7. I do not know

13. How often do you add salt to any food at the table?

- 1. Never
- 2. Rarely
- 3. Sometimes
- 4. Usually
- 5. Always
- 6. I do not know

14. What type of breakfast cereals do you most often eat?

- 1. I do not eat cereals
- 2. Corn flakes
- 3. Oat flakes
- 4. Corn and oat flakes
- 5. Other
- 6. I do not know

Appendix 4.

Path analyses for 3 separate drinking outcomes: antecedent or mediator role of OC in relationship between ERI and drinking outcomes

Table 1. Path analysis for antecedent or mediator role of OC in relationship between ERI and binge drinking in men

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 → OC wave 2	1.80	0.346	0.316	< 0.001
ERI wave 1 \rightarrow ERI wave 2	1.50	0.237	0.242	< 0.001
$OC \rightarrow ERI \rightarrow Binge drinking$	1.05	0.027	0.021	< 0.001
OC wave 1 → ERI wave 2	1.31	0.148	0.146	< 0.001
ERI wave 1 \rightarrow Binge drinking wave 2	1.39	0.183	0.143	< 0.001
<u>ERI \rightarrow OC \rightarrow Binge drinking</u>	1.01	0.006	0.005	0.077
ERI wave 1 \rightarrow OC wave 2	1.16	0.084	0.079	< 0.001
OC wave 1 → Binge drinking wave 2	1.15	0.076	0.058	0.054
Tests of model fit		RMSEA= 0.055	CFI= 0.861	TLI= 0.786

Table 2. Path analysis for antecedent or mediator role of OC in relationship between ERI and problem drinking in men

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 \rightarrow OC wave 2	1.80	0.346	0.316	< 0.001
ERI wave 1 \rightarrow ERI wave 2	1.50	0.237	0.242	< 0.001
$OC \rightarrow ERI \rightarrow Problem drinking$	1.03	0.019	0.014	0.001
OC wave 1 → ERI wave 2	1.30	0.147	0.145	< 0.001
ERI wave 1 \rightarrow Problem drinking wave 2	1.26	0.128	0.098	< 0.001
<u>ERI \rightarrow OC \rightarrow Problem drinking</u>	1.01	0.007	0.005	0.048
ERI wave 1 \rightarrow OC wave 2	1.16	0.082	0.078	< 0.001
OC wave 1 \rightarrow Problem drinking wave 2	1.16	0.084	0.061	0.031
Tests of model fit		RMSEA= 0.056	CFI= 0.882	TLI= 0.791

Table 3. Path analysis for antecedent or mediator role of OC in relationship between ERI and heavy drinking in men

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 → OC wave 2	1.80	0.347	0.317	< 0.001
ERI wave 1 \rightarrow ERI wave 2	1.50	0.238	0.242	< 0.001
$OC \rightarrow ERI \rightarrow Heavy drinking$	1.02	0.012	0.010	0.009
OC wave 1 → ERI wave 2	1.31	0.148	0.146	< 0.001
ERI wave 1 \rightarrow Heavy drinking wave 2	1.15	0.081	0.065	0.005
<u>ERI \rightarrow OC \rightarrow Heavy drinking</u>	1.00	0.002	0.001	0.475
ERI wave 1 \rightarrow OC wave 2	1.16	0.084	0.079	< 0.001
OC wave 1 \rightarrow Heavy drinking wave 2	1.04	0.021	0.016	0.453
Tests of model fit		RMSEA= 0.056	CFI= 0.848	TLI= 0.742

· .					
-	Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
-	OC wave 1 → OC wave 2	1.84	0.359	0.358	< 0.001
	ERI wave 1 \rightarrow ERI wave 2	1.51	0.242	0.251	< 0.001
	$OC \rightarrow ERI \rightarrow Binge drinking$	1.04	0.023	0.018	0.007
	OC wave 1 → ERI wave 2	1.32	0.155	0.159	< 0.001
	ERI wave 1 \rightarrow Binge drinking wave 2	1.30	0.146	0.113	0.005
	<u>ERI \rightarrow OC \rightarrow Binge drinking</u>	1.01	0.004	0.003	0.341
	ERI wave 1 → OC wave 2	1.15	0.076	0.077	< 0.001
	OC wave 1 \rightarrow Binge drinking wave 2	1.09	0.050	0.038	0.328
	Tests of model fit		RMSEA= 0.062	CFI= 0.837	TLI= 0.739

Table 4. Path analysis for antecedent or mediator role of OC in relationship between ERI and binge drinking in women

Table 5. Path analysis for antecedent or mediator role of OC in relationship between ERI and
problem drinking in women

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 \rightarrow OC wave 2	1.84	0.360	0.359	< 0.001
ERI wave 1 → ERI wave 2	1.51	0.242	0.252	< 0.001
$\frac{OC \rightarrow ERI \rightarrow Problem drinking}{OC wave 1 \rightarrow ERI wave 2}$	1.03 1.32	0.018 0.156	0.015 0.160	0.048 < 0.001
ERI wave 1 \rightarrow Problem drinking wave 2	1.23	0.116	0.091	0.042
ERI → OC → Problem drinking ERI wave 1 → OC wave 2	1.01 1.15	0.005 0.076	0.004 0.077	0.276 < 0.001
OC wave 1 → Problem drinking wave 2 Tests of model fit	1.13	0.069 RMSEA= 0.061	0.053 CFI= 0.840	0.253 TLI= 0.726

Table 6. Path analysis for antecedent or mediator role of OC in relationship between ERI and heavy drinking in women

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 → OC wave 2	1.83	0.357	0.356	< 0.001
ERI wave 1 \rightarrow ERI wave 2	1.51	0.244	0.253	< 0.001
$OC \rightarrow ERI \rightarrow Heavy drinking$	1.02	0.011	0.009	0.037
OC wave 1 → ERI wave 2	1.32	0.156	0.160	< 0.001
ERI wave 1 \rightarrow Heavy drinking wave 2	1.13	0.069	0.056	0.033
<u>ERI \rightarrow OC \rightarrow Heavy drinking</u>	1.00	0.001	0.001	0.608
ERI wave 1 \rightarrow OC wave 2	1.15	0.076	0.077	< 0.001
OC wave 1 \rightarrow Heavy drinking wave 2	1.03	0.018	0.012	0.541
Tests of model fit		RMSEA= 0.062	CFI= 0.818	TLI= 0.708

Path analyses for 3 separate drinking outcomes: mediator roles of PC and ERI in relationship between OC and drinking outcomes

Table 7. Path analysis for mediator roles of PC and ERI in relationship between OC and binge drinking in men

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 \rightarrow Binge drinking wave 2	1.12	0.063	0.049	0.084
$OC \rightarrow ERI \rightarrow Binge drinking$	1.07	0.037	0.029	< 0.001
OC wave 1 → ERI wave 1	1.54	0.253	0.247	< 0.001
ERI wave 1 \rightarrow Binge drinking wave 2	1.30	0.148	0.117	< 0.001
$OC \rightarrow PC \rightarrow Binge drinking$	1.02	0.010	0.008	0.013
OC wave 1 \rightarrow PC wave 1	0.85	- 0.092	- 0.090	< 0.001
PC wave 1 \rightarrow Binge drinking wave 2	0.82	- 0.107	- 0.084	0.006
ERI correlates with PC		- 0.051	- 0.085	< 0.001
Tests of model fit		RMSEA= 0.062	CFI= 0.857	TLI= 0.775

Table 8. Path analysis for mediator roles of PC and ERI in relationship between OC and problem drinking in men

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 \rightarrow Problem drinking wave 2	1.13	0.067	0.051	0.055
<u>OC → ERI → Problem drinking</u> OC wave 1 → ERI wave 1	1.07 1.52	0.035 0.246	0.027 0.240	< 0.001 < 0.001
ERI wave $1 \rightarrow$ Problem drinking wave 2	1.32	0.142	0.240	< 0.001
$\underline{OC} \rightarrow PC \rightarrow Problem \ drinking$	1.02	0.010	0.007	0.007
OC wave 1 \rightarrow PC wave 1	0.85	- 0.089	- 0.086	< 0.001
PC wave 1 → Problem drinking wave 2	0.82	- 0.109	- 0.085	0.003
ERI correlates with PC		- 0.047	- 0.078	< 0.001
Tests of model fit		RMSEA= 0.064	CFI= 0.863	TLI= 0.782

Table 9. Path analysis for mediator roles of PC and ERI in relationship between OC and heavy	
drinking in men	

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 \rightarrow Heavy drinking wave 2	1.03	0.018	0.014	0.526
OC → ERI → Heavy drinking	1.03	0.019	0.015	0.010
OC wave 1 → ERI wave 1	1.52	0.248	0.242	< 0.001
ERI wave 1 → Heavy drinking wave 2	1.15	0.077	0.061	0.009
$OC \rightarrow PC \rightarrow Heavy drinking$	1.01	0.006	0.005	0.032
OC wave 1 \rightarrow PC wave 1	0.84	- 0.094	- 0.092	< 0.001
PC wave 1 \rightarrow Heavy drinking wave 2	0.89	- 0.067	- 0.053	0.025
ERI correlates with PC		- 0.045	- 0.075	< 0.001
Tests of model fit		RMSEA= 0.065	CFI= 0.861	TLI= 0.766

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 \rightarrow Binge drinking wave 2	1.11	0.059	0.046	0.208
$OC \rightarrow ERI \rightarrow Binge drinking$	1.06	0.035	0.026	0.003
OC wave 1 → ERI wave 1	1.51	0.242	0.236	< 0.001
ERI wave 1 \rightarrow Binge drinking wave 2	1.30	0.145	0.112	0.003
$OC \rightarrow PC \rightarrow Binge drinking$	1.01	0.008	0.006	0.048
OC wave 1 \rightarrow PC wave 1	0.86	- 0.086	- 0.087	< 0.001
PC wave 1 \rightarrow Binge drinking wave 2	0.84	- 0.098	- 0.072	0.037
ERI correlates with PC		- 0.049	- 0.084	< 0.001
Tests of model fit		RMSEA= 0.062	CFI= 0.834	TLI= 0.745

Table 10. Path analysis for mediator roles of PC and ERI in relationship between OC and binge drinking in women

Table 11. Path analysis for mediator roles of PC and ERI in relationship between OC and problem drinking in women

Parameter	Odds Ratio	Unstandardized coefficient	Standardized coefficient	P value
OC wave 1 \rightarrow Problem drinking wave 2	1.13	0.070	0.055	0.158
$OC \rightarrow ERI \rightarrow Problem drinking$	1.06	0.030	0.023	0.026
OC wave $1 \rightarrow \text{ERI}$ wave 1	1.50	0.238	0.233	< 0.001
ERI wave 1 \rightarrow Problem drinking wave 2	1.25	0.124	0.097	0.021
$OC \rightarrow PC \rightarrow Problem drinking$	1.02	0.009	0.007	0.045
OC wave 1 → PC wave 1	0.85	- 0.090	- 0.091	< 0.001
PC wave 1 \rightarrow Problem drinking wave 2	0.83	- 0.102	- 0.079	0.029
ERI correlates with PC		- 0.048	- 0.083	< 0.001
Tests of model fit		RMSEA= 0.060	CFI= 0.855	TLI= 0.752

Table 12. Path analysis for mediator roles of PC and ERI in relationship between OC and heavy drinking in women

Parameter	Odds	Unstandardized	Standardized	P value
Falalletei	Ratio	coefficient	coefficient	r value
OC wave 1 \rightarrow Heavy drinking wave 2	1.03	0.018	0.012	0.533
$OC \rightarrow ERI \rightarrow Heavy drinking$	1.02	0.013	0.010	0.095
OC wave 1 → ERI wave 1	1.51	0.241	0.236	< 0.001
ERI wave 1 \rightarrow Heavy drinking wave 2	1.10	0.054	0.043	0.091
$OC \rightarrow PC \rightarrow$ Heavy drinking	1.01	0.005	0.004	0.070
OC wave 1 \rightarrow PC wave 1	0.86	- 0.085	- 0.086	< 0.001
PC wave 1 \rightarrow Heavy drinking wave 2	0.90	- 0.061	- 0.043	0.065
ERI correlates with PC		- 0.049	- 0.082	< 0.001
Tests of model fit		RMSEA= 0.064	CFI= 0.837	TLI= 0.726

Appendix 5. Published paper relevant for the thesis

During the period of my PhD study, I have published three academic papers in the field of social epidemiology (as first author), mainly focusing on work stress and health. In particular, one published paper relevant for the thesis has focused on the associations between OC, ERI and dietary outcomes in the HAPIEE study.

Published paper relevant for the thesis:

Chen SW, Peasey A, Stefler D, Malyutina S, Pajak A, Kubinova R, Chan JH, Bobak M, Pikhart H* (2016) Effort–reward imbalance at work, overcommitment personality and diet quality in Central and Eastern European populations. *British Journal of Nutrition* 115 (7): 1254–1264.